Drawing as a Tool for Thought:

The development of the ability to use drawing as a design tool amongst children aged 6 - 8 years.

Gillian Hope

Goldsmiths College, University of London

Thesis for submission for award of Doctor of Philosophy

2004



Gill Hope (2004) Drawing as a Tool for Thought

Abstract

My research into young children's understanding of using drawing to support designing was undertaken in a rural First School (children aged 5-9 years) across the years 1998-2002.

Since little previous research had been conducted in the field, the first phase of the research aimed to discover how young children could use drawing to support designing, through analysing drawings produced in Design and Technology lessons. It appeared that below age 8, although children could record design ideas, they did not use drawing to support their design thinking or develop their ideas towards making. Understandings gained through reading, led to the belief that the metaphor of design drawing as both a Container and a Journey could be used to teach younger children to use drawing as a design tool.

The second phase of the research, therefore, involved devising a Programme of four school term's duration, for a Year 2 class (average age 6.10 at start of Programme), that embedded the Container / Journey metaphor. These children's developing capability with design drawing was compared at intervals with that of a parallel class who did not receive the Programme.

The evaluation of the Programme used both qualitative and quantified analysis to assess both process and products of using drawing to support designing. The analysis instrument for the products was based on a holistic view of the design process, placing Understanding the Purpose of the Drawing at the centre of capability and Dimensions of Design Drawing through which such capability were expressed in drawing as emanating from this central understanding. Besides demonstrating the success of the Programme, the analysis showed how children adapt their use of drawing for different activities (problem-solving or product design) and revealed the importance of discussion whilst drawing for the development of viable design ideas.



Acknowledgements

My thanks for support and encouragement throughout the whole process of researching and writing this thesis go to:

My family:

My husband, Ray, without whose unfailing interest and enthusiasm the whole project would have been impossible, especially in helping to develop the necessary computer skills. Nothing has ever been too much trouble in terms of practical or emotional support, from the cups of tea that have quietly appeared to being willing to listen to me talk endlessly about methodology or design theory. My son, Ralph, and daughter, Rachel, who have never known Mum not to be studying; especially Rachel, who at one stage became my research assistant.

Kay Stables & Professor Richard Kimbell of Goldsmiths College, University of London:

Kay's attention to detail, as well as her knowledge and experience of research in Design and Technology, has been much appreciated. So too has Richard's encouragement as well as his ability to ask the searching (and, occasionally, difficult) questions.

The staff and children of Halfway Houses Primary School:

Gerry Wetherel, Headteacher, who tolerated me researching as well as teaching;

Sue Hammond, good friend of long-standing, Literacy Co-ordinator, whose own research into young children's developing writing skills contributed to my understanding of place-holding meaning; Thérèse Smith, Special Needs Co-ordinator, whose knowledge of cognitive development assisted my thinking about symbolic manipulation skills; Maggie Robinson, Art Co-ordinator, who, in 1996, initially suggested that "Flat Stanley" might be a suitable starting point for some small-scale research into Design and Technology;

Chris & Caroline, Claire & Hilary: Teachers of the Focus and Comparison Classes of the Structured Phase of the research, who agreed to me teaching their classes, and to teaching mine whilst I did so; Teaching Assistants : Wendy, Sheron, Jackie, Donna, Carol and Julia, who became so well-tuned to just how much help (or not) to give.

And, not least, the children, especially my Focus Class, on whose willingness to become part of what I was doing, the success of the whole project depended and to whom saying Goodbye was difficult.

Finally, for help and encouragement along the way:

Dr. Bridget Egan of King Alfred's College, Winchester;

Prof. Clare Benson and Julie Mantell of the University of Central England;

Dr. Malcolm Welch of Queen's University, Ontario, Canada

and members of the T.E.R.U. research group at Goldsmiths College: especially Dr. Rob Bowen, Dr. Dan Davies, Dr. Sandra Horne-Martin, Tony Lawler and Dr. Olefile Molwane

Contents Lists :

Contents of Thesis	5
Examples of Children's Work	10
Figures, Tables & Charts	11
Colour Conventions for Charts	14
Appendices	15

Contents of Thesis

SECTION 1 - Introduction	16
1.1 Overview of Thesis	
1.1.1 The Elements of the Study	
1.1.2 Thesis Outline	19
1.2 Starting Places	20
1.2.1 How Adults use Drawings	21
1.2.2 Drawing in Conversation	25
1.2.3 What do I mean by "Drawing as a Tool for Thought"?	26
SECTION 2 - Research methodology	27
2.1 Introduction to Section 2	28
2.2 Research Paradigms	
2.2.1 Modes of Engagement	
2.2.2 Making Observations	
2.2.3 Multiple Perspectives	
2.2.4 What Counts as Data?	
SECTION 1 - Introduction 1.1 Overview of Thesis	37 39
2.3 Epistemological Metaphors	
2.3.1 Ontology and Epistemology	41
2.3.2 Validation	
2.3.3 Credibility	
2.3.4 Transferability	
2.3.5 Ethics	50
2.3.6 Application : Myself as both Teacher and Researcher	51
2.4 Choosing a Methodology	53
2.4.1 Research and Design	53
2.4.2 "Hands on" or "Plan ahead"	57
2.4.3 Applicability to my Research Methodology	59
2.5 Research Organisation	61
2.5.1 Research Structure	61
2.5.2 My Kesearch Data	כט רא
2.6 Reflections on Section 2	

70

S. FINROQUCION TO SECTION S	71
3.2 Literature review: Drawing as product or tool	
3.2.1 Using Drawing as a Tool for Thought	73
3 2 2 Children Drawing	
3.1 Introduction to Section 3. 3.2 Literature review: Drawing as product or tool. 3.2.1 Using Drawing as a Tool for Thought 3.2.2 Children Drawing 3.2.2 Drawings as Maps of Development 3.2.2 Drawing for Drawings. 3.2.2 Children Drawing of Objects. 3.2.2 Childrens Planning Drawings. 3.2.3 Drawing for Developing Ideas. 3.2.3 Childrens Planning Drawings. 3.2.3 Drawing for Developing Ideas. 3.2.3 Lindrens Planning Drawings. 3.2.3 District Planning Drawings. 3.2.4 Implications of the National Curriculum for Design and Technology. 3.2.4 Distructions of Section 3.2 3.5 Design as Internal Process. 3.2.6 Making for Play. 3.2.7 Reflections on Section 3.2 3.3 Mapping the Field. 3.3.1 Mapping the Field. 3.3.2 Distinguishing Drawing Types. 3.3.2 Distinguishing Drawing Types. 3.3.3 Classifying Children's Design Drawings into "Drawing Types". 3.4 The Outliers. 3.5 Classifying Children's Analogy in the Construction of Meaning. 3.4.1 Modelling Concept Relationships. 3.4.2 The Role of Metaphor & Analogy in the Construction of Meaning. 3.4.3 The Container / Journey Metaphor.	
3.2.2b Canonical Drawing of Objects	81
3.2.2c Analogue Drawings	83
3.2.3 Children's Planning Drawings	84
3.2.3a Drawing for Developing Ideas	84
3.2.3b Sketch Books and Process Diaries	85
3.2.4 Implications of the National Curriculum for Design and Technology	86
3.2.4a "Extrapolation Downwards"	86
3.2.4b Observations of Young Children Drawing for Designing	88
3.2.5 Design as Internal Process	92
3.2.6 Making for Play	
3.2.7 Reflections on Section 3.2	97
3.3 Observational Constructs	
3.3.1 Mapping the Field	
3.3.2 Distinguishing Drawing Types	
3.3.2a Through Stan98	
3.3.2b Through Pandy98-9	110
3.3.4 The Outliers	114
3.3.5 Classifying Children's Design Drawings into "Drawing Types"	116
3 A Theoretical Constructs	4 7 7
2 4 1 Modelling Concept Bolationshins	400
2.4.2 The Pole of Metaphor & Applogy in the Construction of Meaning	
3.4.2 The Container / Journey Metanhor	
3.4.4 The Story behind Designing	
3.5 Reflections on Section 3	136
ECTION 4 - Structured Phase	138
4.1 Introduction to Section 4	139
	140
4.2 Planning for the Structured Phase	8 6 8 8 8
4.2 Planning for the Structured Phase	1/0
4.2 Planning for the Structured Phase	
4.2 Planning for the Structured Phase	140 140 142 144

SECTION 3 - Exploratory Phase

Contents

4.3 Programme Delivery	
4.3.1 The Structure of the Programme	154
4.3.2 Programme Outline	
4.3.3 Teaching the Container / Journey Metaphor	164
4.3.4 The Comparison Class' Experience	167
4.3.5 External Events that Affected the Programme Delivery	169
4.4 Evaluation of Teaching Input to Focus Class	
4.4 Evaluation of Teaching Input to Focus Class4.5 Reflections on Programme Delivery	171
 4.4 Evaluation of Teaching Input to Focus Class 4.5 Reflections on Programme Delivery	171 179 179

SECTION 5 - Analysis of the Assessment Tasks 184

5.1 Introduction to Section	. 5	185

5.3 Qualitative Evaluation of Children's Performance	.189
5.3.1 Assessment Tasks	189
5.3.1a Assessment Task 1: Design a pizza	189
5.3.1b Assessment Task 2: Frosty the Snowman	191
5.3.1c Assessment Task 3: Easter Egg Holder	194
5.3.1d Assessment Task 4 : Surprise Card	200
5.3.1e Assessment Task 5: Pandy's Suitcase	202
5.3.1f Assessment Task 6 : Theseus' Maze	205
5.3.2 EmergentThemes from Qualitative Evaluation	208

5.4 The Quantitative Analysis Instrument	212
5.4.1 Creating the Analysis Instrument	212
5.4.1a Starting Points	212
5.4.1b Grid and Ticklist	213
5.4.2 Refining the Instrument	216
5.4.2a The Purpose of the Design Drawing	217
5.4.2b Representing Progression	219
5.4.3 Re-defining the Analysis Instrument	221
5.4.3a A Holistic Analysis Tool	222
5.4.3b A Multi-layered Analysis Tool	225
5.4.3b(i) The Purpose Continuum and the Drawing Types	226
5.4.3b(ii) The Dimensions Continua	228
5.4.3b(iii) The Techniques Ticklist	231
5.4.4 Summary of Section 5.4.	233

5.5 Understanding the Purpose of the Design Drawing	234
5.5.1 Comparing Class Profiles	
5.5.1a Mean scores	234
5.5.1b The Range of Results	236
5.5.2 Developing Understanding over Time	239
5.5.3 Emergent Themes from the Purpose of Drawing Analysis	242
5.5.3a Cross-task Comparisons	242
5.5.3b Cross-class Comparisons	

5.6 Drawing Types	
5.6.1 Analysis of Drawing Types Used	
5.6.1a Viewing by Type	
5.6.1b Viewing by Task	
5.6.2 Relating Purpose Continuum to Drawing Types	
5.6.3 Comparison to 1998-9 Results	
5.6.4 Emergent Themes from the Analysis of Drawing Types	255

5.7 Dimensions of Design Drawing	257
5.7.1 Comparing Capability in all Dimensions over Time	258
5.7.2 Comparing Capability in each Dimension across Tasks	264
5.7.2a Comparing Mean Scores for each Dimension	265
5.7.2b Problems and Products	267
5.7.2c Differences in Performance on Dimensions Continua	269
5.7.3 Relationship between Purpose Continuum and Dimensions	272
5.7.3a Within Tasks	272
5.7.3b Across Tasks	275
5.7.4 Emergent Themes from the Dimensions Continua	276

5.8 The Techniques Ticksheet	.279
5.8.1 The Role of the Techniques Ticksheet	279
5.8.2 Information from the Techniques Ticksheet	280
5.8.2a Generating and Developing Design Ideas	280
5.8.2b Satisfying Constraints vs. Exploring Possibilities	283
5.8.2c Communication Techniques	284
5.8.2c (i) Annotation of Drawings	284
5.8.2c (ii) Level of Detail	286
5.8.2c(iii) Patterns, Motifs and Logos	287
5.8.2c(iv) Recording Materials	288
5.8.2d Planning Construction	289
5.8.2e Evaluating Ideas whilst Drawing	290
5.8.2f Relating Drawing to Making the Product	292
5.8.3 Emergent Themes from the Techniques Ticksheet	293

5.9 Gender Differences	
5.9.1 Quantifiable Differences	
5.9.2 Accounting for Differences	
5.9.2a Discussion whilst Drawing	
5.9.2b Designerly Play	
5.9.2c Accessibility of the Container / Journey Metaphor	
5.9.3 Reflections on Gender Issues.	

SECTION 6 - Conclusions and Reflections	305
6.1 Overview Summary of my Research Journey	306
6.2 Key Themes that Emerged from the Research	308
6.2.1 Children can use Drawing to Plan What They Want to Make	308
6.2.2 The Programme was Successful	
6.2.3 Why Children Choose to Draw As They Do	
6.2.4 Understanding the Purpose of Drawing for Designing	
6.2.6 Dimensions of Design Drawing on a Journey	
6.2.7 Generating and Evaluating Creative Solutions	
6.2.8 Multiple literacies : Discussing, Drawing and Writing	
6.3 Recommendations based on the Research	323
6.3.1 Teachers as Researchers	
6.3.2 Developing Children's Design Drawing Skills	
6.3.3 Developing Children's Design Capability	
6.4 Agenda for Future Research	333
6.5 Drawing as a Tool for Thought	334

3	31	r.	
1	3	3	35

Examples of Children's work

Frontispiece:

Design and Product : Pand	y's Suitcase; Yr.2 (2001)	. 1-2
	,	

SECTION 3 - Exploratory Phase

3.2 Literature review: Drawing as product or tool	
Ex. 1 (Section 3.2.2a) Yr. 3 : Drawing genres: Glen's Klee	78
Ex. 2 (Section 3.2.2a) Y2. 1 : Patterning	78
Exx. 3-5 (Section 3.2.2b-c) Yr. 1 : Delayed Language Development	81, 83, 84
Exx. 6-7(Section 3.2.4b) Yr.2 : Role of Task Structuring	90
Ex. 8 (Section 3.2.5) Yr.1 : Narrative genre	92
Ex. 9 (Section 3.2.5) Yr.4 : Design genre	
Ex. 10 (Section 3.2.7) Yr.R : Place-holding meaning	98

3.3 Observational Constructs

Ex. 11 (Section 3.3.1) Yr.1 : Flat Stan (1996)	
Ex. 12 (Section 3.3.1) Yr. 4 : Insects (1997 & 2000)	
Exx.13-15 (Section 3.3.2a) Yr. 2 : Flat Stans (1998)	
Exx. 16-17 (Section 3.3.2b) Yr.2 & Yr4 : Pandy's Suitcase (1999)	
Ex. 18 (Section 3.3.1) Yr. 4 : Surprise Tube (2001)	
Exx. 19-20 (Section 3.3.3) Drawing Type : Picture	
Ex. 21 (Section 3.3.3) Drawing Type : Single-draw	
Ex. 22 (Section 3.3.3) Drawing Type : Multi-draw	
Ex. 23 (Section 3.3.3) Drawing Type : Multi-design	
Ex. 24 (Section 3.3.3) Drawing Type : Progressive	
Ex. 25 (Section 3.3.3) Drawing Type : Interactive	
Ex. 26 (Section 3.3.3) Yr. 4 : Flat Stan (2000)	

SECTION 4 - Structured Phase

Ex. 27 (Section 4.3.3) Examples shown to Focus Class	
Ex. 28 (Section 4.3.4) Moving Pictures	
Ex. 29 (Section 4.4.1) Completed Wagwum Set	
Exx. 30-2 (Section 4.4.3) Focus Class Stans	
Ex. 33 (Section 4.4.4) Marble Runs	
Ex. 34 (Section 4.4.4) Sandals	176

SECTION 5 - Analysis of the Assessment Tasks

Exx. 35-6 (Section 5.3.1b) Frosty	
Exx. 37-44 (Section 5.3.1c) Easter	
Ex. 45 (Section 5.3.1d) Surprise Card	
Exx. 46-7 (Section 5.3.1e) Pandy's Suitcase	
Exx. 48-50 (Sections 5.3.1f & 5.3.2) Theseus' Maze	

Figures, Tables & Charts

Figures

SECTION 1 - Introduction

Fig. 1 (Section 1.1.1) Research Structure	18
Figs. 2-14 (Section 1.2.1) How Adults use Drawing	21-4

SECTION 2 - Research methodology

Figs. 15-16 (Section 2.2.1) Research Paradigms	
Figs. 17-18 (Section 2.2.3) Triangulation.	34-5
Figs. 19-23 (Section 2.4.1) Models of Researching and Designing	54-5
Figs. 24-5 (Section 2.4.1) Strategy Knowledge	
Figs. 26-31 (Section 2.5.1) Research Structure	61-3
Figs. 32-3 (Section 2.5.1) Research Subjects	64-5

SECTION 3 - Exploratory Phase

Fig. 34 (Section 3.4.2) Design Drawing Concept Web	
Fig. 35 (Section 3.4.2) Strategy Knowledge for Design Drawing	130
Figs. 36-38 (Section 3.4.3) Extrapolating Lakoff & Johnson (1980)	132-3
Fig. 39 (Section 3.4.3) Container / Journey Metaphor	134
Fig. 40 (Section 3.5) Transition to Structured Phase	137

SECTION 4 - Structured Phase

Fig. 41 (Section 4.2.3) Research Subjects (Structured Phase)	.142
Fig. 42 (Section 4.2.4) Primary and Secondary Data	.148
Fig. 43 (Section 4.2.4) Distribution of Data Collected	.149
Fig. 44 (Section 4.2.4) Excerpt from Log Book	.151
Fig. 45 (Section 4.2.4) Distribution of Data Sets used for Analysis	.152
Fig. 46 (Section 4.2.4) Management of Data in Structured Phase	.153
Fig. 47 (Section 4.3.1) Programme Structure	.154
Fig. 48 (Section 4.3.2) Teaching Input and Assessment Tasks	.158

SECTION 5 - Analysis of the Assessment Tasks

Fig. 49 (Section 5.4.1a) Kimbell et al.'s Dimesions of Capability	212
Figs. 50-1 (Section 5.4.2b) Non-linear Progression across Drawing Types	219-220
Fig. 52 (Section 5.4.3a) Pascal & Bertram's (1989) Assessment Model	
Fig. 53-4 (Section 5.4.3a) Purpose of Drawing Holistic Assessment Model	223
Fig. 55 (Section 5.4.3a) Dimensions Wheel	224
Fig. 56 (Section 5.4.3b(i)) Transforming the Purpose Rings into a Continuum	
Fig. 57 (Section 5.4.3b(i)) Relating Purpose Rings to Container / Journey	228
Fig. 58 (Section 5.4.3b(ii)) Rings on Dimensions Wheel	228
Fig. 59 (Section 5.7) Abbreviations for Reading Dimensions Charts	

SECTION 6 - Conclusions and Reflections

Fig.	. 59-60 (Section 6.2.3) Developing Understanding of Design Drav	wing
Fig.	. 61 (Section 6.2.3) Metaphorical Reasoning about Design Drawi	ng
Fig.	. 62 (Section 6) Relating the Dimensions to Making the Product	
Fig.	. 63 (Section 6) Moving Forward Design Ideas	
Fig.	. 64 (Section 6) Contributors to Ideas on a Journey	

Tables

SECTION 3 - Exploratory Phase

Table 1 (Section 3.3.2a) Research subjects : Stan98	104
Table 2 (Section 3.3.2a) Flat Stan database: Form view	
Table 3 (Section 3.4.2) Taxonomy of Design Drawing	
Table 4 (Section 3.4.2) Booth's Family Resemblances	

SECTION 4 - Structured Phase

Table 5 (Section 4.2.3) Qualitative & Quantitative Data	.145
Table 6 (Section 4.3.2) Programme Objectives	156
Table 7 (Section 4.3.2) Programme Sessions	.157
Table 8 (Section 4.3.4) Comparison Class Experience	.167

SECTION 5 - Analysis of the Assessment Tasks

Table 9 (Section 5.4.1b) Individual Marking Grid	214
Table 10 (Section 5.4.1b) The Techniques Ticksheet (Version 1)	215
Table 11 (Section 5.4.2a) The Purpose of the Drawing	217
Table 12 (Section5.4.2a) Purpose of Drawing Grid	218
Table 13 (Section 5.4.3b(i)) Purpose of Drawing & Drawing Types	227
Table 14 (Section 5.4.3b(ii)) The Dimensions Continua	230
Table 15 (Section 5.4.3b(iii) The Techniques Ticksheet (Final Version)	232
Table 16 (Section 5.6.1b) Static Drawings vs. Moving Ideas	249
Table 17 (Section 5.8.1) Ticksheet Example to show Progression	
Tables 18-19 (Section 5.8.2b) Task Constraints	
Table 20 (Section 5.8.3) Diversity & Richness	
Table 21 (Section 5.9) Numbers of Children by Gender	
Table 22 (Section 5.9) SATs Scores by Gender	

Charts

C	Colour Conventions for Charts14	4
SEC	CTION 3 - Exploratory Phase Charts 1-3 (Section 3.3.2) Stan98	8
SEC	CTION 4 - Structured Phase Chart 4 (Section 4.2.2) SATS scores14	3
SEC	CTION 5 - Analysis of the Assessment Tasks	
5.5 l ((((Jnderstanding the Purpose of the Design DrawingChart 5 (Section 5.5.1a) Means of Purpose of Drawing ScoresChart 6 (Section 5.5.1b) Purpose of Drawing Profiles23Chart 7 (Section 5.5.1b) Developing Design IdeasChart 8 (Section 5.5.2) Changes in Understanding of Purpose24Chart 9 (Section 5.5.2) Growth in Understanding at Class Level24	5 7 8 0 1
5.6 I (((Drawing TypesChart 10 (Section 5.6.1a) Drawings by Type24Chart 11 (Section 5.6.1b) Drawings by Task24Chart 12 (Section 5.6.2) Relationship between Purpose & Drawing Type25Chart 13 (Section 5.6.3) Pandy's Suitcase: Comparative Results25	7 8 0 4
5.7 I	Dimensions of Design DrawingCharts 14-15 (Section 5.7.1) Dimensions : Mean Scores - all Tasks25Charts 16-18 (Section 5.7.1) Dimensions : Mean Scores260-1, 26Chart 19 (Section 5.7.2a) Mean Scores for each Dimension26Chart 20 (Section 5.7.2b) Changes in Dimensions Means across Time26Chart 21 (Section 5.7.2b) Progress in use of Drawing over Time26Chart 22 (Section 5.7.2c) Differences in Performance in each Dimension27Charts 23-25 (Section 5.7.3a) Relating Purpose Continuum to Dimensions272-Chart 26 (Section 5.7.4) Dimension Means from Easter Onwards27Chart 27 (Section 5.7.4) Comparing Purpose to Dimensions27	845890467
5.8	The Techniques TicksheetChart 28 (Section 5.8.2a) Generation of Ideas28Chart 29 (Section 5.8.2a) Development of Ideas28Chart 30 (Section 5.8.2c(i)) Annotation of Drawings28Chart 31 (Section 5.8.2c(ii)) Level of Detail28Chart 32 (Section 5.8.2c(iii)) Decorative Features28Chart 33 (Section 5.8.2c(iv)) Recording Materials28Chart 34 (Section 5.8.2c(iv)) Recording Materials28Chart 35 (Section 5.8.2c) Evaluating whilst Drawing29Chart 36 (Section 5.8.2e) Reasons for Changes29Chart 37 (Section 5.8.2f) Relatiionship to Product29	1256789012
5.9 (Gender Differences Chart 38 (Section 5.9.1) Gender Differences: Purpose of Drawing29 Chart 39 (Section 5.9.1) Gender Differences: Drawing Type	17

Colour Conventions for Charts

= Interactive

 Where the Assessment Tasks are represented:

 = Pizza
 = Frosty

 = Card
 = Suitcase

 Where the Drawing Types are represented:

 = Picture
 = Single-draw

= Comparison Class

Where the Dimensions of Design Drawing are represented:

Where the two classes are represented:

= Focus Class

= Multi-design



= Progressive

Where other data are represented, other colours have been chosen.

Appendices

Appendix A: Taxonomy of Design Drawing	348
Appendix B: Programme Lesson Plans	350
Appendix C: Target Groups : Pizza	
Appendix D: Powerpoint Presentation of Snook's Animals	368
Appendix E: Target Groups : Frosty	
Appendix F: Focus Class Target Group: Stan series	371
Appendix G: Target Groups : Easter Egg Holder	372
Appendix H: Easter Egg Holder : all Products	374
Appendix I: Target Groups : Surprise Card	378
Appendix J: Focus Class : Sandals Project	382
Appendix K:Target Groups : Pandy's Suitcase	384
Appendix L: Target Groups : Maze	386
Appendix M: Log Book	
Appendix N: Development of Quantitative Assessment Instrument	
Appendix O: Quantitative Data	412
Appendix P: Published works arising from Thesis	425

SECTION 1 - Introduction

1.1 Overview of Thesis	
1.2 Starting Places	20

1.1 Overview of Thesis

This Introduction to the thesis briefly indicates the perceived problem that I wished to investigate, the question that I formulated and sought to address within my chosen setting, the organisation of the task and how it was structured and how the thesis is written to reflect this.

1.1.1 The Elements of the Study

<u>The Research Problem</u>: There was little research into the way in which young children could use drawing to support their design thinking and yet the implications of the National Curriculum for Design and Technology were that they would do so.

<u>The Research Question</u>: To what extent could young children use drawing to support their design thinking and could this be enhanced through teaching?

<u>The Research Setting</u>: All investigative work for this study was conducted at a rural First School in Southern England where I worked for most of the duration of the research, as Year 1 Co-ordinator and Co-ordinator for Design and Technology. I began by looking at drawings in Design and Technology lessons across the whole school (ages 5-9 years) and then conducted a programme lasting four terms with children from the beginning of Year 2 (mean age 6.8 years) through to the end of their first term in Year 3 (mean age 8.1 years), with the aim of improving their facility with design drawing.

<u>The Constructs</u>: In order to answer both halves of the research question, I needed not only to discover what young children *could do* but also *why* (or *why they did not*) in order to discover how to enhance their performance. I needed to discover what was at the root of being able to use drawing for designing, not just as skills that could be observed but also in relation to cognitive development and learning.

The theoretical constructs centred on the metaphorical nature of drawing and how new learning can be created by metaphorical extrapolation from the known to the unknown. By using a specific metaphor for design drawing (drawing as both Container and Journey), I sought to enable the children to use drawing to support the generation and development of design ideas.

The building of theoretical constructs came after the answering of the first half of the research question (to what extent could young children use drawing to support their design thinking), thus dividing the research into two phases: an Exploratory Phase (1998-2000) in which I

investigated what children could do and a Structured Phase (2000-2003) in which I attempted to implement a Programme to enhance performance:



Fig. 1 Research Structure

<u>The writing</u>: The way the thesis is organised reflects those two phases and the style in which it is written reflects my constant interaction with my material, whether reading the literature or observing the children. It talks about children's design journeys but it is also the story of my research journey. An end came when I needed to draw a line under the process and write about it. That it will be a spring-board for more enquiry is shown in the key themes that emerged and are discussed in Section 6: Conclusions and Reflections, each of which could become the start of another new journey.

1.1.2 Thesis Outline

Section 1, the Introduction to the thesis sets the scene by illustration, rather than by trying to tightly define, the area of investigation which I set out to explore.

Section 2 discusses the research methodology issues as it pertains to my study: paradigms, epistemology and choices affecting research design and organisation.

Section 3 details my first forays into researching young children's design drawings. The account of this Exploratory Phase is organised under three major headings: the literature search, the observations of young children (aged 5-9 years) drawing for designing and the theoretical constructs which evolved as a result of the interaction between the two.

Section 4 explains how this led into the more Structured Phase of the research and briefly outlines and evaluates the Programme delivered to one class of children (Focus Class) in Year 2-3 (aged 6-8 years) and how their capability would be compared with a parallel class who did not receive my input (Comparison Class).

Section 5 contains the analysis of the Assessment Tasks that were conducted with both classes during the Programme. Qualitative analysis of the children's performance is followed by the account of the quantitative analysis instrument by which the drawings were analysed. The results of that analysis are then given, divided into sub-sections relating to the layers of the analysis instrument. Also within Section 5 is a brief discussion of gender differences within the sample.

Section 6 pulls together the themes which have emerged from the thesis and draws conclusions from the research undertaken.

1.2 Starting Places

People use drawings in a whole range of contexts; even people who say they never draw. Since beginning to look at the way children might be taught to use drawing as a modelling tool to explore and convey their ideas about objects they wish to make, I have become increasingly aware of the use people make of sketches in the course of everyday life, quite apart from workplace use of plans and diagrams for buildings, electrical circuits, flow charts of productivity or traffic movement, layout and product design:

Sketch maps are frequently drawn to give directions.

- People can be seen walking around DIY stores clutching sketches of ideas and diagrams of room sizes.
- Drawings are frequently used to aid explanations: when our hot water tank needed changing, my husband came home with a diagram from his brother showing him what to do; when making clothes for my daughter when she was little, I used to draw her the range of possibilities for the garment: puffy sleeves, square neck etc.; when I want to tell a teaching assistant what I want the children to make, I draw the parts and use arrows and captions to indicate how it fits together.
- Abstract ideas are frequently modelled by drawings and diagrams. This can be generative as well as illustrative.
- The use of CAD packages has had its impact on drawing. I have found that manipulating screen objects can be just as fruitful a means of generating new ideas and seeing new combinations as pencil sketching.
- And the area of my greatest interest to generate and record ideas to support thinking and planning.

But like artist's sketch books, these drawings rarely get seen by others. They are redundant once the product is made and apart from professional designers who might want to keep them for future reference, they are discarded. On the following pages are some examples of these uses of drawing that I have collected since beginning to research drawing for designing. They illustrate rather than define or justify. They demonstrate the facility with which many people use drawing as a tool to support their thinking.

1.2.1 How adults use drawing

This section illustrates some of the ways in which adults use drawing.

Explanatory Drawings

These are frequently drawn as the person is talking and serve to illustrate and support a verbal explanation.







The example below, Fig. 5, also by my husband, is part of an explanation to a site engineer of the layout of the air-conditioning pipe-runs in a building. The paper was turned round partway through the explanation.



Drawing to model and develop abstract ideas

Fig. 6 was a response to the model of Performance Management with which we as a school staff were being presented. I disliked the "top down" approach and muttered as much to my colleague next to me, drawing diag.1 and saying that we worked as a team (diag.2). She responded with diag.3: "Now... this is us all going off in different directions." Which led me to create an alternative model of staff development (diag.4).



The following (Fig.7) is one of a series of diagrams which I created using a CAD package to help me work out the relationship between the various cognitive factors involved in using drawing for designing:



The following is the first of four pages filled whilst discussing the nature of design drawing with a fellow researcher:



SECTION 1 - Introduction

Drawing as a planning tool

Fig. 9 is my thinking drawing for a display at the exhibition at the DATA Millennium Conference. I then worked out an assembly order for the card panels, to be attached by slots and tabs.



When I wanted to play with some ideas for re-designing part of the garden, I measured it out and drew it on a CAD package. I then experimented with ideas based on the shape of the plot. There were several sheets of these, of which Fig. 10 is one:



Drawing for Exams

Excerpts from my daughter's GCSE Design & Technology Course Work:



The following is a plan for a mobile home suitable for disabled people. She produced a whole series of different views and projections of this and a beautifully-made 3D model from balsa wood complete with furniture, fittings and soft furnishings.

Fig. 12:



Completely missing are all her initial ideas sketches. All the scraps of paper onto which she played with ideas, crossed things out, redrew them and tried again have long since been assigned to the bin. The Course Work folder is itself a work of art.

That she did use drawing to play with ideas about real things that she intended to make is preserved in the spare pages at the back of old school exercise books.



In the back of her history notebook was a series of designs for new bedroom furniture that she might make. Fig.14 shows one page on which details of sizing and costing were calculated for the inside of the wardrobe. After costing, she decided that cutting up her brother's old wardrobe to make the shelves was the right option. She and her Dad spent a happy weekend making it:



Comment:

To me all these examples show adults using drawings to support real thinking about real spatial, organisational and conceptual problems. It was these skills that I aimed to encourage amongst the young children I taught.

1.2.2 Drawing in conversation

From my colleague Sue Hammond's study of Early Years children engaged on Independent Group Activities comes this snippet of four children (aged 5 years) interacting and negotiating their drawing experiences. The children are engaged in making a "Family and Friends Dictionary" :

Robert: Ip, dip, doo

Kathy: Eeny, meeny, miny, mo..... square, Mrs.H. I want 'square'.

Mrs.H.: No K., you're doing family and friends - remember?

Kathy: Oh, yes.

Emma: If you do a square you could do some arms and legs and make an Iron Man. **Robert:** I'm going to draw his whole body. (*Draws round figure on his page, watched by A.*)

Alex: When my brother was football training, this boy had all paint on him. He said it was chicken pox. *(Laughs)* Easy, peasy, lemon squeezy, Apple pie, take your squeezy.

(Robert put dots of pen on his face.)

Emma: You've got chicken pox. My mum's going to have chicken pox. (*Drawing Mum.*)

Robert: I'll tell my Mum and Dad I've got chicken pox. Who likes brown the best? **Emma:** Not me. I like pink and red the best.

Robert: I've got an alien bedroom.

Kathy: You haven't.

Robert: This is Henry [his twin] with chicken pox.

Emma: My Mum's going to have brown chicken pox.

Robert: Who wants brown? Who wants brown?

Emma: I'm going to do `lp, dip, doo'.

It is clear from this short excerpt that the children are thinking interactively as they draw. The serendipity of Kathy's desire to draw a square sparks Emma's memory of the Iron Man; Alex's anecdote of football training begins the whole cycle of chicken pox spots; What made Robert think of his bedroom? Despite Kathy's put down, the thought of home and the image of their shared room sparked the idea of drawing his twin brother Henry with chicken pox. Emma wants to add his dip rhyme which began the excerpt as a caption to her picture.

The phrase which I gleaned from Sue's study was "place-holding" (Hammond 1997). As shown by Ex.10 (Section 3.2.7), at the emergent writing stage, young children, use pictures and single letters, occasionally odd words, but often just marks and squiggles to place-hold their ideas so

that they can record and later recount the ideas which were flowing in their heads at the time. This caused me to wonder: Why could that facility not be harnessed for understanding designing? Surely sketching and recording design ideas is a form of place-holding? Could not this intuitive understanding of place-marking of developing ideas function as a spring-board for understanding how to use drawing to develop a design for making a product to fit a specific purpose?

1.2.3 What do I mean by "Drawing as a Tool for Thought"?

All of the above examples illustrate the way in which drawing is used to support thinking and communicating thought. The range of contexts, ages and intentions of those who drew these examples were varied. The common thread is the need to record visually and graphically that which could not be considered, manipulated or communicated by words alone.

I presented the phrase "drawing as a tool for thought," as the thesis title at the very beginning of my exploration of children's use of drawing to support designing. However, it was over a year before I felt I knew for myself what I meant by that phrase. I needed to read and to observe young children in action.

Drawing occupies a middle ground between the imagination and the real world. Even drawings of minimal clarity can be discussed and explored as if they were real, as they extend and make visible the inner thought processes of their creator. By objectifying these inner thoughts and images, the drawing enables these to be observed by the thinker. The imagination becomes visible and takes form. Changing and developing ideas now have something tangible on which to work, allowing review and reflection, return another day, with other ideas both new and old, which can be incorporated with the ideas recorded.

I perceive there to be a distinction between drawing as *product of* thought and drawing as *tool for* thought. The first describes an artefact, the second a process. The first brings closure to the activity on the completion of the act of drawing, the second describes a way of recording thoughts in action. They frequently occur together, especially in the action of designing. Several completed drawings may form a chain of products which together map out the path that thought has taken. It is not an "either / or" dichotomy, rather a "together / and" interaction, which supports thinking in process and, through the creation of visible products, enhances reflection and evaluation of thoughts and ideas.

It is this process to which I attach the phrase "drawing as a tool for thought". It was the development of this facility amongst young children that I set out to study, document and, once I felt I sufficiently understood it to explain it to them, to enhance capability.

SECTION 2: Research Methodology

2.1 Introduction to Section 2	28
2.2 Research Paradigms	.30
2.3 Epistemological Metaphors	.41
2.4 Choosing a Methodology	.53
2.5 Research Organisation	.61
2.6 Reflections on Section 2	. 68

2.1 Introduction to Section 2

Morgan (1983) provides a "framework for analysing the logics of different research strategies" under the following three heads:

- Constitutive Assumptions (Paradigms)
- Epistemological Stance (Metaphors) and
- Favoured Methodology (Puzzle-solving)

I have adapted Morgan's headings to frame this section of the thesis:

Research Paradigms (Section2.2)

This section looks at the way in which researchers engage with their research and its subjects. It does not provide a conventional overview of research paradigms, but rather discusses issues which are common to all research endeavours and discusses how researchers within different communities of practice have sought to deal with them. The concluding section (Section 2.2.6) indicates how these considerations relate to my research by addressing the question "Where do I fit?"

Epistemological Metaphors (Section 2.3)

Claims to knowledge through research lay open to question unless certain safeguards are applied. This section of the thesis briefly discusses some of the major issues: ontology and epistemology (Section 2.3.1), validation of research findings (Section 2.3.2), credibility (Section 2.3.3) transferability to other populations (Section 2.3.4) and ethical issues (Section 2.3.5) The final sub-section (Section 2.3.6) discusses how these issues relate specifically to my situation: conducting research in the school in which I was also a teacher.

Choosing a Methodology(Section2.4)

As a researcher of designing, I could see parallels between the two processes and was pleased to find that this had also been observed by others. This section explains how I have applied my reflections on designing to my research processes and provides a resolution of the dilemmas posed in Sections 2.2.6 & 2.3.6

Research Organisation (Section 2.5)

My research process fell into two distinct phases, one exploratory and one more structured, as indicated in Section 1.1.1. This, together with the way in which the data have been collected and handled, led to the decision that the way in which the thesis would be written would reflect these two phases of my research process.

Section 2.6 concludes Section 2 with Reflections on the issues raised throughout Section 2.

2.2 Research Paradigms

The chapter from which my framework for Section 2 of the thesis is taken (Morgan, 1983) has as its sub-heading "Modes of Engagement." I have chosen this as the title of Section 2.2.1, which attempts to dis-engage from the dichotomous presentation of research paradigms that many "how to do research" textbooks for novice researchers frequently present. Since making observations is the primary activity of researchers in the field, this seemed to be the next topic to discuss (Section 2.2.2). Multiple perspectives are frequently advocated to help eliminate a one-sided view of events and this issue is discussed in Section 2.2.3. Amongst competing and frequently conflicting philosophies and methodologies the question of what counts as data is often not far below the surface. Section 2.2.4 discusses some of these issues. All this must be placed in context for educational research and so the penultimate sub-section of this part of the thesis (Section 2.2.5) looks at the purpose of educational research. The Applications (Section 2.2.6) draw these discussions together into a brief indication of my personal position.

2.2.1 Modes of Engagement

The multiple *modes of engagement* in educational research range from the collation of nation-wide statistics of exam results and league tables to detailed descriptive/analytical studies of one child's development in a single area of learning. Each of these is underpinned by beliefs about the *values* of research, about what counts as data, about the role of the researcher, about how the research should be presented or disseminated. These are congruent with the pragmatics of what is available at the time, both in terms of the researcher's own knowledge and expertise and in terms of the opportunities which present themselves or are able to be sought. There were times when, contrary to the order in which textbooks are written, I felt like "researcher in search of a paradigm" and that "where I fit" was a conclusion I came to as part of the process of researching.

Paradigms of research in the social sciences are often classified as simple either-ors:



Fig.15: Research Paradigms (a)

The "how to do research" textbooks usually deal with either quantitative or qualitative methodologies. Quantitative methodology books typically write as if the other camp does not exist; qualitative methodology books imply that they have the higher moral ground because they are not using numbers to analyse or report results.

From the experience of being a psychologist by day and an avid reader of literature by night, Bruner (1962) asserted that there exists two ways of knowing, which he calls the *narrative* and the *paradigmatic* (the positivist viewpoint), which he perceived as incompatible because the positivist seeks validity through the pursuit of abstract truth (of mathematics, logic and science), whereas the narrative mode is validated through its truth-likeness and the ability to perceive universals through parallels. This difference, claims Bruner, precludes corroboration of the one by the other.

The reality of research is that the boundaries are much more blurred :



2.2.2 Making Observations

All science, whether physics or social science, is based on observation. The basic tenet of empiricism is that only observations made by the senses counts as research data; ideas, viewpoints, intuition and other internal states do not. Thus scientists attempted to gain a "pure" view of external reality. However, by Heisenberg's (1958) uncertainty principle, observation is affected by the presence of its observer and failure to realise this has led social science researchers of all persuasions into difficulties.

The positivist tradition within the social sciences has its roots in the desire to be seen as scientific. Its consequent methodology entails the collection of data against which a hypothesis is tested, with statistical techniques often applied to test the validity of the hypothesis. The experiment and its environment is set up and controlled by the researcher and rigorous measures are undertaken to exclude unwanted influences.

The dangers of attempting to ignore the differences between the context of the experimental laboratory situation and everyday life are inherent in the construction of problems "off-stage" (Lave 1988). Because the "construction process" is hidden from the experimental subjects, they are then deemed to have "failed" if they do not produce the intended response:

"This absence of a normatively defined response as failure is so central a hallmark of experimental (and school) practice that it may be surprising to note that there are substantive alternatives in most other social situations"

(op cit.: 36)

Thus the positivist paradigm is often rejected by teachers undertaking research, for example:

"The motivation for engaging in research of this kind is a search for answers: it is not reliant on preconceived notions, but a pragmatic response to an issue of concern or interest which will result in praxis. It is imperative that the researcher is responsive to the patterns and surprises that emerge from the intensive examination of seemingly familiar events."

Hammond (1999: 23)

The interpretive tradition, as a reaction to laboratory-based work, had its roots in anthropology and ethnography, aiming to observe and report on a natural human situation, frequently using descriptive analysis rather than numerical data. Researchers went out of the laboratory to become "participant observers", mixing with the people being observed as an honorary participant. The researcher did not originate from and was not part of the culture being described, which, they assumed, gave them an objective and, hence, scientific perspective. That they, as "trained observers", should be able to interpret their observations in an appropriate manner, even if this conflicted with the interpretations of the observed, raises issues of privilege, status and power.

Gitlin, Seigel & Boru in "The Politics of Method" (1989) observed that researchers from the interpretive traditions were just as likely as the constructor of the laboratory experiment to write themselves out of the script, frequently using their privileged position to say what things mean, since understandings of the situation expressed by subjects to the researcher are treated as data on which the researcher acts as arbiter. In their discussion of Apple & King's (1979) studies of Reception classrooms, they comment that the reader is expected to take the researchers' interpretations as givens. Despite studying the social constructs of the children and teachers, they do not consider their own. Apple and King are not alone, say Gitlin, Seigel & Boru, of being guilty of:

"a naive realism by editing themselves out of their text. they assume that non-reflexive, spectator-like research is possible and even essential to the writing of thick descriptions. In a sense these researchers use the language of traditional positivist research."

(Gitlin, Seigel & Boru, 1989:.203)

For example, one of the contributors to Ely et al. (1997) found herself in immediate problems when she, as a college administrator tried to be unobtrusive in a professor's seminar (pp.26-8) and concluded that to be a participant observer in one's own workplace was just "too close to home". Another of Ely et al.'s contributors commented on the role conflict:

"To observe is an unintrusive role. All my training has been geared to an active, participatory role." (op cit.: 47)

If "participant observation" implies "fly-on-the-wall" (and many interpretive studies adopt that stance), then the researcher is essentially a "non-participant observer", whereas teachers conducting research in their own classrooms are almost certainly acting as full participants. Most teacher research, including my own, is not of unobtrusive observing but of observing whilst participating, frequently as the power figure in the situation. In its original connotation (in contrasting field methodologies with laboratory methodologies) the term "participant observer" had meaning, but early researchers of all persuasions did not anticipate their subjects turning into researchers and researching themselves and their own setting, nor of seeking to change it as a result of their own findings. This led to the evolution of "action research" (practitioners researching action in which they are involved) and has important implications for issues of validity and reliability (Section 2.3)

2.2.3 Multiple Perspectives

Using different methodologies to examine a phenomenon is generally agreed to be a good thing. Trochim (2002), for example, expresses a commonly held view that multiple methods or viewpoints are essential to claims to validity. Triangulation is advocated as the way to reduce the risk of distortions inherent in the use of a single method.

However, there is little agreement between researchers of different traditions as to the nature of validity, let alone whether different methods either corroborate or contradict each other. Thus "methodological triangulation" (Denzin & Lincoln, 2002), defined as "a complex process of playing each method off against the other so as to maximise the validity of field efforts"

(p.304), leading to a reduction of "threats to internal and external validity" (p.308), becomes reduced to a mix-and-match of qualitative data collection techniques.

Brewer & Hunter ("Multimethod Research" 1989) begin their first chapter:

"diversity of methods implies rich opportunities for cross-validating and cross-fertilising research procedures, findings and theories..... we must develop more cosmopolitan research strategies."

(p.13)

Qualitative perspectives and methodologies, however, get no mention and form no part of this proposed cosmopolity. Thus Brewer & Hunter (1989) can discuss differing research methods as differences in *style* (p.11) and warn that in multimethod research one must ensure that mutual biases are not being re-inforced, without considering that any mix-and-match of methods from within one tradition will automatically re-inforce the biases of that tradition, unless those biases are made explicit and presented as part of the research setting.

Methodologically, there are two ways of viewing triangulation: as a process of cumulative validation or as a means of producing a more complete picture of a phenomena. Campbell & Stanley (1963) borrowed the triangulation metaphor from surveying. As can be seen by examining three different triangles (below), the closer A and B are to each other, the harder it is to determine the exact position of point C.



Denzin (2002) was only partly successful addressing some of the issues (practical as well as theoretical) inherent in the triangular metaphor, mainly because he looked only within the quantitative tradition for its application. From a theoretical perspective, it would seem almost incumbent upon every researcher whose aim is the production of a complete picture of a social situation to use as wide-ranging techniques as possible.

The ability of the two dominant traditions (qualitative and quantitative) to talk to one another certainly seems limited and attempts to create dialogue flounder (claim Fielding & Schreier, 2001) on the quantitative researchers' lack of understanding of qualitative methodologies and

perspectives. However, from my reading of texts from both sides, the lack of understanding (and respect) appears mutual.

However, it seemed to me that in the social research context, both ways of knowing could be desirable and mutually supporting. Among the things that qualitative data can contribute to quantitative research are depth, an idea of the range of core concepts, and the ability to solve puzzles that quantitative data cannot address (Sieber, 1979). This can provide context, interpretation and reasons behind the trends which quantified data can identify and, perhaps, possible solutions or ways forward from dilemmas thus identified as significant. Quantitative data can, in turn, enhance qualitative studies through giving a breadth of view, counteracting the tendency to focus on the easily accessible, to pursue topics of personal interest or bias which could threaten the validity of the analysis, through salience, ease of recall and previous cognitive schemas (Taylor, 1982).

Cross-paradigm approaches are, however, fraught with difficulty. Creswell (1994) points out the purely practical point of researcher time and energy expended in immersion in each in order to do both. The idea that research results produced across different paradigms can be used for mutual validation assumes the existence of a common epistemological framework (Fielding & Fielding, 1986), who argue that, although such combinations of methods may add breadth or depth to the analysis, the results may be no more valid. They emphasise instead the potential complementarity of qualitative and quantitative research methods, as does Flick (1998):

"Triangulation is less a strategy for validating results and procedures than an alternative to validation (...) which increases scope, depth and consistency in methodological proceedings." (p.230)

Extending the triangulation metaphor, devised by researchers within the quantitative tradition as a way of espousing multimethod research, to also include qualitative methodologies, may in fact make determining point C on the triangle just as difficult through distance of points A and B from each other:



By focusing on how the two methods might be complementary, such aspirations ignore the issue that far from being complementary, underlying ideologies, and hence research methods, may actually be contradictory. The debate over the relationship between philosophy and

methodology lies at the heart of the question of what counts as data (the subject of Section 2.2.4, below).

In the Exploratory Phase of the research, I was trying out a range of ideas, whilst also trying to maintain a consistency of approach so that results could be directly compared. I discovered that these two aspirations were not mutually compatible within the limitations of time and opportunity available to me. The drawings from whole school studies (Section 3.3.2) were subjected to quantified analysis, and my understanding of the range of techniques that children were likely to use at certain ages and in response to certain activities developed as a result of this work. The deeper understanding of reasons behind children's choices came through the evaluation of a wide range of activities undertaken by a large number of children across several years of their school life, supported by reading of the literature on children's drawing development and cognitive growth.

In the Structured Phase, the issues of validity and reliability were more focused on maintaining consistency in Assessment Task presentations and marking, whilst using a multi-method approach to evaluation (Section 4.2.3).

2.2.4 What Counts as Data?

Data are observed phenomena, which become informative when infused with meaning, through being interpreted by an observer who endows them with significance. The significance which is attached to the observation, and hence to the data, is coloured by the observer's previous experience, training, beliefs and world-view. Frames of reference hold the lens through which observations are made, recorded and classified.

Quantitative studies reduce all data to numbers. Qualitative research, essentially, tries not to, but the problems of having the research accepted by the wider research community as valid and reliable, leads many qualitative studies to have quantitative elements bolted on, for fear of attracting such criticism as that articulated by Hammersley & Atkinson (1983), that when such analyses are challenged, researchers claim the unassailable "ethnographic authority" based on "they were there" and having the esoteric expert's view on the meaning of the data. This fails to gain support amongst the quantitative community's concerns for establishing systematic, externally-validated analytical procedures.

Equally disparaging of the opposing camp's ideology and methods, Cupchik (2001), from within in the social constructivist tradition, typifies empirical research as:
"analytical in orientation and, while it acknowledges the facticity of social phenomena, it fractionates them and reduces them to simpler and more or less analogous models.With an emphasis on productivity to ensure advancement within the field, the experimental paradigm can become *functionally autonomous*, floating free of its original mooring in ecologically meaningful processes. Reference to the original phenomenon that first attracted the community of researchers may be lost." http://www.qualitative-research.net/fqs-texte/1-01/1-01cupchik-e.htm accessed March 2002

At the heart of the issue, it seems, but rarely stated by either side of the debate, is the distinction between data and information and the point of entry (or exit) of meaning. Information is meaningful; data becomes meaningful by employment. Information can be disputed; data is brought on as evidence to buttress or demolish. Problems emerge in this dialogue when each party's frame of reference does not allow the admission of the other's evidence by inability or unwillingness to grant it the status of data.

This has more often been a problem for qualitative researchers attempting to gain recognition for their chosen methodology than for those engaged in quantitative research. The previously established traditions within science militated against qualitative researchers' desire for recognition whilst unable or unwilling to assert their methodology as scientific. Thus the criteria which positivist researchers use to evaluate their own research strategies (reliability, repeatability) continue to be applied to other forms of research, without regard to the relevance to either the underlying ideology or the research setting.

2.2.5 The Purpose of Conducting Educational Research

As an advocate of action research, McNiff (1988) is wary of the direct application of either empirical and interpretive research traditions to educational research:

"It is not part of their methodological design to ask such practical, problem-based questions and it is not part of their conceptual repertoire to answer them. They can make predictions and give descriptions of the phenomena of social settings. They cannot give ... explanations for the events within those settings."

(McNiff,1988: 18)

Her objections rest on the purpose of the research, that neither empirical nor interpretive researchers aim to make direct changes in the lives of their research subjects, whereas in most

research in education, the burning question is frequently, what changes can be made here to improve children's learning?

Stenhouse's (1975) work was highly influential in advocating action research. However, his view of curriculum research was that of pedagogical development, of improving the teacher's practice. Elliott (1991), who was part of the Humanities Curriculum Project discussed by Stenhouse, voiced several objections to the latter's analysis of the project. One of these objections was the focus of Stenhouse's team on the quality of the teaching rather than on the quality of the curriculum materials which the HCP team was developing. This is hardly surprising given Stenhouse's own definition of curriculum as

"a particular form of specification about the practice of teaching and not as a package of materials or a syllabus of ground to be covered."

(Stenhouse, 1975: 142)

In educational research, there is no clear-cut division between research and development (Roberts, 2001). The action researcher expects their research to be useful, initially to themselves, and then, perhaps, also to others, in informing future professional practice:

"Educational action research is a form of educational research which places control over processes of educational reform in the hands of those involved in the action." Kemmis (1988) in Hammersley (ed.), 1991: 189

In the three years between the publication of Kemmis' paper in Keeves (ed.) (1988) and its inclusion in Hammersley's selection for the Open University's Course Reader (1991) lay the Educational Reform Act and the National Curriculum. As Hammersley pessimistically comments in the closing paper of this book:

"..the influence of research on the local and national policies that shape schools is extremely limited. If this were not so, the educational reforms of the late 1980s would not have occurred, since they involve assumptions that run counter to the results of most research over the previous thirty years." (p.220)

Over ten years later, the weakness of the links between practitioners and researchers is commented on by Foray & Hargreaves (2003) but these authors ignore the effects of a government-issued curriculum through which innovation is severely curtailed (the National Literacy and Numeracy Strategies being its most extreme manifestations). It is against this background that teachers (including myself) attempt to question, research and initiate changes.

2.2.6 Application : Where do I fit?

The issues discussed here in Section 2.2 are those of concern to all researchers and which relate to their ideology rather than to specific methodologies. Inevitably the two are inter-related and, coupled with the complexity of the mixture of ideologies, it is often easier to typify researchers by the strategies they use than by the ideology they adhere to, and which is so frequently unstated.

I found that I did not know where I fitted.

Both the empiricist and interpretative traditions assume that the researcher is an outsider. As teacher I am the power figure. Initially, action research did not fit me either, because improving my classroom performance was not my focus. I wanted to know if young children could access drawing as a modelling tool for Design and Technology. I then wanted to know what was behind the ability to do so and whether that shed any light on the age at which a child could be expected to use drawing to support designing.

Having a two-phase research project meant, I came to realise, that I had a paradigm shift mid-way through the research. At that axis point, I wrote:

"What excites me is finding out what is going on inside these little heads and whether they are able to grasp these big ideas about planning and modelling and taking ideas on a journey across the design sheet. I am on a journey of discovery, not about me and my professional performance, but about the children and their developing understanding of designing. I would, therefore, describe myself as a participant researcher: a participator who is also researching, a researcher who participates in the research scenario."

Now at the completion of the thesis, I would say that is only half of the story. From researching what existed, I moved into researching how I could enhance children's capability. Improving my teaching, although not an aim of the research, was an inevitable result of knowing more about children's learning. My purposes became those of the action researcher, whilst my methodological preference remained for mathematical modelling rather than descriptive analysis.

Methodological design and its associated conceptual repertoire determine the questions which can be addressed (McNiff, 1988). So as I took ideas and methods from different traditions, I needed to be careful that they would contribute appropriately to answering the questions I

wanted to answer. My awareness of the link between ideology and methodology grew across the duration of the Exploratory Phase, together with a realisation of the need to formalise moderation procedures in order to ensure validity and reliability in the Structured Phase. The evolving nature of the validation process, from a series of observations about children's drawings with which the Exploratory Phase began, to an multi-method analysis tool for making comparative statements about children's understanding of design drawing based on their work in Assessment Tasks conducted over a period of fifteen months, required not just planning, delivery, assessing and moderating, but also reflection on my place in the ideological spectrum.

The paper that I presented at IDATER 2001 (from which comes the above quotation) was entitled "Participant Research from the Point of View of a Participant Researcher" (Hope, 2001). I took the term from Phil Roberts' Keynote address at IDATER 2000 in which he advocated the necessity for research to be carried out by educational participant researchers rather than by observers.

If I am to answer the question "Where do I fit?" then I have to use Phil Robert's term "Participant Researcher". I am a participant in this research, the power-figure in the classroom, the designer of the Programme. I am not one of Ely et al.'s (1997) non-participating participants, although once the children are working I become the taker of field notes, operator of the video and the interviewer. I am not an empiricist because I believe that the interviewer / researcher affects the data by the constructs implicit in the hypothesis, the choices about what to observe and record, as well in the subsequent methodology employed in data handling, yet I used quantitative analysis in both phases of the research. I did not begin as an action researcher because I was not researching my own actions but the actions of the children and the nature of their learning. I am still not sure that the Structured Phase does not have more in common with empiricism than with action research paradigms.

So, with so much uncertainty about where I might fit, I accept the "Participant Researcher" title quite gladly. It allows me to be a full participant in the educational situation and does not tie me to any particular stance or tradition, with the embracing of favoured methodologies or the eschewing of others. It allows me to be myself: a teacher conducting research into children's learning.

2.3 Epistemological Metaphors

The way in which words are used is determined by our beliefs about their meaning. Part of the problem of initiating debate across paradigms, or of seeking combine insights from differing paradigms, is in unpicking the meaning of words, especially those words used in common to mean different things. Heated debates rumble around what a concept *is* rather than exploring *what the word means* in the contexts in which it is being used.

I have attempted, therefore, to stand back from these debates and have sought for words to use for headings which express the essence of the debate, rather than terms which particular traditions use to validate their own work and to judge others.

2.3.1 Ontology and Epistemology

Eisner (1993) bases his analysis of the issues surrounding the problem of objectivity on the distinction made by Newell (1986) between ontological and procedural objectivity, that philosophical debate about the essence of being and the nature of reality is different from the debate about methods employed in order to make credible observations about that reality.

"Objectivity is one of the most cherished ideals of the educational research community. In fact, it is so important that if our work is accused of being subjective, its status as a source of knowledge sinks slowly into the horizon like a setting sun.....To use the vernacular we want to see and tell it like it is."

(Eisner, 1993: 49)

What we believe about what we can know will inevitably colour what we attempt to discover. Our beliefs about valid means for doing so will drive the search for strategies to find out what we want to know. For example, an awareness of the way the presence of a researcher can alter the social balance, coupled to a dis-belief in the validity of statistical models, will sway a researcher towards a qualitative methodology. Whereas a belief in mathematical models as good analogies for representing knowledge, plus a belief that random sampling by impersonal questionnaires would eliminate reactions to the researcher, would probably result in qualitative methods being chosen.

Eisner (1993) paraphrases Kant as saying :

"Percepts without frameworks are empty, and frameworks without percepts are blind.... It is in the transaction between objective conditions and personal frames of reference that we *make* sense."

(op. cit.: 53)

and calls for a recognition of the constructed nature of knowledge based on the questions we ask and the frameworks which we apply, judged for validity by the application of reason and experience.

Jung's (1966) idea that humans meet themselves in the symbolic constructions that they create to negotiate the social world, is parallel to Nietzche's (1968) view that human treat the "frozen" words that they create as if they were the realities to which they relate, to Wittgenstein's (1969) "language games" and "seeing as" and also to Heisenberg's (1959) idea that the scientist confronts only themselves, not reality, since it is how one has engaged with the world that is being investigated. The photon in the box exists alone only in Einstein's thought experiment. In reality, if measured, it would not be alone. If these things are important for theoretical physicists to realise, then it is even more important for those who research the behaviour of people.

On the basis of multiple ontologies, Morgan (1983) speaks of "knowledges" (p.389) in his chapter entitled "Knowledge, Uncertainty and Choice." Uncertainty about what *is* has major implications for what we can claim to *know*. However, he does not view this as a problem, so much as an opportunity for celebrating multiple viewpoints on social reality: "uncertainty ultimately involves choice":

"in choosing a research strategy the scientist in large measure determines how the phenomenon being studied will be revealed, and indirectly, the consequences of the knowledge thus generated."

(Morgan, 1983: 391)

For the researcher, however, this is not just a matter of personal choice in order to come to a rounded viewpoint and deeper knowledge of a social situation. For research to be authenticated by others, issues of validity and reliability must be convincingly addressed. If claims to knowledge are to be made, especially if recommendations about the actions of others are based upon it, then the basis of that claim to knowledge needs to be firmly established. What begins as a personal quest for knowledge metamorphoses into a statement of personal belief based on observations or experimentation and then the researcher "goes public" and the questions come in: How do you know? Can you prove it? How valid, reliable, generalizable are the results?

Knowing for oneself and believing in the honesty of one's own endeavours is only the first stage in a claim to knowledge. Throughout the research process, I attempted, to the best of my ability, to put in safeguards against bias and halo effects. However, in the Exploratory Phase, when I was mapping the field and trying to discover what children could do, I swiftly came to the conclusion that for the sake of consistency and comparability between classes and ages of children, I needed to present and supervise the activity. Other teachers' interpretations of the activity and the level of help given to children would only add an unwanted variable to the data. In order to place this work in context and be able to claim a measure of objectivity in what I was doing, I needed to collect sets of children's drawings from a range of other activities conducted by colleagues.

Discussions of objectivity seem often to assume that the observer stands outside the situation, apart from the observed, in order to make evaluative judgements about them. Its inverse, subjectivity, is pejorative in the empirical tradition and qualitative researchers try hard to establish that their research cannot have this accusation made against it. Hence, Ely et al.'s (1997) advice on the difficulties inherent in researching in one's own place of work. For the teacher, this is inevitable. McNiff (1988) contrasts the professional researcher coming from the outside to observe and report on what they see and the teacher in the classroom whose heuristic knowledge and professionalism is inevitably part of the data set.

Claims to knowledge are not predicated on observations of the senses alone, as empiricists traditionally claimed. The knowledge that one brings to the situation is the driver behind the choice of what to observe as well as the interpretation that it is put upon it. For the professional researching their professional concerns, the richness of their professional knowledge and expertise cannot be discounted and the judgements that they make within that context cannot, I would contend, be pejoratively dismissed as subjective.

Such issues have impact on claims to validity and reliability of research methodology and findings.

2.3.2 Validation

"How validity is defined and treated varies according to what researchers do, what tasks they are undertaking, and in what phase or stage of the research they are in...Consequently, although we urge scholars to discover and formulate what their research philosophy is, we believe that it is only one factor contributing to how validity is defined"

(LeCompte and Preissle, 1993: 325-6)

As Polanyi (1958) observed, knowledge communities establish for themselves a language and methodology for establishing truth, by which claims to truth by other communities are measured, judged and, with unfortunate frequency, dismissed. This hinders communication between communities of practice, diminishes mutual respect and limits opportunities for dialogue. The way in which educational researchers from different practice communities classify and codify observations on phenomena and, hence, the way in which validity and reliability issues are viewed, is coloured by the paradigm in which they work.

Researchers from within the empirical tradition typically seek quantifiable data from which relationships can be calculated. Validity and reliability are closely linked to replicability and predictability. Texts on conducting quantitative research deal with topics of internal, external and construct validity, prior to discussing random sampling, questionnaire design and (numerical) data analysis.

For example, Judd et al (1991) give the following definitions of reliability and validity:

"The *reliability* of a measure is defined as the extent to which it is free from random error components. In turn, *validity* is the extent to which a measure reflects only the desired construct without contamination from other systematically varying constructs. Note that validity requires reliability as a prerequisite."

(p.51)

Large random populations are the most desirable source of such data, since Tversky and Kahneman's (1982) "law of small numbers" demonstrates the problems inherent in small-scale studies, in that the smaller the sample, the more likely it is to deviate from the norms of the whole population (in the same way that tossing a coin ten times is far less likely to produce 50% heads than tossing the same coin 100 times). Thus extrapolation from any small-scale study to the whole population it represents, based on mathematical modelling, is methodologically suspect.

That "other systematically varying constructs" can be isolated from the system being researched and their variance within that system plotted and their effects calculated and accounted for, seems a tall order for investigations within classrooms. The "contamination" might be the richness, the interesting corollaries, the reasons for action by actors within the social construct. The basis on which such constructs are dismissed as contaminating are, frequently, not mathematically determined but decided by the researcher's personal goals, interests and frames of reference. Unless these are declared and made clear as pertinent "systematically varying constructs," the data lacks validity by Judd et al.'s (1991) definition.

However, the belief that validity and reliability are as defined by Judd et al. (1991) is itself a "systematically varying construct."

The qualitative approach views itself as holistic, searching for patterns within the phenomenon as a whole system, producing descriptive analyses of complex relational structures in order to strengthen understanding of both the complexities and of the underlying structures. Texts on how to conduct interviews, collect field notes and make observations (for example, Ely et al., 1997) discuss the development of the researcher's own skills in social analysis and awareness.

Respect for each others view of what counts as validity seems hard to find, although interpretive researchers are more likely to have a working knowledge of some aspect of quantitative method (and use it in their research) than the other way around. Many studies begin with descriptive methods then lead into quantification of observations, perhaps because of a wavering of faith in the validity of qualitative research unsupported by quantified data. LeCompte and Preissle (1993) caution that moving from qualitative to quantitative processes without making explicit the inherent changes in ideology, prompts a single consolidated definition of validity, in which the quantitative paradigm becomes the arbiter of validity, and hence, of truth. I wanted to be able to use methodologies from both qualitative to quantitative paradigms without such implications.

During the Exploratory Phase of my research, databases were constructed to analyse the children's drawings, which proved to be a useful means of maintaining records of children's developing design capability. The database fields emerged in process of analysis of the drawings and were changed and added to over time in response to analysing more drawings. This is documented in Section 3.3.2. At the same time, I was making more detailed qualitative observations of children, such as those documented in Section 3.2.4b. This trend continued into the Structured Phase, in which both qualitative and quantitative methodologies were employed in order to give as full an account as possible of the development of the children's developing capability across the duration of the Programme. Section 5 reports the results of this dual method analysis.

At the start of the Exploratory Phase (see Section 3.3), I had a measure of awareness of the issues surrounding validity, especially in relation to the need to document the circumstances in which different ideas were tried. In the whole school studies, I attempted to keep the format of the lessons the same each time, and devised a script for task presentation so that each class received exactly the same introduction and instructions. I needed to make decisions over levels of assistance I would give to children who did not understand what to do. However, since I was essentially mapping the field, variations abounded. This was, however, an important part of the research process and, I believe, appropriate to my aims at that stage.

The Structured Phase, however, required a much tighter framework and for extraneous variables to be eliminated as far as possible. To make each Assessment Task comparable, for example, it was decided that the basic material of construction should be paper or thin card that could be easily cut and joined by all children. A small team of moderators was recruited to blind-mark a selection of children's work from each Assessment Task to ensure that the quantified data accurately represented the children's achievements. We met regularly throughout the Structured Phase to discuss both the children's work and the development of the analysis instrument (Section 4.2.3).

Section 4.2.4 details the data that was collected in the Structured Phase, as well as the way in which the qualitative and quantitative data were to be related. Although for clarity, Section 5 reports the qualitative and quantitative analyses separately, the report of the quantified analysis (Sections 5.5 - 5.9) is strongly related to the qualitative observations of children at work.

2.3.3 Credibility

The acceptance, either of concepts of validity and reliability as defined by quantitative research or of the definition of social reality created by the qualitative observer, could imply an acceptance of an ideology which hides the beliefs and viewpoint of the researcher behind the shield of expertise. Although a research project based on systematic data collection can appear to have more credence than one that is not, it does not necessarily follow that, because it is methodologically reliable, the results sufficiently meet the criteria of credibility. In order for research results to be credible, therefore, multiple perspectives are vital.

LeCompte and Preissle (1993) voice concerns for the researcher's audience and how the audience can be assured of the validity of results. They assert that researchers should enhance confidence in their results through: collaborative participation with the research subjects, congruency between theory and observation, inter-method and inter-observer checks, and personal reflection and introspection, to reduce the possibility that bias will affect the credibility of the research. Thus researchers must be "up front" about their own ideology, position in regard to the subjects and situation and their involvement and influence upon the data collection process. Each of the concluding sub-sections within Section 2 (Sections 2.2.6, 2.3.6, 2.4.3 & 2.5.3) provide an overview of where I stand with regard to the research that I have conducted. Sections 2.2.6 and 2.3.6 discuss the issues involved in being both researcher and teacher within the school.

SECTION 2: Research Methodology

The story of my research is related to the story of my developing understanding of Design and Technology education and my own capability in subject delivery. Whilst trying things in classrooms in the Exploratory Phase, I was finding out what worked in terms of delivering Design and Technology as well as what worked for research. This teaching journey was in partnership with colleagues from my own school and, through termly "Bridging Meetings", in tandem with colleagues from other local schools.

I was concerned that the constructs that were created from this exploratory work were reliable and were resultant from real Design and Technology lessons. The range of Design and Technology activities that contributed to the understandings that emerged during the Exploratory Phase are detailed in Section 3.3. Not all of these were from lessons that I taught, although they were all from within my own school. One of my first "Flat Stan" lessons was formally observed by the Deputy Headteacher as part of Quality Assurance.

I was fortunate in having colleagues who were themselves interested or involved in research, with whom I could discuss ideas and issues as they arose. But colleagues who did not see themselves as developing a research interest, were, nevertheless, interested in hearing about mine and contributing opinions and perspectives as well as becoming directly involved in trialling ideas. At the same time, I tried hard not to give the impression that I felt I was doing something of greater import than what they were doing. After all, their minds were totally focused on teaching their own class, whereas I was frequently elsewhere with other groups of children or inviting them to join my class and taking endless photographs of what they were making.

During the Exploratory Phase, I was given several class-sets of drawings from lessons that colleagues had devised and delivered. I was also obtained drawings made during lessons that I observed as part of my Co-ordinator role. If colleagues were especially pleased with lessons they had conducted, I would be invited to come and view the results. I organised annual Design and Technology competitions that would give me an overview of children's capability as well as encouraging children's enthusiasm for the subject.

However, not knowing the level of assistance that had been given to the children in lessons that I was not conducting myself, made direct comparison between lesson outcomes difficult. This was highlighted in a particular activity in the disparity between the performance of one Year 3 Year 3 class and that of the other two classes. It was intimated by one of the other teachers that this colleague would have helped her class rather more, so that they produced the best drawings. I arranged to teach all three classes as soon as possible and discovered that this class had the weakest understanding of the role of drawing for designing. Such

experiences convinced me that the only way to ensure parity during the Structured Phase was to conduct all the lessons myself.

The Structured Phase Programme utilised many of the activities devised and conducted by myself and others during the Exploratory Phase (Section 4.3.1) in order to enhance the reliability and transferability of the research findings. For example, the Easter Assessment Task was developed from an idea by colleague on my moderation panel who worked at a neighbouring school. The Surprise Card Assessment Task was the post-input assessment task from the Enriching Literacy Through Design and Technology Project (Stables et al., 2001) and was conducted by one of the researchers from that project.

2.3.4 Transferability

In the light of Bruner's (1962) two ways of knowing (the narrative and the paradigmatic) and the parallels to the working methods of the interpretive and the empirical traditions, it is important to consider the ways in which both paradigms judged transferability of results and conclusions to other situations and populations. It appeared to me that the basis of the two camps claims to generalizability was largely based on their view of how Truth was to be known, and as McNiff (1988) comments, the concerns of educational researchers is not simply to document what is, however that "is" might be conceived. Educationalists conduct research with transferability in mind.

The positivist reliance on the ideals of science attempts to seek the truth of a situation through controlled environments, large numbers of randomly selected subjects yielding numerical data to be manipulated by statistical means. Applicability and transfer has largely centred on *"generalising to and across other populations"* (Schofield, 1991) and largely equated with repeatability: if a given study was conducted by someone else with equivalent subjects under the same test conditions, would the results prove to be the same. The larger the population studied, the more statistically accurate the conclusions may be claimed to be, and hence more likely "true". Thus, by Tversky and Kahneman (1982)'s "law of small numbers," in order for results to have generalizability, there is a requirement for the subject sample to be representative of a much larger population.

The interpretative tradition (which includes, by default, the action researcher with no access to laboratories and statistical method) has emphasised the narrative (Bruner's truth-likeness), relying on the observer's reflective and analytical skills to give meaning to field notes, seeking by "rich descriptions" to capture something truth-like in their observations which can be

recognised as applicable to other similar social situations. Although the number of subjects in many such studies have been small, the attempt has been made to capture the essence of the situation, which illustrates the general human condition, or part of it. The microcosm speaks to the macro. In this sense, representativeness is not a statistical measure but a statement about the truth-to-life of the observations and subsequent analysis.

Although devised to answer the difficulties inherent in claiming generalizability from smaller studies using quantitative methods, Campbell's principle of Proximal Similarity (as applied in Trochim, 2000) seems a useful concept and equally applicable to qualitative research. Campbell's basic premise is that the greater the similarity between the sample studied and the population to which conclusions are to be applied, the better the claim to transferability. For example, the conclusions of study on 5 year olds can be more readily be applied to 6 year olds than to a 15 year olds. When applied to qualitative research, the principle remains the same: truth-likeness is transferable to similar social groups; similar children in similar classrooms should react in similar ways.

Across the duration of the research, children's capabilities in Design and Technology rose throughout the school. This was evidenced by the results of conducting the same activities several years running with the same age group. For example, the Insect Kit was conducted with Year 4 children in 1997, 1998 and 2000 as well as with my Structured Phase Focus Class as Year 3s in 2001 (see Sections 3.3.1 & 4.4.7). Perhaps the higher profile that my research brought to the subject as well as direct conversations with colleagues had an enhancing effect on standards of attainment and that the things that I was discovering during the Exploratory Phase and sharing with colleagues about teaching Design and Technology and the role of drawing to support designing, were permeating across the school.

Since my research is grounded in real Design and Technology lessons, I believe that the findings of the research are directly transferable to other school settings. The "Emergent Themes " sections of the analysis of the Structured Phase (Sections 4.5.3, 5.3.2, 5.5.3, 5.6.4, 5.7.4 and 5.8.3) that then feed into Section 6.2 and 6.3, demonstrate how the findings of the research built towards the identification of key issues and recommendations that speak from both the qualitative and quantitative data. Section 6.3, in particular, discusses the way in which I believe young children's understanding and use of design drawing can be enhanced, based on the research that I have conducted.

2.3.5 Ethics

".... we must remember that we are, as researchers, stepping into others' lives - and our actions must make sense to them. In our research ethics, we must need to move beyond an egocentrism into an empathetic perspective. To this end, children and others with whom we research have a right to understand, and have some control over, our subjective research intentions, however honourable these may be."

(Dadds, 2002: 13)

The discussion of the school records data in Section 4.2.2 highlights the ethical dilemma in being both teacher and researcher in the school where I worked. Although I could have pursued informal routes to obtaining more detailed data on children's performance in other subject areas and on entry to school (Baseline Assessment), I had to respect my Headteacher's concerns that information which he felt was confidential between school and parents did not enter the public domain. He was, however, consistently supportive of my research activities.

At the Structured Phase planning stage, the Headteacher and Year 2 teaching staff were given a copy of my Programme outline and rationale. All were quite happy for me to conduct the Programme of Lessons and Assessment Tasks. As all the lessons were conducted as part of the normal school day and, in the Structured Phase, constituted the Design and Technology Programme of Study for the Focus Class, informing parents of the detail of the Programme was not necessary, since I had the Headteacher's permission to conduct the lessons. If I used children's work in a published text or for display at conferences and exhibitions I gained their parents' permission to do so, especially if this involved photographs. Without exception, parents were delighted and proud when their children's work appeared in publications or was taken to conferences and exhibitions.

During the Structured Phase, however, discussing what I was doing became more difficult, especially since I was directly comparing the achievements of children who received specific teaching input to the work of those who did not. I was concerned that Miss N., the teacher of the Comparison Class, might perhaps have felt that her ability to teach Design and Technology was being compared to mine, especially when I could begin to see a clear difference between the two classes. We had worked as colleagues for some time and I had previously discussed my research with her freely. The opportunity arose to contribute some photographs to the magazine "Five to Seven" and I submitted some of the work she had done with her class, which were subsequently chosen for publication.

At the beginning of the Programme, I explained to the Focus Class children that I was using them to learn about the way that children learn to do Design and Technology. This meant that I would be taking photographs of them while they were working from time to time and occasionally use a video camera. I would be borrowing their drawings because that was the part that I was especially interested in. The Comparison Class children were also aware that when I was conducting lessons with them it was to find out about how children used drawing for designing, in order to maintain transparency with the children regarding my purposes.

I ensured that work from both classes was included in all publications and displays. I showed the children their work in print and brought back photographs of displays from exhibitions and conferences. I spoke to both classes always in terms of what "we" were doing which was "so good other people want to know about it" to encourage them to give of their best and feel part of a joint venture. At times I felt incredibly dependent on *them*. If they had not responded well or produced work which fell far short of their best, or even behaved badly whilst I was there, this could have posed problems, especially during Assessment Tasks.

2.3.6 Application : Myself as both Teacher and Researcher

Having worked at the school for 15 years, I was accepted and trusted by my subjects and their families and knew the children extremely well, unlike many researchers whose only knowledge of their subjects' performance relates to the tasks set to them. Disadvantages could perhaps be seen in my knowledge of my subjects. It could be argued that objectivity might be a problem if I know the children so well. Would I be looking at them and their work through rose-coloured glasses?

However, objectivity is not the same as detachment. That I know and like the children does not impede making appropriate judgements about the work that they do and the products which they create. If my relationship to them was social rather than professional, then this criticism might have validity, but by the very nature of my job my role was to constantly evaluate performance and attempt to improve it.

I tried to minimise the impact of my personal teaching style on the results so that it could not be said that this was the cause of any observable improvement in the work of the Focus Class in the Structured Phase (who received the Programme) rather than the content or method embodied in the Programme: firstly, by using activities that I had conducted previously with other classes and, secondly, by using my 1999-2000 Year 1 class as the Comparison Class, who would begin Year 2 completely saturated in my way of working.

SECTION 2: Research Methodology

I was not able, however, nor would I have wanted, to stifle the evolution of relationship and shared meanings with the Focus Class. Teaching them weekly for over a year, made it inevitable that a relationship developed between myself and the children. My way of presenting tasks, the vocabulary, syntax and body language I use to explain things, became part of the shared meaning which became established between us as the weeks progressed. I was aware that, as time elapsed since I taught the Comparison Class full-time, their heuristic grasp of my way of doing things was slipping away and that as each Assessment Task came and went that the Focus Class were at an advantage, and even though I purposely avoided using words such as "design journey" in my presentation of Assessment Tasks, these things were there, still understood between us, even though I did not say the words. The Comparison Class had no way of second-guessing such meanings from my words. I knew this was inevitable but I had not expected to be so aware of it experientially.

School environments are such that children are constantly expected to grasp and make sense of things through the establishment of shared meanings with their teacher. Where they do not understand fully, they create a temporary mental construct which they can manipulate until fuller understanding arrives. The scaffolding for this mental place-holding is provided by the teacher who supplies the terminology, the syntax and context for new learning. As the children begin to understand for themselves, so they become independent of the teacher's supporting role. I relied on this fluidity to develop the Focus Class' understanding of design drawing.

2.4 Choosing a Methodology

"The designerly mode of enquiry is entirely appropriate to the study of education through Design and Technology."

(Archer, 1992:13)

Since my interest in researching children's design drawings began, I have been aware of the close similarity, duality almost, between design and research, creation and discovery. Both are teleological and similar processes take place en route: the form of the result of the process is not clear from the start; sources are consulted for information and ideas, which are generated, considered, played with, discussed with peers and experts, modified, rejected, and re-emerge in differing forms further down the line; the final product is a result of the interaction of all these previous processes:

"Accumulation, parallelism. reinvestigation and the whole idea of development as multilinear, multifaceted and of unpredictable pace, speaks to a world of mental flexibility and diversity that artists and other creative practitioners of all kinds can empathetically recognise."

(Paine,1992: 7)

2.4.1 Research and Design

Design, as well as research, has seen attempts by theorists to be seen as scientific. For example, Cross, Naught and Walker (1986), in discussing the need for the term "design science", suspected that:

"this attraction lies not so much in the *method* of science, but in the *values* of science. These are the values of rationality, neutrality and universalism."

(p.20)

The close parallels between the articulations of the action research process by theorists in support of an alternative methodology and the defining of the design process has produced similar graphical representations (Bowen 1998), as shown overleaf in Figs. 19-23 : Models of Researching and Designing.



Fig.19: Models of Researching & Designing (a)

closely resembles design process theory, for example (Fig. 20), Kimbell (1986):



Fig.20 : Models of Researching & Designing (b)

and the model shown in Fig.21 developed by Kimbell et al. (1991) to attempt to capture the interaction between internal and external aspects of designing also can be seen as having parallels to research activity :



Fig.21: Models of Researching & Designing (c)

The centrality of reflection in designing, equally true in research, is shown in the following model (Fig.22) by Rogers & Clare (1994) :



Both design and research are forms of Rittel & Webber's (1974) "wicked problems," social and systems problems whose goals are unclear at the start and indeterminate in character; the kind of problems that Ackoff (1979) calls "complex" and "messes". Middleton (2000) provides a model (Fig.23) for solving such ill-defined problems based on Newell & Simon (1972) which substitutes the latter's "goal state" with a "satisficing zone" and portrays the route to achieving this as the "search and construction space." Again, this is of equal application to research, since most research problems are ill-defined and gain definition, even in the mind of the researcher, in process of researching and there is rarely a "one right answer".



Fig.23: Models of Researching & Designing (e)

My personal view of how problems are solved is an expansion and development of Ryle's (1949) division of knowledge into *knowing how* (skills) and *knowing that* (facts). Factual information about appropriate processes affects the *knowing how* to approach a task. There seem to be inseparably woven threads combining *knowing how* and *knowing that* in the developing mind. One cannot say which comes first, how they might be separated or how they interact.

In problem-solving of any sort, I believe that *knowing how* and *knowing that* are combined in *strategy knowledge:* knowing a particular procedure will work best in the given circumstances. The procedure itself might be classed as *know how* but *know that* is needed to apply it. It is the development of this *strategy knowledge* in relation to using drawing to support design thinking that is the focus of my research, but I have been always conscious of a similar process happening in relation to my own research journey.

This view of the problem-solving process (equally true of design or research) summarises the duality which I perceive and experience in both. The parallelism between the two processes led to the creation of a epistemological model (Fig. 24), which I have adapted and developed to support a range of topics (for example, Hope 2000a, 2003a). This adaptability supports my belief in its validity and generalizability :



Fig.24 Strategy knowledge (a)

For example, children can be taught certain procedures (labelled diagrams, for example) at quite a young age but do not access this procedural knowledge because they do not see its applicability to problem solving. Thus, the problem-solver also needs to *know that* this knowledge can be harnessed to the solving of the current problem. *Knowledge that* only has power if linked to *knowledge of relevance* to the problem. Then *knowledge how* is needed. This may be a physical skill, a mental strategy, or previous experience of solving like problems.

This combination of *know that* and *know how* form the basis of the strategy to be employed to solve the problem. The choice of appropriate strategy in any problem situation depends on the depth and salience of the *know that* and *know how* which support it. Ryle's (1949) *knowing that* needs to be extended to cover not just factual information but also concepts and understanding, including perception of similarities to previously experienced problems and analogical insight.

2.4.2 "Hands on" or "Plan ahead"

Of design research strategies, Pye (1964) comments wryly:

"It must be emphasised that design, of every kind, is a matter of trial and error...We have to make the things we have designed before we can find out whether our assumptions are right or wrong...Research is very often a euphemism for trying the wrong ways first."

(op. cit.: 36)

Strategies for designing and making could be typified as two approaches: "hands on" or "plan ahead". The former is more characteristic of children and un-schooled adults. The latter is the way of industrial practice and National Curriculum Design & Technology.

The hands-on approach:

By "hands on" designing, I mean that design decisions are made through direct engagement with the materials of construction. It is the one-shot approach, in which designing and making happen altogether. At its most extreme, decisions about what the product might eventually be or look like are only partially formed in the mind's eye and re-imaging of the design solution occurs as the product itself takes shape.

The advantages of this approach are in its immediacy. It is tactile, satisfying to the senses and appears to make instant progress towards goal. The real object can be manipulated for size, shape, fit, match. However, the disadvantages are that it eliminates other choices once started (other approaches may have been better). It is potentially wasteful of materials and therefore could be expensive if it does not work, leading to disappointment and the project being abandoned. If the project has to be continued at a later date, it could end up looking nothing like original intention because what was being made, what was to be used, and so on, has been forgotten. The project cannot be continued by others.

The plan-ahead approach:

This approach involves a much greater measure of decision-making before engagement with construction materials. There might be some drawing, note-taking, discussion with peers or experts, making of mock-ups or practising of techniques.

Advantages here are that this method of working takes account of material requirements before cutting which enables costing and resource management. There is a clear sense of purpose and direction to the activity so there is less chance of messing things up. Several ideas, techniques or materials can be tried ahead of task, which can reduce the possibility of major failure later on. The project can be continued at later date without forgetting what had been decided already and the making can be performed by others.

The disadvantages are in the level of cognitive development required to utilise such a technique due to the knowledge base required with regard to the handling of materials (and their properties), techniques, measurements, calculations, where to look for information not to hand - even the knowledge of not having all the information required. It seriously delays the start of the activity, which is the sensually satisfying part of the task, and appears to delay the final completion of the task - time is spent *doing* nothing. It could mean sticking rigidly to a design even though it is clearly not working. Finally, having solved the problem mentally, is there the incentive to carry it out?

What usually occurs in practice is a task-appropriate interaction of both, which derives from the experience and knowledge of the practitioner, i.e. their personal *know how* and *know that* which allows them to start at a particular point on a continuum between the two:



Fig. 25 Strategy knowledge (b)

The knowledge base of the expert allows them to apparently adopt a "hands on" approach but the decisions being made are based on prior knowledge and experience, which enables decisions to be made about the planning process. External modelling, whether by drawing or practice piece, is used to clarify uncertainty about the best procedure to be adopted. Novices and children are far less likely to have appropriate internalised models to which they can relate the present problem and thus are less able to gauge (and hence under-estimate) the level of external modelling support that their design ideas require. Being unable to manipulate a visual image of the problem and identify areas in which modelling would aid success, they tend towards using "hands on" strategies, with subsequent lower likelihood of success.

As I observed children at work on design activities, I could see parallels to my own research activities and, through reading accounts of other researchers, felt that these two approaches were also in evidence in research activity.

2.4.3 Applicability to my Research Methodology

It could be said, therefore, that the "hands on" approach typified the first phase of my research. The activities which I conducted with the children before December 1999 were exploratory. I tried out activities across the whole school or with single classes to see how the children would respond, what strategies they would use and whether there would be progression, patterns or features which emerged at different ages. The insights which resulted from these activities formed the foundations of my theoretical understanding from which a pre-planned Programme developed. I was certainly a research novice. My procedural knowledge was low and I was trying things out in classrooms and reading research literature all at the same time.

As I amassed data and came to conclusions, I formed a hypothesis which I wished to explore and for which I needed to devise a Programme to test the hypothesis - "plan ahead". The Structured Phase.

This was surprisingly hard and extremely daunting. I could not now just try something out and see how it worked. I had, to my view, a "big idea", and it needed extensive development. On the basis of my hypothesis I had to design a Programme to try to prove it. I spent several weeks wandering around with the word "How?" hovering spectrally in my head.

Many of the tasks used as part of the Exploratory Phase could be re-used. The same task, presented differently, would give me a strong feel for the validity of my hypothesis. But I felt concerned about how far this would swing me towards an empiricist methodology, when I felt that one of the strengths of the "hands on" approach in which I had so far proceeded allowed me to be reflexive along the way and react to insights which emerged. More importantly: what if my hypothesis was wrong?

The Exploratory Phase was not devoid of number crunching (an empiricist trait). The tasks had been analysed through the use of databases to produce reports and spreadsheets to produce charts. These had yielded the evidence of the characteristics of children's design drawings on which much of my understanding was based. My original intention was, therefore, to continue to use such techniques to analyse the data produced by the Programme. However, on further consideration, they did not give the detailed view that I wanted. They had produced broad categories from data spanning ages 5-9 years. I was now looking closely at 6-7 year olds and trying to find out exactly what was going on. A new analysis instrument needed to be devised that incorporated the understandings gleaned from the original studies.

The Programme and its analysis methodology was, therefore, designed ahead of task, in April 2000. There were, inevitably, changes along the way and these are documented in Section 4.3.5 (Programme) and Sections 5.4.2 & 5.4.3 (Analysis Instrument).

SECTION 2: Research Methodology

One feature of qualitative research methodology which I did not want to abandon in my "planned ahead" Structured Phase was the ongoing analysis of the data, which could influence the course of the research. I did not want to have set a Programme which I would feel duty bound to stick to rigidly even if it was not working. This, to me, is the biggest disadvantage of the "plan ahead" approach and of the positivist model: analysis comes at the end, as a summary of what happened. This might be acceptable for molecules but it could not be so for children, especially since I was delivering a programme which would form a significant part of their overall educational experience in Year 2 (5% of their lesson time).

Therefore, it was important to me at the start to remain flexible and be prepared to change aspects of the Programme in the light of analysis, informal as well as formal. In practice, however, changes were made due to circumstances which could not be predicted ahead of task, rather than in response to analysis.

2.5 Research Organisation

The word "structure" suggests organisation and planning. At the beginning of my research I was not planning anything more organised than a single lesson or two that could be carried out with different classes around the school to find out how children responded. However, as the project grew, so structure and organisation had to be imposed on it in order to avoid a management crisis. This section of the thesis outlines the structure which developed, the kind of data collected and the decision to structure the writing of the thesis to reflect the research process.

2.5.1 Research Structure

The classic positivist research structure can be likened to an hour-glass (Fig.26) :



Fig. 26: Research Structure (a)

http://trochim.human.cornell.edu/kb/strucre.htm accessed June 2003

Researchers from other traditions, would not recognise these stages. For example (Fig.27), McNiff's (1988) used spirals to model the way that action research might explore issues which caught the interest of the researcher and proceed on parallel paths and contribute insights to the main issues :



These two models represent very different, and in many ways conflicting, views of the research process. Trochim (2002)'s model funnels down, not deviating from its pre-determined goal. McNiff's spreads out, almost in danger of losing the plot.

My research, however, has been in two phases, the first of which (the Exploratory Phase) cascaded into the subsequent Structured Phase and, although the hour-glass analogy is useful, it needs some adaptation to describe my personal research endeavours. The Exploratory Phase was far more like McNiff's spirals, following my own interests and opportunities, trying things out and changing and adapting in the light of observations. The Structured Phase, however, was much more like the hour-glass. There was a Programme to deliver and an analysis instrument against which to evaluate children's work. I could no longer follow interesting spirals off into other realms.

Fig. 28: Research Structure (c)



These two, contrasting methods were determined by the nature of the research process within each Phase. The primary motivation behind the Exploratory phase was investigative. I wanted to find out how children used drawing to support design thinking by examining large numbers of design drawings from across the 5-9 years age range. The reading in which I was also engaged at this stage led to the development of a theory about the role of drawing in the design context which could be applied to teaching young children to use drawing to support design thinking.

The Exploratory Phase of the research was based on Inductive Reasoning (Fig.29) :



Fig. 29: Research Structure (d)

I then devised a Programme which would use this theory as the vehicle for explaining the nature of design drawing.

Thus the Structured Phase was based on the Deductive Reasoning (Fig. 30) which developed from the theoretical conclusions of the Exploratory Phase:



Fig. 30; Research Structure (e)

I did not go back to the Exploratory Phase once I had moved into the Structured Phase. And although the development of my understanding involved the constant re-assessment and re-application of previous understandings, it was more like a snowball rolling downhill than a carousel. There came a time when the exploration had to stop because I had formulated a hypothesis and I needed to structure the research to prove it. The neck in the hour-glass represents that point. That the way I structured the account of my research should reflect this reality is something that I felt strongly.

To return to the hour-glass model to unite these two modes of working: although each of these phases were discrete, the emergence of theoretical understanding allowed the tentative hypotheses of the Exploratory Phase to cascade into the firmer working hypotheses of the Structured Phase, which in turn enabled me to make focused observations, to be analysed in a more rigorous way to provide confirmation of my theoretical understanding, which underpinned the hypotheses which the Programme was designed to test.



Exploratory Phase

Structured Phase

Fig. 31: Research Structure (f)

In his appropriately entitled paper "Self-doubt and Soft Data", Ball (1993) describes the development of theoretical sampling from the naturalistic:

"Theoretical sampling involves the use of analytical insights derived from data collected up to a particular point in time in order to make decisions about the collection of further data"

(op. cit.: 41)

which leads to data collecting focusing around these emergent categories, being part of a

"secondary process of progressive focusing whereby other sorts of data are no longer collected...Choice indicates control and reflexivity." (ibid.)

This seems to describe the transition from the Exploratory Phase of my research to the Structured Phase. My understanding of the development of young children's design drawing capability emerged within the first phase, from analysing and reflecting upon the mass of data generated mainly from tasks conducted across the whole school, which enabled focusing on one age group and devising a programme by which to test my hypothesis in the second.

Despite not using positivist methodologies, I found some of Trochim's (2002) accompanying diagrams helpful in clarifying the kinds of experiments that I was conducting. This was not an attempt to dress up my endeavours as somehow more "scientific" but the utilisation of a useful technique to aid thinking.

The initial Exploratory Phase of my research, in which I was trying a range of ideas with a range of classes and observing the results, was of the Post-test Non-equivalent Groups Non-experiment type (Fig. 32) :

N 1	X 1	O 1
N 2		O 2
N3	X2	O3
N4	V	O4
N5 Ne	▲3	
5 F T T		

(*N* = Non-equivalent groups; *X* = treatments; *O* = observations or measures)

Fig. 32: Research Subjects (a)

Note that some groups have "treatment" and some do not, some classes I had taught at some time in the past, and some I was teaching regularly whilst researching and some I met for the first time to carry out an experimental task.

The Structured Phase, on the other hand, in which I was delivering a Programme to one class (N1, the Focus Class) and comparing the results of Assessment Tasks conducted with both them and a non-participatory class (N2, the Comparison Class) was of the Pre-test/Post-test Non-equivalent Groups Quasi-experiment type, which, since I was conducting Assessment Tasks throughout the duration of the Programme, could be represented (Fig.33) as:



Fig. 33: Research Subjects (b)

The two classes studied in the Structured Phase are Non-equivalent Groups since they were parallel classes within the same year group at the same time. Children were assigned to classes on entry to school in the Reception Year when little was known about their capability. By conducting an Assessment Task at the beginning of the Programme, I attempted to ascertain pre-programme differences in design capability between the classes and used the Key Stage 1 SATs test results as a proxy pre-test of academic attainment.

However, in adopting this way of considering my research subjects, the differences in the aims of my research and those of non-participant observers must be made clear. Those who employ such terms such as "predictive validity" assume that the phenomena that will be observed are phenomena in which they themselves have no part. In the Structured Phase I was making predictions about changes I believed would come about through a Programme I had designed and was to deliver. I predicted that it would be successful and I believe the data suggested that it was. However, as McNiff (1988) observed, teachers researching teaching and learning have a different agenda to those who are simply observing others doing their job.

2.5.2 My Research Data

The input to children was at group (whole class) level. The analysis, however, began at the level of individual drawing in response to input. This is in common with how teachers normally evaluate both children's achievement and their own effectiveness. However, as well as drawings being compared with each other and the cumulative achievement of children compared within their own class, whole class comparisons have been made. This has been true in both the Exploratory Phase, in which it was being established what children could do, and in

the Structured Phase in which two classes were being compared more closely across the course of the Programme.

Thus there were three levels of analysis: the drawing, the child and the class:

Within each phase, each *drawing*, regardless of task, was subjected to the same analysis instrument. So that, although a better instrument was developed during the second phase of the research, there was internal consistency within each phase.

In both phases, the *child* was the basic record-keeping unit. In the Exploratory Phase I maintained records across three years of many children's time in school, which enabled me to form an impression of underlying skills that aided facility with design drawing. In the Structured Phase the quantified system enabled me to rank-order the children's work and examine their achievements in relation to each other on different kinds of tasks and different aspects of tasks.

At *class* level in the Exploratory Phase, I was aiming to find out if there were differences in response to different tasks, different ways of presenting the same task to children of different ages (reported in Section 3.3.2). In the Structured Phase, I was looking specifically to compare the achievements of the Focus Class, who received the Programme, and the Comparison Class, who did not (reported in Sections 5.5 - 5.9).

From these research activities, the *primary data* collected were predominately children's drawings produced during Design and Technology lessons, along with final products or photographs of them. This was turned into *secondary data* by quantifying the qualitative judgements made about the drawings. In the Exploratory Phase, the classification system based on these secondary data (which became more detailed and refined across time) were derived from observations made about the drawings. In the Structured Phase, the secondary data were based on a series of continua derived from a holistic view of children's design drawing capability that developed as a result of reflection on the analysis process during that phase (Section 5.4).

Qualitative data was used to support the quantified analysis (photographs, comments made by children about their work, etc.) in order to create as rounded a picture as possible of how the children were using drawing to support their design thinking. For both phases, databases and spreadsheets were used to collate and manipulate the data. My capability with these computer tools increased dramatically across the course of the research and contributed to the very different way in which the quantified data was handled in each phase.

2.5.3 Telling the Story

The realisation of the close parallels between research and design led me to consider the way in which knowledge is created and that the format in which such knowledge is disseminated should reflect the epistemology. Specifically, the way I presented my research should reflect the way the research proceeded and reflect my understanding of the way in which the two processes are parallel.

"It may suffice to say that where we stand about research, how we see our place in that endeavour, translates to how we write and to the form we select in that writing." Ely (1997:158)

Consideration of how the story of my research would be told related to how this might reflect its content. In the tradition of the Bauhaus, form and function needed to be successfully wedded. If I was to tell the story "as it was" then the way in which the story was framed should enhance the telling and reflect the interplay between *hands-on* and *plan-ahead* which seems to characterise both research and design.

I had a problem with terminology. I wanted to reflect this two-stage research process in the way that I wrote the thesis, *"To tell it like it was."* but I needed a better pair of terms for these two phases. Hence they came to be called the Exploratory Phase and the Structured Phase. These seemed a reasonable choice. At first I was exploring the field: reading, thinking, trying things out in classrooms. Once I reached my watershed, then the work was structured by the Programme: its creation, delivery and analysis. That the analysis instrument was still evolving at this stage did not feel to me to be a problem. I was remaining reflective and not falling into one of the traps I had identified with too much planning ahead, i.e. sticking rigidly to a design when it clearly needs changing.

2.6 Reflections on Section 2

This brief summary reflects upon the applicability to my research of the issues raised in each of the preceding sections of Section 2, in preparation for Section 3, which begins the account of my research activity.

Research Paradigms (Section2.2)

In attempting to clarify issues related to the multiple paradigms of educational research, I discovered I was unable to find a niche in which I felt happy, mainly because (as I was to realise once the process was complete) I had effectively changed paradigms part-way through the research. The Exploratory Phase was dedicated to finding out *what is*; whereas in the Structured Phase I was committed to discovering *what I could improve.*

The hybrid nature of what I was doing, in terms of educational paradigms, meant that, although I did not fit in any one, I found myself using ways of understanding and methodologies from each tradition. Concerns that the interpretive tradition required the participant observer to be a non-participant remained, as did those regarding the way in which action research could be coupled to a deficit model of professional development and personal target setting. I needed a new category for people like me. "Practitioner researcher" sufficed.

Epistemological Metaphors (Section 2.3)

As the teacher, I could not claim detachment from my subjects in the way that both classic empiricist and interpretive traditions required. Indeed, I was directly responsible for their learning. Even in the Exploratory Phase, when I was investigating what the children knew, with no attempt to change it, I was most often the person planning, delivering and assessing the results of the activity. In the Structured Phase there was the additional problem of comparing what I was doing with what someone else was doing, on the basis that I believed I knew something about children designing that others did not.

My understanding of issues of validity and reliability evolved over time. Initially my concern was for consistency so that I felt sure that I was comparing like with like in order to depict as accurately as possible what children could achieve at different ages and in response to different kinds of Design and Technology activities. As I moved into the Structured Phase, I was more aware of the need for the involvement of others in order to validate both the analysis instrument and the evaluation of children's work based on it.

Choosing a Methodology(Section2.4)

My awareness of the parallels between my own research processes and those of the children I was researching in design situations, led me to typify two approaches: hands-on or plan-ahead.

In the Exploratory Phase, I was frequently engaged in "hands on" activity. I would read about someone else's research or have reflected on an idea I had tried already and then go and try out something new. For example, on one occasion Richard Kimbell and I discussed a fan-making activity I had tried with a Year 3 class and, in response to his suggestions, I said "I'll go and try that next week." My constant access to potential research subjects enabled me to be reactive to new ideas and suggestions.

There was, however, a definite watershed at the time that I sat down and wrote the Programme for the Structured Phase, when I realised that I had to make all my decisions up-front and that once the Programme had started I could not go off and try other things. Having, therefore, identified these two distinct phases to the research, I wanted the way that the thesis was written to reflect that.

Research Organisation (Section 2.5)

I came close, at one point, to discarding the Exploratory Phase as a pre-amble to "real" research and had to personally reclaim it as part of my research process. As I moved into the Structured Phase, I felt that all I had was a hypothesis and that now I was setting out to conduct the research. I felt almost petulant in insisting that those early fumbling had value. And yet, the Structured Phase only came into being to "prove" the conclusions which I came to in the Exploratory Phase, from which the hypotheses were informed by observation, experimentation, reading and reflection. That the Structured Phase did considerably more than support my previous observations will be seen in Sections 5.5-5.10. New knowledge ensued from a new way of doing things. Both paradigms and methodologies were equally important.

I felt strongly that the thesis structure should reflect both halves, giving recognition to the Exploratory Phase as "real" research, despite the fumbling nature of my activity, and acknowledging too that, despite the planning ahead, the exploration and development of ideas was continuing throughout the Structured Phase.

SECTION 3 : Exploratory Phase

3.1 Introduction to Section 3	71
3.2 Literature review: Drawing as product or tool	73
3.3 Observational Constructs	99
3.4 Theoretical Constructs12	23
3.5 Reflections on Section 31	36

3.1 Introduction to Section 3

Constructing a framework for understanding children's design drawings has been, for me, an evolving process with its earliest beginnings in 1996 with a small scale study conducted as part of my Master's Degree with the Open University. Joining Goldsmiths College to begin my M.Phil/Ph.D. in January 1998 was part of a seamless process, not just in terms of time, but also in the development of my understanding of children's use of drawing for designing.

Section 3 of the thesis comprises the account of the Exploratory Phase of the research:

- a review of the literature on children's design drawing (Section 3.2)
- personal observations of children's design drawings (Section 3.3)
- theoretical constructs resulting from reading and personal observations (Section 3.4)

Literature review: Drawing as product or tool (Section 3.2)

The greater part of this work was conducted during the Exploratory Phase of the research. Having continuous access to opportunities to try out ideas with children, this section of the thesis is not just a theoretical review of research conducted prior to my own, but a reflective, reflexive account of seeking answers in the literature for my classroom observations and seeking confirmation (or otherwise) in the classroom for things I had read in the literature. Thus my informal classroom observations are interwoven with my reflections on what I have read.

Observational Constructs (Section 3.3)

My early attempts to investigate how young children use drawing to support design thinking was essentially exploratory. I knew from the literature that this had not been researched deeply and so I had no model to guide me. I was convinced of the importance of *grounded research:* that theoretical constructs come from sorting and sifting the data to see what emerges, especially in the early stages of investigating a new field.

Therefore, I simply went and taught Design and Technology lessons and collated the results. I asked other teachers to conduct the same lessons and collected their results. I tried out different ways of presenting the same task to same age children; same presentation to different aged children; different presenter, same script; different presenter, same task, own way of presenting. I taught Design and Technology to a Year 3 class weekly for a year. I taught my own Year 1 classes continuously throughout the duration of the research. I collected evidence from Design and Technology happening anywhere and everywhere throughout the school. I

even invaded the Year 4 Art Club. There was a constant to-ing and fro-ing between reading and observation and my continuous access to classes of children made this easy to do.

As a result of this work, I amassed a collection of design drawings from over 350 individuals across three years, many of whom contributed drawings at various stages in their school career. My analysis of these design drawings led to the development of a classification system which became called "Drawing Types" that moved on with me into the Structured Phase (see Section 3.3.5).

Theoretical Constructs (Section 3.4)

The theoretical constructs resulted from this reflective interaction between reading and observing during the Exploratory Phase and asking the questions :

- What is the role of drawing in designing and how does it work?
- What pre-requisite understandings / concepts are necessary for children to usefully access the genre?

I came to believe that if I could answer these two questions, I could perhaps devise a scheme of work or programme of lessons which would enable children to use drawing more effectively to model design ideas. Putting together a taxonomy of design drawing enabled me to identify key concepts for answering these questions.

Transition from exploration to structured investigation

The transition from this Exploratory Phase to the Structured Phase came as I began to feel that I had sufficient understanding of what was involved in drawing for designing and that I could explain this to young children with sufficient clarity to make a difference to their ability to use drawing for designing.

Part of this process was to identify of "Drawing Types" (which emerged as categories from my sorting and sifting of my collection of drawings), to give me the beginnings of a framework against which to evaluate success. The other part was the understandings created from reflective interaction with the literature, which enabled me to piece together a theoretical framework that became part of the underpinning of the more structured investigation that was to follow.
3.2 Literature Review: Drawing as Product or Tool

Since the major part of my literature search was conducted at the same time as the observations of children in classrooms, it was part of the ongoing weaving of my developing understanding with my observations of young children. I was questioning the literature and comparing it to observations, as well as making observations and looking in the literature for other people's answers. The way this literature review is written reflects that interactive, iterative process.

3.2.1 Using Drawing as a Tool for Thought

"A tool... is an object which has first been chosen, then adapted to function as part of a skill"

(Hodgkin, 1985: 39)

My first questions about the topic were simple ones:

- Why model ideas by drawing?
- Why plan on paper?

Put simply: why draw anyway?

In the normal course of their lives, adults do not always draw out what they are going to make, so why should it be seen as important to teach this to children?

Welch & Lim (1999) assert :

"Sketching is a form of thinking and the fundamental language of design" (p. 136)

quoting Tipping (1983) as saying that sketching ability may be:

"the single most important factor in developing any general design ability." (p. 45)

If this view is correct, then to teach this skill to children is of vital importance in developing their design capability. Welch & Lim (1999) list the role of sketching as: clarifying the task and understanding the design problem, encouraging the designer to play with ideas without the time-consuming and costly experimenting with real materials, facilitating evaluation of ideas and identifying and evaluating possible problems, and "because sketching is a language" enabling communication with the self and others. However, they appeared to discover that children rarely choose to draw their ideas ahead of engagement with materials.

The importance of developing pupils' metacognition is of importance far beyond the confines of the Design and Technology curriculum but the subject is ideally placed to develop these skills (Kimbell & Perry, 2001) . Straddling the arts and the sciences, using techniques, skills and knowledge from both sides of the traditional divide, Design and Technology can foster creativity, reflexive thinking and develop "thinking about thinking". Part of that process requires thoughts to become visible in order to be subject to appraisal by self and others. Drawing frequently performs this function.

It would follow from this, therefore, that children should be taught to record their ideas by drawing but as Smith (2001) said in his DATA lecture, more research and curriculum development is needed into children's design drawing The following review of the literature shows how little there has been, especially with regard to younger children.

A sensible place to start, therefore, seemed to be to look at the practice of adult design professionals and see how their practice might illuminate the process that might be taught to children.

Phrases used for the process of using drawing to support design thinking included "thinking with a pencil", "thinking through my fingers" (quoted by Kimbell, 1998) who also used the phrase "portfolio as mental print out" which represents the designer's train of thought as the project unfolds. Oxman (2000) calls this process "design emergence".

Goldschmidt (1994), observed that architects frequently use drawing, not just as a means of symbolic representation, but to actively generate ideas. This equates to my distinction between drawing as *product of* thought and drawing as *tool for* thought. Although many drawings, whether by designers or others, merely record thoughts already in the head but Goldschmidt was interested in visual thinking and tied this to "*fabrication* of visual displays" (her italics; p.162), via reference to Wittgenstein's concept of imaging as doing rather than receiving. This idea of *imaging as doing* also fitted into the concept that I was building of what I meant by *drawing as a tool for thought*.

The importance of distinguishing between *modelling for* and *modelling of* was stressed by Roberts (1992). A *model of* something, whether it be a flow diagram or a medieval castle made of wood, is a completed product. It is the end, not part, of the designing process. *Modelling for* is future orientated. Design drawing is a tool to support thinking about future action. To treat drawing as *modelling of* is to place closure on the procedure.

Likewise, Garner (1993) discussing the relationship between drawing and visual literacy, portrayed drawing as a tool for understanding, provoking responses and an important way of playing with and communicating ideas throughout the process of designing. However, his paper described the activities of adult design professionals and so this might not necessarily apply to young children.

We can only manipulate what we can name or compare and that our understanding of our environment is limited by our ability to name and make comparisons (Williams, 1985), who asks whether a thing be understood if un-named, let alone communicated, and if scientists missed potentials in their discoveries if they had no names for their observations and that the wider the vocabulary the better quality, more precise the communication and greater the understanding.

One could ask how far this would also apply to visual literacy. We cannot make what we cannot visualise, but it might also be true that we cannot design what we cannot model. The development of drawing has a role to play, therefore, in the development of design vocabulary:

"Visualisation, as expressed through the use of drawings, is almost essential in designing physical things well ...And in design it is not until one backs it up with the visual mode that he can see whether he is fooling himself or not."

(Adams 1974: 73)

Although Adams' discussion was of the practice of design professionals, my observations suggest that amongst young children, drawing ahead of task can appropriately support idea generation but not detailed construction techniques. Scrivener (1998) terms these first freehand externalisations a designer produces "idea sketches". Harrison (1978) applies the phrase "Letting the tool do the job" to the use of drawing as a tool and a springboard for future designing. Once the mental image is put on paper, the material image begins to do the job, as each objectification becomes the springboard for the next thought.

"More often than not" children are asked to "draw me one then make it." observed Constable (1994) but they do not see this drawing as "an essential vehicle for channelling thoughts" and the drawing appears to be more of a hindrance to the real task of making. It is this interaction between drawing and thinking which professional adult designers find so easy which is so hard-won for the child.

I wanted to know:

• to what extent young children could utilise their drawings as a *tool for* designing.

 whether there were essential skill(s) they need to have mastered in order to do this effectively.

However, I was aware that it is pointless to ask children to do something if it does not make "human sense" (Donaldson, 1978) in relation to the task in hand or to the children's perception of the task. Although drawing is part of the design repertoire, it is not a necessary part of something called "The Design Process" which is appropriate to every "Design and Make" task. Drawing should only be used where appropriate to the task and to the age/stage of the children, which might be different for different sorts of tasks, or even different sorts of children.

Therefore, I needed to clarify:

- what kind of tasks
- what age / stage / sorts of children

My informal observations of young children led me to believe that, below age 6, children do not generally understand that a drawing can be for planning a future activity but by age 8 or 9 they can. I wanted to know what happens in between, whether it is simply that they do not know that they could, or whether there is a barrier to understanding. If such a barrier exists (perhaps maturity or development of specific cognitive or manual skills), then I wanted to know what it was and whether it could be overcome by teaching.

If these issues could be answered, then those answers could enable teachers to help children to use drawing for designing more effectively (or, perhaps, at an earlier age) to develop design ideas and which might have knock-on effects in other areas of learning.

3.2.2 Children Drawing

That drawing is almost an intuitive act on the part of young children has been assumed by many writers, for example:

"(Children) draw before they can write, and they associate their drawings with thought even before they can draw anything recognisable." (Silver 1978; 51)

Young children want to draw. Making marks on paper seems to be satisfying in its own right. However, to deduce from this that designing on paper is something they can do without specific teaching makes assumptions about children's understanding of the potential of drawing for designing. Although ideas are flowing whilst a child is drawing a picture, the completion of the drawing concludes the event. Understanding that a drawing could be a plan to make something else or that drawing could be used to develop ideas about something that might be made, is not the same as drawing to portray either real or imaginary events or characters.

The literature on children's design drawings is, I discovered, slim. Historically speaking, the overwhelmingly greater part of research into children's drawings is into drawing as "finished product", rather than into drawing for intent to make. As Outterside's (1993) comments, of the three major forms of modelling (iconic, symbolic and analogue) identified by Baynes (1989), only the development of the first has been extensively documented.

3.2.2a Drawings as Maps of Development

Children's graphic development was assumed to be *teleological*, leading towards the accurate representation of reality as observed; Lowenfeld (1947) and Kellogg (1959) being influential in portraying children's drawings this way. Both were based on a stage-theory view of child development and of drawing as recording observation accurately on paper. Goodenough's (1926) Draw-a-man test had set the stage for the idea that children's artistic ability and intelligence could be measured by such "camera shots" of the world around them, which was assumed to be biologically determined and purposefully creative. Such plotting of graphic development seems highly dubious on three counts.

Eirstly, this view of the nature of art discounted the non-representational adult artistic output within Western culture (some of the best known works of Picasso, Mondrian and Klee would all have failed the tests). The view of artist as recorder of visual observation that was applied to children's output, was not in tune with mid-twentieth century art theory, which viewed the artist as translator of an inner image into external form to communicate affectively to others, often relying on symbolism and suggestion. Klee's *A Young Lady's Adventure* (1921) uses lines, not to express observed reality, but to explore the dark and shadowy side of the "adventure" undertaken. Kandinsky combined conventional placement (for example, the triangular arrangement of many Renaissance Madonnas) with abstract shapes to express his view of the essence of art.

The mis-trust of non-representational art in society as a whole (which persisted throughout much of the twentieth century) and the continued preference given to representational art as a model to which children should aspire, has led to the marginalisation of other forms and uses of drawing that children might be encouraged to produce, some of which might have more potential for using drawing as a design tool than the "picture" (which was always seen as

finished product). As shown by seven year old Glen's "Self portrait in the style of Paul Klee" (Ex.1), children express themselves in range of drawing genres.



Ex. 1 : Yr. 3 : Glen's Klee

Egan (1996) classified three styles of drawing produced by young children (representation, narrative and patterning) suggesting that narrative drawing contributes most to design capability. I too have observed many young children who can produce almost virtuoso drawings but cannot use drawing for design, perhaps because they see drawing only as a finished product, to be made as aesthetically pleasing as possible, rather than having potential for recording of changing, developing ideas, as the example below by a Year 1 girl (Ex. 2) shows. She could produce beautifully patterned pictures but did not become adept at design drawing, even by Year 4.



Ex. 2 : Yr. 1 : Patterning

<u>Secondly</u>, in the research setting, the child was frequently asked to draw a picture for which no contextual or stylistic cues were given, but the product was assumed to be typical of the child's artistic production. This showed little understanding of drawing in context. Children learn and utilise a whole range of genres (pattern making, cartoons, diagrams, sections, exploded diagrams, scientific diagrams) within both representational and non-representational art.

Assumptions were made about the development of children's drawings with little consultation with the children. It was assumed that the child implicitly understood the adult's agenda. The child produced what they believed the adult wanted, in a genre which might perhaps be called *drawing a picture to please an adult* : no rude bits, blood or guts. From my teacher's knowledge of young children, these abound, especially in boys' drawings (including Glen), from quite a young age and their absence reflects the power ratio of child to unknown adult.

Einally, the assumption that drawing *comes from the inner life of the child*, untainted by social pressure, became popular as early 20th century artists became enamoured with Primitivism and child art, which led to the paralleling of children's art with so-called "primitive cultures". This denied the social context in which children of all cultures develop their mark-making. Eng (1931) (for example) paralleled the development of child art with the development of primitive and folk art (Palaeolithic & Bushmen in alternate breaths).

Although to a more modern viewpoint this seems naively Eurocentric, the underlying assumptions of development towards attaining Western conventions of pictorial representation led to equating this skill with intelligence. Lowenfeld(1947) listed six stages of development in child art, which he saw as natural aspects of human development through which the child must pass, extrapolated from observations of American children striving to please the teacher and conform to their perceptions of the norms of their own society. Part of the problem lay in not understanding the cultural determination of children's drawings and artistic output.

Behind such schema are several unstated assumptions:

 that children's drawings are rooted in the nature of human cognition and not in the nature of society;

• that the social functions of drawings (and art in general) in western society is typical of all human societies and has not changed across time;

• that the attainment of point perspective was the pinnacle of artistic endeavour and the one towards which graphic education should aspire.

It is only recently, as a result of the National Curriculum for Art and Design, that young children have been shown examples of non-representational art as a regular part of their artistic experience in school and produce such work as Glen's Klee.

These issues remain un-addressed in more recent work on children's drawings. Matthews' (1999) detailed study of the artistic output of a small group of young children (focusing particularly on that of his son Ben) remains firmly within the confines of the genre of drawings as products. There are no plans-to-make drawings, design drawings, yet Ben, with his level of graphic competence must have produced design sketches, scientific diagrams, maps, flow-charts and a whole range of other drawing types that do not feature in the book at all. Matthews' viewpoint is of drawing as product. Even where it is a process towards resolution of inner image, the process stops with the completion of the drawing.

Wilson (1992) asserted that children from an Egyptian village with few outside influences had a very restricted visual vocabulary and showed far less diversity in their drawings than those of western culture, where the influence of television and especially cartoons influenced drawing style. Wilson did not, however, appear aware of Islamic disapproval of representions of humans and animals. Nor did he compare the Egyptian children's abstract pattern-making (a highly developed Islamic art form) with Western children's output in this genre.

For designing as well as picturing, due credence needs to be given to the cultural influences and expectations of children's drawings. Children function as part of a multi-layered socio-cultural system (Rogoff, 1996) as learners, peers and teachers; as transmitters of culture to each other as well as receivers of culture from older children and adults. They are actively seeking competence and identity within the overlapping and interacting cultural milieux they inhabit.

The booklet "Start Drawing" produced by The Campaign for Drawing (2002) demonstrates a greater diversity of uses for drawing (Perception, Communication and Manipulation) including a section on solving problems, which are predominantly Design and Technology, which comes close to my view of using drawing as thinking tool:

" reflexive oscillation" between impulse, ideas and mark, receiving feedback from the marks appearing on the page, which prompt further thought and mark-making. Usually the drawing is one of a series, where ideas are explored, repeated, refined, practised, worked over, discarded, combined, where alternatives are sought and alternative possibilities explored." (op. cit.: 2)

Children's drawing skills, like all aspects of child development, are far more fluid and adaptable than was once thought but if children are to access and utilise the genre of design drawing, then it needs to be demonstrated and taught along with other genres, otherwise we are relying on children's intuition to grasp the teacher's expectations heuristically.

Children's lack of experience or knowledge of a range of drawing genres may mean that their response to demands for a design drawing is a demonstration of their prowess in drawing rather than to produce a range of ideas sketches of something to fit the demands of the task. Children are less adept than adults at adjusting the level of specificity according to perceived purpose or context (Van Sommers, 1984). I have observed Year 4 girls drawing fashion details such as pockets and zips on designs for a puppet, which could not be made with the materials in the tray in front of them on the table.

3.2.2b Canonical Drawing of Objects

4-5 year olds often produce canonical drawings, i.e. they draw a cup with a handle even if shown a cup with its handle facing away from them. My observations of Year 1 children suggest that children want to produce the "best view" of the object: children sitting with a side view of a teddy turned it towards them so they got front view, to the howls of protest from the child seated with the best view of teddy's front. Drawing a teddy means a front view of teddy. Drawing his side view does not satisfy the child's idea of what constitutes a teddy.

There are two sorts of children who do not fall into this trap. The more sophisticated see it as a party trick - *Can you put this strange view onto paper*? and children who have conceptual difficulties. For example (Ex. 3), 6 year old Andy's back view of my rocking chair (below). Andy had a delayed language development.





The accuracy of his drawing is probably due to the limitations of his verbal labelling skills. Such children record the shapes and angles of the scene that they see before them, without verbally labelling the objects. Nadia, studied by Selfe (1983) ceased to produce her exceptional drawings once she started to speak. Canonicality, I would argue, is not an aberration of children's drawings which they need to outgrow, but essential to designing and thinking: the inner visual image which accompanies a word label.

By Year 2 Andy was already losing his advantage to other more verbally competent children, although his class teacher tried to foster and encourage his artistic skills. By Year 3, he could no longer record his design ideas except in the most basic way: a single drawing of the object to be made. Perhaps he had reached the canonical stage, which other children achieved two years earlier and perhaps I had captured the progress of a child whose drawing ability enabled him to reveal the pre-canonical stage. When he left the school, aged 9, he was still a long way behind his peers with regard to literacy skills and had not become a good designer. This would suggest, perhaps, that internalisation and inner labelling, the connection between language and perception is vital for manipulating any symbol system, including drawing.

As part of their normal development, children come to understand that symbols are cut free from the concrete experience, that names of things are not part of the thing and can be used and manipulated in their own right. This must be true too with graphic thought, a time when children can begin to use drawing as a tool for thought, development and communication because the drawing has taken on an abstract reality of its own, no longer tied to the particular thought or object that inspired it, but can be changed, the symbols used, redrawn or crossed out because they are only a staging post towards an end. This involves conscious exploitation and manipulation of the symbol system; graphic metacognition.

Drawing made from observation shows the defects and idiosyncrasies of the particular whereas the canonical drawing reflects the generalised inner image and can be used as a basis for designing. Arnheim (1995), describes the process of creative designing as an interaction of arguments and moves. By making a sketch, the designer provides the mental image with an optical image that loses the disadvantages of the mental image whilst being still fluid enough to be re-interpreted by the eye and manipulated by the brain.

3.2.2c Analogue Drawing

Children's first graphic products are analogue: they look nothing like what they represent, although they have meaning for the child (Outterside, 1993). Analogue drawing is an abstract system of representation so, perhaps, children who produce such drawings are possibly more adept at using their drawings as a tool for modelling ideas than those who see drawing as a representation of observations.

Andy produced the analogue drawing (Ex. 4) at 4 years old, soon after his entry to school. He told an incoherent story as he drew, concerning a day out at the beach with his parents and brother. It was stored as "evidence" in the Special Needs Co-ordinator's filing cabinet and given to him to take home some 18 months later. He became very excited and recounted to me the day on the beach and pointed out what the various parts of the drawing represented.



Ex. 4; Yr.1: Delayed Language Development (b)

"It's a map. We went on a boat on it. There's big waves, like that big. I saw people writing with fingers on sand, saying "Liam" - and that. (pointing to writing on right) And a boat nearly sank.. And then we did, got fishing out. We got some worms and we catched some fish and got our fishing rods and put them in the sea and we fished."

To me, there were three noteworthy aspects to this episode: firstly that he could read it so long afterwards (he was still unable to read print). Second that he called it a map; to an adult it looks like so much scribble, but his reading of it gives it total sense. Thirdly, Andy was one of the few children who could use drawing as a recording tool for his design thoughts by the end of Year 1.

Andy's design for "Wiggly Worm's House" (Ex. 5, overleaf, annotated by me) took account of materials and construction detail:



Ex. 5; Yr.1: Delayed Language Development (c)

As a non-writer, Andy was still using picturing to place-mark stories by the end of Year 1 and this helped his designing since he thought visually as first choice. However, as indicated above, this prowess did not continue as he got older.

3.2.3 Children's Planning Drawings

"Drawing is not habitually demonstrated [in the classroom] as a useful tool for organising and representing ideas.... our education system rarely offers examples of adults modelling drawing as a tool for thinking."

(Anning, 1993; 38)

This is probably because most teachers don't! Teachers have had to use writing as the medium of expression in order to become teachers, even those whose main subject is art or design and who might naturally think more visually. Many teachers feel a sense of inadequacy about their drawing skills. This militate against the use of drawing to organise information or communicate ideas in classrooms.

3.2.3a Drawing for Developing Ideas

Matthews (1999) claims that children's repetition and reworking of images and themes over a period of days or weeks constitutes an editing and re-drafting of ideas. This might be harnessed to encourage children to maintain a planning note-book.

The spontaneous use of drawing to draft a picture becomes more common as children get older. At the start of Year 1, children draw an outline shape in colour and then fill it in but they

soon begin to draw in lead pencil and then colour this. They have realised that the lead pencil gives them the freedom to rub out and change parts of the drawing with which they are unsatisfied. By Year 3 the pencil line drawing is perfected before painting commences.

I quickly discovered the need to give children a black biro so that I did not lose valuable evidence. The frequent occurrence of multiple drawings of the same object amongst Year 2-3 children was, I realised, due to their desire to improve their drawing. There was no progress of ideas, only adjustments of lines. They regard first attempts as "wrong" (i.e. wrongly drawn) rather than as stages towards a final outcome.

It would seem that the desire to get reality correctly onto paper inhibits designing at ages 7-9. Time and effort is wasted both on correcting unnecessary details and also on drawing to an inappropriate level of detail, even in collage work where it would be more sensible to cover the whole background first or simply copy the plan without drawing it on the background at all.

3.2.3b Sketch Books and Process Diaries

Robinson (1995) suggested that children should make their own sketch books to record their drawings which would then become a cumulative record of knowledge, thus encouraging children to mimic the practice of real artists. This could encourage children to see drawing as a developmental tool rather than the simpler "drawing as picture". This seemed a good idea so we tried it in Year 4 Art Club but many children did not bring them each week and had lost them by part-way through the year. Although my own observations during the Exploratory Phase indicated that children as young as 8 years could begin to use drawing as a developmental tool, it was not something they did spontaneously or would necessarily continue outside the classroom.

Children frequently are unaware that adults do not produce works of art at first attempt, whether written, visual or musical. Hammond (1997) comments on the importance of young children knowing that redrafting their writing is not just acceptable but vital. She speaks of the earliest attempts at conveying story on paper as "place-holding". The child can retell the story based on the marks made on the paper. This seemed to me to be a pre-design skill as well as a pre-writing skill. If they can place-mark for story-telling, I wondered, why not place-mark ideas for designing?

Vygotsky (1978) asserted that children's early drawing is a form of graphic speech, a precursor to written language. Harste, Woodward & Burke (1984) concluded that there exists a reciprocal

relationship between young children's drawings and writing. At this early stage, graphics and emergent writing skills combine and support each other. When one ceases to be useful, the child moves to an alternative communication system to placehold meaning (Harste et al). As children become more competent writers, drawing takes a secondary role, from which it is difficult to resurrect it. But what makes them become recorders of frozen reality rather than fluid users of graphic expression? I think the answer is social pressure and a desire to copy the finished products they see around them. They so rarely see drawing being used to support thinking that they are unaware of the possibility.

The Process Diary (Rogers & Clare, 1994), recorded in a variety of media: words, photographs as well as drawings, enabled children to record and think about the process of design through recording significant moments in the development of their project. They can then reflect on the decisions made at various stages. Rogers & Clare identified reflectiveness as the most important aspect of this work and making children metacognitively aware of their own thought processes to inform their future thinking.

3.2.4 Implications of the National Curriculum for Design and Technology

It came as a surprise to me, some 8 years after the publication of the first National Curriculum Orders for Design and Technology, to discover that there had been no research into young children's use of drawing for design purposes prior to its publication. Yet in 1992, 7 year olds were required to be tested in their competence in these uncharted skills. Any literature relating to drawing for design by children younger than 10 appeared to be an investigation into why the children seemed unable to fulfil the requirements.

3.2.4a "Extrapolation downwards"

My previous Headteacher read this phrase in one of the early circulars about the National Curriculum. It was nowhere more true than in the Design and Technology Order. Had no one heard of Piaget and that little children see the world differently from older ones? It was as if the whole body of child development understanding in which Infant teachers were grounded had been wiped away with one stroke. The serendipity of playing with materials which became something exciting in small hands had now been swept aside by identifying needs, generating ideas, recording possible solutions and making mock-ups of them to be evaluated before they fall apart before the next lesson. Medway (1992) called it the "academicization" of practical

activities: the doing is only allowable within the overall context of the communicating, evaluating and other intellectual skills.

Articles and books that gave advice to teachers about how to develop designing skills frequently used the blanket word "children" with no indication of age. For example, Ritchie (1995) :

"The ability of children to develop their ideas through drawing needs to be developed throughout the curriculum from an early age - so that "drawing an idea" becomes second nature"

(op.cit.: 82)

with the warning that:

"As children get older....they and their teachers can put too much emphasis on finished drawing quality." (ibid.)

What is this "early age"? Four? Ten? and when do these children "get older"? seven? eleven? or sixteen?

It has been assumed that the practices of design professionals were of educational application, and frequently the age of the children is unspecified, for example:

"Graphic representations, in the forms of drawings, graphs and charts, are used to convey the design technology process and its results. The child grapples with the difficulty of transferring an idea to a two-dimensional format. Sketching freezes elusive ideas and provides a format for mental rehearsal as the child mulls over possibilities ... Just as a designer or engineer works with multiple drafts, so the child... the project will evolve, possibly through several drafts... a final two-dimensional rendering will capture the resulting changes in the original design."

(Dunn & Larson, 1990: 34)

The "design" side of the Design and Technology Orders appeared to be heavily dominated by making explicit things that had previously either been assumed to happen inside children's heads, or had not been considered in relation to young children's craft work (as it was previously known) at all. The newness of the subject in the school curriculum, together with muddled thinking about the role of drawing for design and the capabilities of children at this age has produced mixed messages about both.

Teachers, let alone children, did not assume drawing to be a procedural tool prior to the National Curriculum. There could be none of Bruner's "scaffolding" since teachers had little perception of the intricacies of the structure. Neither, it would it seem, did the Curriculum writers. Yet this unfortunate document became a yardstick against which children's capability were now judged. There appeared to be mis-matches between what teachers knew about small children, the demands of the document and what researchers knew about designing.

McCormick et al (1993) considered the linear model of the design process promoted via the National Curriculum not only to be a poor model of how people solve problems but that its imposition on children leads to lack of ownership of the task.

Baynes (1998) criticised the effort expended in (and since) the National Curriculum on getting children to produce a design drawing and then carry it out. He feels that by giving drawing a role which it cannot fulfil, the National Curriculum has undermined the role of drawing in designing. He cites a bird house drawn by an 8 year old. The product is very different to the drawing, because she carried on designing as she made it "as adult designers do" (says Baynes). There is a mis-match between the National Curriculum model of the design process and the practice of real designers.

3.2.4b Observations of Young Children Drawing for Designing

Not surprisingly, many early papers and articles focused on what young children could not do, and whether or not what they could do was what the National Curriculum writers had in mind all along. The lack of research into young children's design skills prior to the publication of the document made its instructions a cause for anger or despair among teachers and frustration for the researchers who now entered the field. The following line by Constable (1994) :

"Although there is nothing in the Order which states that children need to approach the ATs in a linear fashion - heaven forbid..." (op. cit.: 13)

prompted an instant reaction which I wrote in the margin: "Why weren't we told?"

She continues:

"I would like to reassure KS1 teachers that this articulation of ideas need not necessarily be on paper..."

(ibid.)

The reassurance did not reach us either, since:

"Th(e) inappropriate use of drawing is partly due to the unfortunate linear approach to D&T which is encouraged by numbering the (old) attainment targets 1 to 4, thus suggesting that the complete design needs to be "generated" before making can take place."

(op cit. p.10)

Constable also highlighted teachers' hazy perception of the role of evaluation, which, again due to the numbered list of Attainment Targets, was seen as the last lesson in the scheme of work. The iterative nature of designing had not been conveyed to teachers and hence was not being conveyed to children. Garvey and Quinlan (1997) observed that Year 2 children regarded their design drawings as "wrong" if the teacher suggested improvements.

The examples of children's work shown in papers and articles such as Samuel (1991) showed that Year 3 children could use drawing to record design ideas but, as the article makes clear, few teachers felt confident as to how to encourage children to do so. Chalkley & Shield (1996) reported Year 5 children being unclear how drawing could support designing.

In Key Stage 1, similar observations were made. Stables (1992) observed Year 1 children completely ignoring their drawings of a "home for a spider" once they began making them. Anning (1993) described two 6 year olds who thought they were giving their drawing to the hamster as a present; they did not see the drawing as a sketch of something they would make. My own observations as a teacher confirmed these. Children at this age appear to see a drawing as a product. Their agenda for the use of drawing is mastering the genre of conveying 3-dimensional objects in a 2-dimensional medium whilst also creating pictorial balance on the paper and aesthetic pleasure in the colours and lines.

I agreed with Egan's (1995) comment:

"certain approaches or intentions while drawing would lead more naturally into design modelling than others." (op. cit.: 9)

But which, and how? She concludes:

"Design drawing.. is drawing to explain rather than to depict, and as such has more links with the narrative.... It is possible that concentrating on the pictorial reinforces the concept of the drawing as an end in itself." (op. cit.: 14) Constable (1994) described the design drawing as needing to be a simple line drawing, probably annotated, with views from different angles and smaller detailed parts drawn separately, but few Year 2 children chose to use the techniques even though they could do them. My observations in 1998 suggested that few children of this age could satisfy Constable's criteria, except in a very structured task, such as that shown in Ex. 6 in response to a mental manipulation task conducted with two classes from each of Years 1-4 (246 children in total).

I showed the children a cardboard box and told them to imagine they were going to make it into a car. They were told to draw side, top, front and back views separately and indicate the extra materials they would need. The children were not asked to make the car and so the detailed plans may not have been perceived as blueprints for making.



Ex. 6; Yr.2: Role of Task Structuring (a)

Shown below (Ex. 7) is the same child's design for making a suitcase for a toy panda, an unstructured task. Her "design" was typical of Year 2 children. The drawing (on the left) shows two attempts to perfect a simple outline drawing. rather than develop ideas about suitcases. The prototype (on the right, mounted on black) shows little continuity of thinking from the drawing.



Ex. 7; Yr.2: Role of Task Structuring (b)

From observations of adults and children, Welch & Lim (1999) concluded that since neither opted to develop their ideas through drawing, that drawing was not a necessary part of the design process. However, drawing does improve design efficiency (consideration of options and possibilities prior to engaging with materials of construction) and that some of their subjects used drawing to clarify the terms of the task and to establish mutual understandings of a possible solution. As an educator, I would argue that what people do naturally is not necessarily the criteria on which decisions should be made about what children should be taught. With the hindsight of conducting my own research, I would query whether the isometric drawing techniques taught to Welch & Lim 's subjects imparted sufficient understanding of the *role* of drawing as a designing tool, or whether this was simply a learnt technique for which the subjects saw no application in the design task.

Mantell (1999) suggests using designing techniques familiar to teachers in other curriculum areas (mapping, listing and flowcharts), referring to the work of Wray and Lewis (1997), which could enable children to use graphics and text interactively. My results in the Structured Phase showed a much greater level of annotation of drawings than in the Exploratory Phase and I felt that the introduction of non-fiction texts in the National Literacy Strategy for Key Stage 1 had given children appropriate techniques to use in a Design and Technology situation.

Egan (2001) encouraged children to record their design ideas through drawing, so that the ideas could be viewed and discussed by others, thus enabling communication and clarification of ideas discussed in small groups. After some teacher-led practical tasks to provide knowledge and understanding of materials and processes, the children chose to re-image their ideas and produced more focused drawings of what they intended to make. Again these were older children than those whom I taught, as were those studied by Ching & Hulsbosch (2001).

The "Enriching Literacy through Design and Technology" project conducted in the education Action Zone in Middlesborough demonstrated enhancements in children's ability to record and develop design ideas. In a paper delivered to the Centre for Research into Primary Technology's Third Conference (2001), Rogers & Stables reported that Literacy and Design and Technology had proved to be mutually enhancing. The activity that they used for Year 2 post-input assessment was to become incorporated into my Structured Phase (see Section 4.3.2 & 5.3.1d). The findings of this study (improvements in generating and developing ideas, addressing the task and identifying user needs) were also to surface within my own results (Section 5.7).

Design drawing in the primary school, remains, however, an under-researched area. In his DATA lecture, John Smith raised the following questions:

SECTION 3 : Exploratory Phase

If sketching is an important modelling aid for designing then surely more research and curriculum development should be undertaken in how to develop pupils' and students' sketching skills which provide opportunities for ambiguity and hence an opportunity for creating new ideas? What age should learning sketching techniques be started and to what depth? Do pupils understand that one reason for sketching when designing is to assist in the generation of more ideas through the ambiguity of the sketches and the juxtaposition of ideas?

(Smith, 2001: 8-9)

He provided no answers and quoted no research into this area which might suggest appropriate answers. One assumes that he considered these questions still to be open and un-researched.

3.2.5 Design as internal process

Manual skills apart, my observations suggested that there appears to be a mental block on the idea of using a drawing as a blueprint for making which is not satisfactorily bridged at least until age 8. Before this age most children see a drawing as a product, a picture. It has no bearing on the making task for which they have been told that it is the plan. The potential of the analogy between drawing and making needs to become conscious in order to see that a particular drawing can equate a possible answer, and only one among many.

But little children do not play with their drawings in this way. They do not want to have several tries on one sheet. They want to produce a picture, including what the weather was like behind. Ex. 8 shows another Yr. 1 "Wiggly Worm's House" from the same class as Andy, cited in Section 3.2.2b & c.



Ex. 8: Yr. 1 Narrative genre

This left me with a list of questions for which the literature on children's drawing and designing provided few answers:

• Is children's drawing ability too rudimentary, insufficiently developed to record what they want to make, particularly when combined with their lack of fluency in the medium in which they are going to make the final product?

- Or is it a lack of awareness of the potential uses of drawing?
- Or is it due to the children not having realised the symbolic nature of drawing?

The problem appeared not to be whether they could do the drawing but whether they could model in one medium (the drawing) and then make a product in a different medium which matches, in its essential characteristics, the drawn model.

Year 4 children appeared to be able to use the drawing to support thinking about what to make. Ex. 9 is a Year 4 girl's development of a "Surprise Box".



Ex. 9 : Yr. 4 Design genre

The initial idea was a handbag with mice and insects inside. The word "mouse" made her think of a laptop carrycase, but what would go inside? Still on insects, she thought "worm in an apple" and partway through writing the word "water" she broke off to think "drain". I noticed that she was sitting staring into space and sat down next to her. She drew picture 3 as we talked - she was linking insects to bugs - computer bugs. What else had a double meaning? apple/Applemac, mouse, chips. She thought for only a few moments and confidently drew her final design - a plate of chips with a bug sitting on top and mice attacking from all sides. She

made it from a shoe box. It was a computer on the outside with a plate of chips, bug and mouse on the inside.

Children in Key Stage 1 frequently do not understand that there can be a connection between what they can draw and what they can make with some other material. They see no analogy between the drawing and the future product. They will conform to the teacher's instructions - make a drawing, make a model - but the drawing does not inform their making unless they are constantly supervised and kept on task. The children may be able to draw, they may be able to see that someone else's drawing is a plan for action, but the conceptual difficulty is seeing that *their* drawing could become a blueprint for *their own* actions.

If a child has not grasped the idea that drawing is symbolic and can become context-free, then they are unlikely to be able to use drawing as a design tool. They may be able to make realistic models and even be good at drawing (in the usual representational sense) but until they see that ideas can be developed by drawing, or from the drawing, then this mode of designing is closed to them. This is part of a larger issue concerning the development of symbolic manipulation. Piaget & Inhelder (1969) saw the emergence of symbolic thought at age three in the development of symbolic play: the "as-if' quality of play emerges and children pretend the materials are something which they are not (Craft, 1997)

Donaldson (1992) reported finding that 6 year olds did not appear to accept the limits imposed by the problem or the information she had provided, adding other characters to the situation ad hoc. She observed that:

"They did not have a clear conception of *this problem* - this one and no other - which they could hold on to and use in deciding when the problem had been successfully dealt with, so that thinking about it should cease." (p.135)

Perhaps this is one reason why, for young children, the drawing does not necessarily relate to future action. They do not perceive the drawing as in any way providing *the* design solution. It is *one* design solution, the model which they make later is another.

The ability to handle both aspects of designing (addressing the problem whilst imaging possibilities) were to become emergent themes in the Structured Phase of the research.

3.2.6 Making for play

In his discussion of modelling, Archer (1992) observed that :

"...the human mind is predisposed to seek similarities within and between its accumulating conceptions, and to assign these to categories... (plus) the predisposition assign symbols to represent conceptions, categories and relations. The use of symbols permits abstraction in inner thought, and the externalisation of thought for recording or communication purposes."

(op. cit.: 5)

This predisposition towards use of analogy and symbolism emerges early in life. The symbolism which accompanies the fantasy role playing of small children is vital to the abilities which underlie design as manipulation of symbols. Those children who have rich imaginative play are better at visualisation and hence design tasks. Those who do not play so imaginatively do not manipulate symbols, make one thing stand for another, and so cannot image solutions. This might imply that there is a strong link between design capability and the use of found objects in play.

A baby picks up a plastic bottle, for example, to *explore its properties*, the pre-schooler will make it *be* something, the infant school child might make it *into* something and finally, in middle childhood, the bottle will be a *component chosen for its physical properties*. This goes beyond, Bruner's (1978) enactive, iconic and symbolic representation in play into the "making for play" activities in which children of school age engage, e.g. making clothes for dolls, constructing hides and dens, which are closer to the activities of Design and Technology.

In this "making for play" children are prepared to do a fair amount of pretending. A cardboard box become a den one day and a train the next. Total realism is not the aim, but enough to satisfy the requirements of the play. These are tools for play, part of the tool box for a game - a fantasy world which mirrors reality but occupies a different plane of existence. When the game ends, the tools are discarded. For adults, design is needs driven, whereas children are playing. The object forms part of fantasy world and the solution does not need to fulfil the criteria in a realistic way. A hat for teddy to go home in the rain can be made from ordinary paper. Properties can be re-assigning as a part of play.

Winnicott (1971) concluded that human play arises from the capacity to make bridges imaginatively between our own inner reality and the external. Children combine objects from outer reality with ideas from inner reality to create a "dream potential". This seems to me to be

SECTION 3 : Exploratory Phase

the same skill as designers use. Even something as simple as a concertina-ed strip of paper with a string attached and called an Inch-worm can become a spring-board for play for small children and stimulate the imagination further. Bailey (1971) closely allied to the child's scientific search, seen first in an infant's grasping at objects, feelings of pleasure in exchanges with significant adults and the need to explore :

"....the basis of creativity is need, a need to know and express feeling, a need to come to terms with what is already known and what is only partly known." (p. cit.: 118)

Aided by the acquisition of language, this leads to planning, comparing, categorising, which in turn become the foundations of creativity. Craft (1997) perceives "possibility thinking" as the core element in creativity; involving play, asking questions and motivation.

Language-mediated play, as the culmination of representational play, is essential for school success. Schomburg (1999) conjectures that one of the reasons that children with good representational skills do so well in school is that their opportunities for play are not cut off. These continuing inner fantasies are crucial to taking designerly playing into a more mature form of designing. This would accord with my observations of Andy (cited in Section 3.2.2) who had not achieved Schomburg's language-mediated play until at least two years after his peers.

The relationship between children's play and the adult trait of playfulness was explored by Lieberman (1977), who compared the results of a set of "Divergent Thinking Tasks" to a "playfulness" scale for teacher assessment, on which she found correlation both for small children and adolescents, concluding that playfulness continues beyond childhood to become an adult personality trait. One of her "Divergent Thinking Tasks" (product improvement) was a classically Design and Technology task: "How could you make this doll/toy dog more interesting to play with?"

Coghill (1989) also observed making and playing to be early indicators of design capability. Like playing, designing involves the use of cognitive maps, ideas and representations to consider the means towards a "not yet fully perceived end, making meaning through action or imagined action." Like Bailey (1971), Coghill perceived curiosity as a spur to meaning-making, often embedded in, or projected into, aspects of physical reality which act as holders for thought and action, so they can be worked on or changed.

Yet drawings seem to have no such function in young children's minds. The children in my Year 1 class did not use their drawings as a tool for planning what might be made with other materials. Once a drawing was finished, the thinking had finished. It was not a springboard for something else. Kress (1994) described the way a drawing changed its function once a child had cut it out, to become an object for play, something I observed often in my Year 1 classroom. For example, one day a boy brought in a "ghost" he had made with his childminder. It was a piece of tissue paper, draped over a second piece rolled into a ball, tied round with a piece of thread to make a neck, dangling from another length of thread attached to the top of the head. By 10 o'clock half the class had one and were playing with them and conducting conversations through them.

In their playing and their making such young children use their perceptions of the similarities between things, the analogies which they perceive all around them, sometimes by serendipity, sometimes by intent, using and combining them playfully and creatively to design a self-propelling, shared world. In this, they are acting in exactly the same way as adult designers. Hence the term employed by Baynes (1989): designerly play.

Smith (1992) asserts that pattern-recognition prevents the imagination running out of control and confusing reality and fantasy. He sees this as the mechanism as the brake on fantasy running out of control. This pattern-recognition is socially learnt and practised in play. Children with a rich fantasy life are often the most adept at creative and design tasks. They have learnt to exploit mental fluidity, yet they have a strong sense of what would really work.

3.2.7 Reflections on Section 3.2

By the end of the Exploratory Phase, I had searched the literature on children drawing and made connections with my informal observation of children. I was also reading in related areas: problem-solving, creativity, language development. I was becoming interested in generic cognitive functions which expressed themselves in the capacity to model ideas and record these in drawings. Observations that I made in classrooms gave me a perspective on my reading, and vice versa. For example, trying out the canonical cups with about 100 children for whom I had design drawings gave me a good feel for this measure of inner imaging among 5-6 year olds.

There seemed to be few studies of consequence of children's drawing for any purpose other than picturing or of children using drawing for planning to make. What was required was a longitudinal study of a convincingly large cohort of children, so that an audit of skills at different ages could be compiled. My own explorations (detailed in the following Section 3.3) seemed to be among the only ones being conducted.

SECTION 3 : Exploratory Phase

I was becoming convinced, via my colleague Sue Hammond's M.A. study on emergent writing, that the place-holding of ideas through a mixture of graphics and text-symbols, which appeared as a staging-post towards literacy, had potential for recording of design ideas. If children just past their fifth birthday (as was Shelley who produced Ex. 10) could use drawing and text so interactively to tell me about playing in her garden, then surely, I reasoned, this skill can be harnessed for developing design ideas at quite a young age. However, most research into children's drawings continues to be from the viewpoint of drawing as art, as the finished product and not as a plan for future action. It was to be the literature on cognitive development, language and emergent literacy that would give me the most useful insights into how young children might be enabled to access the genre of design drawing and use it as a tool for thought.



Ex. 10: Yr. R Place-holding meaning

3.3 Observational Constructs

From January 1996 to July 2000 I was exploring and trying out ideas with different groups of children across the school as well as collecting drawings from other teachers' lessons. Those activities were to form the basis of my understanding of how young children's use of design drawing evolved over time. For my findings to be reliable, it was important that the drawings I considered were not just from lessons conducted by myself or that were presented in just one particular way. I wanted to know what children across the 5-9 age range could actually do, in terms of using drawing to support designing in a range of contexts. I developed a classification system, which was to be carried forward into the Structured Phase as part of my analysis instrument (Section 3.3.5).

As Design and Technology Co-ordinator, my role was to act as subject leader, to write the subject policy documents and the long-term and medium term plans and liaise with colleagues from other local schools and with Paul Shallcross, Kent County Advisory Teacher for Design and Technology. As part of the school's quality assurance procedures, I was required to work alongside colleagues in a supportive role and to observe their Design and Technology lessons, from which I gained insights into children's capabilities at different ages. Some of my colleagues' lesson ideas became part of my Structured Phase Programme. An initial survey was conducted in Summer 1997: a Punch & Judy Theatre as a series of three lessons, to develop colleagues' confidence in teaching Design and Technology whilst, at the same time, enabling me to observe children designing.

Throughout my research, observations, feedback and discussion with colleagues were important in developing my understanding of what young children could do. Sue Hammond, the Literacy Co-ordinator, was conducting research into children's emergent literacy skills and Mrs. R., the Art Co-ordinator, began a Masters Degree in Expressive Arts. This meant that we were constantly sharing ideas, discussing issues of methodology or cognitive theory and suggesting useful texts to read. The Special Needs Co-ordinator was often part of these discussions, with her expert knowledge of children's cognitive development.

The Kent model for teacher appraisal was by peer review and so, for example, the Deputy Headteacher observed one of the Stan96 lessons as part of this process. Her transcript of my introduction of the activity became the basis of my script for introducing the activity across the school in 1998. Throughout the Exploratory Phase, therefore, my activities were under constant peer review and scrutiny. This not only enhanced the validity and reliability of the research but gave me confidence in the evaluative judgements that I was making.

3.3.1 Mapping the Field

My understanding of methodological issues at this stage was rudimentary. I knew that for the sake of validity, I needed to keep as many variables as possible the same in each task presentation. Yet, for the sake of reliability, it would be better to have several colleagues involved. However, in order to maintain a measure of consistency, all activities were introduced by me, even though, once working, many children were supervised by teaching colleagues or other adults.

I believed that it was unlikely that children would use drawing in a very different way during a single lesson that I was delivering to that to which they were accustomed with their own teacher. This was borne out across the three years period of the Exploratory Phase, as I began to identify varying levels of understanding of designing among my teaching colleagues through taking their classes for these occasional sessions.

For all activities the children were provided with sheets of white "kitchen paper" for drawing ideas, throughout both Exploratory and Structured Phases. In the Exploratory Phase, I wanted to discover how children used drawing to support designing and felt that providing a pre-printed worksheet might guide them in a particular direction and that I would not get a true sense of what they would choose to do unaided.

"Stan96"

Mrs.R., the Art Co-ordinator, had read "Flat Stanley" by Jeff Brown to Year 1 and suggested that children working in pairs to design and make a puppet to go in an A5 envelope as a suitable task for the small-scale research project which was part of my M.A. course. I conducted the activity with all children in Year 1 (96 children), some of whom worked independently in peer pairs, some helped by an older child (Yr.4), some by teenagers and some by parent volunteer helpers. All pairings with older helpers were audio recorded, as were an equal number of peer pairs. In general, the Year 1 children relied on the older partner to tell them what to do. The parents allowed the children least lee-way in making their own independent design decisions. whereas my daughter became extremely skilled ("Do you want this group guided or unguided, Mum?").

The Year 1 children had little idea why they were planning on paper before making the puppet. "Why are we doing this twice?" became a question which, for me, would not go away. With hindsight, they should, perhaps, each have made their own puppet, even if they had collaborated on the planning. Some pairs solved the problem by one child decorating the design sheet (Ex. 11) whilst the other made the puppet.

They paid scant attention to the construction materials whilst drawing their plans. Some children took a selection of materials to their table and then chose from amongst this selection once they had drawn the figure on the card and some drew the figure onto card first and then went back to the resource table and chose their materials. Whatever they were doing on the design sheet, it was not planning construction. Feedback from my teenage helpers suggested that guiding the Year 1 children towards using drawing for planning what they would make was difficult since the children had no concept of what was being asked of them. One of my Year 1 colleagues, Miss S., who had previously taught in Year 4, felt that even these older children had limited understanding of design drawing.



Ex. 11 : Yr. 1 : Flat Stan (1996)

"Insects97"

As a result of Stan96, I believed that I had established that Year 1s did not have much understanding of planning and designing, but what about Year 4? I began with the expectation that Year 4s would be more likely to articulate their understanding of design when they were in an organisational role. Sue Hammond selected some "good designers" from her Year 4 class to each work with two Yr.1 children.

In order to study the design aspect separate from the making, a task was needed for which this separation would not be too contrived: putting together a kit for a younger child to complete unaided later. Thus although the input of the Year 4 children terminated at the planning stage,

they had to think ahead to what the Year 1 child would need for task completion. Working with the Year 1s who would be assembling the kit would, hopefully, enable the Year 4s to realise what was needed.

The children showed no inclination to use drawing as a tool for thinking and seemed unaware of drawing as a way of developing understanding of the problem, exploring or communicating ideas, or supporting their, frequently rich, discussions. They were relying on developing shared meanings with their Year 1 clients through talk. This also appeared to get in the way of the production of a template for creating a model based on the design. They did not see either the drawing or the template as stages towards final product that the Year 1 children would make but as discrete products in themselves to be produced because they were on the instructions. The template was frequently treated by the Year 1s as a base to decorate, not a pattern to draw round, as also were some initial drawings.

The following year (1998), I repeated the task but kept the children apart once they had decided on the insect to be made. The Year 4s developed the kits in their own classroom, coming to me each day for feedback and collecting materials. The results seemed remarkably similar to the previous year's and so the presence of Year 1s for the entire process had not hampered the Year 4s in 1997. Since the Year 4s were not present for the making, they were not aware and, therefore, unable to trouble-shoot problems with the kit that they provided. We had one Year 1 boy in tears because he could not make the ladybird because he could not find the red cloth. He found it hard to accept that older children could have made such a simple mistake as forgetting to put it in the envelope.

In hindsight, I think that part of the problem was that in 1996-7 Design and Technology was still a relatively undeveloped subject area in school. I was feeling my way and design drawing was not being taught consistently in school. Some teachers gave their classes sketch books to record ideas, but most did not, including the class used for the Insects97 task. Over the years, I have noticed that design skills generally across the school have improved as teachers' understanding of Design and Technology has increased and children's capabilities have been developed as they pass through the school. The examples overleaf (Ex.12) show the difference between the use of drawing for designing between the Year 4 children involved in Insects97 and the Year 4 Art Club 2000 children, who were involved as Year 1s in Stan96. Both are representative of average capability. The 2000 example shows clearly that this girl was thinking of a 3-dinemsional product as she drew, the 1997 girl simply drew an insect and the way she used the drawing led the Year 1 child to treat it as a flat shape to decorate.

Ex. 12 : Yr. 4 : Insects (1997 & 2000):





3.3.2 Distinguishing Drawing Types

I began my investigations for my M.Phil./Ph.D. by building on the small scale studies and the understandings I had gleaned from them, together with the associated reading from the M.A. course. These materials were heavily weighted towards Secondary Schools and my own endeavours had led to finding only a handful of texts, even in 1997. No wonder teachers lacked expertise. Therefore, as well as reading as widely as possible around the subject, I began to conduct my own experiments within school. These were to lead to the identification and naming of different types of drawing used by children in response to design tasks.

3.3.2a Through "Stan98"

The aim of Stan98 was to try Flat Stan again and add a follow-on task "Round Stan" in as many ways as possible across as many classes as possible to find out what worked best and what commonalties existed regardless of presentation. 279 children made Flat Stan, of whom 197 also made Round Stan (Table 1).

	Flat Stan				Round Stan			
Year Group	1	2	3	4	1	2	3	4
No. of Classes	2	3	3	2	2	1	3	2
No. of Children	39	83	94	43	28	28	83	58

Table 1: Research Subjects : Stan98

I wanted to try as many formats and ways of presenting the task as possible in order to find out what worked best, as well as how well the children understood the role of drawing for designing. I did not question children closely about their intentions since I was looking quite specifically at what children recorded on paper. I attempted to devise ways of handling quantitative data using databases that evolved as I looked at an ever-increasing number of factors as my ideas about how young children use drawing for designing developed.

Initially, beginning with the Flat Stan task, I looked at:

- Satisfying task criteria (Was Flat Stan flat? Did he fit in the envelope?)
- The use made of the drawing (Did they pre-draw? Was there more than one attempt at drawing?)
- Recording materials (Did they draw the puppet as if made from the materials supplied?)
- Relationship of product to the drawing (Did it look like what they had drawn? Had they used the materials provided or had they simply coloured a cardboard cut-out?)

Inclusion of the Round Stan task, and the desire to compare the two, led to extensions and refinement of some of these categories:

• The use of the drawing (For children who did more than one drawing, was there a range of ideas or had they simply tried to improve the drawing of a single idea?)

• Recording materials (Had they indicated intended colours? Had they indicated intended materials?)

- Appropriate level of detail (too much was a bad as not enough!)
- Labelling (arrows or lines linking words with drawing)

Throughout the analysis process I used a simple 1=Yes, 0=No for the database field entries with summation fields for more global skills (satisfying task criteria, use of drawing, recording o materials and relating making to drawing) which attempted to answer the question "To what extent?" with regard to capability in each skill area.

These skills summation fields were then summed to make a Total Score for the task. Table 2 shows the Form view of the Flat Stan section of the database by which all these early trials were analysed (see Appendix N: AN1.1 & AN1.2).

FLAT STAN:		
F.CRITERIA: 1	F FITS: 1	FLAT: 0
F.DRAW SCORE: <u>3</u>	F PRE-DR: <u>1</u> F VIEWPOINTS:	F MULT DR: <u>1</u> F MULT DES: <u>1</u> F DR DET: <u>0</u> F EXPANDED: F LABEL:
F.RECORD MATS: _1	F COLOURS: 0 F INSTRUCTS:	F DRAW MATS: 0F WRITE MATS: 1 F EQUIP:
F.MAKE SCORE: 2	FMATS: 1 FR	ESEMB: 0 F EXACTLY: 0 F MK DETAIL: 1

Table 2 : Database, Form view (the "F" Prefix indicates that this is the Flat Stan section)

This Year 2 child has a total score for Flat Stan of 7, composed of:

- Criteria score of 1 : the puppet fitted the envelope but was not flat.
- Draw score of 3 : they pre-drew their idea on scrap paper; there was more than one item on the paper; these represented different ideas.
- Recording materials score of 1: they had written the intended materials to be used.

• Make score : they used the same materials as indicated on the drawing; the level of detail in the finished puppet was appropriate to the materials provided (i.e. they did not make a detailed drawing to the card and then cover over the drawing with fabric or sticky paper, which was common amongst Year 2 children) What this child did not do that other (probably older) children did:

- · Draw in sufficient detail to convey design intentions.
- Indicate colour or draw the puppet as if made from the materials provided.
- The puppet did not resemble the drawing.

This scoring technique was, admittedly, crude. However, the findings fed into my growing understanding of children's design capabilities. Appendices N includes the full form view of the database and spreadsheet for all these early whole school activities. The Total Scores for all the Stan tasks (Chart 1) produced an approximation of a normal distribution curve, which felt reassuring, since this data represented the work of children across the 5-9 age range, exposed to a variety of task presentation methods, and supervised by different adults.



Chart 2 shows the results for each year group for Flat Stan only, since there were less children involved in Round Stan and a greater variation of task presentation. The Stan96 children were the Year 3 cohort in this study in 1998.



By distinguishing between the number and types of drawn items on the paper, I began to get a sense of different uses of drawing to support designing. At this stage, I was considering this to be a linear progression, since findings such as those represented by Chart 2 seemed to suggest a growth in understanding with age. This over-simplification was not to be resolved until the Structured Phase (Section 5.4)

Key Stage 1 children (and less able older ones) tended to produce a *single drawing* quite quickly of what they wanted to make (Ex. 13). If they were thinking about the realities of making the product as they drew, these were not indicated in the drawing. The conversations between children centred on clarifying the task more often than on developing ideas about a product. Some children cut out their drawing and stuck it onto a lolly stick "handle", despite the unsuitability of the flimsy kitchen paper for a final product. I tried hard to convey the message that this was scrap paper for playing with ideas.



Ex.13 : Yr.2 : Flat Stan (a)

There was a distinction between *Multiple Drawings* and *Multiple Designs* (Exx.14 & 15 overleaf). Some children re-drew their one idea more neatly whereas others recorded several different ideas. This also seemed to be age and ability related. A schematic of the finished puppet demonstrating its parts and / or materials appeared to be a distinguishing characteristic between drawing a picture and designing a product.



Multi-drawing Ex.14 : Yr.2 : Flat Stan (b)



Multi-design Ex.15 : Yr.2 : Flat Stan (c)

More Year 3 children drew their Round Stan figure as if made from the materials or wrote down the materials than for Flat Stan. I was unsure whether this was because Flat Stan was seen as a decorated card copy of the drawing and so they had not thought about materials until "decorating" stage or whether it was due to practice effect. Older children, especially Year 4 girls, put in too much detail e.g. zip and pocket details, indicating that they had not understood the drawing as planning to make *with the materials provided*. Some re-drew their Flat Stan figure onto card in great detail and then pasted fabric on top.

Drawing as a means of designing appeared generally to be beyond the capability of Year 1, even if supported by an adult, although several ways were tried. They were successful planners if they told an adult or a tape recorder what they were going to make, but they could not *draw* and make. They were not using drawing to image ideas to be made in another medium.

The importance of seeing as well as hearing was re-inforced through comparing the responses of two parallel Year 2 classes for whom the only difference in experience was in seeing the illustrations in the Flat Stanley book rather than simply having the story read to them (Chart 3).


This was sparked by fellow researcher saying to me, "If I was a six year old and you showed me the pictures in the book, I would just copy them." This proved not to be the case. Children who saw the pictures produced a greater range of ideas for their puppet than those who did not, possibly because seeing the pictures enabled them to understand that the puppet had to be flat and so they used drawing to play with ideas for making a flat figure. Those who did not see the pictures tended to produce single stereotypical human figure drawings.

The importance of seeing as well as hearing was re-inforced through considering the drawings of the third Yr.2 class, who were shown preliminary drawings done by two teenagers (who were able to be compared with a Yr.4 class who had the same Round Stan introduction). Both classes produced clearer, more designerly drawings than their respective peers, leading me to conclude that it was important to show children what I meant by a design drawing. This Yr.2 class also made a "Stan Buggy" in which I asked them to choose the main components from the recycled materials box before starting to draw. They produced a single, well-focused drawing and then made what they had drawn, implying that providing the major components might limit the range of ideas but that this aids thinking towards construction.

Although making a puppet is a common D&T task, a host of questions arose:

- How truly "Design and Technology" was it?
- There was no client or user to consider. Did it matter?
- Did the children view it as D&T or Art? Did that matter? Is the design process the same?
- Did the fact that the product was a human figure make a difference? Would there have been more varied ideas for a different product, less likely to produce a stereotypical response? How many children just drew their current representation of a person?
- For Flat Stan, was there a problem with it being flat and made of card? Many children were simply copying their drawing onto card and then decorating it.

I had purposely chosen something simple to make so I could concentrate on the drawing but I worried that there was not enough scope in the task, not enough of a problem for the children to tussle with. I wanted to do a trial run with something else, which had a client, could not be confused with art, not prompt stereotypical responses and have some constructional issues with which to grapple.

I had identified three kinds of drawing that children produced in response to design tasks: *Single Draw, Multi-draw* and *Multi-design*, which implied: a single quick sketch, more than one drawing but subsequent drawings are simply neater versions of the first, and recording of several different responses to the task requirements.

3.3.2b Through "Pandy98-9"

In July 98 I tried another idea with my own Year 1 class and a Year 4 class - Pandy's suitcase. The children were to design and make a suitcase for my toy panda to take on holiday. The handle of the suitcase must fit over his arm, and not be too big that he could not carry it nor too small that he could not get his belongings inside. The modelling (working out with the paper) was to include making sure that what they made fitted by trying it on the toy panda. This concept seemed even harder for Year 1 to understand than "draw before you make".

Year 1 mostly drew the suitcase and items of clothing, sunglasses, bucket and spade etc. on one piece of paper and then cut them all out. Many first attempts were drawings of suitcases, with little regard to the size of the panda, which were cut out, and declared finished, even after other children who had a better idea of what to do had made considerable progress towards success. What seemed to be stumping them was the idea that I wanted a real suitcase that Pandy could put things in, not a picture of a suitcase. *"That's the suitcase"* they kept saying, showing me their cut out drawing. *"How do you get the clothes in?"* I kept asking. *"In there"* they replied, pointing at the suitcase. Some of them folded up the edges of the suitcase, and I demonstrated the cut out clothes falling off of it. They were sent to look at my handbag and school bag. The overall solution was to cut out two flat, suitcase shaped pieces and staple them together. Those who still did not understand what they were doing put the staples through the middle rather than round the sides.

For example, Avril cut out two suitcase shapes, complete with handles, stapled them together and cut large holes in the body of the suitcase:

Me to Avril: "Why's it got a hole in it?" Avril: "So he can hold it" Me: "Won't his things fall out?" Avril: "No" Me: "Have you ever seen a suitcase with a large hole in the middle?" Avril: "Mmm" (uncertain).

What came across was that they had not thought in terms of making a real thing that worked (suitcase) or that they were planning real things to go in it (holiday items). They were making things that could be pretended to be the real thing. "Does it fit over his arm?" could be managed; they tried it on. "Does it hold things?" could not because they had not realised it was meant to. Cutting out pictures of the holiday items that were minuscule in comparison to the size of the panda, were flat and made of paper did not bother them at all. They were willing to

pretend that the panda could wear them. My protestations that it could not was greeted with amusement, puzzlement, confusion, even dismay that I was not as delighted with their results as they were.

This experience led, inevitably, to more questions:

- At what age do children start to make real things for their dolls and action figures?
- Why do they not do it earlier?

And some tentative answers:

• It is not to do with motor skill.

• It appears to be to do with their pre-occupation with play and fantasy. Their imagination would make up the shortfall on reality of the items they had made.

• This would appear to stand on its head the idea of children progressing towards symbolic representation. What they had made were symbolic representations of suitcases, *rather than* suitcases.

In contrast, Year 4 had little difficulty with the task. Making it fit the panda was solved by measuring. They were engaged in the reality of solving the task. No one cut out pictures as final products, although some appeared to be drawing what they wanted to make in order to clarify their ideas about how it would look. Some appeared to be engaging in the ritual of draw and write instructions. They did not exhibit the spontaneous *"I know how to do this"* that had been in evidence in Year 1 (although such confidence was largely mis-placed).

Secondly, they showed much greater flexibility in choice of technique, looked at and assessed each others progress and made subsequent adjustments to their own. They were confident in finding materials they needed (e.g. sellotape, treasury tags) without asking me. They also asked for more paper as and when they needed it. They were less inhibited about starting again if it went wrong.

The task appeared to be sufficiently challenging whilst being within their capability. I had wondered about its suitability, especially for Year 4 boys. However, no one quibbled about making such an item. Perhaps this was aided by the initial discussion about keeping such toys from early childhood (the toy panda is genuinely mine). I was pleased with the results. It was a step closer to producing a real artefact than Stan had been. So Pandy's suitcase was conducted across whole age range, 2 classes per year group in the following Spring term.

I knew that we were into the grey area between fantasy and reality. I was asking the children to design a real suitcase to fit a real toy panda in which he could put a real plastic mac but

pretend he could go on holiday. I decided that I would resist telling them it needed two sides but would prompt them to think about it with the question "How will he put the mac inside?" or "Can he put his mac inside?" I did not want them failing the design task simply because they had misperceived the reality/fantasy divide inherent in the task. I would, therefore, positively intervene where a child told me they had finished when they had only a cut out picture of a suitcase and not allow them to pretend that they could put the mac inside. If this made no sense to them I would leave it, but I thought it unfair to discriminate against those who had consulted me (or had even been sent to ask) while others gleaned the information from observing successful peers.

Observations in process:

Year 2 - No rulers in evidence here, except to rule straight lines. Quite a lot of "How will he put his mac inside the suitcase?" prompts needed.



Ex.16 : Yr.2 : Pandy's Suitcase (a)

The design sheet in Ex.16 has become rather crumpled and smudged due to lying on the table whilst the suitcase (right) was made. The suitcase was made in pale yellow card (hence mounting on black) but was single-sided. Effectively, the child had perfected their drawing on the design sheet, copied this to yellow card and cut it out, without realising that a real suitcase to hold Pandy's mac was required. Imaging of real suitcase had ceased once perfecting the drawing had taken over as the priority in the child's mind.

Year 3 - Many spontaneously got rulers and passed the mac around to measure it. Their design drawings were done to these measurements. I told those who had done this to put the measurements on the drawings so that I would know they had done so. Some children, however, put more effort into designing different pictures and logos for the outside of the suitcase than how it would be made.

Year 4 - Comparable level of sophistication to pilot study class : for example, "*Can I draw a net for the suitcase as my design?*" but also, as with Year 4 girls on the Stan task, their interest in fashion details often overrode the reality of the materials provided for making the item.

Despite drawing just one idea (after a couple of false starts) the child whose work is shown in Ex. 17 made an exact copy of the shape and logo as drawn on the design sheet. Construction was not recorded in the drawing. The suitcase has been taken apart so that each half can be seen separately.



Ex.17 : Yr.4 : Pandy's Suitcase (b)

Reflections:

• More sophisticated drawing techniques (multiple viewpoints, labelled diagrams and expansion to show small detail) were used by Year 3 and Year 4 children than had been in evidence in the Stan task. Pandy had triggered more designerly thinking.

• These older children were grappling with the reality of the task (e.g. spontaneous use of a ruler) whereas younger (Key Stage 1) children were sincerely pretending.

Appendix N: AN1.3 & 1.4 show the database through which evaluations of the children's drawings were quantified. As a result of this analysis, I needed a new label (beyond *Multi-design*) to describe the work of children who had used their drawings to develop a design solution: *Progressive*. They had begun with a basic idea (or a range of ideas), sketched this and then used drawing to record how they would make it, perhaps through one or more labelled diagrams or by drawing it from several viewpoints. This appeared to me to be a major stage in their understanding of drawing for designing. They were imaging a real object they wished to make and using drawing to work out how to make it.

3.3.4 The Outliers

Interactive Drawings

Among the many smaller samples of children's drawings collected across a range of less structured settings, the experience of helping to run an after-school Art Club with Year 4 children added most to my understanding.

One of the projects was a "Surprise Box" for which the children had to make the inside of a box into a surprising interpretation of the theme of the outside of the box. The children's use of drawing as they grappled with this play of ideas and double meanings led to identification of a new Drawing Type: *Interactive*. A few of the children were evaluating and combining several drawings to create new solutions to the design problem. These Interactive drawings had several drawings on a page, a clear thread of thinking could be determined across them and they combined features of previous drawings in new ones. The children were becoming genuinely evaluative about what they had drawn. An example of one of these is shown in Section 3.2.5 as Ex.9.

The following year, we did a "Surprise Tube" for which the children had to make a card tube into an object whose contents were surprising, a sort of Jack-in-the-box (Ex.18, overleaf). This time there were more children who were beginning to use drawing in an interactive way, suggesting that the standards across the school were going up.



Ex.18 : Yr.4 : Surprise Tube

Non-designers

There were, however, children from every age group, regardless of task or its means of presentation, who did not use the planning sheet for planning. They either cut out the drawing of Flat Stan and pasted it onto card or stuck the puppet's stick directly onto the cut-out drawing. There were some who used the planning sheet to draw Pandy going on holiday carrying his suitcase. They were oblivious to the different working methods of other children around them.

Providing all children with paper for planning at the start of the session, obliged them to record something before making and I labelled these as *Picture*. The difficulty in knowing if a single simple drawing of a human figure was a picture or if it was intended as a design for a puppet led me to realise that puppets (and specifically Stan) were not suitable as an assessment task.

3.3.5 Classification of Children's Design Drawings into "Drawing Types"

Throughout the Exploratory Phase, the features of the children's design drawings were recorded on databases, which enabled the classification of the drawings into the following "Drawing Types". All examples are taken from Stan activities.

The Picture

The child sees the drawing as an end in itself, rather than future-planning. The child may includes features of narrative or representational drawing which are inappropriate to the genre of design drawing.



Ex. 19 : Drawing Type : Picture (a)

The child is not addressing design problems and client needs, they are drawing a picture which relates to the subject or problem. The drawing is perceived as a product, a completed activity, which does not cascade into the making process. Therefore, the drawing may either be abandoned completely and something entirely different be made, or the picture is decorated to make a collage of the subject instead of making a product.



Ex. 20 : Drawing Type : Picture (b)

Single-draw

The drawing is seen as a record of an idea which might be made, to show the teacher before going and making it or something like it. The genre of design drawing, an object disembedded from its background or context, has been grasped but the drawing is not used to develop design ideas. Once allowed to handle the materials, the drawing is frequently forgotten, although copying it exactly without any subsequent development or modification is equally common. There is no record of constructional issues having been considered.

Ex. 21 : Drawing Type : Single-draw





Progress in understanding of the purpose of drawing for design then seemed to take one of two alternative paths, which I called Multi-draw and Multi-design.

At this stage of the research I was uncertain as to the relationship between Multi-draw, Multi-design and Progressive. Multi-draw appeared to be more closely related to Single-draw and I wondered if the children were simply re-drawing their idea more neatly, so that it looked better; improving the drawing rather than enhancing the idea. I was uncertain whether to classify drawings that were essentially the same except for surface decorations as Multi-draw or Multi-design, or how much different a second drawing needed to be to count as progressive. <u>Multi-draw</u> - The child seeks to perfect their drawing of a single idea by redrawing several times rather than using drawing to develop and explore design ideas. There is evidence of understanding of the needs of the client, but only one real solution to the problem is recorded. Drawing is not used to explore and develop a range of design ideas. Evaluation relates to the appearance of the drawing rather than to the practicalities of construction or alternative design solutions. Surprisingly, after spending time perfecting the drawing, it does not necessarily inform the making since the child has not really seen the role of the drawing as a way of modelling real outcomes.

Ex. 22 : Drawing Type : Multi-draw





Multi-design

The child sees the role of the drawing in designing as a means of brainstorming ideas. The design sheet will be filled with different ideas, some related more closely than others. The object made may even be yet another different idea. The child has grasped the idea that the paper can be used to try out lots of ideas related to client needs and to working out solutions to the design problem, but without thinking too much about constructional issues or evaluating how any of the ideas would work out in practice. The product to be made may well be selected on the basis of "best drawing", even though it may not represent the most fruitful or practical idea.

Ex. 23 is quite a sophisticated Multi-design by a Year 4 boy, in which body parts have been drawn, almost as a mix-and-match selection. Possibly, he was planning to make the puppet fully articulated. In the final product, only the head slides up and down.

Ex. 23 : Drawing Type : Multi-design





Progressive

Although they may arrive by differing routes, all children need to reach this phase in their understanding of design drawing. This is the point at which they realise that they can use drawing to develop an idea and work out how the object will be made or fit together. Labels, verbal descriptions, expanded drawings to show small or separate details, diagrams which attempt to show different viewpoints or results of movement. The product is a realisation of the final drawing. It should be appreciated that this phase does not necessarily "follow on" from Multi-design. Children more frequently opt for one good idea and develop it into an action plan.

Progressive drawings frequently show a combination of words and graphics but the example shown here is by a dyslexic Year 3 boy, who drew the figure and then used drawing to record his ideas about how he would then make the figure. A clear design path, including the figure's pose, can be seen between the drawing and the product. It remains my all-time favourite Round Stan puppet and hung from its string in my room for a long time.

Ex. 24 : Drawing Type : Progressive





Interactive

At this point the child begins to have a conversation with the drawing. The child sees the drawing as a means to work out what will be made and how to make it. This phase can almost be seen as a combination of Multi-design and Progressive. More than one design idea is recorded, which are then thoughtfully evaluated and discarded or developed through more drawings, combining and discarding elements of several drawings. Several related ideas, styles or construction methods are considered and combined to develop a product based on this process. Evaluation occurs as part of the total process. Further ideas about previously drawn solutions may be recorded after other solutions have been developed as the child begins to combine ideas (in the example below, the comment at the top left was added last)

Ex. 25 : Drawing Type : Interactive





<u>Comment</u>

Although generally speaking progress is age-related, the examples used here are chosen to illustrate that this is not necessarily the case. The *Picture* example is by a Year 2 girl with receptive language difficulties. The *Single-draw* is by a S.E.N. Year 3 boy. The *Multi-draw* is by a precocious Year 1 boy. The *Multi-design* is by a Year 4 boy who demonstrated a high level of design capability from Year 2 onwards. The *Progressive* drawing shown (Year 3 boy) is without text but during the Exploratory Phase, this was often one of the distinguishing features between Multi-design and Progressive. The *Interactive* drawing is by Avril (the girl with whom I had the conversation about Pandy's suitcase when she was Year 1) but here she is three years later in Year 4.

Finally, Ex.26 is by a Year 4 boy in 2000 whose work shows a clear progression of ideas, with combinations of ideas from earlier drawings into the final design, expansion (of the head) to work out small details and sensible use of words (recording a design decision *"he is having glasses"*) and the colour scheme as a list. This is a long way from my comments on the 1996 Year 4s who seemed to have no more idea than the Year 1s as to why they should use drawing to develop their design ideas. The standards of design capability had risen considerably across the school in the intervening three years.



Ex. 26 : Drawing Type : Interactive: Flat Stan : Yr.4 (2000)

3.4 Theoretical Constructs

As a result of my reading and observations of children at work in Design and Technology lessons, I began to create personal theories about what was happening inside children's heads in design contexts and to pull these together into a more coherent picture. This led to the creation of two taxonomies of design drawing: one of external assessable evidence and the other of internal cognitive skills. From this process developed the major theoretical underpinning of the Structured Phase of the research: the analogical reasoning inherent in design drawing.

3.4.1 Modelling Concept Relationships

My readings and reflections were catalogued under various subject headings stored as "Thoughts files" (database with related text documents). I began to think through how these topics related to each other by creating a series of concept webs, placing different areas of interest at the apex and re-arranging the other elements in relation to each new "key word." I found this a very powerful yet simple way to develop my understanding of the way the various aspects of design cognition related to one another. There was nothing propositional about these concept webs and I was not trying to define precisely what the elements meant or how they related to each other. At this stage, I was merely playing with ideas. One of these concept webs is shown in Section 1.2.1 as Fig.7.

There are many ways that the elements could have been arranged, even with the same choice of "key word." The concept webs were a tool for thought, stimulating and generating new ideas and understandings. However, this way of representing concepts that were emerging as being important for children's designing skills enabled me to see not only the relationships between them, but also which were of more central importance. Put simply, some concepts turned up in almost all the webs whereas others hardly featured at all.

As a route to further clarifying what I already knew about young children's designing through drawing, both from reading and from personal observations within the Exploratory Phase, I attempted to construct a taxonomy of design drawing. The methodology was simple: I highlighted every statement in the "Thoughts files" directly related to children's design drawing skills and collated it. The Taxonomy headings are to be found overleaf (Table 3) and the full Taxonomy, showing sources for the headings is in Appendix A.

Table 3 : Taxonomy of Design Drawing :

PURPOSE drawing as "finished product" communicate to others/aide memoire for child

INVESTIGATING NEED perspective taking simple prediction forward planning

CONCEPT DESIGN brainstorming generation of ideas deciding what to make recording of the design stage

DETAIL DESIGN investigating production method deciding on production method recording production method sequencing production drawing with respect to materials produce product specification

TYPES OF DRAWING analogue/iconic/s ymbolic symbolic (stereotypical) manip of inner image drawing in the style of.. labelled diagram canonical drawing 2D & 3D plan drawing/viewpoints appropriate level of detail

DRAWING & MAKING make what they draw pattern development recording of finished product COGNITIVE DEVELOPMENT Age/maturity Other cognitive factors

ADULT INPUT

Nature of tasks Explanation given Amount of help given

KNOWLEDGE BASE Previous experience with materials Previous experience of designing Knowledge of techniques Visual literacy

GRAPHICACITY

Ability to draw Understand drawing as process tool Use drawing to develop thoughts & ideas Partial occlusion

MODELLING

Understand relationship between draw & make Fluency in symbolic manipulation Use drawing for solution in other medium Use metaphor & analogy Reality/fantasy Interact with the drawing

IMAGING

Make changes from drawing Imagine/record future state Imagine/record future intentions Imagine & record the possible

REL TO OTHER WAYS OF THINKING Linguistic/narrative Creativity/imagination Play Conation

Two things became immediately apparent from this work :

• the taxonomy needed separating into external, observable skills and inner, cognitive processes

• I had far more instances from my own observations than the literature search had yielded.

The field of children's design drawing had not been previously explored in any depth.

Whilst I was working on the almost mundane cataloguing task of highlighting and filling in the Taxonomy, I was simultaneously creating a Concept Web (Fig. 34) on a separate piece of paper on the table beside the computer. This was not a mechanical *"this fits here and that fits there"* process as I worked through the Thoughts files but a creative event that recorded the leaps of understanding and realisations of connections, which occurred in parallel to the task in which I was engaged. At one point I completely suspended compiling the taxonomy in order to give complete attention to the web.



Fig.34 : Design Drawing Concept Web

This was the most important result of the taxonomies in terms of moving my understanding forward. In creating the taxonomies I realised that I was more interested in what was going on in the children's heads and asking why they could or could not produce different kinds of drawing for different purposes rather than simply cataloguing the kinds of drawings used. It seemed to me that if the cognitive processes involved in using drawing for designing could be identified, then it would be possible to determine whether or not, or at what age or stage in their development, young children could access and use the genre. Before this was done (and it appeared to me not to have been) I could not see how anyone could make recommendations about what young children should be taught to do.

This concept web enabled me to identify the factors which I considered to be the most important. The relationship between drawing and making seemed paramount. By this time I had a considerable collection of children's drawings from a range of Design and Technology lessons and this seemed to be the key issue: Did they understand that they were using drawing to *plan* what they were going to make?

Many children, even in Year 1, could make a drawing of something they wanted to make or had been asked to make, but the drawing was being used simply as a recording device, as if the child was using it as a ticket to gain permission to make. It was a statement of *"I want to make one of these"* a cursory drawing which contained little real meaning in terms of representing design thinking.

The web also enabled me to place, in relation to everything else, the central questions of "What is a design drawing?" "What function does it serve for design practitioners?" and "Is this relevant to teaching small children?"

It was at this point that I began to consider the metaphorical or analogical nature of drawing, in the way that it acts as a go-between, negotiating the gap between the inner image and the outer world of materials and products. Drawing is viewed and discussed as if it is the real thing, whereas it is more like a mirage on paper.

From the start of my research I had been aware that the ability to manipulate symbols was highly pertinent. In Jan.1998, Mrs. S., our Special Needs Co-ordinator, and I were discussing this and she wondered if the ability to use drawing for designing was linked cognitively or developmentally to the ability to read beyond the words and make inferences about a text, which she used as an informal marker for "really reading" rather than simply decoding text. Her observations were that the average child achieved this during Year 2.

Was she right? Was there a second order symbolic manipulation skill that was the key to using drawing for designing that was akin to reading for meaning?

Related to the ability to appreciate analogies is the juxtaposition of the incongruous which is basic to humour, art, literature and design. Koestler (1974) coined the word "bisociation" - the ability to put two things together to create something original. Thus the question might be asked: "Does the age at which humour and ability to understand and tell jokes emerge also match the age/stage at which children can start to create designs as opposed to the serendipitous playing with materials?" This ability would then parallel the analysis/synthesis

aspects of designing. My observations of 5-6 year olds would lead me to speculate that those children with a lively, flexible mind, who appreciate humour and can "read between the lines" of a story, are able to begin to record their planning for a making task by the end of Year 1 whereas those whose world is more literalistic have far greater difficulty.

These thoughts had been there right from the start, gone underground and resurfaced several times. I now thought that I knew what this skill was: the ability to use symbols metaphorically, to use the symbol as if it were the real thing. The *analogical nature of drawing* was to become the central tenet of my understanding of what design drawing was, how it functioned and, ultimately, how the skill could be taught to children at the point at which they develop Mrs.S.'s "really reading" ability, i.e. in Year 2.

All I needed to do now was prove it!

3.4.2 The Role of Metaphor and Analogy in the Construction of Meaning

The perception of drawing for designing as involving metaphorical or analogical understanding placed me in unfamiliar territory. I began with an internet search of American university materials on-line which gave me an overview and the names of the major players in the field. The insights gained from these writers on semantics meshed with understandings gained from writers on child development, creativity and cognition, and of course, Design and Technology. This interweaving of ideas from apparently disparate fields can be seen in this account of my developing understanding of the analogical nature of design drawing and led, eventually, to what was for me a major conceptual breakthrough.

"Invention can only be done deliberately if the inventor can discern similarities between the particular result which he is ensivaging and some other results which he has seen and stored in his memory...An inventor's power to invent depends on his ability to see analogies between results..."

(Pye, 1964: 27)

Gick & Holyoak (1985) demonstrated the importance of analogy in problem solving and concluded that successful transfer of learning involves overcoming contextual barriers, perhaps by focusing attention on the abstract character of potential solutions. This insight seemed useful to me in that the ability to use drawing as a design tool hinges not just on seeing the analogy between the designing medium and the making medium and being able to transfer

seamlessly from one to the other, but on being able to manipulate one symbol system to develop ideas about something to be created in another. Perhaps I could find a way of getting children to perceive the spatial similarity between the mental image, the drawing and a real object to be made.

In his discussion of the difference between the way the brain and the computer solve analogical problems, Arnheim (1969) asserts that the brain is geared to perceiving topological features which "inform the organism of the typical character of things" (p.77), which suggests a predisposition towards the abstraction necessary for successful transfer between 2 and 3 dimensions inherent in design drawing. Arnheim concludes that analogical perception is basic to intelligent behaviour and that the topological skill which enables similarities and analogies to be made is what makes productive thinking possible.

Researching the role of analogy in the development of scientific concepts, Gentner (1982) sees models as "structure mappings" from one domain to another. But she struggles to find a suitable ordinary word which does not have other connotations and does not lead into theoretical arguments over semantics and thereby distract from the real issues:

"There is no good term for "non-literal similarity comparison". The term "metaphor" conveys an artistic or expressive non-literal comparison of a certain form; the term "model" conveys an explanatory-predictive non-literal comparison, often mathematically stated."

(op. cit.: 107)

Thus Gentner chose the term "analogy". I wanted to use the word "mapping" to indicate that we are talking relationship not semantics, then the word "structural", "analogical" or "metaphorical" could be used as a prefix to describe what kind of relationship is currently being discussed. I think that "structural mapping", "metaphorical mapping" or "analogical mapping" (depending on the precise context) sound quite good but smack of esoteric tautology. Perhaps Gentner was sensible to just use "analogy".

Gentner discusses scientific "explanatory analogies" and compares them to "expressive analogies", which are the province of the semanticist. I wanted to know if there are not also such things an *exploratory analogies* (like drawing) which are used to explore ideas, which would also, perhaps include allegory and paracosm and even day-dreaming.

Her term "structural mapping" seems to be parallel to Veale's (1999) "conceptual scaffolding" which he describes as

SECTION 3 : Exploratory Phase

"an architectural guide, or blueprint, for the assembly process, but may not constitute an element in the final edifice. That is to say, conceptual scaffolding possesses a transient existence to serve as a temporary representational purpose."

http://www.compappdcu.ie/~tonyv/papers/CogSci.ps,gz; accessed Jan.2002

Tourangeau in Miall (1982) speaks of "creating a parallel system" (p.25) which also seems to describe the process of being able to think completely in one system and come to logical conclusions about the parallel target system; to think in it, manipulate it, make parallels and juxtapositions and jump back and forth from one to the other. What Wittgenstein (1969) simply called "seeing as".

Parallel to Wittenstein's "language games" (which, in conjunction with his "seeing as", is what metaphors allow us to play) is Booth's (1978) suggestion that instead of attempting a formal definition of metaphor, "family resemblances" should be sought. I attempted to plot parallels between each of Booth's family resemblances (Table 4) and design drawing and felt that the family resemblance between the two are sufficiently strong to be persuasive:

Table 4: Booth's list of family resemblances	Applicability to design drawing
Intentionality	Design drawing begin from the intention to find a solution to a problem or opportunity
Context-dependent	Design is always rooted in a specific context
Persuasive purpose	If the client is to be shown this stage
One of many ways of expressing same idea	Other media could be used
Can be paraphrased using another metaphor	Other drawings could be made of same idea
2 things, not 2 words, are being compared	The drawing relates to other real or imagined objects
Stable: "once understood, no further act of interpretation is required."	Equally true of drawings
Local & finite: tied to this situation	The design context is frequently specific and particular
More is communicated than the words say	Design drawings frequently suggest lifestyle and marketing niche
A change in perception of the situation takes place	Design drawings objectify inner images of the intended product. Changes in understanding of the problem and its solution frequently occur as a result of drawing.

Weldon's distinction between difficulties, puzzles and problems (cited by Bruner, 1962) refers to the way a problem is solved or a discovery made when we impose a puzzle form on a difficulty to convert it into one with which we can deal. Discovery consists of knowing how to impose a workable "puzzle form" on various difficulties. Dufresne et al.(1995) observed that novices cannot use analogies to solve problems due to difficulties identifying which of the problems that they have already solved are conceptually similar to the one they are currently trying to solve.

Because the links between concepts and problem situations are bi-directional, analogies are an extremely useful problem-solving tool for experts, who, by classifying problems according to the same umbrella concepts, can translate problem situations into appropriate procedures. It would seem, therefore, that a central skill in problem-solving is to be able to link the appropriate strategy knowledge to the specific domain knowledge, which, I would assert, is by analogy with previously encountered problems. For the experts, with rich previous experience in solving like problems, the analogies are not far away. For novices, learners and children, the search is too wide to provide useful links.

For drawing for designing, it is necessary to see that



Fig. 35 : Strategy Knowledge for Design Drawing (see Fig. 24)

"I think intelligence cannot develop without content. Making new connections depends on knowing enough about something in the first place to be able to think of other things to do."

(Duckworth 1987: 14)

Knowing that by drawing it, a design problem can be solved, is an imposition of a known puzzle form in Weldon's sense. However, realising that the task is bigger or more complex than can be visualised mentally and that external support is needed, whether by doing a drawing, making a list or other place-holding device, involves a level of self-awareness or metacognition which Key Stage 1 children lack.

"If the mind cannot solve a problem by terms dictated by the situation, then it will do so in terms of some other but similar situation. Thus invention is the emergence in the mind of novelty under the control of system."

(Blanshard, 1964: 148)

Children are unaware of the limits of their visualisation skills. They think they have the answer and start to make something, leave it half done because it does not work or change it completely at a whim. By teaching children to objectify and record their mental images, visualise onto paper, we are teaching methodological efficiency for use in a whole range of contexts.

The ability to use drawing as a design tool hinges on seeing the analogy between the designing medium and the making medium, on being able to transfer seamlessly from one to the other and on being able to manipulate one symbol system to develop ideas about something to be created in another.

Arnheim (1969) called these "pictorial analogies" which

"fulfil a mediating position between the world of sensory experience and the disembodied forces underlying the objects and events of that experience."

(op cit.: 148)

In speaking of the adolescent "having discovered that art may be consciously manipulated as metaphor," Matthews (1999: 144) limits metaphor to the psychological, objectifying inner perceptions of mental states and abstract concepts. However, I believe that this is only one form which graphic metaphor takes. Design drawing requires the same skill, seeing one thing as another, but the metaphor is of a concrete reality to be constructed in another medium. Interestingly, Matthews links this stage in his son's artistic development to his playing of fantasy "Dungeons and Dragons" type games, making parallels with the emergence of infant art and the role-playing of young children.

The metaphorical (or, perhaps, more accurately, metonymical) nature of drawing for designing, as a way of *seeing as* in Wittgenstein's sense means that we view the drawing as if it were the real object and discuss and adapt our ideas about a mental image of a real object in the light of the drawing of that mental image. Seeing the similarities and patterns in things enables us to make the leap from one area of knowledge to another or from one symbol system to another. Analogical fluency allows us to construct in one symbol system a pattern for construction in another: to draw what we will make.

3.4.3 The Container / Journey Metaphor

Lakoff and Johnson (1980) believe that all human thought is based on metaphor and the central tenet of their position is that new ideas and concepts are not just built from previously stored ones, but from the metaphors in which prior concepts are couched. These newly constructed concepts they call *metaphorical entailments*. Their examples are taken from language use and their main example throughout the text is the concept ARGUMENT (I have followed their convention of capitalisation of examples here) for which they produce the diagram shown as Fig.36.



Fig. 36 : Extrapolation of Lakoff & Johnson (1980: 96) (a)

My prior seeing drawing as metaphor enabled me to realise that this model, with its specific metaphors of JOURNEY and CONTAINER can be generalised to include all process verb / product noun pairs, e.g. trust, work, plan, design, etc. (as shown in the Fig.37). Some of these verb / noun pairs do not share exactly the same word, but the metaphorical connection remains. For example, the verb "make" has no directly attached noun, but the process of making and the object that is made have the same JOURNEY and CONTAINER metaphors entailed in them. In making an object, we undertake a *journey* of thinking and planning and doing. The object we create *contains* all those thoughts and plans and actions.



Fig. 37 : Extrapolation of Lakoff & Johnson (1980: 96) (b)

The word "design" fits neatly into the pattern. "To design" is a process which is creative and intellectual *journey* which we undertake. "The design" is the thing that *contains* our thoughts and plans. Likewise, "drawing". Thus drawing for designing also fits the pattern. When we use drawing for designing we take our thoughts, along with our pencil, on a *journey* and produce "a drawing" which is then the *container* for those ideas. Applied to the specifics of design drawing:



Fig. 38 : Extrapolation of Lakoff & Johnson (1980: 96) (c)

I further simplified the diagram to the following form (Fig. 39) for use with children (using colour as metonymic to suggest interaction and merging) and, because it is conventional to read from left to right, it seemed more logical to put the CONTAINER metaphor on the left, as being the static starting-place for the JOURNEY:



Fig.39 : Container / Journey Metaphor of Design Drawing

This model not only transformed my understanding of the role of drawing for designing but also gave me a narrative, a story, in which to embed an explanation of the process to Year 2 children: It's like going on a journey with a carrier bag full of ideas and every now and then you stop, take your ideas out of the bag, look at them, re-arrange them, pick up something else interesting lying around, put them back in the bag and off you go...

3.4.4 The Story behind Designing

The whole issue of the development of early design skills seems interwoven with the issues of play and fantasy and the ability to manipulate inner mental images, not as simple discrete constructs, but as complex free-flowing, changing, kaleidoscoping, transforming and interconnecting with fuzzy boundaries which can collide, combine, spark off the new and the novel and create a whole new world of meaning and seeing.

Piaget made a seminal contribution to the understanding of child development but where he was less successful was due to his view of the child as scientist, whereas the child was looking for the story and expecting the story to make sense. Although I believe his central tenet of assimilation and accommodation to be correct, I believe that he mistook the motivation and the mechanism, due to his own background as a scientist in an age and culture in which rationalistic science was considered to lead the perception of reality, rather than being one of many ways of making sense of the world around us.

The child's motivation is story: making sense of the world as narrative. Small children love stories and the more stories they are told, the better their ability to construct abstract concepts.

Children from story-rich homes, generally speaking, seem to do better in school. They know that meanings slide. They know that words can be used in several ways and can be interpreted at different levels. They know about metaphor and can extrapolate from the literal to the figurative. Their heuristic knowledge of the way we can use and interpret language enables them to access and utilise the symbol systems of the classroom, mathematical as well as linguistic.

The mechanism is analogy and extrapolation: fitting new percepts into the inner story already created and stretching and extending those inner constructs to assimilate new ones where possible or to rearrange or even discard those constructs to accommodate new ideas which will not comfortably fit. Experience and knowledge provide the base ground for assessment of new percepts which are added to knowledge base by finding an analogy to already stored perceptions, which the brain stores as like/non-like. Language or visual labels shortcut the process and enable storage and classification by label. Perception or recollection of a label may trigger or bring out a whole raft of concepts.

I wondered how far is this manipulation of mental schema from the understanding of allegory or the construction of paracosms and whether children who can create a whole fantasy world in their head are generally better at designing. I knew of children for whom both abilities existed and wondered if they were related. The ability to create and think completely in a system and come to logical conclusions in that system, whilst fully aware that it is a created system which parallels another system, seems common to both fantasy role playing and designing. Whether using drawing as a design tool or living out a role in an imaginary world, the player can think in it, manipulate it and make logical domain-appropriate decisions within it by juxtaposition with a domain-parallel system and mentally switching back and forth from one to the other.

People are meaning-makers (Wells, 1986) and the primary vehicle of making-meaning is language. That the richness of metaphor could be used to convey understanding of an abstract idea such as the use of drawing to model design ideas seemed to me highly likely and one which I was prepared to try. The theoretical underpinning of the Structured Phase was an extrapolation from these thoughts.

3.5 Reflections on Section 3

Following the framework of Section 3, these Reflections are sub-divided according to the Sections of this part of the thesis: the literature, my observations and the creation of the taxonomies.

Literature review: Drawing as product or tool (Section 3.2)

My reading around the subject of drawing and of young children designing revealed that little had previously been researched with respect of drawing for designing, mainly due to the newness of Design and Technology as a primary school subject. The most useful insights came from writers on play, creativity and other contributory skills towards design capability. I found I was casting the net widely, looking for strands which would be applicable to young children's design skills, in fields as diverse as cognitive modelling and artificial intelligence to the development of language and pre-literacy.

Observational Constructs (Section 3.3)

Between Jan.96 and July99 I had looked at the work of 371 individuals across the 5 - 9 age range, 96 of whom had been in the original Stan 96 study and had been tracked across their whole time in the school. Some of these older children, especially those with whom I had considerable contact, were using their drawings to develop a design solution. I also had an intuitive feeling that standards were going up across the school and I doubted that my comments about the Year 4 helpers for Stan96 (that they did not understand why I wanted plans on paper any more than Year 1) would be true of the 1999/2000 Year 4s. As a result of all this work, I had devised a classification system (Drawing Types) which could be used to determine capability in design drawing.

Theoretical Constructs (Section 3.4)

The construction of the Taxonomies and the Concept Web finally crystallised my thinking. This meshed with my reading on language development to look for parallels with the development of visual literacy, mean that I encountered the idea of the role of metaphor in the construction and transmission of meaning, which was to become central to my understanding of the drawing as metaphor. I also became convinced that children who were more able to access metaphorical language were likely to be able to use drawing to model design ideas, because of the parallels between the two processes.

Extrapolating from Lakoff and Johnson's (1980) metaphorical entailments diagram enabled me to construct a model for explaining the role of drawing in designing to Year 2 children: the

Container / Journey metaphor. It was this understanding that was to be enshrined in my Programme of lessons in the Structured Phase of the research.

Transition from Exploration to Structured Investigation

I felt that to push Year 3 children over the brink into using their drawings to progress design ideas would be easy. I had managed this with many of the children in a Year 3 class I had taught weekly in 1998/9, whilst I was still trying to identify what children were doing. The year group for whom I had least data, for purely practical reasons, were Year 2. I knew that it would be conceptually impossible for Year 1s to understand what I meant; I had been trying with every class since 1996 to no avail. But I thought I now had sufficient understanding of what I was looking for and also what I meant by using drawing as a tool for thought to attempt to improve Year 2 children's design understanding, although both the literature and my previous personal observations suggested that this would not be easy.

This transition can best be represented diagrammatically:





SECTION 4 - Structured Phase

4.1 Introduction to Section 4	139
4.2 Planning for the Structured Phase	140
4.3 Programme Delivery	154
4.4 Evaluation of Teaching Input to Focus Class	171
4.5 Reflections on Programme Delivery	.179

4.1 Introduction to Section 4

The transition from the Exploratory to the Structured Phase was not an easy time. I had an idea about the metaphorical nature of design drawing and a belief that new knowledge is constructed by analogy and that this inherently human skill is invoked constantly in teaching, but initially I had little idea how I might demonstrate this. I also believed that I had a specific metaphor (Container / Journey) that could be used to explain the use of drawing for designing to children, possibly as young as Year 2. I came to the pragmatic conclusion that if I wrote a Programme in which such a pedagogy was embedded, then its success might prove the hypothesis.

Planning for the Structured Phase (Section 4.2)

As discussed in Section 2.3, changes in methodology affect issues relating to the generation and validation of data and these need to be made explicit. Delivering a pre-planned Programme rather than mapping the field meant that there were issues of validity and reliability (addressed in Section 4.2.3) that pertain specifically to this phase of the research. Section 4.2.4 details the range of data generated as a result of the Programme and how both qualitative and quantitative data were used in the multi-method analysis.

Programme Delivery (Section 4.3)

Section 4.3 details the structure of the Programme, and provides an outline of the Teaching Input and Assessment Tasks. More detailed lesson plans can be found in Appendix B. The embedding of the Container / Journey metaphor (Fig.39) within the Programme is explained in Section 4.3.3. The Comparison Class' experience of Design and Technology is outlined in Section 4.3.4. Events that occurred within the 15 month duration of the Programme, external to the Programme itself, that affected its delivery are detailed in Section 4.3.5.

Evaluation and Reflections on the Teaching Input (Sections 4.4 - 5)

It was decided, for the sake of clarity, to locate the evaluation of Teaching Input to the Focus Class separately from the outline and also from the analysis and discussion of the Assessment Tasks found in Section 5. Section 4.4 provides an evaluative commentary on the delivery of the Programme and Section 4.5 reflects on the learning: both mine as researcher and that of the children.

4.2 Planning for the Structured Phase

In planning the Programme to be delivered in the Structured Phase of my research, I was much more fully aware of the practical issues which impinge on research activities. I was no longer "trying things out in classrooms" as I had been during the Exploratory Phase but was seeking to implement a Programme of four school terms' duration which had a specific overall aim: that of improving Year 2 children's capability in design drawing.

The choice of year group was based on the Exploratory Phase analysis. Year 2 appeared to be the youngest age group with whom the Programme might have success, as Year 1 classes consistently showed little understanding of recording design ideas through drawing. Year 3 children, on the other hand, employed drawing for recording design ideas and were beginning to use drawing for developing their ideas. I taught Year 3 children for nine years prior to becoming Year 1 Co-ordinator, and had continued to teach Design and Technology to a Year 3 class until 1999, so as well as having a large collection of drawings from this age group, I also had a heuristic "feel" for their capability. So, although I felt that I wanted to test the Container / Journey metaphor with the youngest children possible, I also wanted to continue the Programme into Year 3, so as to be able to make comparisons with my observations of children of this age. Also, from a practical point of view, I had activities that I had used with Key Stage 2 children, which could be incorporated into the Programme if I continued it into Year 3. I especially wanted to incorporate kit-making (the idea behind Insects 97,98 & 2000).

4.2.1 Aims and Objectives of Structured Research Programme

• To improve young children's ability to use drawing as a design tool through the use of a specific understanding of designing as a Journey and drawing as place-holder (Container) for ideas along the way.

• To attempt to establish a link between ability to use drawing for designing with more general cognitive skills relating to analogical understanding

• To limit the scope of the research to *drawings* within the wider context of designing, and, therefore, to consider annotation of a sketch as part of the whole but to exclude text for which the drawing is merely illustrative.

From previous observations of children's design drawings, I knew that before Year 2 very few children can begin use drawing as a tool for thought but by age 8-9 many of them do. Therefore I wanted to focus on Year 2 children and the development of their designing skills. I believed

that understanding and being able to exploit the analogy between the drawing and the making processes is the key to being able to utilise drawing as a design tool.

In describing the difference between the way that novices and experts solve problems, Kahney (1993) uses the terms declarative and procedural knowledge. Declarative knowledge is at the level of verbal knowledge, following instructions. But:

"In order to achieve skilled performance you need to be able to translate declarative knowledge into actions. A new form of representation, known as procedural knowledge must be established...The expert learns to respond to whole patterns rather than to individual components of a situation."

(Kahney, 1993:.91)

Although aiming to develop the children's ability to use drawing as a design tool, specific drawing techniques (such as perspective) were not taught, neither did every lesson feature drawing. The aim was to impart understanding of design and the part that drawing can play in supporting the design process. I needed to ensure that the difference in experience was the *understanding* of using drawing to support design thinking which was being developed rather than simply inducting the children into a formulaic way of "how to do D&T".

Therefore, I devised a Programme of Lessons which would form a coherent teaching package to last 15 months, from October 2000 to December 2001 to be delivered within the context of Design and Technology lessons to one class of Year 2 children (from the beginning of Year 2 to the end of their first term in Year 3) to extend their understanding of design by making explicit the analogical nature of design drawing. This I would do through utilising the Container / Journey metaphor (Fig.39) to explain the design process. This class is referred to as the Focus Class throughout.

In conjunction with this I devised a series of Assessment Tasks to be conducted termly, concluding in January 2002, which were single lesson Design and Make tasks, most of which I had previously conducted with other classes across the 5-9 age range during the Exploratory Phase of the research. These Assessment Tasks were also conducted with another parallel class (the Comparison Class) to determine whether any gains were made by the Focus Class as a result of the Programme.

4.2.2 Research Subjects for Structured Programme

Two parallel classes with the 2000-1 Year 2 cohort were chosen, one as the Focus Class and one as the Comparison Class:



Fig. 41 : Research Subjects (Structured Phase)

The children were heterogeneous parallel groups (representing a range of academic abilities and yet being within a narrow age band) which helped to address both internal and external validity. The history of both groups was known, in that they had similar experiences of Design and Technology activities in Year 1 since I had overview of planning and delivery in my joint roles as both Year 1 and D&T Co-ordinator. The setting in which they were studied was naturalistic, since the Programme formed part of their school curriculum, they were studied working in their own classrooms and they knew me well. There were 24 children in each class group at the beginning of the Programme, but numbers shrank due to placements in other schools and families moving away, which also affected the gender balance. Thus, although threats to internal and external validity were kept to a minimum through the choice of subjects, the loss of subjects during the study could not be controlled.

Focus Class: with whom I conducted the Programme of lessons, delivered weekly during four consecutive school terms, during which the dual nature of design drawing as Container and Journey was made explicit. These were children from a Year 1 class parallel to my own in 1999-2000. My previous contact with them was only in my role as Year Co-ordinator.

Comparison Class: with whom I would conduct the Assessment Tasks only, at intervals throughout the Programme (also conducted with the Focus Class). These children were my 1999-2000 Year 1 class, plus a new girl who joined the school in Sept.2000.

Target Groups: Six children from each class who would demonstrate a range of approaches and understanding and who would not be camera-shy or react adversely to being questioned about their work, whose drawings would be submitted to my moderating panel for blind-marking. The choice was based on my heuristic knowledge of the children at the beginning of Year 2. The Comparison Class Target Group were chosen whilst they were still my Year 1 class, and the Focus Class Target Group were chosen in discussion with colleagues during the first few weeks of the Programme (Autumn term 2000) to mirror the Comparison Class Target Group.

Gender balance

There were more girls than boys in both classes (14:10 at the start of the Programme) and the three children who left the Comparison Class during the year were all boys. One boy in the Focus Class was absent for two Assessment Tasks and so was excluded from the analysis in order to even up the numbers. Section 5.9 contains the evaluation of the differences between the boys' and girls' performance.

Academic attainment

School Records Data on the children's academic performance was limited to the results of the SATs tests taken by the children in May 2001 and used as proxy pre-test data to indicate the academic performance of the children. I was unable to obtain detailed records of the Baseline Entry profiles which were conducted at the start of the children's Reception Year but know that this cohort scored several points lower than the county average (37 : 45). As can be seen by the SATs scores (Chart 4, below), the children involved in this study are of below average attainment for their age (7.0 to 7.6 years in May 2001, the older half of the SATs cohort) but fairly comparable to each other. The Focus Class appear better at writing; the Comparison Class seem to be better mathematicians.



4.2.3 Validation of Structured Phase Analysis

I began the Structured Phase of the research with greater awareness of the issues of validity and reliability. It was important that, as far as possible, I was able to demonstrate that systems were in place to address these issues. Since I was using a multi-method approach, using both qualitative and quantified data, I was concerned that the terminology and assumptions of one methodological tradition did not become the lens through which both aspects of my methodology were viewed and also that validity checks that were appropriate to large-scale quantitative studies would not be inappropriately applied to my small population. Section 4.2.2 has already indicated the minimising of threats to validity through the selection of research subjects.

Section 2.2.3 outlined the importance of multiple perspectives for addressing issues of validity and reliability of research procedures and findings, whilst also limiting the extraneous variables through maintaining consistency of approach and analysis methodology. At the beginning of the Structured Phase, I was far more aware of these issues than I had been during the Exploratory Phase and I attempted to deal with these issues in the following ways:

Assessment Tasks

The Assessment Tasks were all activities of the type that would form a normal part of a Year 2 Design and Technology curriculum and (with the exception of the Maze Task) had been trialled my myself and others prior to the Programme. Although each task had its own characteristics, the materials of construction could be cut by the children using scissors and the task could be completed within one session. The Assessment Tasks were conducted at intervals throughout the duration of the Programme, thus giving a longitudinal aspect to the study, as shown in Fig. 47 in Section 4.3.1. I was careful that the Assessment Tasks were conducted in the same way in each class and that the same Teaching or Learning Support Assistant was present for both sessions for each Assessment Task.

Trochim (2000) discusses the reliability issues that relate to Test-Retest scenarios, observing that the longer the time gap between the re-tests, the less similar the factors that contribute to error. In my Programme, the tests were spaced unevenly across a period of 15 months. However, the aim was not to compare before-and-after as much as between two populations at each assessment point. The assessment situations were not of the type usually associated with large-scale quantitative studies, they were intended to be the kinds of activities in which Year 2 children would normally be engaged in Design and Technology lessons and, as such, each Assessment Task had its own characteristics that affected the results. This I did not see as a
disadvantage, for although it might have affected the mathematical reliability of a statistically based analysis, it fitted more appropriately with the reality of the classroom which I was attempting to capture.

Although the Assessment Tasks were of broadly of the same type, differences between the tasks to give a range of opportunities for children to display their design capabilities. Issues surrounding the variations between Assessment Tasks not realised at the start of the Programme are discussed in Section 5.2. The extent to which these variations would affect the internal validity of the research was minimised by using comparison between the classes as measure of Programme success rather than relying purely on criteria referencing. Trochim (2002) defines parallel-forms reliability as relating to the consistency of the results of two tests constructed in the same way from the same content domain. The content domain for my Assessment Tasks was Design and Technology lessons appropriate for the age of the children. However, the richness and diversity of variables within that domain was far greater than that to which I could be easily mathematical modelling, even if such were considered valid, given the low number of research subjects.

On reflection, I feel that such variations added to the validity of the study, since it demonstrated another perspective: the way in which children use drawing in different design circumstances. If each Assessment Task had been identical in its demands and required the same range of thinking skills, a much narrower perspective on children's design skills would have emerged. In order to establish parity across Assessment Tasks, I introduced each task to the whole class at the start of the lesson and the children were then asked to develop their ideas on paper before engaging with the materials and making the product. To give a different perspective on the children's design capability, one of the researchers from the "Enriching Literacy; the Design and Technology Evaluation Project" conducted the Surprise Card assessment activity from that study.

<u>Analysis Methodology</u> - both qualitative and quantitative approaches were used in order to give as full a picture of the children's capability as possible. Section 4.2.4 details the range and type of data collected, summarised in Table 5 :

Qualitative	Quantitative
Log book entries - Programme Log (Focus Class) & Child Response Log (both classes)	Assessment Task drawings & photographs (both classes)
Audio & Video recordings - children working (Target Group, Focus Class) & de-briefing interviews (Target Groups, both classes)	School records data - register details & SAT results (both classes)
Drawings and/or photographs from most non-assessed lessons with Focus Class	

Table 5	2	Qualitative	&	Quantitative	Data
---------	---	-------------	---	--------------	------

Within the quantitative analysis, the same marking criteria and recording format was used on every piece of work produced for an Assessment Task. Moderation by others was an integral part of the validation process, both in the development of the quantitative analysis instrument and in establishing a consensus on what constituted a specific allocation to points on a continuum (discussed in Section 5.4.3b) to avoid halo effects, generosity and contrast errors, which, as Judd et al. (1991) point out, can easily lead to biased judgements. Trochim (2002) considers that "inter-rater reliability is one of the best ways to estimate reliability when your measure is an observation." (http://trochim.cornell.edu/kb/reltyypes.htm). My sample was not large enough to support statistical techniques suggested by Trochim, such as Cronbach's Alpha, but the principle remains the same: multiple raters would lead to greater assurance of reliability of analysis findings.

Within school, two colleagues were involved in testing the analysis instrument at several points in its development (detailed in Section 5.4.2). Miss N. (the Comparison Class teacher) was involved in the early stages of developing the analysis instrument and, as a mathematics specialist, gave useful advice on modelling the results. However, since I was anticipating a difference between the two classes to become apparent, it seemed unwise to ask her to become involved with marking. Miss S., the Year 2 Co-ordinator, and Mrs. R, the Art Co-ordinator (now also Year 3 Co-ordinator) took on this role. Both colleagues had many years experience in teaching across the Primary age range and had maintained an interest in my research since Stan96, They blind-marked all the Target Groups' work for each Assessment Task, including re-marking where changes to the analysis instrument had impact on classification.

A member of the Design and Technology Bridging Group, who taught Year 2 in a neighbouring school also offered advice on the wording on the analysis instrument and became the third member of my moderating panel who blind-marked Target Group work for each Assessment Task. Holding meetings with her at the same time as my in-school colleagues were available was sometimes difficult to arrange, since all of us held posts of responsibility within our schools as well as running after-school clubs.

Since two of my moderators were colleagues within my school (especially since Miss S. was Year 2 Co-ordinator) the moderation panel were given photocopies of the children's work with names obliterated so that they did not know which were Focus Class and which were Comparison Class children. I was anticipating differences emerging between the two classes and I did not want any comparisons being made between Miss N. (Comparison Class teacher) and myself, since I was expecting the results to be better than I had achieved previously with children (hopefully comparable to the Year 3 class that I taught weekly in 1998-9) If differences

emerged it would be due to the Programme content, not my teaching style and I did not want comparisons being made about our teaching skills based on Programme outcomes.

Our procedure was for each of us to have copies of the Target Groups' drawings, photographs of finished products, copy of video, audio tape or Log Book notes, which we would then assess by the analysis instrument criteria (Appendix N). On meeting we would compare, discuss and come to an agreed position about each drawing. This agreed evaluation would become the quantified data entered into the spreadsheets on which the quantitative analysis reported in Sections 5.5 - 5.9 was based. This process was complicated by the evolution of the analysis instrument (see Sections 5.4.1 - 2) and there were several occasions when we needed to revise our previous judgements based on revisions of the analysis instrument. Once the analysis instrument metamorphosed into its final form (Section 5.4.3), Miss S. and I jointly reviewed the evaluation of all drawings (not just those of the Target Groups).

An in-school workshop that I conducted in June 2001 on developing young children's design drawing skills provided an opportunity to ask the whole teaching staff to blind-mark the Easter Assessment Task work using the quantitative analysis instrument as it was at that stage. Colleagues worked in pairs with a selection of Target Group drawings. In July 2001, I repeated the design drawing workshop at the Design and Technology Association conference, again using the Target Groups' drawings and asked those attending to evaluate them by the criteria on the developing analysis instrument. It was apparent, however, as I circulated the room, that the shared meanings that had been established with my moderating panel and the heuristic knowledge that my in-school colleagues brought to the evaluation of First School children's drawings, was not shared by this more varied group of people.

Neither of these workshop groups had the benefit of viewing or listening to the video and audio recordings, reading my Log Book notes or of seeing the completed products. My moderating panel agreed that it was difficult to place the drawings without the other data, since the child often revealed their intentions in a comment during de-briefing, or that it became apparent that design ideas were being developed when the product was viewed in conjunction with the drawing. It would seem that there was an inherent danger in assessing drawings, that were intended to support a design journey, as if they were journey's end.

The evolution of the analysis instrument, as well as the assessment of children's work was subjected to discussion over many months with my research supervisor and the research group at Goldsmiths College (Section 5.4.2b). This research group comprised research students of the Design Department at Goldsmiths College, who met termly for mutual support and discussion of progress, chaired and advised by Professor Richard Kimbell.

4.2.4 The Choice of Data to Collect

It was important to make the correct decisions about the data to be generated for analysis: wide ranging enough to give as full a picture as possible, without giving myself an impossible task as regards analysis time and complexity. Spreadsheets and databases were constructed to manage the task of recording and analysing all data, whether evaluations of products or transcripts of video recordings.





Primary Data: All drawings produced during the Programme were kept. Photographs were taken of products of Assessment Tasks and of children working both during Assessment Tasks and some of the Teaching Input sessions. Video recordings were made of children working (Focus Class Frosty and Flat Stan) and discussing their work after completion (Stan Series). The school records data were the children's SATs results.

Secondary Data: I maintained a Log Book throughout the Programme, which included all field notes made during contact time with children. Heuristic knowledge of the children included my knowledge of family circumstances, acquaintances with other family members, relationships built through teaching them or their close relatives.

Data used for qualitative evaluation Data used for quantified analysis Data collected but not analysed



Fig. 43 : Distribution of Data Collected

Drawings produced by Focus Class children during Teaching Input would not be subject to quantified analysed, although observations would form part of the evaluation of the Teaching Input. Despite enabling me to be informed about the work that the Comparison Class were doing, photographs of this class' work outside of Assessment Tasks were not analysed. The school records data was not as extensive as I had hoped, so that conclusions about relationships between design capability and achievement in other curriculum areas were not able to be drawn. My heuristic knowledge of the children informed my choice of the Target

Groups and being well-known to parents enabled permission to be granted to use children's work for publication.

My intention was to video-record the Target Groups at work on each Assessment Task. However, attempts to use the rarely-used school video camera led to frustration over missing parts and flat batteries. Previous experience of transcribing audio recordings of classroom work did not encourage me to substitute audio for video. This media was used successfully for de-briefing interviews with the Target Groups immediately after the Assessment Task sessions. My Log Book, became the main vehicle for recording in-context observations, combined with digital photography to capture critical moments and to record all finished products, became the most efficient and effective means of recording both teaching and assessment sessions.

The advantages of the Log Book were its:

• *spontaneity* - it moved with me to the hot-spots in the classroom; it was not trained on children doing not much whilst something more note-worthy was happening elsewhere.

• adaptability - I could record verbatim speech, personal reflections, make sketches of work in progress (as excerpt shown in Fig.45), write in columns or divide up the page for recording several children's work and add notes to each child's square when I returned later, colour-code, write sideways or across previous recordings, draw circles and arrows to connect themes.

• *familiarity* - it was the system I ordinarily used in the course of teaching to record observations and evaluations of children's learning across the curriculum.

Fig. 45 (overleaf) shows a page from this Log Book. The "Reflective General" comment (just off the scan) reads : *There is a far greater degree of negotiation and sharing of ideas and copying good ideas going on than I had imagined*, which is borne out by my sketches of the work of the main girls' group at the top of the page.

Fig. 44 : Excerpt from Log Book:



These Log Book notes, combined with digital photographs, proved to be the most efficient and effective means of recording children working and capturing the important moments, as I could circulate, photograph, sketch and note comments made by children while they worked. I found that sitting down and making quick sketches of the children's work prompted them to tell me about what they were doing and I would annotate my drawing appropriately and note their comments verbatim. I was thus able to get a cameo of each child at some point during each Assessment Task session, which became vital for interpreting their drawings. Holding a "photo-shoot" at the end of each session was the easiest way to record the finished products.

I had used the Log Book in this way during all the teaching sessions with the Focus Class from the beginning of the Programme and the children would frequently come to me to tell me what they wanted me to write down. I created a Log Book Index database (Appendix M) to enable classification of the comments and sketches. My original intention here was to use the filter function to trace themes and the development of individual children across time. When I attempted this, however, I realised that the context was lost. It was in the reading of all the comments relating to each activity that the richness of the situation could be seen, as well as difference between the two classes.





The qualitative data was to form the basis of the evaluation of the Teaching Input to the Focus Class (Section 4.4) as well as contributing to the evaluation of the children's performance on the Assessment Tasks (Section 5.3). The quantified data originated from evaluation of the children's drawings, which could then be manipulated more easily in numerical format. This is reported in Sections 5.5 - 9.

Fig. 46 shows the data that I had collected by the end of the Programme and how it was managed for analysis of the Structured Phase:



Fig. 46 : Management of Data in Structured Phase

4.3. Programme Delivery

Unlike those of the Exploratory Phase, the Structured Phase activities were planned as a complete Programme across fifteen months (October 2000 - January 2002) to be delivered to a specific group of children who entered Year 2 in September 2000 in the school in which I worked. The Programme was devised in February 2000. This section outlines the Programme, indicates the changes made and gives account of the reasons behind these changes. The full lesson plans can be found in Appendix B. The Programme Outline (Section 4.3.2) briefly describes the projects within the Programme.

4.3.1 The Structure of the Programme



I began to plan the Programme of lessons to be conducted with the Focus Class with the belief that understanding and being able to exploit the analogy between the drawing and the making processes is the key to being able to utilise drawing for designing. This meant devising a Programme to extend their understanding of design by making explicit the metaphorical nature of design drawing. This I would do through utilising the Container / Journey metaphor to explain the design process. At key points throughout the duration of the Programme, I would conduct Assessment tasks with both the Focus Class and the Comparison Class in order to map the development of their design capabilities.

I had a wealth of activities from the Exploratory Phase which could be used but these needed allocating as teaching input activity or Assessment Task. I had extensive documentation of these activities, including method of presentation, children's drawings and design sheets, photographs of children working and products made, observations at time of delivery and analyses of the children's work. I also knew the limitations and pitfalls of these activities, some of which had been ironed out through repeated trials. I especially wanted to use these tried and tested activities for assessment as this would cut down on the unknowns and so increase the reliability of results. At the same time, I wanted to try some new ideas, including some ideas from projects conducted by other researchers.

The Stan activities based on "Flat Stanley" by Jeff Brown could not be omitted; I had too much experience and comparative data on these, yet knew they were not suitable as Assessment Tasks (no client, human figure produces stereotypical drawing response). However, my complete familiarity with a whole range of ways of presenting the activities made them ideally suited to being the vehicle for introducing the central Container / Journey metaphor to the Focus Class. If I produced better results through using this explanation than any previous presentation, then there could well be grounds for attributing the results to the explanation.

Pandy's Suitcase could be used as it was, since it had proved successful in the Exploratory Phase. The kit-making ideas (Insects in the Exploratory Phase) could be adapted to different subject matter for teaching input. Boat-making and work on food promotions, conducted with the Year 3 class I taught weekly in 1997-8, could be used with few changes.

New ideas were trialled with classes around the school during 1999-2000; for example, Easter Bunnies with wheelbarrows in March 2000, but this took too long to be a useful Assessment Task. The Surprise Box of Art Club 1999 had become a Surprise Tube activity conducted with groups of children across the school in May 2000 and could be completed in a single lesson. This was adapted to the Easter Egg Holder Assessment Task which was based in the same wide card tubes.

There was also the need to maintain parity of experience in terms of hand-skills between my Focus Class and the rest of the year group. The sandal project was designed to match textile work (Puppets) and I adapted the food series to bring it more into line with the planning of the Year 3 team. My final Assessment Task (the Maze) was a new activity designed to fit in with Year 3 Literacy Hour texts.

4.3.2 Programme Outline

The Programme began part-way into the Autumn term of the new school year in 2000 (to allow for a "settling in" period for my new Year 1 class) with its final Assessment Task session at the beginning of the Spring Term 2003. The Teaching Input Sessions delivered to the Focus Class were conducted almost weekly until the end of November 2002 (Thursday afternoons whilst they were Year 2, Fridays Year 3). The Assessment Tasks were usually conducted with each class on consecutive weeks, except for the Card and Maze, which were conducted as two consecutive sessions in the same morning.

Table 6 provides an overview to show the teaching objectives of each half term block of the Programme. As well as the modelling skills (with the focus on drawing) which would be developed across the course of the Programme, this Table indicates the practical techniques to be taught and also other skills associated with design capability, such as team-work.

TERM	PROJECT	SKILLS	TECHNIQUES
AUT	Fantasy figures	To make what they have drawn	labelled diagram
2000		Making a pattern before engaging with materials	design sheets/grids
		Evaluation at planning stage through discussion with partner Develop ideas collaboratively	story-boarding
SPR	Visual analogies	Using what they see to stimulate ideas	transfer by tracing
2001		Develop understanding of and facility with visual analogies	
	Ideas on a Journey	Understanding what constitutes clear design communication	clarity of diagrams
		Introduction of Container / Journey metaphor	recording materials
L		Use drawing to support design journey	
SUM	Modelling in other	Evaluate each other's ideas and create a joint product	paper folding for runs
2001	media	Use media other than drawing for planning	bead-making
		Use drawing part-way through design&make process	flat-pack box
	Extended Project	Begin design activity from product analysis	observational drawing
		Carry through ideas across several sessions	pattern development
AUT	Team-working	Use drawing to communicate ideas within a group	design development
2001		Work as part of a team	graphic communication
		Develop meta-cognitive awareness of design processes	
	Designing for Others	Generate & critically review each other's ideas	
		Address needs of a client	
		Communicate & refine ideas	
		Prototype a product	

Table 6 : Programme Objectives

Table 7 overleaf lists each session of the Programme. A brief resume of lesson content follows. Full lesson plans are to be found in Appendix B. Table 7 : Programme Sessions

TERM	WEEK	PROJECT	ACTIVITIES
AUTUMN	1	Assessment	Cups, persons, pizza
2000	2	Reality / fantasy	Wagwums (1) figures
	3		Wagwums (2) clothes
	4		Wagwums (3) vehicle
	5		Wagwums (4) house
SPRING	6	Assessment	Frosty the snowman
2001	7	Visual analogies	Pictures from shapes
	8		Iron's visual analogies (1)
	Q		Iron's visual analogies (2)
	10		Letter Spirit Project
1	11	Ideas on a Journey	Snook's Animals
	12		Flat Stan
	13		Round Stan
	14		Customised Stan
	15	Assessment	Easter Egg Express
SUMMER	16	Non-drawn activities	Marble run
2001	17		Necklaces
	18	Customising	Buggy (1)
	19	-	Buggy (2)
	20	Assessment	Surprise card
	21	Extended Project	Sandals (1) - designing
	22		Sandals (2) - templates
	23		Sandals (3) - making
	24		Sandals (4) - completion
	25	Assessment	Pandy's suitcase
AUTUMN	26	Team-working	Food (1) - menu & poster
2001	27		Food (2) - diarama
	28		Food (3) - sandwiches
	29	Designing for Others	Kit-making (1) - design
	30		Kit-making (2) - prototype
	31		Kit-making (3) - testing
	32		Kit-making (4) - assembly
SPRING 02	3.3	Assacemant	Maria

The Programme had to take account of the development and maturation of the children across 15 months of their lives. Broadly speaking, the activities near the beginning of the Programme were ones which I had previously conducted with Years 1-2 and those nearer the end were ones which I had conducted with Years 3-4.

Additionally, the activities at the beginning of the Programme were "tight", in the sense that the activity was teacher controlled. The children were developing skills under specific guidance. As the programme progressed, the activities became "looser" and there was more room for the children to use and apply the skills gained in the earlier part of the Programme and to develop ownership of the skills as well as the activities.

In considering how to present that account of the Programme, it was decided that a short account of the content of each project should be included here, as an overview. The full lesson plans are to be found in Appendix B. However, since the teaching of the Container / Journey metaphor was central to the Programme, it was decided to discuss this in far greater depth (Section 4.3.3).

It was further decided that the qualitative evaluation of the Teaching Input would be separated from that of the Assessment Tasks, as this would allow the Focus Class' developing understanding of design drawing be discussed in relation to the Programme delivery. Mutually supporting qualitative and quantified data would then be used to compare the capability of the two classes as revealed by their performance on the Assessment Tasks:



Fig. 48 : Teaching Input & Assessment Tasks

<u>Autumn Term 2000</u>

Session 1: Assessment Task with Both Classes: Design a pizza

A Baseline Assessment Task (Design a Pizza) was carried out in October 2000 using an activity that had been conducted in previous years with Year 1 classes in the Summer term but that this particular cohort had not experienced. This Assessment Task was designed to establish base-line capability within both classes. It was a design-and-make activity suitable for children at the beginning of Year 2 in which they were asked to use pencil drawing to generate design ideas for a collage of a pizza to be made from a range of sheet materials, fabrics and small items. There was an initial discussion of favourite toppings on pizzas and how the toppings are arranged to look appetising. The children were told that they would be given some white paper to try out some ideas about what they would make. Progression of ideas across drawings was explained and demonstrated. For both sessions I had the help of Mrs. M, one of our Teaching Assistants.

Sessions 2-5: Teaching Input to Focus Class : Reality / fantasy

This first series of lessons was a fantasy theme based on alien creatures called Wagwums and Foozles. The design-and-make activities (Figures, Clothes, Transporter, House) based around these imaginary creatures and their planet home involved the use of drawing for designing but I wanted to encourage discussion and clarification of their ideas with a partner. I did not want to create artificial situations, in which children had to draw just because I was interested in their drawings, where other means of modelling and developing ideas would be more natural. I believed that by discussing their plans with a partner, the usefulness of drawing to communicate ideas would also become apparent.

I wanted the children to see the link between drawing and making, and also that drawing could be ambiguous and temporal. The children were shown large pictures of the aliens and asked to make them in plasticine and matchsticks. It was impossible to see from the drawings of the aliens whether they were flat or rounded as viewed from the side:



For the Clothes session the children used newspaper to make a paper pattern by trial and error and then draw the clothes they were going to make and attach sample materials to a worksheet. For the Transporter and House, I wanted the children to design by discussion and then record their ideas by drawing and writing prior to making.

Spring Term 2001

Session 6: Assessment Task with both Classes : Frosty the Snowman

The aim was for the children to produce a drawn design and then make a model to demonstrate a means of getting Frosty the Snowman's shopping to him on his hilltop if the shop is on the opposite hilltop and there is a lake in between. The children were shown a model snowman, a picture of the solution of the problem of the Lighthouse Keeper's Lunch story they knew from Year 1 to stimulate discussion on how Frosty might solve the problem and they were reminded how to roll newspaper to make a reasonably strong structure (bridge-making in Year 1). A range of other materials were available: string, card, lollysticks, corks etc., plus they were free to utilise anything else they were allowed to use from their own classroom stock (e.g. contents of Junk box).

Sessions 7-10: Teaching Input to Focus Class: Visual Analogies

Prior to presentation of the Container / Journey metaphor, I wanted to conduct activities that would enhance the Focus Class children's facility with analogical thinking. Four activities were chosen to achieve this objective, only one of which I had conducted before. The others were adapted from ideas found as a result of literature and internet searches on the topic of visual analogy.

a) Circles Activity - this was an ideation fluency activity that I had used around the school, most frequently in Year 3: circles printed on a worksheet, each one to be made into something different.

b) The Abstract Shape activity came from the RIOTT programme by McCracken (1985): each child had a card of roughly horseshoe shape to use as a template and make a picture based on as many of these shapes as they liked.

c) Iron's Analogies (2000) were used as the basis for making a board game. Although intended as a single lesson, two sessions were needed to complete the games.

d) The adaptation of the McGraw's Letter Spirit Project (McGraw & Hofstadter, 2000) to a design-and-make task, requiring the children to extrapolate from letters A - G in a variety of fonts to writing the rest of the alphabet on squared paper and making a name sign for their bedroom door.

Sessions 11-15 : Ideas on a Journey

These four sessions were the focus for the teaching of the Container / Journey metaphor. A fuller discussion of them is to be found in Section 4.3.3.

The first step was to establish in the children's minds what a good "Container" would look like, before attempting to get them to move it forward on a "Journey" through the activities based on the Stan tasks that I used during the Exploratory Phase. The Flat Stan and Round Stan activities were used as the main vehicle for introducing the Container / Journey metaphor, re-inforced by Customised Stan, in which I wanted the children to devise their own adventure for Stanley. In the Flat and Round Stan activities I wanted the children to use drawing to record and develop their design ideas and in Customised Stan I wanted them to move towards recording their developing their ideas as if made in the materials. I knew this would be difficult as, in all my previous experiences, recording the materials has been the aspect that children found hardest.

Session 15: Assessment Task with both Classes : Easter Egg Holder

The design problem involved in the Easter Egg Holder activity was to design and make a holder for an Easter Egg from a 100 cm. wide cardboard tube and other recycled materials, such that a small egg (hard-boiled eggs supplied for measurement) will be held securely inside the tube and the outside is suitably decorated to give to a friend.

I was interested to discover whether my input to the Focus Class would show any immediate effects. The Comparison Class had also been using drawing to support designing during the Spring term but without the Container / Journey metaphor to explain its purpose. The Easter Egg Holder task would be the first test of my metaphor's effectiveness.

Summer Term 2001

Sessions 16-19: Teaching input to Focus Class: Modelling in other Media

In Summer 2001 (first half-term) the Focus Class used media other than drawing for modelling, to ensure that the children had not just learnt the design drawing skill at a declarative level, but could transfer the procedural skill of modelling to other media. I wanted to try modelling in plasticine before making in wood, modroc or other plastic medium. Ritchie (1995) found that the children were much happier to change their designs in a plastic medium than if drawn. This seems likely since no permanent mark-making is involved. It would probably be more interactive in terms of discussing, changing and adapting ideas. The activities chosen were a Marble Run (made in plasticine, card and small found objects) and Necklaces made from paper beads.

The children had yet to experienced any flat-pack box-making activities, in which they could think about customising whilst constructing, and so I planned a two-week Beach Buggy project. Drawing would be used to think about logos and decoration of the product, once the basic box cart had been built. The basic cart was built in the first session. At the beginning of the second session I reminded the children about Containers & Journeys. I showed a side, front and back design drawing for a beach buggy, which "Since it is too small for everyone to see" I redrew on the flip chart and explained to them how my thoughts had gone on a journey.

Session 20: Assessment Task with Both Classes: Surprise Card

This was conducted during the first week after half term by one of the researchers from the "Enriching Literacy through Design and Technology Evaluation Project" conducted the "Surprise Card" from that project This gave an extra dimension to the on-going assessment of the children's capabilities, a activity devised and conducted by someone else to counter possible problems associated with myself as programme writer, deliverer and assessor. It was also the only task with a pre-printed design sheet (see Appendix I).

I assumed that both class teachers would allow the children to make the cards that afternoon and did not discover that this had not happened until the following week, when it was too late. This meant that there was no "relationship to making" for this task. There was also no opportunity to interview the Target Group afterwards.

Sessions 21-24 : Teaching Input to Focus Class : Sandals

This was the longest design-and-make project conducted with the children, to enable them to see the role of drawing in product development. Although related in theme, other activities had not lasted more than two lessons. The main making session was conducted by Mrs R (class teacher) as I was away at conferences in the last week of June.

This project followed on well from the Surprise Card task, since both developed from product evaluation. In the first session of the sandals project, the children engaged in discussion of the different sandals in pairs or small groups and recording features they wanted to incorporate in their own designs. In the second session, a class discussion of good sandal design (and what was possible to make) lead on to making templates to fit their own foot. The sandals were made across two sessions.

Session 25 : Assessment Task with both Classes: Pandy's Suitcase

The Task was to design a suitcase / travel bag for the toy Panda to take on holiday, as developed during the Exploratory Phase.

From previous experience, solving this task involves grasping the interplay between fantasy and reality and accepting my terms of reference as to which parts of the task are "real" and which are "pretend" : design a travel bag, pretending that card is a suitable material, for a real toy panda in which he could put a real plastic mac in the context of pretending he could go on holiday. In 1999, many Year 2 children produced a picture of a suitcase and insisted that he could put his mac in it. I hoped that fewer children in the Focus Class would fall into this trap and misperceive the reality/fantasy divide inherent in the activity after the sandal project.

Autumn Term 2001

Sessions 26-28: Teaching Input to Focus Class: Food

Autumn 2001 (first half-term) focused on food technology, incorporating activities that I had previously conducted with Year 3 children (designing menus and making a diorama to advertise their healthy meal) with sandwich-making activities from the QCA scheme of work being followed by the rest of Year 3. I felt that my Focus class might feel deprived if they did not eat sandwiches too. Therefore I changed the Programme, spending two weeks on healthy food dioramas and the third week making and eating sandwiches. In practice, this faster pace for the dioramas worked well.

Sessions 29-32: Teaching Input to Focus Class: Kit-making

The final half-term block was aimed at bringing together all the strands on which we had worked: team working, addressing needs of a client, consideration of materials, as well as the use of drawing to support these activities: to generate and develop ideas, to communicate these ideas to others (both between working partners and to the client) and to record the final product.

I have conducted kit-making activities with Year 4 children in previous years, always in the Summer term and related to Year 1's topic of mini-beasts. Designing a Christmas present for Mum would be a suitable activity at this time of year and to which Year 1 would relate quite happily.

The Focus Class needed to put together a kit of parts for the Year 1s to make their gift, to include sufficient (but not too much) of each material, labelled with its intended use, any templates needed, a poster illustrating the final product and an instruction sheet.

The Year 4s had made simple errors such as putting decorative items (e.g. spots) into the kit but no base cloth on which to stick them, which caused considerable distress to the Year 1s, who assumed that not being able to make the kit was their fault. This time my Focus Class would test each other's instructions before giving them to their clients. They would produce a simple poster to explain their idea to the rest of the class, so that any difficulties could be identified through peer review. This would aid the reflective skills of all members of the class. The whole process required the children to think about making construction processes explicit and think what a younger child will be able to do and understand. They would need to have internalised the making process and use drawing to support their explanations.

Spring Term 2002

Session 33 Assessment Task with Both Classes : Theseus' Maze

Since I had not taught the Focus Class since the end of November, this Task would help to indicate that real understanding had taken place rather than just acquisition of taught techniques.

The activity was not one that I had conducted before. I had planned to conduct a three week block activity with both classes (Puppet Theatre) which I had used in 1997 across the school. However, since I was now working at Canterbury Christ Church University College, I did not have the time to visit the school weekly in order to do this. To do something which fitted in with their Literacy Hour work (Greek myths) seemed most time-effective and I found a pop-up book of Greek myths containing a 3 dimensional maze to illustrate Theseus and the Minotaur. The children were to use drawing to plan a 3-dimensional model of the maze to help Theseus escape from the Minotaur and show him how to use the string.

4.3.3 Teaching the Container / Journey Metaphor

The Container / Journey metaphor was introduced in the second half of the Spring term (as shown in Fig. 47) using the Stan activities I had used extensively in the Exploratory Phase of the research. I wanted to introduce the Container / Journey metaphor (Fig.39, Section 3.4.3) as early in the Programme as possible but I believed that children need some experience on which to base and build understanding of the new concept. I also believed that in order to absorb the idea, internalise it and make it their own, the children needed to be exposed to the idea in a range of contexts over a period of time. I did not just want adoption of a drawing technique, I wanted understanding of how drawing could support design thinking.

SECTION 4 - Structured Phase

Mantell stresses the importance of making design techniques and strategies

"explicit to the children, so that it might become part of their learning and over a period of time, be absorbed into their repertoire of techniques for designing." (Mantell, 1999: 91)

The vehicle for introduction of the Container / Journey metaphor was to be a series of three "Stan" lessons, as these were activities which I knew worked well with children of this age. Although much of the Exploratory Phase analysis was centred on these activities, I had become convinced that it was not a suitable task for assessment. Children of this age tended to produce a stereotypical figure drawing and I was concerned that I had frequently captured their current representation of a human figure, rather than real design thinking. I was confident, however, of its suitability as a vehicle for teaching input.

The advantage of using a task with which I was so familiar (and of whose possible drawbacks I was aware) were that there were few unknown variables. I had delivered the activity so many times that I could keep everything else the same except the explanation involving the metaphor and I planned to make an audio recording of my introduction to the children to compare with 1996 and 1998 deliveries.

I placed the Stan series at the end of the second school term of the Programme because I believed that I would have prepared the children through introducing the use of drawing for designing (Autumn) and heightened their awareness of analogical thinking (first half of Spring term). I wanted to introduce the concept as early in the Programme as possible but I believe that children need some experience on which to base and build understanding of a new concept. It cannot be introduced "cold".

The first session was introduced through a PowerPoint presentation of illustrations from Snook (1974). This lady was my art teacher at school, who wrote several books on a range of craft topics, using examples from pupils' work (including my contemporaries). There is considerable variation in her annotation, clarity of drawing and expansion to show helpful small details, which would help or hinder another person following the ideas: the teaching point I wished to convey to the children, to draw *"so someone else could make it"*. (The PowerPoint presentation used for this session is in Appendix D.)

Mrs R., the class teacher, read "Flat Stanley" to the Focus Class in advance of the three sessions to be based directly on the book. In the first lesson (Flat Stan), the children were

shown the four examples in Fig.27 from my Exploratory Phase collection of drawings, two static and two showing progression of ideas.

- Fig. 27 Examples shown to Focus Class

The children had little difficulty distinguishing between the genres and the Container / Journey metaphor was drawn on the flip chart and explained. The children then used drawing to develop their ideas for a Stan puppet, which they then made. In the following session (Round Stan), my aim was to re-inforce this learning and move the children further into using drawing to develop design ideas. I was able to show the children a range of examples from last week of ways they had used drawing to support their design journey. In practice, the timing seemed about right. I had built up sufficient relationship with the children for them to believe that what I was saying was important for them to learn and that I was interested in what they had to tell me about what they were learning.

Knowing that children needed to see as well as hear and to articulate their understandings as they develop, I planned to make my explanation as visual and interactive as possible. As a lead-in to the series and so the children would understand the features of clear drawings, the cycle would begin with a PowerPoint presentation (Appendix D) of illustrations from Snook (1974), who was my art teacher at school and the author of many books on art, craft and embroidery, frequently using examples from her pupils' work including, in this work ("Making Birds, Beasts and Insects"), that of my school friends, in order to capture the children's interest. The considerable variation in Snook's annotation, clarity of drawing and expansion to show small details would all be used as basis of discussion on the importance of clarity of communication.

By showing examples of children's work in the Stan sessions, I hoped to enable the children to feel confident that this was something they too could achieve. My tactic was to show them examples of drawings as products and examples of design drawings and ask them to distinguish between the two genres. This would then lead into sharing the Container / Journey metaphor, drawn as I talked, onto a flip chart. The children would immediately put this into practice by producing their own design drawings for a Stan puppet to go into an A4 envelope.

The Container / Journey terminology would then become part of the shared meanings established between me as teacher and the children as learners. The terminology would be used as part of classroom talk throughout the rest of the Programme, so that the children would extrapolate, from its constant usage in a range of applications and situations, the nature of design process.

My aim was that "Through good teaching the child can become self-consciously aware of his or her design capacity and be able to make deliberate use of it." (Baynes, 1992: 42) and I believed that if told how and why drawing could support their designing, children would be able to access the process as well as the genre.

4.3.4 The Comparison Class' Experience

Term Autumn (2000) first half	Project Badge-making
Autumn (2000) second half	Christmas activities
Spring (2001) first half	Snow-mobiles (related to topic on transport)
Spring (2001) second half	Moving pictures (QCA)
Summer (2001) first half	No D&T - SATS
Summer (2001) second half	Puppet-making
Autumn (2002) first half	Sandwich-making (QCA)
Autumn (2002) second half	No D&T but did Christmas activities in last 3 weeks

Table 8 : the Comparison Class Experience

In line with other local schools, my school was in the process of adopting the QCA schemes of work for all foundation subjects. In this transition year, Comparison Class experienced a mixture of school-devised units and units from the QCA scheme (Ex. 28 shows an example from the Moving Pictures Unit).



Ex. 28 : Moving Pictures

As Design and Technology Co-ordinator, I was keen that the school should continue to deliver successful Design and Technology units that had been devised and developed by year group teams, as I felt that this encouraged teachers to think more clearly about the objectives and outcomes of their own teaching. I could see a danger in simply "doing the QCA" at declarative level of teacher thinking and assuming that delivery of the practical content would automatically lead to learning of the lesson objectives. Although this was not the case in Year 2 (Miss N. is a very reflective teacher), I was less happy with the ethos of the Year 3 team.

Making observations of the teaching input to the Comparison Class would have involved arranging supply teacher cover for my Year 1 class, which, unfortunately, was only available for monitoring Literacy and Numeracy lessons. I had copies of the year group's medium term plans as part of my role as Design and Technology Co-ordinator, which indicated design objectives in broad terms, in line with QCA guidance. Miss N. did not make detailed personal notes for teaching the individual lessons.

The Comparison Class were encouraged to draw design ideas before making a product but did not receive the same explanation as the Focus Class as to the purpose of design drawing in terms of the Container / Journey metaphor. They had workbooks in which design ideas were recorded, which their teacher marked and made written comments on. In contrast, my Focus Class worked on loose paper, stored in folders once no longer needed by the children. Feedback on design ideas was informal, on-the-spot and verbal and the drawings were frequently annotated in consultation with the child to ensure I understood their design intentions. By using my 1999-2000 Year 1 class as the Comparison Class, I hoped that factors relating to my personal teaching style would be minimised and even at the end of the Programme in Jan.2002 I still had a viable teaching relationship with those who had been mine in Year 1. That their Year 2 classroom was just around the corner from my room and we saw each other daily throughout Year 2 helped to continue this relationship and I made short informal visits to see what they were doing in Design and Technology and take photographs. However, it was clear to me as the Programme progressed that the shared meanings which I was establishing with the Focus Class were not shared with the Comparison Class. This was more an instinctive feel than something than could be documented or quantified. I could just sense by their reaction to my explanations that the Comparison Class and I were no longer quite on the same wave-length.

4.3.5 External Events that Affected the Programme Delivery

The external events outlined here form the backdrop against which the Programme was conducted. Ely et al. (1997) stress the "certain emotional sturdiness" required of those who embark on naturalistic research. The contrast between the real happenings in the world of people and the positivist paradigm of controlled experimentation on large samples of subjects could not have been starker.

There was a long-drawn out muddle about how the 1999-2000 Year 1s would transfer to Year 2 in Sept.2001, involving differing interpretations and applications of the Government's "no more than 30" ruling for a Key Stage 1 class and the problem that the school's admission number of 96 would involve mixing Key Stage 1 and Foundation Stage children. The Local Education Authority agreed to an extra teacher which would mean the four Year 1 classes of 24 children remaining as such into Year 2. These were three parallel classes of September-May birthdays and a younger class of Summer-born children. My Focus and Comparison Classes were two of the older classes. However, the departure of another member of staff at Christmas meant splitting the Summer-born class to make an inevitable 32 in each of the other three. Although these extra children were present for all the sessions from January onwards, I did not include them in the research, which had begun without them.

The Comparison Class group did not remain as 24 children. One moved away over the summer break, during the research period another emigrated after a severe house fire, another gained a place in a small private school and a child with mild learning difficulties and behavioural problems was excluded when he became a danger to others. A child who remained in school was diagnosed as having Muscular Dystrophy and became confined to a wheelchair but I continued to include him in the data since he did not deteriorate cognitively across the Programme and had sensitive adult support.

SECTION 4 - Structured Phase

In January 2000, the most disturbed child I have ever taught joined my Year 1 class. He seemed to be living out the fantasy roles of violent computer games and was excluded from school in March. At the same time, another child in my Year 1 class was terminally ill with cancer, dying in April. Both these situations deeply affected the rest of my class and there were occasions when I needed to give them priority over my research. The session planned for the day on which she died was, obviously, cancelled at short notice. I needed to be with my class when they were told.

In Autumn 2001 the sessions were changed to Friday afternoons (not my preferred time-slot) to fit in with the Year 3 timetable. The Ofsted inspection in the last week of the first half-term in Autumn 2001 caused two sessions to be lost: the Focus Class' teacher did not want to swap on the Friday afternoon before the Inspection and nobody, including myself, wanted to do anything other than celebrate with their own class on the following Friday once it was all over.

For the final Assessment Task I had originally planned to use an extended, three-week activity (making a puppet theatre) which would enable me to ascertain the children's ability to use drawing to plan, develop and adapt their plans across the duration of a longer project. This was not to be, since I was by then working at Canterbury Christ Church University College and was not able to visit the school weekly. The single session needed to fit in with work in other areas of the curriculum and so the final Assessment Task was related to the topic of Greek myths that formed part of Year 3's Literacy Hour: design and make a model of King Minas' maze to help Theseus get to the Minotaur and escape as easily as possible. This was a task which I had not conducted before and I worried a great deal about the wisdom of it afterwards.

However, despite all these pressures and unforeseen circumstances, the Programme was completed. I felt that, on balance, the unforeseen circumstances added to, rather than, detracted from the validity of the research. It was not conducted in a laboratory setting, away from the real world of children's real lives or the reality of corporate school life. That we began one session with a child asking "Miss, are you sad about S.?" or that I had to defuse the ongoing rivalry with the class next door (which degenerated into racist name-calling and fist-fights one Thursday lunchtime) did not contaminate the data in the way that it might have done if my view of educational research and data gathering had been different. If my Programme was going to make a difference to how the children used drawing for designing and if being involved in my research was going to have more than a marginal effect on their lives, then I could not deny the entry of the rest of our lives in through the classroom door just because I was now "conducting research".

4.4 Evaluation of Teaching Input to Focus Class

The success of the Programme was to be formally assessed through the Assessment Tasks and the Focus Class' (hopefully) greater learning about designing would be revealed by comparing their achievements with those of the Comparison Class. Table 5 indicated Programme objectives, project by project. This section of the thesis, therefore, addresses the question : how far were these objectives met in the course of the Programme?

The major source for this material was my Log Book, which was maintained throughout all sessions of the Programme, which contained notes on children's working and an evaluative commentary on each session written immediately after delivery. Themes and issues which surfaced across the Programme (summarised in Section 4.5.3) were also to emerge as important differences between the two classes, suggesting that these were specific outcomes of the Programme.

(Sessions 2-5) Fantasy Figures

When shown the large pictures of the aliens and asked to make them in plasticine and matchsticks, the children immediately produced variations, calling out "*Mine's got …*" (whatever they had drawn that was different to the picture) and were surprised when I praised those who had copied the figures exactly. They did not realise they were being given specific instructions to follow rather than being asked to explore their own ideas. Donaldson (1992) also found that 6 year old children did not adhere to "this problem and this problem only"

The Clothes-making session began with a demonstration of how to use newspaper to make a pattern which fitted the alien (by trial and error). However, the children began making straight away in cloth, rather than choosing their fabrics and making newspaper patterns. So I stopped the whole class and explained again about why we need to plan - waste of cloth, disappointment when it doesn't fit, and so on.

Again, with the Transporter activity, following the instructions was the most difficult part. I said *"Talk to your partner and then each draw your best idea"* but the children began drawing immediately without discussion. I was clearly using different models of designing to that to which they were previously accustomed. Designing and making the House was less difficult as they were beginning to realise that they needed to listen carefully to instructions and I was beginning to pick up the role of language in designing:

Need to be less fixated about recording in drawing if writing in sentences is more natural for the children and appropriate for the activity. They had told each other how they would make it - why not write that down?



Ex. 29 : Completed Wagwum Set

(Sessions 7-10) Visual Analogies

This work in the first half of the Spring term was intended to develop analogical fluency as a generic skill. On reflection, I became less convinced that it would and more convinced that analogy-in-context is a basic human skill, which exposure to abstract analogical reasoning tasks would do little to improve. However, I think that these activities enhanced the children's learning in an appropriate way, but not, perhaps, for the reasons for which I had originally perceived.

This was especially true of the Iron's Analogies Games, where problems emerged when they were asked to devise their own games. On reflection, I realised that the game I made for them to play had no "wrong" pieces, so it could be solved by shape matching and not by analogy. Although my explanation and summary at each stage emphasised the visual analogy, it was not what the children were seeing in the activity.

Continuing the game-making for a second week was the right decision. The children had half-understood and were able to devise their own games, played them together and were keen to take them home to play with their families. The activity taught children the need to accept the rules of a game and to reason within those rules. This was to prove to be an important contributor to the Focus Class' design capability.

(Sessions 11-15) Ideas on a Journey

My initial reaction to the series was one of real satisfaction. Even without formally assessing the results, I could see that they were considerably better than anything I had achieved with Year 2 before. I was looking at Year 3 work to make comparisons. Certainly some of them had produced work that was comparable to the best of the Year 3s in 1998. I felt "*This works*".

There were two different approaches emerging on opposite sides of the classroom, centred around the main boys' and girls' friendship groups. Gender differences are discussed in Section 5.9 but this was the only point during the delivery of the Programme at which I was really aware of a difference between the boys' and girls' approaches to designing.

The girls tended to opt for a single idea which they progressed towards making (Natasha's Round Stan (Ex. 30) is typical) usually annotated and sometimes using writing rather than drawing where this was more appropriate (for example, Jolene wrote a list of crisp flavours that Stan might choose from). Their level of discussion with each other whilst drawing was cursory. They would trade ideas but were working in parallel rather than together.

Ex. 30 : Focus Class Stans (a: main girls' group)





The main boys' group, however, seated at the other side of the room designed by discussion, as the debriefing transcript that follows Ex. 31, Randal's Flat Stan, overleaf reveals:

Ex. 31 : Focus Class Stans (b: main boys' group)





Randal: I did that one and then I thought of that one.... and that one...erm....I got that one from Craig. Then I did that one. Craig didn't like that one. He thought it looked like a crocodile (giggles). So he said why don't you do Superman. So I did. But I did that one. (Pointing to the first drawing in the second row - the spirals on the drawing are there -under the puppet's clothes!)

How far this was gender-related or due to the seating arrangement in the room was difficult to say. The main boys' and girl's friendship groups were seated on the opposite sides of the room, with the younger Summer-born children in the middle. This might mean simply that cross-fertilisation of ideas was not happening across this barrier. Martin (Ex. 32), who was not part of the main boys' group, shows many of the characteristics of the girls' work :



Ex. 32 : Focus Class Stans (c)

To appreciate the difference between the Focus Class' work at this stage of the Programme (immediately after the Container / Journey metaphor had been used to explain designing) and that of the Exploratory phase children, Natasha's work should be compared both to Ex.9 (by her older sister Nikki as a Year 4) and then to typical Exploratory Phase Yr. 2 work (Exx.13-15).

(Sessions 16-19) Modelling in other Media

For the marble run, socially competent children worked well together, the less social did not. Some pairs tried to have a board each or have "my half and your half". Randal and Carl built a wonderful joint product which changed and evolved continually for the whole session, which was at its best halfway through; I wish I had taken the photo then.



Ex. 33 : Marble Runs

I had planned for making beads from air-drying clay but no clay in school meant that paper beads were made instead. The children, boys and girls, thoroughly enjoyed the session, possibly because it was novel, simple and produced instant results. I wrote in my Log Book: *They all wanted to wear their necklaces to story-time, so they definitely had fun.* At the end of the term, I asked the children which were their favourite activities and this one scored highly. To me, this raised the issue of the need for an external client in Design and Technology for this age group. I wondered whether the children had enjoyed it more because they had made something for themselves.

These "no drawing" sessions enabled me to apply the Container / Journey metaphor to designing as a whole and not just to the drawing, which was used only for the development of decoration and logos for the sides of the Buggy. for which I produced a side, front and back design drawing on a small sheet of paper, which "*Since it is too small for everyone to see*" I redrew on the flip chart and explained to them how my thoughts had gone on a journey. The terminology seemed to have become part of our shared design vocabulary.

My Log Book comments reflect the child-centred view of a car :

They worked purposefully and well - where did the time go? Many of them made really good seats etc. inside, but some had them in funny places - lower than the axles. Interesting that they don't really think about where parts of a car are in relation to each other. Steering wheels were far away from seats. No thought of an engine - the steering wheel went into the front of the car.

(Sessions 21-25) Extended Project - Sandals

Appendix J documents this project.

The children's choice of drawing or text for recording their product analysis was interesting. They wrote about the sandals that they observed but drew their ideas for sandals they would like to make. Words were used for observations; drawings for ideas generation and futuring.

Many children had difficulty understanding that the prototype made by drawing round their foot was not part of the final product. Some made two complete sandals in this thin card and seemed to find it impossible to grasp that it was just a pattern. It seemed that making a 3-D mock-up was harder for them to understand than drawing. I had this problem with Pandy's suitcase with Year 1 in 1998.

Everyone had a pair of fitting sandals by the end of Session 5 and the children were delighted with the results and walked around in them for the rest of the afternoon. I think they felt they had made something for *themselves* rather than the teacher, which casts interesting light on the *addressing client need* aspect of D&T; perhaps young children *need* to be their own client. Certainly much customer satisfaction in evidence. A good end to the Summer term.



Ex.34 :Sandals (the main girls' group are to the right of the photo with their hands in the air)

(Sessions 26-28) Team-working : Food

A range of team-working approaches were used, from arguing their way through the whole menu to delegating a different role to each team member. One group allocated roles, who then produced (and drew) several ideas which the team then discussed and then decided their final menu. Once making the dioramas, most groups delegated different parts to individual members. The presentations at the end of the session showed that the children were able to give account of their working methods and how they had worked as a group. I felt this was more valuable than the details of their models.

From previous experience, I had anticipated problems with recording of materials, although I stressed this loudly several times - *"or I won't know what to bring next week"* Consequently, there were embarrassed giggles when I handed back their discussion sheets and they realised they had forgotten what they would need (followed by relief when I produced two boxes of "possibly suitable" materials from behind the desk).

Making the sandwiches, I believed, would give the children opportunities to for thinking about presentation, as well as having an enjoyable afternoon testing different breads and spreads. The health issues associated with preparing food in an ordinary classroom meant that sharing each others sandwiches and trying each other's food combinations would, unfortunately, not be possible. I need not have worried about this. Once they had made their plate of sandwiches, they had no intentions of sharing them with anyone else.

(Sessions 29-32) Designing for others

There were three main areas of improvement based on prior learning:

- They understood prototyping and mock-ups: "Now I can do a really good one for my Mum."
- They had definitely got the materials message, every group had considered and recorded what they would make the gift from
- They could provide constructive feedback : they tended to say what they liked and made helpful suggestions rather than *"you haven't..."*

Trialling each other's instructions and gift led to changes in their own work as they realised that others had better ideas or produced more workable instructions. Although initially they were resistant to the idea of swapping instructions and making each other's, rather than making their own idea immediately, there was a genuine sense of enjoyable challenge once the first few pairs had swapped and comments were bandied across the room. It was all done in remarkable good humour and no one got upset when their "trialler" came back saying "I can't make this because.."

The role of the real client kicked in when I insisted that they put together the kit for Year 1 before finishing their own gift or the younger children would not have time to make theirs. The Focus Class could finish their own gift with own teacher later. My Log Book records:

Panic ensued as they realised that their work really was going to Year 1 and that the younger children would have a view on their success!

The Focus Class worked in groups to design the kits, but then each child produced their own kit for a Year 1 child. Many of them wanted to know the name of the child to whom the kit would be going, so that they could put a message inside. I had envisaged the work going from a group of Focus Class children to a group of my Year 1s. I had not anticipated this level of personalisation and it seemed to be the more capable designers who asked for a named child. This might indicate that the level of engagement with the task was related to having a specific client in mind (as Stables (1993) observed). A hypothetical "Year 1 child" did not enable them visualise their kit in use.

The Year 1 children, in turn, wanted to know the name of the older child who had assembled the kit. Some of the Focus Class came to my classroom to show to me their own completed gifts and were delighted to find some of my Year 1s were working on theirs: *"They're really making them!"*

4.5 Reflections on Programme Delivery

Planning the practical aspects of the Programme delivery was relatively straightforward. From the Exploratory Phase I had a range of activities which could be used either for Teaching Input or for Assessment Tasks. Designing and evaluating teaching programmes was part of my work as a teacher and so I did not feel daunted by issues such as breaking down overall aims into identifiable objectives or ensuring progression across the Programme. This was equally true of evaluating the children's learning whilst teaching. Devising validation strategies and informing stake-holders were organisational tasks which needed to be done but I did not consider these especially problematic. The external events that affected Programme delivery (outlined in Section 4.3.4) were the kinds of things which affect any long-term teaching plans, being adaptable and flexible in the light of such changes are part of being a teacher.

Making the right decisions about the kind of data to collect ahead of conducting the Programme was more tricky. In non-participant research strategies, a small-scale pilot study is frequently conducted between identifying the problem and conducting the full study, both of which are conducted within the same paradigm. I had shifted from an exploratory to a delivery-plus-test model, which meant that although I took my conceptual understandings with me into the Structured Phase, much of the data-collection detail was aspirational at planning stage.

4.5.1 Evolution of Theoretical Understanding

It was inevitable that across the delivery of such an extended Programme that my understanding and ideas would continue to develop and refine as I reflected in praxis. As an experienced teacher, such reflection is second nature. As a researcher, such reflection revolved around the theoretical foundations on which the Programme was based.

I began the Structured Phase believing in a causal relationship between analogical reasoning ability and design drawing capability and although I still hold to that view, my understanding of how that relationship works changed across the period of time covered by the delivery of the Programme. When I designed the Programme, I believed that conducting activities in the Spring Term 2001 to develop the Focus Class' analogical reasoning skills would enhance their facility with design drawing. By the time I had conducted these activities, I no longer believed that the skill development worked that way. What they learnt from these activities, I came to believe, was accepting the rules of the game, which was to prove to be essential learning for the Focus Class.

SECTION 4 - Structured Phase

What I had come to believe in the interim was that all learning is an extrapolation from what is already known. Green (2000) argues that teachers assume in practice that young children can and do learn through analogy and my own teaching experience supports his stance. Skills and knowledge transfer is predicated upon analogical reasoning. My Container / Journey metaphor for design drawing would work, therefore, because it was an appropriate metaphor for the design process, not just because it was an analogy. Improving children's analogical reasoning skills would not improve the appropriateness of the metaphor. However, teaching them to *play the game within the rules as given* would improve both their analogical skills and their design capability by teaching them to look for solutions within the problem as defined.

However, more important differences appeared to surface at Easter, after my input to the Focus Class of the Container / Journey metaphor. I felt that there was considerable learning of the nature of designing at the first presentation of this metaphor. I went back to my own classroom afterwards, flushed with success, and said to one of my Teaching Assistants, *"I now know how to teach D&T"* to which she replied *"Well, you've been doing it a long time"*.

From informal observation, it seemed that, once the Focus Class had grasped the idea that design was an on-going thing and that change and development were part of the process to which drawing could contribute, that they were able to build on and apply that understanding to a range of media. I had hoped that would be so and was gratified that, despite the inevitable interruptions to the Programme, the children did not lose this.

4.5.2 Emergent Themes from Evaluation of Teaching Input

The Container / Journey Metaphor:

The key concept I wished to communicate with the Focus Class was that of seeing design as taking ideas on a Journey. I believed from my informal observations of the children's response to the Stan activities in which the Container / Journey was embedded that this was immediately successful. Using these activities meant that I had instant feedback on the success of the metaphor, through comparison with findings from previous presentations.

From the introduction of the metaphor onwards these children felt more like Year 3s than Year 2s. I was aware that and I was speaking to them at the level I had previously used with the Year 3 class to whom I taught Design & Technology weekly in 1998-9. It would seem that the explanation of the nature of designing in terms of a journey and how to use drawing to record stages along the way had enabled them to access the genre.
Learning to play by the rules of the game:

Problem-solving, I believed from my reading of Wittgenstein (1969), was much to do with exploring the rules of the game and seeking to resolve the discrepancies between the rules and the problem. Liddament (1996) applied Wittgenstein's language games to designing. I had come to believe that creative designing was embedded within the resolution of the rules / problem dichotomy.

If my Focus Class were going to learn how to use drawing for designing, they had to learn to play by my rules. The pragmatics of the tight time schedule of the Programme required that they did so quickly. They needed educating into Donaldson's (1992) "this problem and this problem only" that she identified as so difficult for 6 year olds. My tightly teacher-led tasks at the beginning of the Programme were designed to ensure that this happened. This then became the foundation for accepting my definition of what design drawing was all about and enabled them to successfully access both the genre and the underlying concept.

<u>Language:</u>

a) Shared Design Vocabulary: I used and the children adopted appropriate vocabulary with which to discuss their work. Asking them to explain their thought process encouraged their reflective skills and meta-cognitive awareness. They began to learn what was important to me and that I would want to note down in my Log Book. Increasingly, they would bring their drawing or come and tell me the conclusions to their discussion. They were learning that designing as a process was valuable. They were not tempted to throw early attempts in the bin as "wrong", they would come to me with several drawings and explain how their thoughts had developed across several iterations.

b) Design by Discussion: The main boys group mainly designed by discussion, frequently producing almost identical design sheets. Debriefings and in-context questioning enabled me to understand that this was not "copying" but co-operation and that the apparent randomness of some of the ideas recorded on paper was due to this swapping and sharing of ideas. Craig was always full of helpful suggestions to others and his input appeared in several other children's drawings.

Not only did this repartee generate and stimulate ideas, it also provided a safe environment in which to hone evaluation skills. It enabled children to find out each others strengths and abilities, which was important for team-work. Not surprisingly, pairings and groupings from within the chatty, sociable boys' group produced successful designs. There were dangers in

this, however. Randal and Craig's marble run was never finished. It became part of an on-going conversation that they did not want to dismantle at the end of the lesson.

c) Multiple Literacies: Although I wanted the children to use drawing to support their thinking, I did not want to fall into the trap of making writing less valued or of taking an artificially contrived role. The children used more writing than I had expected (or that I had observed amongst the 1998-9 cohort of the Exploratory Phase) and I endeavoured to encourage this to be used in an interactive, exploratory way (listing variations for example). I hoped that the children would be able to use drawing and writing to enhance each other.

They used writing for lists (e.g. of materials), labelling of parts and descriptive or procedural text (e.g. "First I will cut out...") Writing sometimes seemed to be used when they were less confident of their graphic capability, for example, labelling to clarify communication, to tell me what part of a drawing was meant to be. I was interested, yet should not have been surprised, in the division between use of writing to describe the sandals they were shown and the use of drawing to record their design ideas. Discussion of the sandals' features translated directly into writing, whereas part-formed images in their heads were recorded as graphics.

Communication

I wanted the Focus Class initially to be able to use drawing to support their own design thinking. Later in the Programme (Autumn term 2001), they were asked to work collaboratively and at this point drawing for communication became important. Since they freely discussed and shared ideas, many children were already using drawing in this way. The video transcript of Craig and Noel, for example, shows their ability to interpret and discuss each others drawings.

One of my key teaching points in the showing of Snook's animals was clarity of communication and I specifically used the words "so someone else could make it". Ways of enabling another person to do so were discussed and this input had immediate impact on the quality of their drawings. Despite the message needed repeating in Autumn 2001, they were able to take this on board and the materials were specified without the need for prompting in the kit-making project.

Evaluation of ideas

My view of design as Container / Journey implies constant feedback and iterations, which drawing can support by enabling opportunities to reflect on half-formed ideas. I wanted the

children to learn to evaluate as they generated design ideas and reflect upon possibilities and constraints in order to minimise trying to adapt a half-constructed idea which is not working.

This was written into the Programme through the team-working of the Food project and also the trialling of each others' designs for the Kit-making. However, the ability to do this was built on the foundation of product evaluation (Slippers) and on the expectation that designing would involve thinking of several ways of solving a problem or moving a possibility towards a design solution, which was inherent in the Container / Journey metaphor.

Addressing the Needs of the Client

The children showed the greatest engagement with the activities that resulted in something for themselves: the name plate for their bedroom door, the necklaces and the slippers. They became highly focused on the kit-making activity once I was setting time limits for completion so that their clients could assemble the kits.

Stables (1993) identifies three "clients" within the school setting : consumer, sponsor (usually the teacher) and designer (the child) and argues that the roles of each need to be made explicit in order to maximise children's motivation through their understanding of the purpose of the task. My observations accorded with Stables', that where children could clearly identify all three they showed greatest engagement with the task. But this was especially so if the consumer was themselves.

This caused me to wonder about the role of the external client and to feel that it is important to give children of this age a balanced diet of activities. They need real clients (parents for Christmas gifts, for example), they need to be stretched to imagine the needs of people they will not meet, but also they need to make things for themselves. I came to believe that it was an important part of Design and Technology education to have experience of making things for themselves and feeling the satisfaction of doing so.

SECTION 5 - Analysis of the Assessment Tasks

5.1 Introduction to Section 5	185
5.2 Evaluation of Assessment Tasks	187
5.3 Qualitative Evaluation of Children's Performance	189
5.4 The Quantitative Analysis Instrument	212
5.5 Understanding the Purpose of the Design Drawing	234
5.6 Drawing Types	246
5.7 Dimensions of Design Drawing	257
5.8 The Techniques Ticksheet	279
5.9 Gender Differences	296
5.10 Reflections on Analysis of the Assessment Tasks	303

5.1 Introduction to Section 5

The Assessment Tasks were the main vehicle for evaluation of the success of the Programme and were the only part of the Programme submitted to quantified analysis. In the account of the qualitative analysis and in each section of the reporting of the findings of the quantified analysis, emergent themes are identified, which contribute towards the Key Themes that Emerged from the Research (Section 6.2).

Evaluation of Assessment Tasks (Section 5.2)

Although, for each task, the children were to use drawing to record and develop design ideas prior to making, each of the Assessment Tasks had different characteristics and had slightly different demands. This allowed for some measure of comparison to be made between the way the children responded to different kinds of task. Section 5.2 outlines these differences.

Qualitative Evaluation of Children's Performance (Section 5.3)

Video recordings of children at work, together with audio-recorded de-briefing sessions and the use of a Log Book to make notes as they worked, all contributed observations about children's performance. The original intention was to analyse these data using a series of databases. It quickly became apparent, however, that the overall sense of growing capability was embedded within each situation and the loss of context through filtering and sorting hindered rather than enhanced reflective evaluation. The excerpts from the qualitative records quoted within Section 5.3 are selected as illustrative of the reflective reporting in context.

The Quantitative Analysis Instrument (Section 5.4)

The evolution of the quantified analysis methodology and the development of the analysis instrument is recounted in Section 5.4. Qualitative judgements on children's design drawings were converted into quantified data whose results were recorded and maintained as a set of child-level spreadsheets, which were then collated for class-level analysis. Section 5.4 traces the development of this quantitative analysis instrument from a Grid and Ticksheet, containing, respectively, aspects of design capability and specific skills, to a multi-layered analysis tool, to analyse and compare children's understanding of the purpose of design drawing, their choice of Drawing Type, their use of drawing with respect to different dimensions of design drawing capability and their demonstration of specific skills, such as annotation.

Understanding the Purpose of the Design Drawing (Section 5.5)

In the holistic assessment model that emerged (Section 5.4.3) the child's understanding of the purpose of the drawing is not only seen as paramount to the way that they subsequently use the drawing to support their designing, but as central to the assessment process. Section 5.5 presents the analysis of the quantified data relating to the way in which the children each of the two classes understood the purpose of drawing to support the development of their design ideas.

Drawing Types (Section 5.6)

The classification of children's drawings into Drawing Types developed during the Exploratory Phase (Section 3.3.5) is applied to the drawings of the Structured Phase. This is used to compare the children's choices about how to use drawing to support designing, as well as to examine the relationship between the choice of Drawing Types and the children's understanding of the purpose of drawing for designing (Section 5.6.2).

Dimensions of Design Drawing (Section 5.7)

Section 5.7 presents the findings of the analysis of children's ability to use drawing to support the development of their design ideas with respect to eight dimensions of design drawing. These were developed from the dimensions of design capability identified by Kimbell et al. (1991) in the light of observations made about children's design drawings in both the Exploratory and Structured Phases of the research.

The Techniques Ticksheet (Section 5.8)

Specific skills (such as the use of annotation or the recording of colour or materials to be used) were recorded on the Techniques Ticksheet. Section 5.8 examines these data and makes comparisons between the two classes' use of such techniques.

Gender Differences (Section 5.9)

Although not a main focus of the research, Section 5.9 examines the differences that emerged between the gender groups in each class.

5.2 Evaluation of Assessment Tasks

The Assessment Tasks were originally planned to occur at the end of each school term but various external events prevented the strict adherence to this plan. The opportunity to add the Surprise Card task to the Programme could not be missed and needed to be delivered at the same time of year as it had been to the children of the Enriching Literacy through Design and Technology Evaluation Project. The same considerations held for the placement of Pandy's Suitcase at the end of the Summer term.

However, this meant that three tasks (Easter, Card and Suitcase) were conducted quite close together and then there was a six month gap before the Maze task in January 2002. This was unavoidable given the external circumstances. Ofsted inspectors were in school at the end of the first half of the Autumn term and Christmas activities dominated the curriculum in December. I was determined to conduct the kit-making activity with the Focus Class as it was an up-dated parallel to the Insects activity of the Exploratory Phase and I wanted to evaluate the changes I had made to the activity as much as their designing skills. Delaying the final Assessment Task, however, meant that it acquired the status of a test of sustained changes beyond input to the Focus Class.

The design drawings produced by both classes in response to this final Assessment Task (the Maze) were very different to those produced in the previous three tasks (Easter, Card and Suitcase). I was especially frustrated that the my Focus Class' drawings lacked all the sophistication and clarity of communication I had endeavoured to teach them during the Autumn term. The most sophisticated Focus Class Maze drawings, for example, used a legend next to bird's-eye-view drawings to show which colour lines represented the string but no construction detail beyond a list of materials. The same children had included measurements in centimetres, a range of decorative features and detailed construction method for both the Easter Egg Holder and Suitcase.

However, during the process of the quantified analysis of the Assessment Tasks, it became clear that there were important differences within the tasks that I had not anticipated would affect the way that children use drawing to support their design thinking.

Frosty and Maze tasks were *problem solving scenarios* whereas the other three activities required the *design of a product*. The objects that the children were asked to produce for the Frosty and Maze activities were 3-dimensional models of a problem solution, which led to much less recording of materials, colour and construction technique. They used the drawing to clarify and solve the problem and then worked out how to model the solution directly with the

SECTION 5 - Analysis of the Assessment Tasks

materials. For product design tasks, the children were more likely to be planning a complete design solution, including colour, decorative features and construction details. It would seem that the children dealt differently with solutions that they regarded as "models" to those which they regarded as "products". Fortunately, the teaching input to both classes had majored on designing products and so there was parity of experience and one class were not at a disadvantage with respect to the other.

In her consideration of young children's problem-solving strategies, Roden (1997) wondered whether they might exhibit a different range or combination of strategies for different situations and which strategies might be common and which might be task-specific. This was a question I could apply to my data :

- comparing Pizza, Easter & Suitcase for development in drawing for product design;
- comparing Frosty and Maze for development in drawing to support problem solving;
- comparing structured delivery to less structured
- comparing these sets for similarities and differences.

In terms of Programme success, however, it was the *comparison between the classes* not the *comparability of the activities* which would indicate the hoped-for greater progress in my Focus Class, and their (hopefully) richer responses in a range of situations would be compatible with my aim of improving their design drawing capability regardless of task or presentation.

5.3 Qualitative Evaluations of Children's Performance

My hopes for the qualitative observations of children working was that they would give me an understanding of children's processes in using drawing to support designing and that this would provide the context in which the quantified analysis of those drawings would be pursued. These observations enabled me to determine the children's understandings of the role of drawing for designing, especially in terms of its relationship with making. The aim of this section of the thesis is to demonstrate the ways in which the two classes responded to the Assessment Tasks (supported by quotations from transcripts and Log Book entries) and to indicate the overall trends within these observations.

5.3.1 Assessment Tasks

In the excerpts from my Log Book, used as examples in this section, italics for observed actions and normal font for reported speech.

Drawings and products by Target Groups from both classes can be found in the Appendices: Appendix C (Pizza), E (Frosty), G (Easter), I (Card), K (Suitcase) and L (Maze).

5.3.1a Assessment Task 1: Design a pizza

For this first session with each class in October 2000, I had the assistance of Mrs.M., a Learning Support Assistant who worked with me in my Year 1 class. The Comparison Class, had the advantage of being familiar with me and my expectations and appeared to do better.

Evaluation of Session from Log Book:

Throughout session, Mrs.M & I commented how much more interesting they were than Focus Class. And how different & diverse their ideas were. In discussion with Mrs.L (Year 1 Teaching Assistant) later, decided it was not just that they were more creative, but also they knew me so well and knew what they were allowed to do bend the rules a bit. Focus Class had simply followed the instructions. Comments from my Log Book on individual children also bear this out:

Focus Class

Ellie	Cut out chosen design from design sheet			
Louise	That's pepperoni! Pipe cleaner bent round			
Chloe	Decorated design sheet. Later added "apples" to design sheet after I asked her if			
	that was it - she'd just covered the pizza in yellow "cheese"			
Noel	Has stuck black tissue underneath pizza and nothing on top.			

Comparison Class

drawings.

Zara	Arranged matchsticks on pizza, marked their position "So if anyone knocks it I'll
	know where they're meant to be" Later, tried to put red tissue between match sticks
	(already stuck on) to be the sauce. To Mrs.M: "I want to do more detail." Re. Putting
	base colour on round the top layer - seen this with Year 4s - insect kits.
Kirsty	Modelling by drawing: Me "Which one are you going to make?" Kirsty "The cheese
	one" Me "Which one is that?" Kirsty pointed to first. Me "What are the others?" -

- "Forgotten now" Emma To Mrs.M "Is this pepperoni colour?" Wanted me to cut circles - how many? Counted the circles on her design drawing. Only child to meticulously colour all
- Garth I can't make it that size. *pointing to drawing. Only child to show any concern about size discrepancy.*

The relationship which we both had with the Comparison Class children can be seen in the recording of exchanges, whereas Focus Class children did not readily discuss their work with us. However, the quantified analysis of their work would show little difference in achievement for the Pizza task. This would seem to bear out Taylor's (1982) argument for the merits of quantitative methodologies (referred to in Section 2.2.3). It would have been tempting to view the Comparison Class as better designers, whereas in fact they were more confident communicators.

The drawback of recording and reflecting on the interesting is that it can be viewed as typical, whereas it is atypical of the whole group. The following is a Log Book entry in which I recorded reflections as well as observations but was careful to note as non-typical:

Wendy, Ellis and Alice : Copied design by drawing it on pizza. Sitting together. Everyone else (both classes) drew the outline of the pizza and filled with materials. Were these children more sophisticated or was just this their interpretation of how to do the task? Did this technique crystallise the mind's eye image or did it's freshness evaporate in the copying process?

Further observations and quantified analysis of these three girls drawings on other tasks, would now lead me to suggest that the technique was Ellis' idea and the other two girls, as less confident personalities, followed her lead. Such insights into the complexity of social settings require the interaction of both qualitative and quantitative methodologies. Wendy is a confident child who finds designing difficult; Alice is very creative but extremely unsure of herself in unfamiliar situations. She was an elective mute during Years R & 1. Such heuristic knowledge of the children enabled greater depth of understanding of their actions.

5.3.1b Assessment Task 2: Frosty the Snowman

Drawings and products by the Target Groups from both classes can be found in Appendix E.

This was conducted with each class during the first two weeks of the Spring term. The intention was to video the Target Group from each class in order to observe at leisure their use of drawing for designing and the extent to which they consulted these designs once making. This was my first experience of using a video camera and although the Focus Class group were recorded successfully, I was unable to record the Comparison Class group due to problems with the school's video camera. I had my camera available as back-up and so stayed near the Target Group for the duration of the session and took photographs of critical moments.

From the video of the Focus Class Target Group came this moment of two boys (Craig and Noel) discussing Noel's drawing (Ex.35, shown overleaf):

Craig : (prodding Noel's paper): What you could do is....like... have that and then that connected there.

Craig: Yeah. And then.... (moving his pencil about in the air over the paper)Craig: And, and... (waving his pencil over Noels paper in the same way, indicating what Noel should draw) And then that bit....

Example 35: Frosty (a) : Noel (Focus Class)



The video also revealed an alternative reason for children not making what they had drawn: that someone else's idea appears better. Damian had the idea of a tunnel through both mountains and under the lake. Randal spent most of the session trying to help him resolve the practical issues of making a tunnel out of rolled newspaper and making a pod to pass through it. Towards the end of the session, he hastily made himself one. This level of co-operative work was to prove a consistent feature of the main boys' group in the Focus Class.

In contrast to the boys' working together, Jolene and Natasha worked in silence and showed no interest in each others' work. Their only communication was on sharing sellotape or holding parts for each other whilst applying the tape. However, both girls produced successful designs and products. Jolene was the first child in the class to finish her work, having worked conscientiously and purposefully throughout the activity. Natasha used her drawing as a template, laying the sections of her bridge on it to check for size and curvature (Ex.36). The paler patches are where I have obliterated my annotations after scanning.



Ex. 36: Frosty: Natasha (Focus Class)

SECTION 5 - Analysis of the Assessment Tasks

Sitting near the Comparison Class Target Group rather than relying on the video to record the event gave me a much more intuitive feel for the children's actions and choices. The photographs of critical moments acted as an aide memoire for the session. The children appeared to be conducting two separate activities: they would draw one or more suitable ideas and then use the materials to design a different solution, or even several different solutions. Jordan drew a bridge but made several solutions, opting eventually for an aeroplane because his friend Alistair had made one. Holly and Zara did not consult during drawing but once making they bounced ideas off each other and moved their bridge-making through several iterations and false starts, using up most of a box of straws in the process.

These observations were characteristic of the Comparison Class and will be seen repeated across all Assessment Tasks. They were consistently less likely than the Focus Class to use drawing to work out a solution to the design problem, preferring to model directly in the construction materials, yet in Year 1 they had been given more opportunities than had the Focus Class to acquire designing skills. The quantified analysis supported these observations on the Comparison Class.

Reflections in my Log book on the Focus Class make observations which were later to be important in identifying differences in the use of drawing between problem-scenarios and product design activities:

Looking at drawings prior to analysis - Many children appear to have reverted to single-draw. Does this show that I was too specific in my explanation, that showing them the Lighthouse Keepers Lunch stopped them from having their own ideas, or is the use of drawing task-dependent? Perhaps they had their ideas in their heads and discussed them but didn't draw them. Perhaps they simply opted for their first solution and didn't attempt to think of others. Viewing of tape indicates that Noel & Craig discussed ideas as they drew - and did not put pencil to paper to record the ideas they were discussing.

The use of drawing *is* task-dependent. Once quantified analysis was undertaken, the observations made about Frosty would be revealed to be equally true of the Maze, leading to the realisation that these two tasks were problem-scenarios and that children use drawing differently in such activities than in product design.

5.3.1c Assessment Task 3: Easter Egg Holder

This was conducted with each class during the last two weeks of the Spring term. Drawings and products by the Target Groups from both classes can be found in Appendix G.

I used my Log Book to record something of importance about each child's work, either a record of an exchange between us, an observation which lasted more than a few moments or I sketched some of their work in progress. I was able to ask about any part of their drawing which I did not understand, for example: *Louise (Comparison Class): egg holder inside at the bottom.* The page from the Log Book used as illustration in Section 4.4.1 is from this Assessment Task session with the Focus Class.

Even as I explained the task, the Comparison Class children began to say "I know what I'm doing" and seemed to want to make an instant choice, for which they produced a labelled diagram. Once working, they found other things around the room to use rather than using the materials provided by me (some did not even use the card tube that was central to the problem). They seemed to be working on a "free play" model of designing. It looked as if they had been taught a technique (labelled diagrams) which they did not relate to problem-solving or making.

The Focus Class, on the other hand, used the paper to generate ideas which related more closely to their making and used the materials provided. They appeared to understand that they were solving this particular problem, not a related one of their own, and engaged in the task as stated.

Audio recordings were made of the debriefing of the Target Groups for both classes at Easter: with me immediately after the session and later that week by Mrs.J (an experienced Learning Support Assistant studying psychology with the Open University). I provided a list of questions based on the task criteria and on the child's response (e.g. "You seem to have drawn several Easter Egg Holders, can you tell me why?")

The following pages provide examples (Exx. 37-44) and excerpts from four of the debriefing interviews to illustrate the difference in approach within the two classes. The interviews each highlight different aspects of children's design strategies and the role of drawing to support their designing.

Maria (Focus Class)



Ex. 37 : Easter Egg Holder (a); design drawing, Maria (Focus Class)

I had the following conversation with Maria whilst she was making the Easter Egg Holder. She did not appear to have made much progress. There were some pieces of ribbon on the table but she seemed to be doing more talking to Stacey than getting on with her own work. The joining arcs had not been drawn between the items on the page at this stage.

Me:	Hi Maria, what are you doing?
Maria:	That one (Pointing to second picture.)
Me:	Is this a one with stripes, one with a square and one with a triangle?
Maria:	Mm.
Me:	So what are you doing now?

Maria: That one (Again pointing to second picture.)

Debriefing her later, Mrs. J. had the benefit of the arcs and Maria was a more forth-coming:

- Mrs.J.: Tell me about how you have drawn this, Maria.
- *Maria:* That's the tube. That's like the tube with a hole, square hole, I thought like to see the egg in. And that's how I done it.
- Mrs.J.: So what's this?
- Maria: Oh that's like.. I just draw, drew another one.
- Mrs.J.: Why was that?
- Maria: Dunno, cos, like, I was just thinking like what it might be like and I thought, like, if I had a triangle hole would be different.
- *Mrs.J.:* So is this the egg inside?
- Maria: Yeah.

SECTION 5 - Analysis of the Assessment Tasks

It would seem that after her conversation with me, Maria realised that her drawing did not express her intentions. I had seen it as several different ideas, but the first drawing is the outside of the tube and the subsequent drawings are ideas about being able to see the egg inside.

- *Mrs. J:* So did you not do a hole in the end? In the side, I mean.... in the end, did you not do a hole in the side, after all?
- Maria: (giggles at Mrs. J. getting herself in knots here) No it was too tough. The tube. I couldn't get the scissors through. That's why I didn't get as much done as Stacey. Because it was ages trying to do the hole and in the end I thought I'd better do the stripes or I wouldn't finish cos Mrs. Hope was saying there was only 10 minutes. So I got another tube cos... there was a spare one on Miss' desk.... and I did it again quickly and my first one was messed up.



Ex. 38 : Easter Egg Holder (b); finished product, Maria (Focus Class)

By putting together the Log Book notes, the photograph and the tape-recorded debriefing conversation with Mrs.J., the whole picture of Maria's design strategy can be seen. When I spoke to her, she was probably contemplating how to make the hole in the tube. Her reference to Stacey in conversation with Mrs.J. shows her interest in her friend's work (she frequently relied on other children and her class teacher was even concerned about her copying in the SATs tests). She realised that her drawing did not convey her design intentions and so added the joining arcs. This was a strategy she would use again (see Section 5.3.1e). Helping herself to the spare tube as a way of resolving the mess she had got into is interesting. I did not notice she had done this!

Craig (Focus Class)



Ex. 39 : Easter Egg Holder (c); design drawing, Craig (Focus Class)

- Craig: I decided to cut the tube because it's too wide. That's that bit there.. that's scissors. Then sellotape to make it thinner. That's it bended round. That's the top - that circle. That's me decorating it [pencil drawn on design sheet]. I did a handle but it was too thin, the card was too thin. It went ugh, all floppy.
- *Mrs.J*: Are these rabbit's ears? *[on product]* That's different to your drawing. Why did you do those?
- Craig: Someone else had done that and I thought "That's a good idea" so I stuck them on my top.

Craig's drawing provides support for working out how he will make the holder but he does not feel bound to include details that do not work well in practice. Instead he freely borrows a friend's idea. His comment "that's me decorating it" was unique and there was no way of knowing if other children were also visualising themselves making the product as they drew.



Ex. 40 : Easter Egg Holder (d); finished product, Craig (Focus Class)

Jordan (Comparison Class)



Ex. 41 : Easter Egg Holder (e): design drawing, Jordan (Comparison Class)

Jordan: That was going to be a Stickosaurus but I didn't want to do it because it was too hard.

That was a big one with a lid but that was too hard because it fell over. Then I was trying to make a robot one.

Me: Is that the one you made?

Jordan: Yes. Well, no, not really. Cos it was too hard really.

Me: So what did you do?

Jordan: Well I just sort of got all the bits I'd got and ... um ..

Me: And made it out of them?

Jordan Yes, really. And it was like this.



Ex. 42 : Easter Egg Holder (f); finished product, Jordan (Comparison Class)

Zara (Comparison Class)



Ex. 43 : Easter Egg Holder (g); design drawing, Zara (Comparison Class)

- Mrs.J: Have you drawn this twice?
- Zara: No they're two parts. It goes over your arm.
- Mrs.J: Oh, right, so what are these bits?
- *Zara:* That's the string. And you cut a hole there and one there for the egg to go in. In the side of the tube. In both tubes.
- Mrs.J: So why did you make just one half of it?
- Zara: There wasn't time to do both sides. So I put... and it goes over your arm like this, look...



Ex. 44 : Easter Egg Holder (h); finished product, Zara(Comparison Class)

As creative as the two Comparison Class children clearly are, Jordan did not include the card tube that was central to the problem inherent in the task in his final design or product Zara needs two tubes for her design. They were given one each. So she was not modelling ideas with respect to the materials provided either. That this was common within the Comparison Class is illustrated in the photographs of the Easter Egg Holders in Appendix H.

SECTION 5 - Analysis of the Assessment Tasks

Although this debriefing gave me an insight into what the Target Groups *thought* they were doing, I was not convinced that they were sufficiently self-aware to give account of themselves after the event. This was especially highlighted by Lee, whose account both to me and to Mrs.J. contrasted sharply with my classroom observation. I was convinced that the paper he submitted as his "design" was drawn after he made the product so that I would believe his work to be a success. I felt sure that the paper he had in his hand as he wandered around the room had a different drawing and had been quietly disposed of. I cued Mrs. J. to begin the conversation with the words "So this is the drawing you made of your Easter Egg holder once you'd made it." but he stuck to his story.

This convinced me that for accuracy's sake, questions need to be asked of the children immediately, as they worked. Also the recordings were limiting my recorded observations to the Target Groups. I was beginning to want to record other children's understandings of designing.

5.3.1d Assessment Task 4 : Surprise Card

One of the researchers form the Enriching Literacy through Design and Technology Project visited the school and conducted the Surprise Card Assessment Task in the first week of June. I circulated with my Log Book and ensured that I had spoken to every child during the session. The design sheet used for this task, along with the work of the Target Groups from both classes can be found in Appendix I.

Both classes used the examples as a resource when designing their own cards:

Focus Class:

- Michael: Drawing the demo cards & mechanisms- hasn't really designed something of his own.
 - Emily: I told her to draw rather than write "love-hearts & flowers" got rather into it "Miss,
 I've got a pattern" Later: Her drawing is the blue spiral demo + pop-up at top. Me:
 prompt to think about front. Later: Has carried on designing into Box 6

Comparison Class:

- Zara: has drawn round the blue mechanism provided
- Cassy: her flower vase is to be like the demo card with the wobbly head
- Sophie: her first drawing is of the green mechanism example

The Focus Class, however, were more likely to consider how the card would be made :

Focus Class:

Carl: two pieces of paper on top of each other & folded "And I'm going to keep on going down in size"

Natasha: Hers opens like the booklet - 2 flaps revealing a central page

Comparison Class:

- Kathy:Me what's inside? Kathy: I forgot to do the back of it.Chloe:A little girl on the front words on inside left, picture on right
- Nicola: The top bit is a flip-up football.
- Carys: Has changed her design after working out how to make it

The following comment, from the most academically able boy in the Comparison Class, indicates perhaps how the Comparison Class children did not see the drawing as supporting designing of something that they were really going to make:

Alistair: has drawn one huge picture

Later: It's for Gran & Grandad - they like birds - when you open it there will be a bird which springs out really quickly and come alive & fly away. It's going to come out & flap its wings & fly.



Ex. 45 : Surprise Card; Alistair (Comparison Class)

The Design Sheet had a separate space (Box 6) for recording how the card would be made and the children were specifically told to fill this in. There was a difference in reaction to this instruction. Most of the Focus Class children did so within a few minutes, perhaps then going back to their main drawing (Box 5) to finish it off. Many of the Comparison Class ignored this instruction and continued with their main drawing. On second prompt, more of them wrote (rather than drew) something in Box 6 but on reading these, the comments did not take the design idea towards making, they simply described the graphic in Box 5.

5.3.1e Assessment Task 5: Pandy's Suitcase

This was conducted at the end of the Summer Term 2001, using the same format as Pandy 99 (Section 3.3.3b). Drawings and products by the Target Groups from both classes can be found in Appendix K.

I circulated and asked children what they were doing and noted things down, especially for the Target Groups in both classes. These children remained behind after the end of the session, to ensure that I had written down everything they wanted me to know about their suitcase and how they had planned it. This seemed to work well. I did not take photographs of the products of this Assessment Task as I kept the card prototypes of the suitcases as well as their drawings. This enabled me to see whether the children were taking their ideas forward from their drawing into the making.

Many of the Focus Class children measured the Panda with their rulers and although they sometimes forgot the importance of this information once they were making, many of the snippets of conversation centred on whether or not the mac would fit inside the suitcase:

Martin: Immediately measuring mac with a ruler Louise, Lisa, Ellie also get ruler to measure mac Tasmin: measuring Panda's arm

Natasha: Measuring mac, says "7"

Later: Drawing round mac onto card

- Noel: Made one card one & realised it was too small tried the mac in it
- Craig: Has made really good bag but will the mac fit? bit small came apart making another
- Lisa: "Mine fits!" (mac slides right through tube-like structure)
- Carl: Has stuck 2 halves together will mac fit? yes



Ex.46 : Pandy's Suitcase (a) : Maria (Focus Class) Recording Measurements

Or on Construction issues:

Lewis: Single piece of card folded up rather than 2 separate sides.

Ellie: I am going to make a box because Mrs R. taught us how to make a box. (Made really large box)

Kate: Made box by gluing strips to rectangle base

- Stacey:Shows me rectangle & mac it's too small. What to do? Me cut out another.Later:2 pieces of card, right size mac sandwiched how to join together? thinksshe will use sellotape.
 - Later: Put bits of string at sides so it opens further to get mac in.

In the Comparison Class, attempts to make the mac fit were not supported by use of rulers:

Alice: Folding the mac to try to fit onto white paper - it just about fits onto there: "The mac fits"
Holly: Has made a huge box shape

Rhiannon: Super little bag but too small. I suggest making another one bigger

In the Comparison Class I was more aware of their lack of understanding of the role of the white paper for planning the prototype to be made in card:

- Robert: Is cutting up the white paper
- Alistair: "Are we allowed to cut out the paper?" he has drawn a single decorated item. Later: Got card - "It can't be this big can it?" waving card about Later: Is using his drawing as a transfer
- Lee: Shown me white paper "Shall I cut it out?" Later: Came to me with cut out - it doesn't fit over Pandy's arm: "I've done a little hole" Me - Does the mac fit in it? - Don't know haven't tried it. Later: Brought the mac & white single sided cut out: "It fits" Me: What are you going to do now? - "Make it properly" Then went on to make a 2 sided bag - so was the cut-out a prototype? But final product is much larger.

The Log Book comments reveal the Comparison Class to be very similar to Year 2 children in my Exploratory Phase. Several children cut up the design sheet and used this in a variety of ways: to draw round onto the card, to stick onto the card and then cut out both, or to fold up and make into the suitcase (Robert). Lee seemed to have used the design sheet as a prototype but he was alone here, except perhaps for Alistair who used his drawing as a template after asking about cutting out.

Also in evidence in the Comparison Class were the "Single-sided cut-outs" that had been typical of Year 2 in 1999 :

- Sophie: Single sided Me: Does it fit? yes Me: What are you going to do now? : Put little footballs, like on the drawing.
- Chloe: Pink paper stuck to cut out. Me: Did it fit? I don't know I didn't try it. Later: "The middle bit fits, the side bits don't" *Me (unsure what she means as it is single-sided):* Do you want to try again & make it bigger? - yes
- Wendy: Drawing & making a carry stick "You put things in & hang it over your shoulder" Later: Single sided - fits over paw, went over shoulder - not happy - trying again.

Ex.47: Pandy's Suitcase (b) : Wendy (Comparison Class); single-sided cut-out:



Only one girl in the Focus Class had difficulties of this sort:

Kara has 2 rectangles - what next? Me: stick them together? Not sure. Went to look at Jolene's.

Part of the problem appears to lie in the child's inability to perceive the divide between reality and fantasy inherent in the situation. Stables (1992b) cites research that suggests that the ability to handle reality is based on the ability to handle fantasy. On this view, the children who created single-sided cutouts and imagined that the mac could go inside were less able to handle the interplay between reality and fantasy in the situation. In my Exploratory Phase observations that it was the younger children who more commonly made this mistake (Section 3.3.3b). The older children understood and could work within the rules of the game. Stables (1992) relates this ability to handle both reality and fantasy simultaneously as fundamental to design. I was beginning to relate both to being able to play the game by the rules.

5.3.1f Assessment Task 6 : Theseus' Maze

The Maze task was conducted in Jan. 2002. Drawings and products by the Target Groups from both classes can be found in Appendix L.

The Focus Class were bubbly and chatty but as I circulated I realised that they were actually discussing where the Minotaur was, where the walls were, and so on, which then continued into mutual support during making:

Michael & Maria (sitting next to each other) Drew outside view as well as top/down

Noel: Measuring his wall against the plan. Connor: You don't stick it on there, mate. [was he going to?] Later: Noel now cutting out more green to make it same length as yellow base card. Has Connor distracted his correct thought about measuring?

Randal: Can we change it cos it's quite hard to make that.

Lisa: To me "Can I make slots - I can make slots" Later: Holes in card are for Minotaur to go through - bits left are the bridges

Ellie, Carl, Natasha : Large box structures, similar outsides; insides different

Tasmin: Checkerboard floor pattern - copied from L (Non-assessed child)?

Emily: ditto but later: Drawn maze on yellow (different to plan) & then folded & stuck walls on it - though not on the lines - so is this a mosaic floor?

Hayleigh: Doing the one on top left of paper.

Stacey: Has "lost her way". Jolene sorted her out "Look back at your drawings" - folded edge of card for her.

This sharing of ideas was in contrast to the Comparison Class who worked quietly but individually whilst drawing. In contrast to the Focus Class, there is a low level of co-operative work amongst the Comparison Class children. Kathy and Chloe were sitting next to each other and probably used each other's interpretation of the task to inform their own; this was the only instance of working together noted during the lesson.

- Kathy: "Will you draw the walls or make it?" Me: whatever you like. Kathy: I'm drawing it and then put walls.
 Later: I'm going to make figures and maybe a little boat so he can get back [she has no internal walls]... so I can put my bits inside.
- Chloe: Also only has walls "I'm going to make people.." Later: finished - drawn maze not made walls

Kathy and Chloe were just two of the Comparison Class who drew a maze on the card and built walls around the outside edges. The Focus Class made internal walls - and no one struggled to know how to glue them. Comparison Class children needed to be told how to make tabs.

This wall-building was the most obvious difference between the finished products. The Focus Class could use a top-down view to support making a 3-dimensional object, which the Comparison Class could not really do. Once they had drawn the maze the Comparison Class struggled to jump back into 3-dimensional thinking. The lack of internal walls when making, suggests that the drawing had confused them into thinking that they were making a 2-dimensional object. Despite many of them being able to discuss their drawing with me (placement of Minotaur etc.) they did not make this the basis of a 3D model.

Making what they had drawn appeared to be so rare in the Comparison Class that I noted children who did. Much more typical was Ellis, whilst making: "I've just worked out how mine's going to work" and Alistair's belated realisation that what he had drawn was impossible to make:

Alistair : Wanted to do trap doors etc at the drawing stage - I nearly stopped him because I thought it would be impossible to make. Later: "This is really, really hard. Look I've got all that to do. I'll have to do all that."

Peter and Robert were typical of the many boys who focused on the ghoulish:

Peter:Has drawn rope as well - really good drawingLater:Blue triangle in flat base "skulls". Not following drawing at all.

Robert: Has drawn a maze on the card and a single door

Later: Blue fan-folds - bones on top - where the Minotaur has left bones

Robert's drawing was almost a comic-strip telling of the story rather than a design for making:



Ex.48 : Theseus' Maze (a) : Robert (Comparison Class)

The making of fan-folded structures with bones on top, shows the extent to which these children were creatively fantasising as they made their product. There was no sense of solving "this problem and this problem only" Donaldson (1992). This was even more marked amongst those who departed completely from the script:

Jordan: Waterfall (blue) for Minotaur to have a drink

Zara: This is for the crocodile to come in (blue). Lee: "It's a Minotaur" Zara: No, a crocodile is in the garden; I'm doing a boat, walls & a door.
 Later: Playing with it - walking the Minotaur about

My evaluation notes on the Comparison Class session read:

Some of them were in a sort of narrative mode - making bones, water etc. for the Minotaur - on different understanding of what the problem solution might be like. They did not define & solve the problem at drawing stage & then find they needed to make changes (as did Focus Class) but developed ideas in the making - and these then diverged from the "model for Theseus so he knew the way out" scenario. They are making a model of the maze as a personal play object.

I think too that the Comparison Class had left too many possibilities hanging and un-addressed before handling the construction materials, that they were still unfocussed on the problem to be solved and so used the materials to make what they fancied rather than solve the task. I think this observation gave me the biggest justification for the "Why draw anyway?" question that lurked constantly at the back of my mind throughout the Programme's duration.

My "Evaluation of Session" notes continue:

The different understandings and working methods of the two classes were immediately apparent. Some of the Comparison Class were still at the "drawing a picture to define the problem" stage. There were considerable numbers of the Comparison Class who ignored their plans completely, also quite a few drew a maze on the yellow base and then built a set of walls round the outside only. This seemed odd since they had all seen my pop-up maze example. The Focus class were making real 3-D mazes and threading the string through. Regardless of what is shown about their differing use of drawing, the different level of engagement with the task was remarkable.

This last sentence captured the difference between the two classes. It was not in the sophistication of what they recorded on paper, but in what could not be captured except in my

SECTION 5 - Analysis of the Assessment Tasks

log book notes, in the *level of engagement* with the task as given. The Focus Class were problem-solvers. They understood what was to be done and set themselves to satisfy the criteria of the problem. The Comparison Class were not grappling with a real problem about making Theseus a model of a maze; they were doing something else, parallel to that. I think it was making themselves something with maze-like properties or characteristics. Many were playing with the idea of a maze in a freely creative way.

In comparison, the Focus Class were more prosaic and business-like. There were one or two bridges to get over deep water (even these were planned on paper before making) but on the whole they saw the activity as providing a solution to a specific, pre-defined problem. They had their flights of fancy in their discussions before, and in tandem with, putting pencil to paper and then they made their final idea. The Comparison Class drew in almost total silence but then discussed, adapted and completely changed what they were doing and lost the thread of the activity once they had the card in their hands.

5.3.2 Emergent Themes from Qualitative Evaluation

In drawing together the Emergent Themes from the qualitative evaluation of the Assessment Tasks, my aim is to indicate the differences that were apparent between the two classes on the basis of observations made in context and recorded in my Log Book and by video or audio recording.

Some of these relate to the Emergent Themes identified through my informal evaluation of the Programme as delivered to the Focus Class and build on my reflections in Section 4.5.2 "Emergent Themes from Evaluation of Teaching Input." However, other themes emerged from the data, which, it will be seen in Sections 5.5-8, relate more closely to the results of the quantified analysis of the children's drawings.

The Role of the Drawing

There was a distinct difference between the two classes perception of the purpose of the drawing. In the Focus Class it was a discussion document used to develop a design solution, whereas the Comparison Class were just defining the problem to themselves. The Focus Class used the drawing to support the development of a solution, whereas the Comparison Class seemed to be using drawing to record the task or define the problem. Not having sufficiently defined a possible task solution before cutting into the materials, the Comparison Class began to develop their design ideas once they were engaged with the materials and then made a product which was related to, rather than answered, the problem as set.

SECTION 5 - Analysis of the Assessment Tasks

For the Maze task, more of the Focus Class were still drawing at half-time, suggesting that they were using the drawing to work out real ideas for making, They consequently showed a much tighter sense of connection between the drawing and the making. They were using the drawing to plan their maze and modelling real ideas on paper that they then transferred to the card; the Comparison Class were not. I think the Comparison Class almost saw the drawing as a prelude to the real business of designing, which they did with the materials of construction, whereas the Focus Class had learnt that the drawing was a means to designing.

Interesting light was shed on the Comparison Class' lack of understanding here when I discovered that neither class teacher had given the children time to make the Card on the same day as it was designed. They expected me to do this with each class the following week. From my viewpoint on designing, the children were very unlikely to remember what they had planned and be highly likely to abandon their drawn ideas and begin to design afresh. I realised later that the Design and Technology workbook in which the Comparison Class recorded their work also reflected a fragmented model of designing, making and evaluating. This contrasted sharply with my integrated Ideas on a Journey model.

The Role of Discussion in Designing

The Focus Class discussed as they drew, exchanging ideas, sparking off each other but also keeping each other on task. The Comparison Class drew without discussion. Drawing was seen as an individual, quiet activity. This was especially marked in the Maze activity.

The sharing and discussion of design ideas had been encouraged right from the start of the Programme (Section 4.3.2 & Appendix B). Co-operative work also featured throughout, which encouraged children to use each other as a resource for feedback on possibilities. Focus Class children often used drawing in tandem with speech. Drawing was frequently used to initiate or support such conversations (for example, that between Craig and Noel over Noel's idea for Frosty, Section 5.3.1b).

The point at which children declared their design intentions to each other was an important difference between the two classes' designing styles. The Focus Class said *"What I'm going to do is"* to each other as they drew, the Comparison Class delayed these conversations until they were engaged in making. By this time they had fetched their materials and then comments from peers which led to changes in their ideas were not recorded, defined or thought through ahead of cutting into the materials.

Solving the Problem in Hand

The Comparison Class seemed to have a cavalier attitude towards the task instructions. If they could not solve the task, they changed the task to one which they could solve or would prefer to solve, even to the point of discarding the essential elements of the problem. This was evident at Easter, when children were observed omitting to use the card tube that was central to the problem to be solved (*"How do you make a holder out of this tube for this egg?"*), or on the Maze task, making snake pits and piles of bones instead of grappling with how to make walls with doorways leading to the Minotaur.

In contrast, the Focus Class were solving the problem in hand. They accepted the task as set and sought to find a way to solve it. No one discarded the card tube at Easter, cut out a picture of a stick and kerchief or made rivers with crocodiles outside King Minas' castle. This was not just a matter of degree but of complete contrast between the two classes. The Comparison Class appeared to be less able to take on board the idea that the constraints of the task as stated were important success criteria. The Focus Class did, and I came to the conclusion that the Programme had enabled them to understand the rules of the game and, in Donaldson's (1992) phrase, solve "this problem and this problem only".

<u>Reality / fantasy</u>

For Pandy's Suitcase, none of the Focus Class had problems knowing that they were meant to be making something that a plastic mac could be put into. Several Comparison Class children drew a single rectangle with a handle, cut it out and declared the mac would fit in it. An identical scenario occurred with my Year 1 class of 1998 in which I placed the Panda's clothes against the piece of card and let go (Section 3.3.3b). In 1999, I made a conscious decision not to record Year 2s making this mistake as I could not be sure they got the right answer from me or each other. It was clearly a misunderstanding of the reality / fantasy divide as related to this task which was common amongst Key Stage 1 children. It was more remarkable that the Focus Class did not make this mistake.

This seems also to be related to playing the design game by my rules and knowing where the reality / fantasy divide was meant to be. Many Key Stage 2 children appear to be able to do this and the Programme had enabled the Focus Class to do so too. The difference between the two classes in this respect is best illustrated by the photographs (Exx. 48-9, shown overleaf) taken towards the end of the Maze Assessment Task:

Ex. 49: Theseus ' Maze (b) Jolene and Hayley (Focus Class) built the internal walls before attaching walls and roof:



Ex. 50: Theseus ' Maze (c) : Zara (Comparison Class) making a boat for Theseus. The "Maze" has just one internal wall; Kathy's maze (in background) has none - these were simply drawn on the yellow base.



5.4 The Quantitative Analysis Instrument

A more rigorous analysis instrument was needed for the more detailed assessment of children's drawings during the Structured Phase. The databases used for recording results in the Exploratory Phase were themselves exploratory. They had evolved over time and had highlighted aspects of design that I had found interesting or noteworthy. Although they had come from observations and evaluation of the collection of drawings that I had, they were idiosyncratic. I needed an assessment system that was more firmly grounded in the work of others, in order that my work had greater credibility.

5.4.1 Creating the Analysis Instrument

From the Exploratory Phase I had my Drawing Types which I had intended to continue use as the primary means of assessing the children's work, which, together with the qualitative data from observational field notes and audio and video tapes of the Target groups, would give (to my mind in April 2000) a good rounded view of how children used drawing for designing.

The main problem, which I had not seen at that time, was that whilst in the Exploratory Phase I was documenting what *was*, for which six non-linear Drawing Types would suffice, I was now attempting to improve the children's performance and I did not have a continuum by which I could define *"better"*. Section 5.4 documents the development of a more appropriate analysis instrument.

5.4.1a Starting Points

My theoretical starting point was the Kimbell et al.'s (1991) Dimensions of Capability shown in Fig. 49 :



Fig. 49 : Kimbell et al.: (1991: 23, Fig 2.7)

I combined this with the insights gained whilst compiling the Taxonomies of design drawing (Section 3.4). I felt that the relationship between the drawing and the product was important, and something which had been part of my assessment system from the start. I was also influenced by the National Curriculum Attainment Targets, since these were how we were required to think about design capability in school.

I identified, therefore, eight Aspects of Capability in relation to design drawing:

- Generating and Developing Ideas,
- Exploring the Problem,
- Addressing Client's Needs,
- Appearance of the Product,
- Communicating Ideas,
- Planning Construction,
- Evaluating Whilst Drawing,
- Relationship to Making.

There was no implied precedence of one Aspect over another, either in choice or ordering, except that Generating Ideas was likely to come early in the process and Relationship to Making last. Generating and Developing began as two separate categories but were combined since I seemed to be repeating myself once I started to construct the instrument.

5.4.1b Grid and Ticksheet

The analysis tool was first conceived as a series of linked spreadsheets and databases and this continued to be the method of analysis, despite the subsequent re-definition of the analysis instrument and the display of results as radial plots in Sections 5.5-9. Appendix N details the development of the analysis instrument by tracing the evaluation of one child's drawings, as well as explaining how these individual records were collated into the spreadsheets in Appendix O, on which the analysis on Section 5.5-9 are based.

Initially, an Individual Marking Grid was constructed (Table 9 overleaf, and also Appendix N: AN2.1), together with a Techniques Ticksheet (Table 10, and also Appendix N: AN2.2) and a Collation Sheet (Appendix N: AN2.6). This was constructed by considering examples of children's work that I felt typified each of the Drawing Types (Section 3.3.5) from Stan, Panda and Art Club experiences. For example, I looked at a stereotypical Single-draw and composed a single line descriptor for each of these Aspects of design capability.

	PICTURE	SINGLE-DRAW	MULTI-DRAW	MULTI-DESIGN	PROGRESSIVE	INTERACTIVE
GENERATING &	drawing a picture of an	single, simple	several attempts to improve	several unrelated draiwngs	progression of ideas	uses drawings reflectively
DEVELOPING IDEAS	object, not designing	drawing	drawing of single idea	showing range of ideas	across drawings	to generate new ideas
	a product					
EXPLORING THE	not exploring	minimal use of drawing	aiming to improve the drawing	brainstorming onto paper	uses drawing to develop a	combines previous ideas wit
PROBLEM	design problems	to explore ideas	not explore solutions	without developing a solution	design solution	new ones to produce the bes
	· · · · · · · · · · · · · · · · · · ·					solution
ADDRESSING	minimal understanding	drawing conveys partial	drawing shows understanding	several ideas for satisfying	the client's needs and wants	the client's needs and wants
CLIENT NEEDS	of client needs and	understanding of addressing	of the needs of the client	client needs are recorded	are considered as the design	are treated as part of
·	wants	client need		but not developed	proceeds	the iterative process
APPEARANCE OF	views the drawing as	minimal consideration of	only one overall finishing	experiements with	ideas about finishes are	ideas about finishes develop
PRODUCT	the product	final appearance of product	scheme considered	several finishing schemes	added to design during	interactively within overall
		being designed			the drawing stage	design development
COMMUNICATING	use of narrative or	minimal recording of design	several drawings of	quick sketches of a	conveys sense of object	clearly conveys ideas about
IDEAS	other drawing genre	ideas	same idea	range of ideas	to be made, e.g. by labelling,	object to be made e.g.
					instructions etc	multiple viewpints
PLANNING	not planning to make	no evidence of materials or	minimal consideration of	indicates which idea will	materials or	constructional issues
CONSTRUCTION	the object drawn	construction issues	construction	be made, but not how	construction features	considered en route to final
					shown on drawing	design
EVALUATING	no evaluation	minimal evaluation	rejected early atempt(s)	considered and	decisions made about the	changes made as a result of
WHILST DRAWING			at drawing single idea	rejected range of ideas	object whilst drawing	reflecting upon previous
						design drawings
RELATIONSHIP	making an object is	minimal relationship between	object made is the same	object made is one	progression	uses drawings as resource
TO MAKING	viewed as a separate,	drawing and making	as the object drawn	of the ideas drawn	from drawing into making	during making
	new activity	(

Table 9 Individual Marking Grid

The example shown in Appendix O indicates how the marking grids for each task were collated.

The Techniques Ticksheet was intended to capture the "how" and was a compilation of techniques that I had observed children using during the Exploratory phase of the research (e.g. annotation, recording materials), arranged under headings to match the Grid. The Ticksheet was to cause much less trouble than the Grid, as it required simple yes/no answers rather than qualitative judgements. Comparison between the original list (Table 10 below, & Appendix N: AN2.2) and that finally used (Table15 in Section 5.4.3b(ii) & Appendix N : AN2.4) shows how the interaction between Grid and Ticksheet impacted on the Ticksheet.

SATISFIES CRITERIA OF DESIGN BRIEF	PLANNING CONSTRUCTION
as set by teacher	indicates materials by list
begins from task requirements	indicates materials by labelling drawing
maintains task requirements into msking	materials indicated are suitable
as required by client	materials indicated are available
begins from client needs	indicates cuts, folds &/or fixings
maintains client needs into making	indicates measurements
	indicates equipment needed
RELATIONSHIP BETWEEN	indicates parts to be assembled
WORDS & PICTURES	can be made in time available (roughly)
pictures only	technically realistic (child can make it)
thinking recorded mostly in pictures	
thinking recorded mostly in words	EVALUATING WHILST DESIGNING
single words or phrases relatiing to picture	choices related to appearance of product
list (e.g. of materials)	choices related to design specification
full sentences to describe planned product	choices related to client needs
labelled diagram (with) arrows or lines	choices related to material constraints
words/pictures interact to record process	choices related to construction
lengthy verbal explanation	justifies choices made
instructions - words only	
instructions - pictures with few/no words	RELATIONSHIP TO MAKING
instructions - labelled diagrams	same/adapted object
	same/adapted shape
FEATURES OF THE DRAWING	same/adapted size/proportions
appropriate level of detail	same/adapted colour
shows dynamics/movement	same/adapted pattern, decoration, picture etc
various viewpoints	same/adapted materials
expansions to show small details	same/adapted fixing technique
cut away diagram to show inside of product	able to justify changes
idicates how parts will fit together	
APPEARANCE OF PRODUCT (FINISH)	
indicates pattern or motif	
indicates colour	
considers more than one finish	
justifies choice of finish	

Table 10 : Techniques Ticksheet (Version 1), also to be found in Appendix N: AN2.2 :

5.4.2 Refining the Instrument

Since it contained qualitative statements, the Grid was to become the subject of extended refinement through moderation and eventual redefinition through metamorphosis into the multi-level holistic analysis instrument described in Section 5.4.3. As well as involvement in the evaluation of the children's work (as outlined in Section 4.2.3), my moderation panel were involved in the development of the analysis instrument itself. This enhanced the reliability of the analysis instrument as well as the evaluative judgements made about the children's drawings and products. Appendix N demonstrates this process.

I needed to know quickly if the Marking Grid was viable and so the moderation panel each blind marked twelve examples of the Pizza task (Target Group children) using both the Grid and the Ticksheet. We were all used to working to a "best fit" paradigm of SATs scoring and used a grid to assess literacy achievement across Key Stage 1, so that my colleagues had no more expectation than had I that children's work would fall neatly into the columns. However, few children's work spread across as many as three columns.

Where I was certain about my classification on the Grid, my colleagues tended to agree; where I was less certain, they diverged widely. All three teachers said that they needed to know more about the task and my expectations in order to know to what extent the children had satisfied some of the grid criteria. The Ticksheet was much less problematic: children either did something or they did not. This feedback from my moderating colleagues was to become part of the iterative process of refining the instrument. That the instrument needed refining had become obvious very quickly. Because I knew what I meant by the Drawing Type categories, I had paid less attention to how someone else would interpret the individual cell descriptors without having the mental image of what I was looking for. The Grid was an aide memoire for me and did not work as an evaluation tool for others.

Both my moderators and the Research Group at Goldsmiths College questioned the relationship of the Ticksheet to the Grid. I had realised that I was flagging up concern for my marking if the scores differed by more than 2. It became apparent that, for example, children were getting points on the Grid for simply putting a tick by their choice of idea if they were "Multi-designers". This seemed a lot for not much. Referring to the Ticksheet for Communicating Ideas, Multi-design turned out to be rather more than "tick which one of all those ideas you think you'd like to make". If nothing else, Ticksheet was useful for showing up inconsistencies in the Grid.

However, to be useful as an evaluation tool, the Ticksheet should not, I felt, just be a supporting cast for the Grid. It needed a clear reason for its own existence. Thus, any
information which was repeated on the Grid was deleted, except where I felt I wanted to record information in more detail.

5.4.2a The Purpose of the Design Drawing

As part of my attempts at improving the Grid, I consulted the Taxonomies of Design Drawing that I created from my reading and the design tasks conducted with children in the Exploratory Phase (Section 3.4.1). First on the Internal list was "Purpose" which appeared to solve my Grid problem instantly: This is the essential difference in how the children treated the drawing process and is revealed in the drawings they do.

On this view, the Individual Marking Grid was a *Purpose of Drawing Grid*: what do the children see the purpose of the drawing as being:

DRAWING TYPE	PURPOSE
PICTURE	Finished product, unrelated to future making.
SINGLE-DRAW	Record of an idea to show the teacher.
MULTI-DRAW	Improve the drawing before showing the teacher.
MULTI-DESIGN	Brainstorming ideas relating to the design brief.
PROGRESSIVE	Develop an idea towards considering how to make the product.
INTERACTIVE	To work out what they want to make and how to make it.

Table 11 : The Purpose of the Drawing

A "Purpose Row" was added to the Grid and each Descriptor Cell altered in line with its contents. I felt initially that this had radically improved things. Thinking in terms of identifying the child's perception of the *purpose of the drawing* simplified the whole process and my moderators and I could place them on the Grid much more easily. There was only one child (Emma) whose work required considerable discussion between us. The divergence between us was less wide, probably as much due to the discussion following the first iteration which enabled my blind-markers to more fully understand my classifications as to improvements of the Grid. The establishment of shared meanings meant that their interpretation of the Grid was now more likely to be consistent with each other and myself. I hoped, therefore, that future disagreements over placement of children's work would be due to interpretations of the drawings, not interpretations of the words on the Grid.

The Purpose of Drawing Grid (Table 12) overleaf is also to be found in Appendix N: AN2.3

			1			
	PICTURE	SINGLE-DRAW	MULTI-DRAW	MULTI-DESIGN	PROGRESSIVE	INTERACTIVE
CHILD'S VIEW OF	Draw the object.	Record an idea of what might	Make the drawing as good as	Brainstorming	Develop chosen idea and	To work out what will be
PURPOSE OF THE		be made, to show	as possible before		idicate how it might	made and how to make it.
DRAWING		the teacher.	showing the teacher.		be made.	
GENERATING &	drawing a picture of an	single, non-stereotypical,	several attempts to improve	several unrelated draiwngs	progression of ideas	uses drawings
DEVELOPING IDEAS	object, not designing	drawing	drawing of single idea	of a range of ideas	across drawings	reflectively
L	a product					to generate new ideas
EXPLORING THE	design problems are not	minimal use of drawing	aiming to improve the drawin	brainstorming onto paper	uses drawing to develop a	combines previous
PROBLEM	adressed in the drawing	to explore ideas	rather than explore solutions	without developing a solution	design solution	ideas withnew ones
						to produce best solution
ADDRESSING	minimal understanding	drawing conveys partial	drawing shows understandin	several ideas for satisfying	the client's needs and wants	the client's needs and wants
CLIENT NEEDS	of client needs and	understanding of addressing	of the needs of the client	client needs are recorded	are considered as the desig	are treated as part of
	wants	client need		but not developed	proceeds	the iterative process
APPEARANCE OF	views the drawing as	minimal consideration of	only one overall finishing	experiements with	ideas about finishes are	ideas about finishes develop
PRODUCT	the product	final appearance of product	scheme considered	several finishing schemes	added to design during	interactively within overall
		being designed			the drawing stage	design development
COMMUNICATING	use of narrative or	minimal recording	several drawings to express	demonstrates range of ideas,	conveys sense of object	clearly conveys ideas about
IDEAS	other drawing genre	of design ideas	same undeveloped idea	often through series of quick	to be made, e.g. by labelling	object to be made e.g.
				sketches	instructions etc	multiple viewpints
PLANNING	not planning to make	no evidence of materials or	minimal consideration of	indicates which idea will	drawing indicates	constructional issues
CONSTRUCTION	the object drawn	construction issues considered	construction	be made, but not how	consideration of materials or	considered en route to final
					construction features	design
EVALUATING	no evaluation relating to	minimal evaluation	rejected early atempt(s)	considered and	decisions made about the	changes made as a result of
WHILST DRAWING	designing the product		at drawing single idea	rejected range of ideas	object whilst drawing	considering and discussing
						design drawings
RELATIONSHIP	making an object is	minimal relationship between	object made is the same	object made is one	clear progression	uses drawings as resource
TO MAKING	viewed as a separate,	drawing and making	as the object drawn	of the ideas drawn	from drawing into making	during making

Table 12 Purpose of Drawing Grid

The example shown in Appendix O indicates how the marking grids for each task were collated.

However, this was still modelled on the Drawing Types and by April 2001 I had realised that the Drawing Types were themselves becoming a problem and perhaps needed to be separated from the Grid, removing the temptation to decide the Drawing Type and then check boxes with this in mind, rather than viewing each Aspect row separately.

My moderation panel each blind-marked all Target Groups drawings for the Easter Assessment Task and again supplied feedback. At this meeting, it emerged that they were less worried by the Drawing Types problem as I was, since for them this was simply another criterion and they did not have all the encumbrances of pre-conceptions that I had through devising them.

In Summer 2001, I conducted two workshops on design drawing, one in school and one at the Design and Technology Association Conference in Coventry. This meant that moderation of the drawings had been carried out by twelve of the teaching staff at school and by about fifty conference delegates with a range of backgrounds in Design and Technology education, giving me increased confidence in my classification categories.

5.4.2b Representing Progression

However, by mid-summer 2001, other problems with the Grid had became obvious to others (if less so to me): not all my categories were sliceable into a 5-graded scale. For example, the children had either addressed their client's needs or they had not. I had constructed the Grid by extrapolating from the Drawing Types categories by considering the work of individuals who fitted these categories. The cell descriptors were constructs which fitted the examples I had used; but did they generalise? I was constantly changing the descriptors in the light of the drawings and the moderation process.

I took all this to the Research Group Seminar. Some in the group favoured circles rather than grids; a more holistic, less deterministic feel to circles - had I done a Grid to fit neatly onto an A4 sheet? No, it was so it fitted on a spreadsheet (and so judged to be even more suspect).

A more realistic left-right representation could probably be as Fig. 50:



which suggested the progression of understanding and skill but allowed for multiple routes through the Drawing Types, which was emerging from my analysis of the data, although I still felt that Multi-design was more sophisticated than Multi-draw.

The Research Group made other suggestions, for instance:

aspects Descriptors with overlapping characteristics features

Fig. 51 Non-linear Progression across Drawing Types (b)

Perhaps, the Drawing Types are descriptors which have overlapping characteristics, some of which overlap and some of which are discrete (at the time I was using "aspects" as the term of reference for Grid items and "features" for Ticksheet ones). This was all becoming too complicated and the possibilities seemed endless and the Grid seemed to be the least appropriate vehicle for discovering the answers to such questions. Without looking at both the Grid and Ticksheet, it would not be possible to see that the way the drawing is annotated is often an important feature of Progressive drawings.

The problems were, I believe, two-fold:

Firstly, the Drawing Type categories had arisen from the examination of drawings across the 5-9 age range over a period of 4 years, at a time when the teachers' understanding of Design and Technology had been less clear and my interest in design drawing had little impact within the school. Now I was looking more closely at a narrower age range, over a much shorter time period, looking for improvements, and the techniques used by the children were more varied and sophisticated than had appeared amongst Year 2 in 1998-9.

Secondly, there was the difficulty of determining what constituted "improvement". The Drawing Types were not (and never had been) seen as a linear scale, although in essence they represented progress (the "muddle in the middle" between possible pathways through Multi-draw and/or Multi-Design notwithstanding). It was generally felt that I was deciding on the Drawing Type first and then judging the children's performance on the Aspects according to my Drawing Type classification. In order to minimise this effect, I removed the Drawing Type label from the top of the Grid. I did not feel in practice that this was making much of a difference to

my judgements. The Grid had been constructed by teazing out what I meant by my Drawing Types, so that with or without the word on the top of the Grid, I was likely to come to similar conclusions about individual design drawings.

I think that focusing on the instrument and defining what the Drawing Types meant had distracted me from the rationale behind the Structured Phase of the research: improving children's performance. The real difficulty was in finding a way of capturing the essence of growing understanding and sophistication. Analysis should aim to show whether the Focus Class had made greater progress overall and were using a greater range of techniques in comparison to the Comparison Class and, if so, how significant was the Container / Journey metaphor in contributing to their understanding (i.e. was there significant improvement in the Focus class from Easter which was not matched by the Comparison Class and was that difference maintained?).

By the time the Programme was completed (Jan 2002), I had an instrument which could answer the question *"What do children do?"* rather than one aimed at addressing the question *"Are they better as a result of your input?"* What I had not done was to have the courage to say that as a result of the Exploratory Phase I knew what children *do*, albeit broad-brush. This lack of faith in viewing the outcome of the Exploratory Phase as *results* meant that I had gone round in circles trying to work out how to justify the Drawing Types, rather than using them as just one layer in a multi-level comparative analysis of two groups of children: one who received a Programme to improve their design skills and one who did not.

5.4.3 Re-defining the Analysis Instrument

The realisation that my Grid and Ticksheet were not working well was to result in the complete restructuring of the analysis instrument. Fortunately, it did not result in a complete re-marking of all the children's work, since it was only the middle-band children who were affected by the changes in the re-structuring of the instrument. However, Mrs.R. offered to second-mark all the children's work. I was concerned that we ensured that all drawings with identical characteristics had received identical evaluation. This process took longer than either of us expected but gave me increased confidence in the reliability of the resultant quantified analysis.

During the evolution of the Grid, I had identified Purpose of the Drawing as <u>a</u> major factor in determining the kind of drawing the children would produce, But until all the drawings from the Structured Phase had been examined and discussed, I had not identified perception of purpose as <u>the major factor</u>. I had been aware of Bridget Egan's work in this area for some

years (for example, Egan, 1996) and we have had an on-going dialogue at conference venues since 2000. Her interviews with children in Years 1 and 6 revealed that many children did not understand the function of the drawing in the design process. Those who did articulate some measure of understanding tended to see drawing as recording, not development, of ideas. I agreed with her that children's understanding the role of the drawing was important, and had put this at the head of my revised Grid, but I had not considered the import of this being *the determining factor* in how the children used drawing to support designing.

5.4.3a A Holistic Analysis Tool

Parallel to this, I was searching for texts relating to analysis methodologies for evaluation of young children's learning, looking for ways to address the issues raised by the research group, and which might provide an more appropriate model for an analysis tool.

A child-centred framework for defining and assessing quality in early childhood education is described by Pascal & Bertram (1989), (Fig.52). Each segment represents an area of the child's development or competence. It is person-centric (not curriculum-centric) and emphasises that the child is central to any assessment process and that all dimensions of the child's learning are inter-related, not discrete or linear.



Fig.52 : Pascal & Bertram's (1989) Assessment Model

This seemed ideally suited to my needs. Not only does this mesh with my view of the holistic nature of education for young children, but I could see potential in this model for describing the child's view of design drawing, centred on the child's understanding of the purpose of drawing, the arrows suggesting development in the child's understanding (Fig. 53, overleaf) :



Fig.53 : Purpose of Drawing Holistic Assessment Model (a)

Rings could be added to represent different levels of capability :



Fig.54 : Purpose of Drawing Holistic Assessment Model (b)

and, following Pascal & Bertram's model, the Aspects of design drawing would fit into each of the segments of the circle (Fig.55a, overleaf). However, to display assessment results graphically (as radial plots generated by a spreadsheet), the Aspects needed to be placed on the lines (as in Fig. 55b) rather than within the segments:

SECTION 5 - Analysis of the Assessment Tasks



Fig.55 : Dimensions Wheel

This has the added advantage of having eliminated the boundaries between the dimensions, which suggests, even more strongly than Pascal and Bertram's (1989) original model, the inter-connectedness of the dimensions. This is particularly apposite for my view of design drawing: my Drawing Types had assessed the drawing as a whole rather than separating out parts and the Grid had been created from these holistic assessments.

I decided to call this model a Dimensions Wheel, since Pascal & Bertram (1989) specifically use the word "dimensions" and, as in their understanding of a child's developing competencies, all the dimensions of using drawing for design are represented as inter-relating rather than discrete. This change of terminology was not purely cosmetic but represented a real change in the way I was now viewing the centrality of understanding the purpose of the drawing. This was then worked out through the Dimensions on the spokes of the wheel, refinements of the Aspects from the Grid.

In analysis of the Assessment Tasks drawings the concepts of constraints and possibilities had become viewed as a duality. Addressing the Constraints of the Task was to include all task criteria as set, not solely the needs of the user, as previously. This emerged during analysis of the drawings via the Ticksheet, through comparing findings within the "Satisfying Criteria" sub-group (see Table 15 in Section 5.4.3b(iii)). For example, at Easter some children did not include the oversized tube in their design, although this was central to the problem and I felt this was more important that whether they had written "For Mummy" around the rim. Likewise, in terms of "possibilities", Exploring the Possibilities of the Task meant more than producing multiple ideas. Novelty is also important: six different ways of colouring the cardboard tube being less creative than turning it into an Easter Bunny whose ears popped up as the lid was lifted.

SECTION 5 - Analysis of the Assessment Tasks

The arrangement of the Dimensions on the wheel is important. If there is an entry and exit point to this diagram it is that children generate ideas which address the task constraints and possibilities, evaluating as they communicate to themselves and others, plan construction and think about what they product will look like (colour, decoration etc.), which all becomes the basis of the product which they make. The more a child thinks through all these issues, the more successful their design will be. The extent to which they use drawing to record and support thinking about their design as it develops contributes to the overall success of the project. This accords with the Kimbell et al.'s (1991) model of the design process (Fig.21).

In effect, there were two layers to this quantitative analysis instrument based on Pascal & Bertram (1989). An understanding of the *purpose* of design drawing, which is reflected in the way in which children *use* drawing to support design thinking (dimensions).

5.4.3b A Multi-layered Analysis Tool

As neat as the double-layered Purposes and Dimensions wheels were, I actually had four parts to the analysis system:

- Purposes Continuum
- Drawing Types
- Dimensions Continua
- Techniques Ticksheet.

In order for this analysis system to have cohesion, there needed to be a clearly defined relationship between the elements. That the Purposes Continuum and the Dimensions Continua would mesh together into one coherent system was implicit in the way that both had been created from the Grid. There would also be a relationship between the Continua (Purposes and Dimensions) and the Drawing Types, as the Grid had been created from the Drawing Types. However, although I did not want to let go of my Drawing Types, I now knew that they had to be related to the Purposes and not the other way around.

The Drawing Types were part of the external evidence of what was going on inside the children's heads and related to their understanding of design drawing in the same way as the External and Internal Taxonomies related to one another as two sides of the same coin. Meshing them together was not necessarily going to be neat and tidy. I was beginning to wonder if I had unwittingly collected my evidence to reflect this internal / external duality and

that the struggle I was having to organise it resulted from this lack of realisation. It would appear that I had:

Internal	Pupose Continuum and Dimensions Wheel
External	Drawing Types and Techniques Ticksheet

These internal / external relationships were parallel to those inherent in the Understanding Technological Approaches Project (Stables, 1997), which examined the relationship between the modelling strategies used by children (discussing, looking, drawing etc.) and the internal processes of their design intentions (generating, developing, planning, etc.). I had come to a similar way of viewing the judgements I was making on the children's drawings but from a different route: the need to make sense of the multiple layers of my analysis instrument. I was considering these as tentative ideas, worrying that it all seemed too neat to be substantiable, only to find that others had trod the path before.

5.4.3b(i) The Purpose Continuum and the Drawing Types



Fig.56 : Transforming the Purpose Rings into a Continuum

I had come to believe that the way in which children use drawing to support their design thinking is largely determined by their understanding of the function of the drawing within the process of designing. Making comparisons about how the children perceive the purpose of the

SECTION 5 - Analysis of the Assessment Tasks

drawing in each of the Assessment Tasks is, therefore, probably as much a measure of how the children perceived the *usefulness* of drawing to support the task in hand as a measure of their perception of the role of drawing in designing per se.

0	1	2	3	4	5
No drawing	Views the drawing as the product	Clarifying their idea of what is to be made	Using drawing to record design possibilities	Develop idea and indicate how it might be made	To work out what will be made & how

Table 13a :The Purpose Continuum (see also Appendix N: AN2.5)

Table 13b : Drawing Types

Drawing Type						
No drawing	Picture	Single-draw	Multi-draw	Multi-design	Progressive	Interactive
				<u></u>		

Placing the Drawing Types below the Purpose of Drawing Continuum shows how the two might relate to one another. The assumption could reasonably be made that more sophisticated Drawing Type would be paired with a more sophisticated understanding of the Purpose of drawing for designing, and vice versa. It would be expected, therefore, that viewing the drawing as product would relate to either a Picture or Single-draw; clarifying ideas to Single or Multi-draw; recording design possibilities to Multi-draw or Multi-design, developing the idea to Multi-design or Progressive and, finally, working out what would be made to Progressive and Interactive. The results of this comparison are detailed in Section 5.6.2.

The advantage of this holistic model based on the child's understanding of the Purpose of the Drawing was that it cut through the problems inherent in the non-linearity of the Drawing Types classification (and what I now thought of as the "muddle in the middle" of Multi-draw, Multi-design and Progressive categories) and the reality of children using drawings appropriately differently in response to different tasks. Multi-draw straddles Clarifying and Design Possibilities, allowing it to be more than a Single-draw but still clarifying the problem without developing the idea, yet it can also recording a design possibility because the child is doing more than just the tokenism inherent in the Single-draw. For Multi-design it allows for both the almost random brain-storming and the multiple discrete ideas for which a connecting thread can be seen.



Fig.57 shows how this relates to the Container / Journey metaphor :

Solid lines indicate plateaux of understanding. The two innermost rings represent a static use of drawing, with limited relationship to the designing and making of a product. The red shade rings indicate a growing understanding of how drawing may be used to progress design ideas and thus records a design journey.

5.4.3b(ii) The Dimensions Continua

The same quantification was applied to the descriptors on each spoke of the Dimensions Wheel (Fig. 58), with each ring relating to a level of understanding:



These descriptors formed the Dimensions Continua, to which numerical values were applied, as indicated in Table 14, overleaf, and also Appendix N: A2.5). Comparison with the Grids in Sections 5.4.1 - 2 and Appendix N will reveal how the Continua relate to earlier iterations of the Assessment Instrument

The relationship between the child's understanding of the *purpose* of drawing for designing and the way in which they *use* drawing to support designing as measured on the Dimensions can be seen by reading down the Continua. Thus a child who appears to be in Band 1 of the Purpose of Drawing Continuum will probably demonstrate many of the characteristics in Band 1 of the Dimensions Continua. However, it would be too simplistic a notion to expect that many children would fall neatly into the same Band for all aspects of their work, especially as their understanding develops. The degree to which the children's Dimensions Continua scores related to the Purpose Continuum scores is considered in Section 5.7.3.

Table 14 : The Dimensions Continua:

	ping beorgi racao			
1	2	3	4	5
Drawing a picture,	Simple sketch,	Design ideas	Progression of ideas	Uses drawings
not designing a	showing object to	generated but not	across or within	reflectively to
product	be made	developed	drawings	generate new ideas

Generating and Developing Design Ideas

Exploring the Possibilities of the Task

1	2	3	4	5
Design possibilities are not addressed in the drawing	Stereotypical response, showing little creative thought	Recording possible creative solution(s) to the task	Using drawing to develop novel design solution(s)	Combining novel solutions to produce innovative design

Addressing the Constraints of the Task

1	2	3	4	5
Minimal	Drawing shows	Records way to	Task constraints	Task constraints
understanding of	some understanding	address task &/or	considered as the	treated as part of
task / user needs	of task constraints	client needs/ wants	design proceeds	iterative process

Planning the Look of the Product

1	2	3	4	5
Appearance of a	Little consideration	Overall decorative	Ideas about finishes	Ideas about finishes
product is not	of final appearance	scheme considered	are added to design	develop within
considered	of product		whilst drawing	overall designing

Communicating Design Ideas

1	2	3	4	5
Use of narrative	Simple unlabelled	Conveys some	Conveys sense of	Clear enough for
or other drawing	sketch(es);	sense of the object	object to be made;	someone else to
genre	relying on	to be made; e.g.	e.g. working	make the product
**************************************	shared meanings	indicates materials	diagram	

Planning Construction

1	2	3	4	5
Not planning to	Minimal	Drawing indicates	Drawing	Constructional
make the object as	consideration of	some consideration	demonstrates	issues considered
drawn	construction	of construction	consideration of	en route to
	whilst drawing		construction	final design

Evaluating whilst Drawing

1	2	3	4	5
Yet to define the design task	Minimal evaluation at drawing phase	Considered and rejected a range of	Decisions made about product	Changes made as result of considering
		ideas	whilst drawing	design drawings

A Basis for Making

1	2	З	4	5
Making an object	Product relates to	Object is one of the	Clear development	Using drawing as
is seen as separate	ideas recorded in	ideas drawn	path through	resource during
new activity	the drawing		drawing into	making
			making	

5.4.3b(iii) The Techniques Ticksheet

The Ticksheet reverted to being a record of the techniques used by the children in the recording of design ideas through drawing. Having split this into headings relating to the Aspects as perceived under previous versions of the Grid, there needed to be some re-arranging of the way the data were to be collated but fortunately no re-assessing of children's work.

Although less problematic in terms of development, and therefore having had few words devoted to it, the Ticksheet frequently recorded the detail on which was based the decision as to where to place work on the Continuum or the Grid. The more holistic decisions were not made on the basis of intuitive "feel" but on the basis of the firm evidence of the content of the children's drawings. During the checking and collating process, it was frequently the individual Ticksheets that were consulted to verify that equal credit was being given to equal work. Recording related information in two different formats (Grid and Ticksheet) aided consistency of judgements made about the drawings across tasks and children.

Overleaf is shown Table 15, The Techniques Ticksheet (Version 2), to be found also in Appendix N : AN2.4.

The results from the analysis of the techniques used by the children based on the analysis of the Ticksheet results can be found in Section 5.8.

Table 15 : The Techniques Ticksheet

GENERATING AND DEVELOPING	PLANNING CONSTRUCTION		
single drawing of a product to be made	recording materials:		
single idea recorded as working drawing	indicates materials by list		
more than one attempt at drawing same idea	indicates materials within sentences		
develops single idea into a working drawing	indicates materials by drawing		
range of ideas recorded as guick sketches	indicates materials by labelling drawing		
range of ideas, one developed towards making	materials indicated are suitable		
range of ideas, some developed towards making	materials indicated are available		
progression of ideas across drawings	recording construction:		
drawings combined to generate new ideas	indicates cuts, folds &/or fixings		
combines ideas to produce best solution	indicates measurements		
. 	indicates equipment needed		
SATISFYING CRITERIA	indicates parts to be assembled		
as set by teacher	if construction planning suff. recorded:		
begins from task requirements	technically realistic (child can make it)		
drawing used to address task	can be made in time available (roughly)		
aware of task constraints and possibilities			
maintains task requirements into making	EVALUATING WHILST DRAWING		
as required by client/user	attempts to improve drawing of single idea		
begins from client/user needs	considered a range of ideas		
maintains awareness of needs whilst drawing	adaptations made to single initial idea		
several ways of satifying client/user's needs	adaptations made to one of two/several ideas		
increasingly focuses on satisfying needs	changes related to annearance of product		
maintains client/user peeds into making	changes related to task specification		
maintaine onchadoci needo into marcing	changes related to client/user needs		
COMMUNICATING DESIGN IDEAS	changes related to material constraints		
relationship between pictures and words:	changes related to construction issues		
relationship between pictures and words:	changes related to construction issues		
relationship between pictures and words: use of appropriate drawing genre	changes related to construction issues justifies choices made		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relating to picture	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relating to picture thinking recorded mostly in pictures	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing:		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing)		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings)		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials)	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same pattern, decoration, picture etc.		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same pattern, decoration, picture etc same materials		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words instructions - labelled diagrams	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique different from drawing:		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words instructions - labelled diagrams clarity of drawing:	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique different from drawing: different object		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words instructions - labelled diagrams <i>clarity of drawing:</i> not relying on shared meanings to interpret d'ing	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique different from drawing: different object different colour		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words instructions - labelled diagrams <i>clarity of drawing:</i> not relying on shared meanings to interpret d'ing conveys sense of object to be made	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique different from drawing: different colour different colour		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words instructions - labelled diagrams <i>clarity of drawing:</i> not relying on shared meanings to interpret d'ing conveys sense of object to be made clear enough for someone else to make product	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique different from drawing: different colour different proportions		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words instructions - labelled diagrams <i>clarity of drawing:</i> not relying on shared meanings to interpret d'ing conveys sense of object to be made clear enough for someone else to make product <i>level of detail:</i>	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique different from drawing: different object different proportions different promotions same fixing technique different promotions different promotions different proportions different promotions different promotions different proportions different proportions different proportions different pattern, decoration, picture etc different materials		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words instructions - labelled diagrams <i>clarity of drawing:</i> not relying on shared meanings to interpret d'ing conveys sense of object to be made clear enough for someone else to make product <i>level of detail:</i>	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique different from drawing: different object different proportions different proportions same fixing technique different proportions different proportions different proportions different different colour different proportions different proportions different pattern, decoration, picture etc different materials different fixing technique		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words instructions - labelled diagrams <i>clarity of drawing:</i> not relying on shared meanings to interpret d'ing conveys sense of object to be made clear enough for someone else to make product <i>level of detail:</i> sufficient level of detail not inappropriately over-detailed	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique different from drawing: different object different proportions different decoration, picture etc same materials same fixing technique different proportions different decoration, picture etc different proportions different pattern, decoration, picture etc different pattern, decoration, picture etc different materials different fixing technique reasons for changes apparent		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words instructions - labelled diagrams <i>clarity of drawing:</i> not relying on shared meanings to interpret d'ing conveys sense of object to be made clear enough for someone else to make product <i>level of detail:</i> sufficient level of detail not inappropriately over-detailed shows dynamics/movement	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique different from drawing: different proportions different promotions same fixing technique different from drawing: different proportions different pattern, decoration, picture etc different fixing technique reasons for changes apparent adapted from drawing:		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words instructions - labelled diagrams <i>clarity of drawing:</i> not relying on shared meanings to interpret d'ing conveys sense of object to be made clear enough for someone else to make product <i>level of detail:</i> sufficient level of detail not inappropriately over-detailed shows dynamics/movement various viewpoints	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique different from drawing: different object different proportions different from drawing: different proportions different from drawing: different proportions different from drawing: different pattern, decoration, picture etc different pattern, decoration, picture etc different fixing technique reasons for changes apparent adapted from drawing: adapted form drawing:		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words instructions - labelled diagrams <i>clarity</i> of <i>drawing:</i> not relying on shared meanings to interpret d'ing conveys sense of object to be made clear enough for someone else to make product <i>level of detail:</i> sufficient level of detail not inappropriately over-detailed shows dynamics/movement various viewpoints expansions to show small details	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique different from drawing: different object different proportions different object different proportions different from drawing: different proportions different proportions different proportions different from drawing: different pattern, decoration, picture etc different materials different fixing technique reasons for changes apparent adapted from drawing: adapted from drawing: adapted object adapted proportions		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words instructions - labelled diagrams <i>clarity of drawing:</i> not relying on shared meanings to interpret d'ing conveys sense of object to be made clear enough for someone else to make product <i>level of detail:</i> sufficient level of detail not inappropriately over-detailed shows dynamics/movement various viewpoints expansions to show small details out away diagram to show inside of product	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique different from drawing: different object different proportions different object different proportions different from drawing: different proportions different proportions different proportions different proportions different from drawing: different proportions different fixing technique reasons for changes apparent adapted from drawing: adapted from drawing: adapted proportions adapted proportions		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words instructions - labelled diagrams <i>clarity of drawing:</i> not relying on shared meanings to interpret d'ing conveys sense of object to be made clear enough for someone else to make product <i>level of detail:</i> sufficient level of detail not inappropriately over-detailed shows dynamics/movement various viewpoints expansions to show small details cut away diagram to show inside of product indicates how parts will fit together	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique different from drawing: different proportions different proportions different proportions different from drawing: different proportions different fixing technique reasons for changes apparent adapted from drawing: adapted object adapted proportions adapted proportions adapted proportions adapted proportions		
relationship between pictures and words: use of appropriate drawing genre pictures only single words or phrases relatiing to picture thinking recorded mostly in pictures thinking recorded mostly in words equal weighting between words and pictures labelled diagram (with) arrows or lines list (e.g. of materials) full sentences to describe planned product instructions - words only instructions - pictures with few/no words instructions - labelled diagrams <i>clarity of drawing:</i> not relying on shared meanings to interpret d'ing conveys sense of object to be made clear enough for someone else to make product <i>level of detail:</i> sufficient level of detail not inappropriately over-detailed shows dynamics/movement various viewpoints expansions to show small details cut away diagram to show inside of product indicates how parts will fit together detailing finished appearance:	changes related to construction issues justifies choices made RELATIONSHIP TO MAKING same as drawing: same object (single drawing) one of objects drawn (multiple drawings) same colour same proportions same materials same fixing technique different from drawing: different object different proportions different proportions different object different proportions different proportions different proportions different proportions different pattern, decoration, picture etc different fixing technique reasons for changes apparent adapted from drawing: adapted object adapted pattern, decoration, picture etc adapted materials adapted fixing technique		

5.4.4 Summary of Section 5.4

The redefinition of the analysis instrument in the light of Pascal and Bertram's holistic child-centred view of quality evaluation had given me a much needed transformation of my perspectives on the children's achievements. It meshed with my view of child-centred education and provided a more holistic model for the assessment of the children's design drawings. The Continua could still be analysed by spreadsheets and radial plots could be used to display the results in a format that reflected the structure of the model.

As appears often to have been the case, the strands which came together to create the final analysis instrument were things that I had perceived, put to one side and then picked up again and re-combined when I found that I already had the elements of something useful. The creation of the analysis tool for the quantified data had taken over two years. It had needed testing on the real data to find where it fitted and where it did not, as well as being subjected to moderation and theoretical evaluation. Dealing with these issues not only transformed the analysis instrument but also how I viewed the analysis of my data.

5.5 Understanding the Purpose of the Design Drawing

To aid readability, certain conventions have been adopted and used consistently throughout Sections 5.5 - 5.9:

- In all charts with shared borders, the axes legends are directly below the chart title,
- The number of children is indicated as n=sample number,
- The sample size on the charts is expressed as percentages,
- The colour conventions are as indicated on p.14 in the Content Table.

As indicated in Sections 5.4.2a and 5.4.3, I had come to realise that the child's understanding of the purpose of the design drawing was central to the way in which they used drawing to support the development of their design ideas. Therefore, this became the primary criteria on which the drawings from the Assessment Tasks of the Structured Phase were analysed.

By examination of each drawing, supported by Log Book notes, photographs, video and audio recordings, the children's understanding of the purpose of the drawing in each Assessment Task was ascertained according to the criteria on each ring of the holistic analysis model (Section 5.4.3b) as agreed by my moderation panel (Section 4.2.3). This was converted to the numerical equivalents indicated in Table 13a and recorded on individual child spreadsheets. These numerical data were collated into class spreadsheets, from which the charts shown in this section were produced (see Appendices N & O).

5.5.1 Comparing Class Profiles

Comparisons were made between the classes on both a means and range basis. The mean score for each class' score on the Purpose Continuum for each task was calculated and compared. This is presented as a radial plot in Section 5.5.1a, below. The ranges of results for each tasks are displayed as bar charts to show the percentage of children within each band of the continuum in Section 5.5.1b.

Section 5.5.1a Mean Scores

Chart 5 overleaf (created from the spreadsheet in Appendix O: AO1, Sheet 1) shows the mean scores for each class on the Purpose of Drawing Continuum, from which the difference in understanding between the two classes can be seen. The mean score for each class for both

Pizza and Frosty tasks demonstrate similar levels of understanding of the purpose of drawing for designing in both classes, both in product design and problem scenario tasks. The slight differences can be accounted for by the micro-factors of individual variations in capability and even, perhaps, who was absent on the day.



That the Focus Class made little progress between Tasks 1 & 2 helped to allay my fears about the "me factor" in the situation. It demonstrates that there was nothing special about my teaching style or way of doing Design and Technology that would necessarily make this group of children perform considerably better than the Comparison Class being taught by my highly competent colleague, Miss N. The results of both tasks are consistent with my expectations based on observations of Year 2 children in the Exploratory Phase.

Immediately after the explanation of the Container / Journey metaphor, however, the Focus Class children demonstrate a great leap forward in understanding, whereas the Comparison Class made only a slight gain, perhaps due to the difference between the Frosty and Easter Egg Holder tasks. The progress made by the Focus Class at Easter takes them clearly into the next band of the Purpose Continuum. Rather than using drawing to clarify their idea of what is

SECTION 5 - Analysis of the Assessment Tasks

to be made, they using drawing to record design possibilities and moving towards developing these ideas, indicating how they might be made. This difference in understanding between the two classes is maintained across the rest of the Programme.

It would be difficult to account for this outside of Programme effects. If there was a difference between my teaching ability *per se* and that of Miss N., then there would have been a difference in the two classes for the Frosty task, as this occurred three months into the Programme but before the Container / Journey input. That the differences between the classes continues across the subsequent Tasks, all with different demands and expectations, demonstrates that real understanding and shift in the children's perception of the function of drawing for designing had taken place.

Section 5.5.1b The Range of Results

Chart 6 (created from Sheet 2, Appendix O: AO1) overleaf shows that:

a) most children in both classes began with some understanding of the purpose of drawing in a design context, both for a problem scenario and a product design;

b) at Easter, most of the Focus Class children used drawing to record and develop design possibilities, whereas in the Comparison Class the percentage who did so is off-set by those who are still using drawing simply to clarify their ideas;

c) in the more structured task Card, the Focus Class were using drawing for designing, whereas in the Comparison Class just over half of the children used the drawing to record an "intent to make one of these", clarifying but not developing design ideas;

d) for the Suitcase, the majority of the Focus Class used the drawing to record a range of ideas, place-marking their design options, with a group of higher achievers developing ideas towards making. Nearly half of the Comparison Class have not moved beyond recording a simple sketch, defining the task as "making a suitcase";

e) for the Maze, a roughly equal number of children in each class used drawing to record design possibilities. It was what the others did that made the difference. No one in the Comparison Class did better than this; whereas only a third of the Focus Class children used drawing for clarification without development.

	C	hart 6 : Purpos	se of Drawing	Profiles	
Conversion o	f Purpose Cont	tinuum to X-Ax	es:		
No drawing	Views the drawing as the product	Clarifying their idea of what is to be made	Using drawing to record desig possibilities	Develop idea n and indicate how it might be made	To work out what will be made & how
0	1	2	3	4	5
y axes =	Percentage of	Children	Focus	Comparison	
Pizza		Frosty			
	both classes n=	-20	Focus n = 23; Comparison n=18		
80			80 _T		
60-			60-		
40	40-		40+		
20-	20-		20-		
0 1	2 3 4	5	0	1 2 3	4 5
	Easter			Card	
Focus n	= 22; Compari	son n=21	Foc	us n = 23; Comp	parison n=21
80			80 ₁		
60-			60-		
40- 20- 0	2 3 4	5	40 20- 0 0	1 2 3	4 5
	Suitcase			Maze	
Focus r	n = 23; Compari	son n=21	Fo	cus n=19; Comp	arison = 20
80 ₁			80 _{\[}		
60-			60-		
				_	
40			40		
20			20		
1					
0			0		

Gill Hope (2004) Drawing as a Tool for Thought

From Frosty onwards, at least 40% of the Comparison Class still remain at the stage of using drawing to clarify what it is to be made and do not use drawing to develop design ideas. The difference between the two understanding of the way in which drawing can be used to support ideas development can be seen in Chart 7 by splitting the Purpose Continuum into "static" (Categories 0-2, i.e. *No drawing, Drawing as Product* and *Clarifying*) and "moving" (Categories 3 - 5 : *Design Possibilities, Developing Ideas* and *Working out what and how*):



The radical sustained change amongst the Focus Class children at Easter, which is maintained across the product design tasks supports the hypothesis that the Container / Journey metaphor was understood by these children and that they learnt to use drawing to support the development of their design ideas. The Comparison Class bars show a gradual increase in understanding, as one would expect across such a length of time but, at most, only half the class' drawings showed any sense of design development on any one task. Despite the unfamiliarity of using drawing for a problem scenario rather than a product design, three-quarters of the Focus Class children still exhibited a sense of movement of ideas across their drawings for the Maze. The Comparison Class remained evenly split, presumably perceiving no difference in the way drawing might be used in either type of task.

5.5.2 Developing Understanding over Time

To ascertain the extent to which the children's understanding of the purpose of drawing for design progressed across the course of the Programme, analysis was conducted at individual child level. Since I had observed that children were using drawing differently in problem scenarios (Frosty and Maze) to the product design tasks (Section 5.2), the results were analysed separately. The hi-lo charts (Chart 8 overleaf, for which Appendix O: AO1.2 shows the numerical data) were created to demonstrate this development over time, using data from the quantified analysis of the drawings from the first and last task of each type. This is not a pre- / post-programme comparison, since the Suitcase task occurred part-way through the Programme, not at the end, and Frosty did not come at the start. However, each pair of tasks were roughly equidistant in time (Pizza - Suitcase 10 months, Frosty - Maze 12 months).

The Assessment Tasks increased in difficulty over time, to keep pace with the children's maturity across the 15 months of the project, so that a retrograde step on the chart does not imply forgetting or regressing. It is more likely that the child has not been able to apply their understanding in new or more complex circumstances. As in indicated in Section 4.3.3, both Teaching Input activities and Assessment Tasks were tailored to the children's age at time of delivery, based on my experience as a teacher and the results of the activities conducted in the Exploratory Phase of the research.

The x-axes in Chart 8 represent each child in first name order for both classes but, due to absences, the tenth point, say, on each x-axis does not necessarily represent the same child. Only children present for both tasks on each chart were considered.

Chart 8 : Changes in Understanding the Purpose of Drawing for Designing

x-axes = Individual Children

n = number of children present for both tasks represented on chart

y axes = Scalar equivalent of Purpose Continuum as per the following table :

	Child's Vie	w of the Purpos	se of the Drawing	9	
No drawing	Views the drawing as the product	Clarifying their idea of what is to be made	Using drawing to record design possibilities	Develop idea and indicate how it might be made	To work out what will be made & how
0	1	2	3	4	5



From Chart 8, it can be deduced that :

 there appears to be more progression of understanding in the product design tasks than in the problem scenarios, and in the Focus Class than in the Comparison Class;

 the most movement can be seen in the Focus Class' product design and the least in the Comparison Class' problem scenarios;

• the problem scenario chart indicates almost no change in Comparison Class' understanding of drawing to support thinking towards a problem solution. In contrast, some of the Focus Class have made considerable progress in understanding the use of drawing for problem-solving.

 the product design tasks: in the Focus Class, all but one of the children are using drawing in a designerly way, whereas only about half of the Comparison Class children have progressed beyond using drawing for clarifying the task

These results were compared by creating class profiles across each pair of tasks (Chart 9), to determine whether this would indicate that the Focus Class gained greater understanding of the purpose of drawing for designing than would be expected simply by maturity across the course of the 15 months of the Programme. This was a form of norm-referencing the two class samples, rather than comparisons to external criteria, since the variables within the tasks were not sufficiently controlled. However, in Chart 9 the profiles favour the Focus Class in both Problem Scenarios and Product Design.



Problem scenarios:

The Comparison Class' profile represents roughly the expected outcome for tasks matched appropriately to the age and maturity of the children, delivered several months apart. The Focus Class' profile represents an increase in understanding of the purpose of design that is

greater than that expected by maturity alone and, I would argue, represents the result of receiving the input delivered through the Programme.

Product design:

Both classes had considerably more practice in product design across the course of the Programme and it is clear that the Comparison Class are happier here. However, the two profiles still demonstrate a greater gain in understanding among the Focus Class children.

If one considers the Comparison Class as representing ordinary maturation and growth in understanding without specific teaching input about the purpose of design drawing, then these profiles would suggest that the Programme and the Container / Journey metaphor played an important role in developing the Focus Class understanding of the purpose of design drawing, that probably could not be deduced by the children without that specific teaching.

5.5.3 Emergent Themes from the Purpose of Drawing Analysis

Section 5.2 suggested ways in which the data could be examined in order to answer questions about the similarities and differences in the children's responses across as well as within tasks and this emerged as a key theme within the analysis of the children's understanding of the purpose of the drawing. This section examines the data from both cross-task and cross-class viewpoints.

5.5.3a Cross-task Comparisons

Comparing Pizza, Easter & Suitcase for development in drawing for product design:

From a position in which roughly three quarters of all the children used drawing to clarify to themselves the nature of the object to be made (Pizza), the profiles of the two classes for the two subsequent product design tasks develop not only differently in respect to each other, but also in respect to the two tasks. In the Comparison Class, nearly half of the children remain at this level of understanding. Even for the Suitcase, they produced a simple outline drawing of the object to be made. Nearly all of the Focus Class children moved on from this level. However, their use of drawing was different for each of the two subsequent tasks.

My initial reaction on viewing the results for the Suitcase was that they had regressed but consideration of the task with which they were presented reveals that they made task-appropriate decisions. For the Easter Egg Holder, the children were each given a card tube, which stood on their desk as they drew. They were asked to make this tube into

SECTION 5 - Analysis of the Assessment Tasks

something, consequently they used the drawing to develop ideas about how they might do this. They played with a range of ideas but their thoughts were firmly centred on using the drawing for the development of how to make the chosen one. For the Suitcase, they were shown a toy panda and his plastic mac and told that he needed them to design him something to carry his mac on holiday. Consequently, they recorded a range of travel bags with different patterns and logos and decided which was most suitable for Pandy. Decisions about which idea they could most easily make were made in their heads.

As part of the teaching input about Containers and Journeys, I had aimed to teach a level of understanding of the role of drawing and how it could support designing, that would allow children to use drawing as a tool for thought, as, when and *how* they felt it to be appropriate. The way that the Focus Class matched their use of drawing to differences within the product design tasks appears to suggest they were capable of making choices based on understanding of permission: not just what could be done but what they were allowed to do with the drawing.

Comparing Frosty and Maze for development in drawing to support problem solving:

Exactly a year separated these two activities. It would be expected, therefore, that the children's understanding of the use of drawing for designing would have moved on. For the Focus class, it can be seen to be so. Only a third of the class used drawing to clarify what a maze is, whilst an equal percentage used the drawing to move their ideas towards planning construction. In the Comparison Class, more children used drawing to record design possibilities than for Frosty, but this was by recording of several different mazes, not the development of a design idea in the sense of playing with or recording design decisions. This accords with the observations reported in the qualitative analysis (Section 5.3.1f) that the Comparison Class children had left too many issues about their mazes unresolved before making.

Comparing structured delivery to less structured :

The Card task was the only Assessment Task not devised by myself and also it was the only one for which a pre-printed design sheet was provided and for which the children were talked through the activity. In my Assessment Tasks, I explained the activity and then gave the children blank paper on which to try out ideas before they started making.

The Focus Class performed roughly at a similar level to Easter, with slightly more children producing a range of design possibilities rather than recording how they would make the product. Since they were given specific instructions to indicate how the Card might be made, I did not count *"It is made from card"* and a picture of a glue pot. There had to be a clear progression of ideas between Boxes 5 and 6 (see Appendix I).

SECTION 5 - Analysis of the Assessment Tasks

It might have been assumed that this structured approach would have enabled the Comparison Class to do better than my blank paper. It could be argued that the Focus Class were now used to my way of working and that they would, therefore, perform better within any Assessment Task that fitted the pattern to which they had become accustomed. The Card task gave all children an equal opportunity to show their design understanding: it was a novel situation with an unknown presenter. This task would enhance the reliability of the Assessment Tasks results through this different perspective on children's performance.

As Chart 5 in Section 5.5.1a shows, both class' performance on the Card task is closely related to their performance on the Easter and Suitcase tasks. The Focus Class have maintained their enhanced understanding of the purpose of design drawing into unfamiliar territory. Many of the Comparison Class were still at the stage of using drawing to clarify their ideas and were not able to use drawing in any more sophisticated way, even when talked through the stages of product evaluation, identifying a client, generating ideas and planning construction. They frequently used the Planning Construction space (Box 6) to describe their drawing in Box 5.

5.5.3b Cross-Class Comparisons

Similarities between both classes across all tasks:

The majority of children from both classes had some sense of the purpose of drawing in a design context; there were very few children drawing pictures or not drawing at all. Equally, there were few children who were using drawing to work out in any detail how their idea would be made. This might be indicated in general terms but overall the children preferred to work out how to make their design solution once they had the construction materials in their hands. On all tasks, most children used drawing to clarify the task to themselves, record design possibilities or to develop an idea towards making.

Differences in task responses, both classes:

In problem scenarios, the tendency was for children to use drawing for recording possibilities about how the solution will be realised but their perception of the outcome of the activity *as a model*, means that they abandoned drawing for modelling in the construction media at a much earlier stage of their idea development than in product design tasks, where the desire to produce a pleasing artefact provided the incentive to think through production before beginning to engage with the materials. The higher achieving children perceived that drawing provides a means of supporting such planning.

Differences in task responses between classes:

Closer examination of the differences between the product design tasks suggested that Focus Class children appeared to able to use drawing differently according to the task parameters and thus have a sense of drawing as a tool whose purpose is to aid their designing. For example, for Pandy's Suitcase, confidence in their ability to make the bag meant greater use of drawing to explore type, shape and decoration than to develop construction methods. Most of the Comparison Class, however, did not appear to reach the level of understanding of design drawing at which they could manipulate the tool sufficiently to make such task-appropriate choices.

Differences between classes regardless of task:

At the start of the Programme, the average child in both classes was using drawing to clarify the task to themselves. By the end of the project, the average Comparison Class child was still doing this, in line with expectations from the Exploratory Phase. Splitting the Purpose Continuum into "static" and "moving" categories revealed that from Easter onwards (after the sharing of the Container / Journey metaphor) the majority of the Focus Class children were using drawing as a design tool.

Within the Focus Class there were a group of consistently high achievers who were beginning to use drawing to work out how their product might be made. The number of Comparison Class children doing this was small.

5.6 Drawing Types

Despite the re-defining of the analysis instrument in the light of my reading of Pascal and Bertram's (1991) and the realisation of the centrality of the child's understanding of the purpose of drawing in designing, I did not want to abandon the Drawing Types that I had identified during the Exploratory Phase of the research. in particular, I wanted to discover how these related to the child's understanding of the purpose of the drawing.

5.6.1 Analysis of Drawing Types Used

The range of drawings with which both classes presented me as a result of the Assessment Tasks stretched across the whole range of Drawing Types identified during the Exploratory Phase of the research (Section 3.3.5). There were, however, a much higher level of annotation among the Structured Phase cohort (analysed in Section 5.8.2c(i)). Within the Comparison Class, one child (Peter) wrote rather than drew in response to two Assessment Activities and there were several for whom the drawing served only as an illustration of their text.

The unlabelled Multi-Draw, which had been quite prevalent in the earlier study, was almost non-existent. Text was used to explain what their drawing was intended to illustrate rather than redrawing more carefully. This gave me many examples of Single-Draw-with-Text that I was initially unsure how to classify. Some of them were clearly in the Progressive category because the text extended the information contained in the drawing but others were static

I decided, therefore, that :

- static Single-Draw with cursory labelling would be designated Single-Draw,
- Single-Draw with extending text would be designated Progressive,
- Single-Draw with explanatory text (which did not extend ideas beyond that recorded in the drawing) would count as Multi-Draw since the children appeared to be using the text in place of redrawing.

Appendix O: AO1.3 shows the Drawing Type analysis as quantified data, based on Table 13b, Section 5.4.3b(i). This was necessary for quantified analysis. In the charts which follow, however, the numerals have been re-converted to the Drawing Types that they represent. The discussion of the analysis of the Drawing Types does so from two viewpoints: by Type and by Task. Each view illustrates different aspects of the children's choice of Drawing Type.

5.6.1a Viewing by Type

Chart 10, which represents this viewpoint shows *stacked percentages*, showing the overall use of the Drawing Types by each class.



Drawing a *Picture* was rare, albeit more common in the Comparison Class. Thus, virtually all of the children understood something about the genre of drawing for designing, even if what they drew was not really informing their planning of an object to make.

Single-Draw was favoured by both classes for Pizza and Frosty. The Focus Class hardly used it for the later product design tasks (absent completely at Easter), although it re-emerges for the Maze problem scenario. Roughly equal numbers of Comparison Class children use Single-Draw in all tasks.

The lower incidence of *Multi-Draw* to support designing can be seen clearly. However, it was more common among Comparison Class children than in the Focus Class, suggesting that the Comparison Class were continuing to use the drawing to clarify the problem rather than develop a solution.

Similar numbers of children in both classes used *Multi-design* overall. In the Focus Class this seems to be spread evenly across all tasks, with just a few more for the Suitcase, whereas for the Comparison Class this was most prevalent for Frosty and Maze, the two problem scenarios. *Progressive* drawings are much more common among the Focus Class children from Easter onwards, including its use for the Maze (where it is completely absent in the Comparison Class). *Interactive* drawings are rare in either class.

5.6.1b Viewing by Task

The stacked percentages in Chart 11 do not exceed 100% as they represent the range of drawing types within each task. Those for the Comparison Class for Easter and Card do not reach 100% due to one child not using drawing at all.



Problem scenarios: The balance of Drawing Types for Frosty and Maze is quite different among the Focus Class, whereas it remains almost unchanged within the Comparison Class. Far fewer Focus Class children use Single-draw for the Maze than for Frosty and Progressive drawings are strongly in evidence.

Product design : The interface between the first three categories (*Picture, Single-Draw* and *Multi-draw*) and the higher categories (*Multi-Design, Progressive* and *Interactive*) is very different in each class from Easter onwards. This distinction between categories represents a more sophisticated use of drawing for design, whether by recording options to consider or recording the development of one idea towards construction.

The 50% line has been indicated on the chart to enable appreciation of the difference between the classes in this respect and Table 16 compares the overall percentage of Single-Draw plus Multi-draw with those of Multi-design plus Progressive drawings for each class across all tasks. Since the Single-Draw and Multi-Draw Drawing Types represent the static, whereas Multi-design and Progressive involve the use of drawing to record and support the movement of design ideas, the Focus Class are strongly pitched towards movement. In terms of the Container / Journey metaphor, they are travelling.

	Focus Class	Comparison Class
Single-Draw + Multi-Draw	37%	47%
Multi-Design + Progressive	60%	49%

Table 16 : Static Drawings vs. Moving ideas

5.6.2 Relating Purpose Continuum to Drawing Types

That more a sophisticated understanding of the purpose of design drawing should result in a more sophisticated use of drawing seems common sense. To relate the Purpose Continuum to the Drawing Types, the two scales were placed alongside each other as shown in Section 5.4.3b(i). The Child's View of the Purpose of the Drawing represents a continuum on which the numbers are ordinal points on a scale not cardinal numeric values, whereas the Drawing Types are discrete and although there is a clear difference in the two ends of the Drawing Types scale, the middle ranges are not necessarily consecutive stages (as discussed in Section 5.4.2b). It is not surprising, therefore, that as shown on the chart overleaf, a range of Drawing Types were employed to support different understandings of the Purpose of the drawing.



This comparative activity acted as a useful double-checking device, as it forced me to re-consider categories and classifications of individual children's drawings as the range of Drawing Types used within each Purpose Continuum band became apparent. More importantly, it enabled me to separate my Drawing Types into Static (Picture and Single-draw) and Moving (Multi-design, Progressive and Interactive), which was a major advance in relating the Drawing Types analysis to the Container / Journey metaphor, with the corollary that bands 0-2 and 3-5 on the Purpose Continuum also fall either side of the watershed.

"Viewing the Drawing as the Product" was rare. Most children understood that they were to draw something that they were then going to make. However, on the Suitcase task, some of the Comparison Class children cut out their initial drawings and sellotaped them together to make the suitcase rather than using the paper for planning a product which was then made in card. These appear as the Single-draw and Multi-draw entries in this category.

"Recording idea of what is to be made" might lead to a Picture (for example, drawing Pandy carrying his suitcase) but it was more likely that this would be a Single-draw, or possibly a Multi-draw if the child were not happy with the first drawing or they might have redrawn to re-inforce to themselves that this was the object to be made. The Multi-design examples categorised here were those who showed just two ideas that were broadly similar and the Progressives showed very limited development of a single idea.

There was a tokenism in these children's drawings, which often masked the detailed discussion they had with peers about their ideas. They were developing design ideas but not using drawing to record them, consequently the details of their ideas were frequently forgotten by the time they had sought approval from the teacher and fetched their materials for construction. They were frequently clarifying the task to themselves rather than using drawing to develop a solution. The single idea, however many times drawn or well-labelled, is static, a statement of what might be made. Design possibilities implies fluidity of ideas.

"Using drawing to record design possibilities" emerged as predominantly the province of Multi-design, despite my determination to include well-developed (possibly labelled) Single and Multi-drawings here. Some children (especially in the Focus Class) produced a single line drawing (a Single-draw) which was so heavily annotated that it was clear that design development had occurred although the idea was not re-drawn.

"Developing design possibilities" could be as simple as re-drawing the same idea and annotating the second drawing (Multi-draw) but was most likely to be through Progressive drawings, where a clear thread of ideas could be seen, perhaps through instructions to make or detailing small parts. The production of a design drawing in which construction as well as a range of ideas were considered was extremely rare amongst these young children. There were some (Focus Class) who developed a single idea all the way through from rough sketch to annotated instructions. There was one child in the Comparison Class (Emma) who recorded a (limited) range of ideas for her card in Box 5, including evidence of combining ideas to produce a final design, and then used Box 6 to produce brief instructions.

Where ideas develop across the page in such a way, then they are clearly Progressive. In the final category ("Working out what will be made and how") could be placed most of the best Progressive drawings from the Focus Class. Of those children who appeared to be using drawings Interactively, there was one who was not considering construction, although her ideas were being developed by mixing and matching previous ideas.

Completely missing was the production of a range of diverse ideas from which one was chosen and developed towards production. Those children who were in the Design Possibilities / Progressive band started with one idea and adapted it through several iterations. They did not follow through and record construction. Those who did record construction had chosen to record just one idea which they developed. It appears to be an either / or : record lots of ideas or develop just one.

The range of ways in which children might do this is best illustrated by examples:

Nicola (Comparison Class) produced a complete set of instructions for making Pandy a suitcase, drew a line under it and wrote *"another way to make a suitcase is"* and produced a second, unrelated set of instructions, almost as if by the time she had produced her first set, she had changed her mind. She did not make either.

Stacey (Focus Class) recorded three false starts for her Easter Egg Holder. Next to each of the sketches she had written *"First I will..."* as if she had started to think the first one through, rejected it, started again, and so on four times. She also made something different.

Jordan (Comparison Class) produced a Single-Draw series at Easter. He drew one idea, tried to make it and was unsuccessful and drew another idea and tried to make that, equally unsuccessfully and repeated this cycle again and then gave up drawing and made his Easter Egg Holder from all the parts left on his desk (the debriefing transcript in Section 5.3.1c accurately describes his process as observed). It was unique. No other child in the Exploratory or Structured Phase worked this way.
5.6.3 Comparison to 1998-9 Results

Informal observations of the children's drawings in the Structured Phase of the research indicated that there was greater sophistication in drawing technique overall than I had observed in 1996-8. The number of unlabelled Single-Draw and Multi-Draw examples was overall much lower than for Year 2 in the previous studies.

I think this was due to two factors. Firstly, my known interest in investigating Design and Technology, especially drawing, had given the subject a higher profile within the school. Secondly, the heavy stress on literacy which now pervaded Key Stage 1, through the introduction of Literacy Hour, meant that many children wrote a great deal, which was completely uncharacteristic of any of the 1996-9 children, regardless of age. Year 1 children were being taught the techniques of non-fiction genres for presenting information through labelled diagrams and lists and by Year 2 they were able to write instructions. This appears to have had a positive impact on the clarity of their communication of design ideas.

However, my instinctive "feel" of the Focus Class whilst teaching them throughout the Programme had led me to believe that they were performing more like Year 3s of previous experience. When I encountered the Comparison Class for the Assessment Tasks, they felt like Year 2s. This was especially so for Pandy's suitcase (which I had conducted as a whole school task in 1999). The Comparison Class children had similar misunderstandings as I had previously observed in Year 2 : single-sided card cut-outs, cutting up the design sheet, etc. Since I had whole school data on Pandy's Suitcase, I decided that this would be the best task to use for comparison between this cohort and the 1999 children. The comparison could only be done by Drawing Types due to changes in the analysis instrument between Exploratory and Structured Phases. Chart 13 (overleaf) shows the comparison between the 1999 data from the spreadsheet shown in Appendix N: AN1.4 and the data from Appendix O: AO1.3.

As can be seen from the ages of the children, the Structured Phase cohort straddled the Year 2 and Year 3 age ranges from the Exploratory Phase and the cohort size is very different, since for Pandy99 there were two classes of Year 2 and three of Year 3. However, useful comparisons can be made.



I did not find it surprising that the Comparison Class' profile is closer to 1999 Year 3 than to 1999 Year 2. I had been aware that Design and Technology standards had risen across the school, which, as subject co-ordinator, I found pleasing. The overall spread of scores, evenly divided between Single-draw, Multi-draw, Multi-design and Progressives, would seem to be an expected outcome for 7-8 year olds in this school.

For the Focus Class, a very different profile emerges, which does not conform to expectations for Year 2 or Year 3. The predominance of the use of drawing to record design possibilities and to develop those ideas to indicate how they might be made is beyond expectations based on previous observations or the performance of the Comparison Class. This would strongly suggest that the Programme had made a difference to the way that they used drawing to support their designing.

5.6.4 Emergent Themes from the Analysis of Drawing Types

Improvement on 1998-9

The comparison between this cohort's results of the Suitcase task and that of 1999 showed an overall improvement in the sophistication of children's design drawings, as judged by their choice of Drawing Types. This was interesting because, although my instinctive "feel" of the Focus Class as I was teaching was that their performance was more like my expectations of Year 3, I had not perceived that the Comparison Class were more sophisticated designers than the previous cohort. Looking at the comparative chart reveals why: the Focus Class were using drawing in a more sophisticated way even than the Year 3s in 1999.

From a teaching perspective, especially in my role as Design and Technology Co-ordinator, this was heartening. Despite the pressures of the National Literacy and Numeracy Strategies, we had raised standards in Design and Technology. From a professional perspective, I had been concerned that Miss N. might feel that her teaching was being compared with mine. This result demonstrated her professionalism, that despite the pressures of being Mathematics Co-ordinator at the time of introduction of the Numeracy Strategy, she was able to provide her class with high quality learning experiences in a climate in which Foundation Subjects, and Design and Technology in particular, were being squeezed to the edges of school's priorities.

Annotation

The level of annotation of drawings was an important difference between the 1999 cohort and this present one. Neither Miss N. nor I had specifically encouraged the children to write on their drawings and yet they chose to do so. This added complications to the assessment. I had to make a conscious decision about whether to ignore this writing or to include it as part of the whole. There had been so little writing in the 1999 cohort that this had not really been an issue. But where writing had occurred, I had always considered it as evidence of design intentions.

I decided I must continue to consider writing as part of the whole. Firstly, because that the children often used labelling appropriately to clarify their intentions for my benefit or that the choice to write rather than draw was often based on common sense (making a list of materials is more sensible than drawing sheets of different coloured card), and secondly, most adult design drawings include words, for similar reasons to those of the children. I would, however, focus my attention on the graphics and not consider greater quantity of writing as automatically "better" just because it made the design intentions clearer. This was an important decision to make at this stage, prior to looking closely at the Dimensions of Design Drawing, especially Communication of Design Ideas, where readability could influence decisions over content.

The Position of Multi-draw

The Drawing Type which occurred across most Purpose categories was the Multi-draw. I had wondered at one stage whether I should eliminate this Drawing Type altogether. It was less common than Single-draw, Multi-design or Progressive and seemed to occur at the boundaries between Purpose categories. Several drawings were too detailed or well-labelled to be a classic Single-draw, but the added details clarified rather than extended the ideas. A neat row of Easter Egg Holders, all identical except for spots and stripes, hardly seemed to be Multi-design or "Recording Design Possibilities".

However, there were some Multi-draws which began to move ideas towards a design solution. Labels that were crossed out and changed or parts of drawings erased and redrawn differently bore witness to a movement of ideas, albeit slight. Frequently, the child would spend a long time on the first drawing, make quick changes and then go and fetch the materials to make the new idea. Ideas were on a journey but the drawing was performing a secondary role in supporting the final decision-making.

5.7 Dimensions of Design Drawing

The findings reported here are based on the analysis of the children's drawings based on the Dimensions Continua (Table 14) and collated via individual child collation grids (Appendix AN2.6) into whole class spreadsheets (Appendix AO1.4).

In the same way that the Drawing Type analysis was presented from two viewpoints to enable clarity of representation and discussion, so in this section too the discussion has been separation into analysis by task and by Dimension. The analysis by task (Section 5.7.1 "Developing Capability in all Dimensions over Time") uses radial plots (Dimensions Plots) to represent the data analysis, in line with the holistic model of design drawing (Section 5.4.3). This mode of representation is not used, however, as the basis of the cross-task analysis in Section 5.7.2 ("Comparing Capability in each Dimension across Tasks").

The abbreviations for each Dimension used on the charts in this section are as indicated here in Fig. 59, at the ends of the spokes of the Dimensions Wheels:



Fig. 59 : Abbreviations for Reading Dimensions Charts

5.7.1 Comparing Capability in all Dimensions over Time





An enlarged view of the three inner rings is used throughout discussion which follows:



It can be seen from the Dimensions Plots in Chart 15 that (as represented by mean scores), that the understanding of the use of drawing for designing in the Focus Class has expanded across the duration of the Programme, whereas that of the Comparison Class has remained fairly constant except where specific task related instructions (Planning the Look of the Card) or taught techniques immediately prior to the task (Planning Construction at Easter) had enhanced their performance.

Assuming that each Dimension is of equal importance, the more circular the Dimensions Plots, the more balanced the use of drawing to support designing. The Focus Class plot gives the greater sense of growing, developing, balanced understanding. I think this suggests the growth of a more holistic understanding of design drawing than that represented by the Comparison Class plot, for whom there appears little overall pattern in their responses to the tasks, perhaps suggesting a developing facility with techniques without an overall understanding of the purpose of the drawing. The inference could be drawn, therefore, that a clear understanding of the understanding for designing enabled a more balanced development of the understandings inherent in the Dimensions. References are made back Section 5.3 to give context to the quantified analysis.

In Charts 16-18 on the following pages, separating out the Dimensions Plots for each Assessment Task (from the spreadsheet in Appendix O : AO1.4 Sheet 1) provides a view of developing capability that enables cross-task comparisons to be made. Juxtaposing the plots for Pizza and Frosty gives a sense of the starting points of each class. Then the mid-programme product design tasks (Easter, Card and Suitcase) are shown together and discussed and, finally, the Maze (as a problem scenario) is presented separately.

There is no Basis for Making score for the Card as the children did not make it.

Planning the Look of the Product (decorative features and/or logos) was not relevant to the problem scenario, so that this Dimension was not considered in the analysis of the Frosty and Maze tasks.



At the Beginning of the Programme :

Chart 16 suggests that many children in both classes were drawing a simple, stereotypical sketch of an object to be made, relying on shared meanings for its interpretation. They showed some understanding of the task constraints and exploration of possibilities but little reflection on their ideas or changes made as they drew. The product related to the drawing but it could not truly be said that the drawing had formed a plan for making. Little attention was paid to arrangement of Pizza foodstuffs to enhance the look of the product.

Mid-Programme Product Designs :



Easter:

Immediately after the Container / Journey metaphor input, the Focus Class' use of drawing for designing has developed considerably and equally across all Dimensions, as can be seen by the almost circular plot in Chart 17. The Comparison Class used drawing for Planning Construction equally with the Focus Class but this did not translate into the making. This might suggest that, although Comparison Class children knew to how to *record* construction, they were not *planning* construction, whereas Focus Class children were seeing the role of the drawing as a way of recording planning for making.

The Focus Class also showed greater facility in generating creative ideas, which they were beginning to develop towards a design solution (for example, Chicks, Bunnies' ears popping out, etc.), together with an acceptance of the constraints of the task (the egg needed to be supported inside the tube) in contrast to the Comparison Class' simple line drawings and free interpretation the task (for example, discarding the tube and building a tower for the egg to sit on top). The ability to reason creatively within the "rules of the game" emerges as a Key Theme across all strands of the Structured Phase analysis.

The clear differences that emerge at this point in the Programme indicate the immediate effect of the teaching input to the Focus Class through the Container / Journey metaphor. The Focus Class demonstrated a greater use of drawing to support their designing in all Dimensions apart from Planning Construction. One surprising difference was the Comparison Class children's lack of consideration of the Look of the Product. My assumption would have been that less capable designers would focus on this aspect of the task but many children produced single annotated line drawings in pencil which they ignored once they began making.

<u>Card</u>

The structured nature of the task enabled many Comparison Class children to give their best performance across more Dimensions than any other task. However, except for the decorative aspects of the task, they were outperformed by their Focus Class peers. Planning the Look of the Product was the most obvious feature of the task and what they were initially told to do. Hence both classes achieved well. However, there were important differences on other Dimensions.

The Focus Class generated a range of creative design ideas (on which they reflected and made changes), containing a surprise element, tailored to the preferences of their client, and conveying a sense of how the card would be made (materials and construction details). Comparison Class children tended to produce a simple drawing which represented a stereotypical response, less closely relate to the preferences of their client or the "surprise"

theme (any card, rather than a surprise card for a specific person), yet recorded in my Log Book that the Comparison Class showed much keener interest in the examples they were shown. This lack of focus on specific problem-solving also impacted on their achievement in communicating ideas, planning construction and evaluating.

The similarity in performance of the Focus Class here to that at Easter suggests that they could transfer their learning to a new situation and thus were utilising domain-general knowledge. For the average Comparison Class child this was not so: the two profiles are quite different, suggesting a lower level of domain-general knowledge and that they were engaged in context-specific problem-solving.

Suitcase

The parameters of this task were much tighter. If the Focus Class found it harder to Address the Constraints of this task, the Comparison Class found it even more difficult. The recording of plans on paper was dislocated from a real understanding of task constraints and was frequently ignored in the making of a product that did not satisfy the task criteria (hold this mac and be able to be held by Pandy).

At this stage the Focus Class children had a working definition of the purpose of the drawing that could be summarised as: it is a means of recording ideas about what they *wanted to make*, what is it going to look like when it is finished and having several attempts at improving and refining ideas. The Comparison Class, viewed the role of the drawing as recording an idea (or perhaps several ideas) of what *might be made*. The creative thought, consideration of decorative features, construction techniques and evaluative decision-making occurred once they were engaged with the materials. This lack of pre-planning ability is typical of children of this age and these comments are not made in criticism of the Comparison Class, rather, it highlights the mature way in which the Focus Class were using the design drawing to plan and refine their ideas before their engagement with the materials.

Post-Programme Problem Scenario: the Maze :

The trend continued into the final Assessment Task (as shown in Chart 18 overleaf): the Focus Class generating design ideas related to solving the problem whereas the Comparison Class were simply drawing mazes, as evidenced the higher scores for Generating and Developing Ideas coupled to Addressing the Task Constraints. In the Focus Class drawings, there was an attempt at conveying ideas about an object to be made (even if as simple as the word-label "string") that was absent from the work of their Comparison Class peer.



The level of evaluation whilst drawing and subsequent relationship of the drawing to the making shown in Chart 18 is also indicative of this difference between the two classes. Many Focus Class children thought about the product whilst drawing and made changes at the planning stage, so that there was a fairly close match between the plan and the product. The Comparison Class children, however, not only generated less ideas but were relating this to the reality of construction. This accords with my in-context observations (Section 5.3.1f) and the results of analysis on the Purpose of Drawing Continuum (Section 5.5).

Comparing the Dimensions Plots for Frosty and Maze supports my hypothesis that domain-general learning had taken place within the Focus Class (suggested also in relation to the Card), who demonstrated their ability to transfer learning to less familiar territory and to still outperform the Comparison Class. In contrast, in all Dimensions, the Comparison Class' use of drawing to support designing a problem solution is at a similar level to that of both classes for Frosty, near the start of the Programme.

5.7.2 Comparing Capability in each Dimension across Tasks

The results of this analysis are displayed overleaf as bar graphs in Chart 19, as this was felt to be a more appropriate display medium for cross-task data. As in the previous section, the unit of comparison is each class' mean score in each Dimension. Empty positions are left on the graphs when a particular Dimension was not relevant to a specific task.



5.7.2a Comparing Mean Scores for each Dimension

Comments on differences between tasks are to be found in Sections 5.7b & c.

Generating and Developing Ideas

The Focus Class appear to have learnt to use drawing more effectively to generate and develop ideas between Frosty and Easter, through being taught the Container / Journey metaphor. The slight slippage in subsequent tasks could be due to the fall-back position as levels of understanding take over from reproduction of recently learnt techniques. The Comparison Class do not reach an equivalent level of capability on any task and their learning profile has increased modestly across the year.

Exploring the Possibilities of the Task

Comparing the two classes' exploration of task potentials shows the Focus Class as more capable of making a novel response to the task. My observations of the Comparison Class children freely interpreting the task (for example, Zara playing with the parameters of the Maze task, Section 5.3.1f) and assumption that they were being more creative was not borne out by the analysis of the drawings. The Comparison Class children who produced a divergent response to the task were more often off-task in their making than creatively exploring possibilities whilst drawing.

Addressing the Constraints of the Task

This was coupled to the exploration of task potential. As noted in Section 5.3.2, the Comparison Class did not appear to understand that satisfying the task criteria were essential elements of a successful design solution (for example, discarding the card tube at Easter). The Focus Class were much more aware of the need to solve the problem as set (and looked for creative solutions to it).

Planning the Look of the Product

The differences between the two classes at Easter and for the Suitcase came as a surprise, as I expected less competent designers to colour their drawings rather than recording material or construction details. However, in seeing the role of drawing as clarifying the task rather than develop design ideas, the Comparison Class were not thinking about finishes, whereas Focus Class children planning a solution were imaging a real product, including its colour or logo. The Card task produced almost identical results because the children were told to draw a picture for the front of the card.

Communication of Design Ideas

The peak in both classes' mean scores for Easter and Card tasks, probably has more to do with the emphasis on clarity of communication in view of the impending SATs tests than with teaching in Design and Technology. However, the Focus Class maintained a greater clarity in communication in the later task than did the Comparison Class.

Planning Construction

There was a similar development of use of drawing to plan construction in both classes, except for the Focus Class greater use of drawing in this Dimension for the Card. Having the tube on the table helped both classes to address construction issues at Easter.

Evaluating whilst Drawing

The Comparison Class' mean score gradually improves across tasks indicating natural maturation, perhaps. The Focus Class' profile shows a change at Easter, which could represent an understanding of the genre of design drawing: that several ideas recorded on one sheet is not just allowable but desirable.

Basis for Making the Product

Throughout the Programme, the Focus Class appeared to have a better understanding of the drawing as planning for making. Although the Comparison Class made something that related to their drawing, it could not generally be regarded as a plan for action. In my view, this Dimension most raises the question of the children's perception of the purpose of the drawing.

5.7.2b Problems and Products

Chart 20 overleaf shows the comparison between the children's mean scores across all Dimensions for the Problem Scenarios (Frosty and Maze) and the Product Designs (Pizza and Suitcase) in the same way as the Purpose scores for these tasks were compared in Section 5.5.2, Chart 8. The mean scores across all dimensions were calculated from the individual Collation Grids (Appendix O: AO2). AO1.4 Sheet 2 shows the data from which Chart 20 was created.

Similar conclusions can be drawn from Chart 20 as from Chart 8:

 most progression appears to be made by the Focus Class in the product design tasks, least by the Comparison Class in the Problem Scenarios.

• many Comparison Class children appear to have made little progress in their ability to use drawing to support their Product Designs, whereas considerable progress has been made by most of the Focus Class.

the numbers of Comparison Class who show similar or less use of drawing to support their thinking about the Maze than they did for Frosty shows that there was relatively little progress in understanding of how to use drawing to support thinking about design problems. The Focus Class seem to be able to transfer skills in design drawing to a different context.



As in Section 5.5.2, these figures were collated into bar graphs (Chart 21) to show class profiles for development over time. Unlike the scale on the x-axis of Chart 8, which represented the discrete bands on the Purpose Continuum, the x-axis here represents an analogue scale. Divisions of 0.5 have been used in order to give greater clarity of representation of the data.



The comments made in with regard to Chart 9 in Section 5.5.2 are also pertinent here. In both types of task, the Focus Class display a greater ability to use drawing to support design thinking, regardless of familiarity of task type.

5.7.2c Differences in Performance on Dimensions Continua

Chart 19 revealed that on many of the Dimensions, the Focus Class appear to make a leap in understanding at Easter, that is maintained into subsequent tasks.

To examine the extent to which this was different from the growth in the Comparison Class' use of drawing for designing, Chart 22 (overleaf) was constructed by calculating the difference in mean score for each class for each Dimension (Focus Class' mean minus Comparison Class' mean). The graphs within Cart 22 represents *differences* between the children's use of drawing to support designing, rather than better or poorer performance. There was little difference between the classes at the start of the Programme.



The immediate effect of the Container / Journey input to the Focus Class can be seen in the Easter chart. This would appear to suggest that the understanding of the purpose of design drawing came as a result of the specific teaching input of the Container / Journey metaphor. This immediate effect is one of the emergent themes from the dimensions analysis (Section 5.7.4). The Focus Class had not received specific teaching on Planning Construction whilst drawing, as my Exploratory Phase observations had revealed that children of this age find it difficult to do this. Therefore, it was likely that there would be little difference between the two classes on this Dimension.

The greatest difference at Easter is in the use of drawing for Planning the Look of the Product. This reflects the choice of many Focus Class children to make the tube into something (e.g. Easter Bunny), which they recorded as a whole potential product, rather than the Comparison Class' response to use the drawing to plan how to solve the construction problem and leave the decision about how to decorate the outside of the tube until engagement with materials. The same trend can be seen for the Suitcase.

Both Card and Suitcase tasks show a less pronounced difference between the two classes. I think that the Easter scores represent immediate application of recent teaching and that the Card and Suitcase results represent longer term changes in understanding. It would be expected that some children would not be as capable as others to apply their learning to different situations. However, the ability of the Focus Class children to maintain their greater understanding across a range of different tasks would perhaps indicate that real learning had taken place. The great difference between the mean scores for Generating and Developing Ideas for the more structured Card task might indicate that the printed design sheet led the Comparison Class children into thinking in terms of "one right answer."

The immediate and sustained difference between the two classes immediately after the Teaching Input to the Focus Class enshrined in the Container / Journey metaphor would suggest that this teaching had an immediate and lasting effect on their perception on how drawing might be used to support design thinking.

Page 271

5.7.3 Relationship between Purpose Continuum and Dimensions

The Purpose Continuum and Dimensions Wheel form two overlying layers of a holistic view of children's design capability (Section 5.4.3). Superimposing a Purpose Plot on the Dimensions Plots enables analysis of the relationship between the two in the children's responses, both within and across tasks. The charts were created from Appendix O : AO1 Sheet 1 & AO1.4 Sheet 1.

5.7.3a Within Tasks

At the Beginning of the Programme both classes started from a low level of understanding of the purpose of design drawing (Chart 23), reflected in their limited use of drawing on any of the Dimensions continua:





Mid-Programme Product Designs :

The sudden expansion in both the Focus Class' Purpose and Dimensions Plots following the Container / Journey input (Easter) shows a strong connection between their understanding of the purpose of the drawing as a design tool and the way that they used it, across all

Dimensions. The Suitcase plot confirms that real understanding of the purpose of drawing for designing had been achieved and maintained, this was most clearly expressed in their generating, developing and evaluating of ideas about the overall look of the product.

In contrast, the Comparison Class' Dimensions Plots swing around within and across the Purposes Plot, suggesting that these children were not developing such a holistic understanding of the purpose of drawing and were responding to cues within in the task on each occasion. Hence for the Card task, which guided them through the process and required them to record something that related to each of the Dimensions, their use of drawing relates closely to their understanding of the purpose in all Dimensions. However, without such support and left to their own devices with blank paper at Easter and for the Suitcase, they demonstrated a much weaker relationship between purpose and use of drawing for designing.

At Easter, clarifying and recording possibilities is most closely related to generating, developing and communicating ideas about planning construction, without considering the final external appearance of their product or confronting the constraints and possibilities of the task potential Neither do they appear to view the purpose of the drawing as planning the making of a product. This accords with the observation that many Comparison Class children viewed the purpose of the drawing as clarification rather than solution of the problem. These trends were also true for the Suitcase. However, the children who used the design sheet as part of the product demonstrated minimal understanding of the purpose of the drawing as related to communicating design ideas, bringing down the mean score on this Dimension.

Post-Programme Problem Scenario: the Maze :



In this problem scenario task, there is less difference in the relationship between the Purpose and Dimensions Plots shown by each class. The Focus Class' greater understanding of the purpose of design drawing is reflected in their greater use of drawing across each Dimension, especially Generating and Developing Design Ideas.

5.7.3b Across Tasks

Points of near contact between the Purpose Plot and the Dimensions Plots:

Most frequently, one of those points is Generating and Developing Ideas. Thus, even at this young age, children can understand the purpose of the design drawing as a means of generating and developing ideas about the product they have been asked to design or the problem they have been asked to solve. That Evaluating whilst Drawing is frequently another close contact point suggests that at this age children can be reflective about their work and seek ways to improve and refine their ideas.

Points of greatest distance between the Purpose Plot and the Dimensions Plots:

These occurred in the Comparison Class' plots. Apart from the Suitcase task, the Focus Class exhibited a close relationship between their understanding of the purpose of design drawing and the way in which they used it in all Dimensions.

Thus it would seem that having a secure understanding of the purpose of drawing for designing enabled the Focus Class children to address each of the dimensions of design drawing with greater success than did the Comparison Class. Smith (2001) asked two questions, which I believe my research has, at least partly, been able to answer:

"What age should learning sketching techniques be started and to what depth? Do pupils understand that one reason for sketching when designing is to assist in the generation of more ideas through the ambiguity of the sketches and the juxtaposition of ideas?"

(pp.8-9)

My answer to Smith's questions would be that the understanding needs to come before the techniques. The Focus Class' understanding of the purpose of drawing for designing appears to have had an immediate effect. Unless children have a secure understanding of the purpose of the drawing, teaching of techniques is futile. The Comparison Class could indicate how to make a product that answered the task, often as a labelled diagram, but their lack of understanding of the purpose of the purpose of drawing to develop design ideas meant that they did not relate this drawing to

the development of a solution to a design problem. In answer to Smith's second question, I would say that even such young children as those in my study can and do understand the purpose of drawing as a means of generating design ideas, but that they need to be taught that this is the *purpose* of using drawing in a design context. The beginnings of reflective interaction with their drawings began to show in tasks in which the Focus Class showed the highest degree of balance across all Dimensions, coupled to (and I believe driven by) an understanding of the purpose of the drawing as planning to make.

5.7.4 Emergent Themes from the Dimensions Continua

Throughout the Dimensions analysis, it has been clear that the Focus Class use drawing much more effectively across all Dimensions from Easter onwards. Chart 26 collates these differences into overall mean scores across Easter, Card, Suitcase and Maze tasks (Section 5.7.1a demonstrated the close parity between the two classes responses at the beginning of the Programme).



Over time, therefore, regardless of specific individual task effects, the Focus Class had achieved greater long-term learning about the use of drawing for designing after being taught the nature of design drawing through the Container / Journey metaphor.

However, within the analysis of the Dimensions, themes and trends emerged that mirrored and enhanced observations from other layers of analysis from the notes made on the Teaching input to the Focus Class (Section 4.4) through to the analysis of Drawing Types in Section 5.6.4. Reflections upon these emerging themes led to new insights on the way in which the Dimensions relate to the Container / Journey metaphor.

Impact of Understanding Purpose on Performance in Dimensions

The immediate effect of the Focus Class' understanding the purpose of drawing for designing is best demonstrated by the superimposition of the mean of the Purpose Plots for the mid-Programme product design tasks onto their Dimensions Plots (Chart 27) :



The Focus Class plot suggests understanding of *purpose* expressed through the *dimensions*, whereas the Comparison Class plots seem to suggest a sense of struggling to come to terms with drawing their design ideas and unsure of the purpose of the activity. The integrity of the Focus Class Dimensions Plots suggest the development of domain-general knowledge which is being transferred across tasks, which cannot be said of the Comparison Class, whose responses appear context-specific. The close matching of the holistic (Purpose) and the discrete (Dimensions Plots) on the Focus' Class' chart would suggest that understanding the purpose of the design drawing is that domain-general knowledge and that the Comparison Class have not yet reached a level of understanding of the purpose of design drawing for this effect to occur.

That the effect on the Focus Class' performance was immediate upon the Container / Journey teaching input would suggest a strong relationship between the two. That it was not caused by maturation or deduction from practice is suggested by the Comparison Class not reaching the same level of understanding or use of drawing for designing across the duration of the

Programme. What had happened within the Focus Class was not simply accelerating a natural process but imparting knowledge that they would not have deduced for themselves within the timespan of the Programme.

Task Potential

Throughout the Dimensions analysis it was clear that the ability to Address the Constraints of the Task and to Explore the Possibility of the Task need to be in balance in order to create a successful design. This was identified as an Emergent Theme during observations contributing towards the qualitative analysis (Section 5.3.2).

Both these Dimensions relate to notions of creativity in Design and Technology contexts and to Donaldson's (1992) "this problem and this problem only". I began to call this "awareness of task potential": looking for creative ways to solve the task in hand or to play with ideas within the rules of the game. The Compression Class' divergence from the task constraints showed low awareness of task potential. For example: making the tube into an Easter Bunny shows high awareness of task potential; a unsuccessful Maze did not aid Theseus in his escape.

Problems and products

The difference between the way drawing is used in problem scenarios and product design was realised in the course of Dimensions analysis of the Maze task, when it became apparent that the children had not recorded any details relating to Planning the Look of the Product and very little in terms of Planning Construction. It would seem that their perception of the planned object as a "model" rather than as a "product" elicited a different use of drawing for designing.

This realisation led to the analysis of each task type separately and enabled the perception of trends within the product design tasks (for example, that Planning the Look of the Product was an important difference between the two classes' use of drawing). There were common trends across task types, however, especially in the dual aspects of Addressing the Task Constraints and Exploring the Possibilities of the Task.

5.8 The Techniques Ticksheet

The re-definition of the analysis instrument in terms of the Dimensions Wheel and the consequent re-working of the Grid's Descriptor Cells into Dimensions Continua meant that much of the information contained in the Techniques Ticksheet was subsumed into the Continua. However, as indicated in Section 5.4.3b, the external evidence for the children's understanding of the Dimensions of Design Drawing was frequently contained in the techniques that they used.

The way that the quantified analysis instrument evolved over time made the to-ing and fro-ing of specific bits of information inevitable. One strength of this was that the four elements of the analysis instrument (the Purpose and Dimensions Continua, the Drawing Types and the Ticksheet) were inextricably interwoven and consequently supported each other rather than clashed. I tried to be vigilant in ensuring that the same information was not recorded twice. The causes were frequently due to closer correlations in the data than I had anticipated The result was frequently a re-thinking of definitions, categories and meanings.

The full Techniques Ticksheet list can be found in Table 15 in Section 5.4.3b(ii).

5.8.1 The Role of the Techniques Ticksheet

Although far less time was spent re-working and moderating and re-defining the Ticksheet than was spent on any of the other layers of the Analysis Instrument, this does not imply that the role of the Ticksheet was just as supporting cast.

It evolved into contributing clarification and definition to the Dimensions Continua and supporting information to the Purposes Continuum, yet it also had its own contribution to make about children's designing. Through teasing out the techniques the children used, I was able to get a much firmer grasp on what I meant by my more global categories. For example, the combination of addressing task requirements and client's needs into a single Dimension as Addressing the Constraints of the Task was decided whilst analysing the drawings using the Ticksheet, as it became clear that these were inter-related.

The Ticksheet categories were refined over the course of time, in line with the development of the rest of the quantified analysis instrument. However, from the start its structure had a hierarchy that implied progress, which enabled the information recorded in the Ticksheet to relate directly to the Dimensions Continua. This was really useful for determining the placement on the Continua of work for which I felt uncertain.

For example, Table 17:

PLANNING CONSTRUCTION				
indicates materials by list				
indicates materials by labelling drawing				
materials indicated are suitable				
materials indicated are available				
indicates cuts, folds &/or fixings				
indicates measurements				
indicates equipment needed				
indicates parts to be assembled				
can be made in time available (roughly)				
technically realistic (child can make it)				

Table 17 ; Ticksheet Example to show Progression

5.8.2 Information from the Techniques Ticksheet

There were two aspects to "progress" which I hoped that my Focus Class would acquire in terms of facility and confidence with using drawing to support design thinking. As well as using the drawing to record and develop their design ideas, I also wanted them to be able to choose the technique which best suited their purpose. I was hoping, therefore, that one thing which would emerge from the Ticksheet analysis would be that the richness and diversity of techniques used by the children, and that the range of such techniques would prove to be considerably greater amongst the Focus Class children.

The reporting of the Ticksheet findings in Section 5.8.2, which follows, is designed to illustrate that richness. The subheadings under which this is discussed relate the Ticksheet to the Dimensions.

5.8.2a Generating and Developing Design Ideas

The ratio of single to several ideas recorded by the children seemed worth examining, regardless of how developed the ideas were. This was not the same as number of drawn items on the paper. Several iterations of the same idea counted as one idea. The 40% - 60% band is indicated on Chart 28 to add emphasis to the difference in the two classes in this respect:



In the Focus Class, almost equal numbers of children (and by no means the same children each time) took either option for each task, suggesting perhaps that they felt comfortable using drawing to record that which they felt to be personally useful.

The Comparison Class show a strong preference for single ideas for product designs but several ideas for problem scenarios. My Log Book records that they were calling out "I know what I'm doing" almost before I had finished explaining the Easter task, which would corroborate the high incidence of single drawings for this task. This impulsive recording of the first idea may, therefore, be a factor in their continued recording of single ideas for the Card and Suitcase. The use of drawing for clarification for the Maze task has expressed itself in several drawings, as they had more than one attempt at defining a maze to themselves.

The development these ideas was then considered. The two kinds of response (Single or Several ideas) are considered separately. The range of responses for the whole class is, therefore, split across the two charts, e.g., for the Focus Class response to the Suitcase task, half of the single idea children produced a simple single drawing, but 65% of the class produced several ideas and the same proportion of these developed one towards making as produced simple singles.

As can be seen from Chart 29 overleaf, the development of design ideas from a single starting point is much higher than for the production of several ideas, to the extent that this seems to be an either/or option on how to use drawing to support designing. In terms of Drawing Type, the developed single idea would include all the Multi-Draws and Progressives. The lower incidence of development across several ideas are the Multi-designs. It would appear that children at this age tend either to record one idea which they develop towards making or record several

options from which they choose to make one. The incidence of "draw several and develop one / several" is very low and the "several" is frequently two: a first idea is adapted then discarded and a second idea developed.



The continued higher incidence of simple single drawings amongst Comparison Class children accords with Purpose Continuum analysis that many were not using drawing to develop a design solution but to define the problem to themselves. The Focus Class moved to the recording of several ideas for the Suitcase. Perhaps they felt confident in their ability to make the product and so were using drawing to place-mark a range of ideas as possible solutions.

5.8.2b Satisfying Constraints vs. Exploring Possibilities

The original Ticksheet was tautologeous and the distinction into "task requirements" and "client needs" unnecessary, since there was negligible difference between the results on all tasks, leading to the definition of one of the Dimensions as Addressing Task Constraints, which combined both client and task requirements. The original criteria list on the Ticklist was collated into the following shortlist (Table 18):

as required by task or client/user				
using drawing to model the product specified				
aware of possibilities				
aware of constraints				
considering construction of product				
product satisfies task / client criteria				

Table 18 : Task Constraints (a)

In tandem with this change, I realised that *awareness of possibilities* was essentially a measure of creative response. Children who produced a single stereotypical response scored low and those who produced a range of interesting and different ideas scored high on the Dimension of Exploring the Possibilities of the Task. Thus, a child who produced a chicken laying eggs into the tube was thinking of a more creative solution than those who drew six slightly different patterns for the outside of the tube. Exploring the Possibilities of the Task became one of the Dimensions of Design Drawing. These considerations led to a clarification of my task criteria (Table 19) to ensure equivalence of application to inform decisions regarding placement on the continuum for the Addressing Task Constraints Dimension:

Task	Using drawing to model the product specified	Aware of task constraints	Product satisfies task & user requirements
Frosty	A drawing of something to be made; No snowflakes, clouds or fish in the lake.	Drawn as if made from materials provided.	A model of a viable solution to the problem
Easter	The tube must be included in the drawing.	Drawing indicates how egg will be held inside tube.	The egg is held firmly inside the tube
Card	There must be a drawing of a card, which must match the likes of the client.	There must be a surprise element to the design	N/A
Suitcase	More than just a picture of a suitcase: evidence of considering range of travel bags, or thinking about size, or how to make it.	Evidence of grappling with both construction issues and size.	Suitcase is: big enough to hold mac; small enough to be carried by Panda; handle fits over paw.
Maze	Must have top-down view of the maze, indicating route into & out from Minotaur	No irrelevant detail, not too complicated to make: i.e. constructionally viable.	Model is 3D; must have internal walls.

Table 19 : Task Constraints (b)

This greater clarity gave awareness of finer details of children's work that I might otherwise have missed. For example, so many children had provided a "way in" and a "way out" with a string going between the two, passing the Minotaur on the way, that I began to wonder if I had accidentally implied this in my task introduction.

5.8.2c Communication Techniques

The surface features of the drawings (annotation, level of detail, decoration, recording of materials and construction) provided the information for judgements about the child's placement on the Communication of Design Ideas Continuum. The level of annotation was much higher than in the drawings analysed in the Exploratory Phase but I was careful not to simply give annotated drawings higher status than graphics only, simply because I could understand their intentions more quickly. The level of detail and decoration might indicate how clearly the children were imaging their solution and the recording of materials and construction details, although muddled, might indicate their thinking through the practical problems of making their idea.

5.8.2c (i) Annotation of Drawings

Prior to Easter, few children annotated their drawings, in line with all Year 2 and many Year 3 children in the Exploratory Phase studies, which might suggest that the study and production of non-fiction texts in Literacy Hour was having an improving effect on recording techniques in other areas of the curriculum. This observation is re-inforced, perhaps, by the highest level of annotation used for the Card, which was conducted immediately after the completion of the SATs tests, suggesting that the combined effects of the National Literacy Strategy and the run-up to SATs testing was increasing the use of writing as a recording medium across all areas of the curriculum.

Rogers & Stables (2001) reported similar mutually enhancing effects within the "Enriching Literacy through Design and Technology" project. I seem to have parallel findings here, despite focusing on drawing and not aiming to improve literacy. Mantell (1999) recommends the introduction of "potential designing techniques" from other areas of the curriculum (mapping, listing etc.) since children appear to be able apply such techniques from other curriculum areas, even without specific teaching.

A range of annotation was in evidence, as can be seen from Chart 30 overleaf, from single word or phrase used as title for the drawing ("A Suitcase for Pandy") to a full account of the

task in hand. Since many children used more than one annotation techniques, the results are shown as stacked percentages which allows for totals exceeding 100%:



I was relieved to find similar levels of graphics-only recording, since this meant I had probably not valued heavily annotated (and, therefore, perhaps easier to interpret) drawings above graphics-only in my Purpose and Dimensions analyses. This is especially apposite at Easter, where the percentage of graphics-only Focus Class drawings was higher than for the Comparison Class and yet it was in this task that they suddenly demonstrated a growth in understanding on all Dimensions.

I attempted to look at all drawings holistically and see annotation as part of the child's communication, seeing it as a proxy Multi-draw where words repeated drawn information and give credit where words were used to enhance the information given (e.g. specifying colour in words rather than colouring in). I felt that this was fair. I had not told the children that they must draw (rather than write) everything and there were many occasions when words were the most sensible communication option, especially with regard to recording materials.

One child, Peter (a very bright boy in the Comparison Class) did not draw for either Easter or Card but defined the task in words. Despite attempts to be as fair to him as possible and look for features of design skills within his writing, his inappropriate choice of medium meant that there was little and he is among the lowest achievers of the cohort. However, his use of writing for task clarification gave insight into the design intentions of children whose drawings were a static recording of the task. His case also highlighted the limitations of words and the importance of drawing for imaging, manipulating and adapting possible design solutions.

5.8.2c(ii) Level of Detail

Details such as different viewpoints, expansions to show small details, cut away diagrams and indicating how parts would fit together were features of product designs but absent from problem scenarios (Chart 31). This further re-inforces the view that children saw a different use for drawing for each task type.



The differences appear as strongly between tasks as between classes, indicating that all the children used a level of detail which they felt was appropriate to the task. However, it can immediately be seen that the range of techniques employed by the Focus Class is much greater, despite not having been specifically taught to do any of these. Despite the Card task being strongly structured towards recording different viewpoints (inside as well as the front of the card), I was surprised how few Comparison Class children did so. Prior to analysis, I deliberated whether inside and outside counted as "different viewpoints" since I assumed nearly all children would do both.

The Suitcase task prompted a range of ideas rather than detailed working out of how it would fit together. Perhaps they were aware that they would be pretending these parts when they made the suitcase and so felt little need to be specific about them whilst drawing. The fantasy element of the task would encourage children to record a range of possibilities rather than develop ideas towards making. Perhaps the greater clarity of design ideas at Easter and the Card was attributable to the children believing they were planning a real product for a real client and so they were being real designers, whereas pretending that Pandy was going on holiday encouraged them to role-play at designing.

5.8.2c(iii) Patterns, motifs and logos

The observation that such details (even colour) were missing from the Frosty and Maze tasks informed the realisation that problem scenarios were treated differently to product designs and hence the non-assessment of Frosty and Maze for Planning the Look of the Product in the Dimensions analysis. Only the product design tasks are considered here, therefore, and, since all children except Peter drew a picture of the front of their Card, only Easter and Suitcase are compared for recording of decorative features.



Chart 32 illustrates the Focus Class' increasing ability to image and record several design possibilities. This was re-assuring at a time when I was still thinking in terms of Progressive automatically being better than Multi-Design and considered such aesthetic features as colour or pattern as peripheral add-ons. This analysis enabled me to realise the role of Planning the Look of the Product in conceptualising a whole possible solution to task requirements.

5.8.2c(iv) Recording materials

Choice of materials available :

Pizza : card, coloured paper and small items such as seeds, matchsticks etc.
Frosty : newspaper (to roll), string, wide range of recycled materials
Easter : the tube, coloured cord and wide range of recycled materials
Card : none provided
Suitcase: Card, cord, treasury tags, paper clips, paper fasteners etc.
Maze : Thick and thin card, string, small sticks, range of fastenings, etc.

These were shown to the children during the introduction to the task and laid out on side table (except the tube for Easter, placed on work tables).



The Frosty task had high potential for recording materials, as a large range was available, and so the sudden increase in doing so, shown in Chart 33, must be attributable to teaching during the Spring term. This could also be a Literacy Hour effect, as most indications of materials is by writing, not drawing. The children were not drawing their idea as if made from the materials provided, suggesting that this was not the image they had in their heads but that consideration of what the object would be made from came with the recording the inner image as a drawing and so was added in writing.

The high incidence of recording materials (especially by labelling) among the Comparison Class children at Easter correlates with their high score on the Dimensions Continuum for Planning Construction. Despite specifically being told to record materials in Box 6 on the Card
design sheet, about half the cohort did not do so. The low incidence of recording materials for Suitcase in both classes may be due to the fantasy elements inherent in the task (as observed with regard to level of detail, Section 5.8.2c(ii)). For the Maze, the incidence of recording materials is much higher in the Focus Class, reflecting their overall greater sophistication in using drawing for designing; some children even used legends:



5.8.2d Planning Construction

The Ticksheet list was not the instrument to capture the holistic "intent to make" sense that I was looking for in order to place children's drawings on the "Planning Construction" Dimension, but usefully recorded specifics for product design tasks (but not problem scenarios).



Despite my intuitive feel that the Comparison Class tended to be drawing a labelled diagram of how to make a product, this more detailed analysis reveals that it was the Focus Class who did so for both Easter and the Card. Consideration of the designs reveals that the Focus Class had more complicated ideas which required parts to be detailed, along with their construction (the Easter Bunnies had ears or the Chicks had wings and beaks, for example). For the Suitcase, the Focus Class' frequently recording measurements on a simple diagram, indicating that they were grappling with the realities of the size of the product, rather than production techniques. No measurements were taken by the Comparison Class.

5.8.2e Evaluating Ideas whilst Drawing

This analysis examines just those drawings for which there was more than one item on the paper. For the Card, both Boxes 5 & 6 were considered but Box 7 (for reflection once finished) was not. My focus was on evaluation *whilst* drawing rather than on completion of the work.

Evaluation whilst Drawing closely reflects the differences between the Drawing Type categories: *improving the drawing* is essentially Multi-Draw, *producing a range of ideas and then choosing* is essentially Multi-Design, *improving ideas* (whether starting from an single initial idea or a range) is Progressive and *combining ideas* is Interactive. This in tern reflects my perception of the centrality of evaluation within designing rather than seeing it as something that happens at the end. This is the view of Kimbell et al. (1991) as expressed in the model cited in Section 2.4.1, as it is also central to Rogers and Clare's (1994) model (cited in the same section).

Since the aim of this analysis was to examine the results of evaluative action, the simple single drawings with no evidence of evaluation are not included on Chart 35.



This analysis confirmed my belief in the close relationship between Generating & Developing Ideas and Evaluating Whilst Drawing. The adaptations, additions and multiple drawings on the paper are the result of an evaluative response to the first drawing. It may be that several other

ideas tumble out of their minds onto the paper (Multi-Design) or that they want to make improvements to the one idea which seems to have answered the problem (Multi-draw, Progressive).

Different kinds of tasks seem to promote different responses: problem scenarios seem to promote the recording of a range of possibilities whereas product designs tend to promote the development of one good idea (except for Focus Class Suitcases). Choices were often indicated by tick or crosses against sketches of ideas (or occasionally the word "best") but the reasons were rarely recorded. In debriefing interviews, children could often recount the whole of their design path and justify choices and changes but these were not recorded on the paper.

The things that were changed across drawings were then examined. The drawings that fell within the category *"improving drawing without developing ideas"* from the first analysis (above) have been excluded. Chart 36 shows a much higher incidence of changes made by the Focus Class overall, and this is most frequently related to task specification, providing evidence for the Focus Class being more likely than the Comparison Class children to be reasoning within the parameters of the task potential.



The category "appearance of the product" denotes re-drawing the same idea but simply changing the decoration or colour. The low level of even these minimal changes amongst the Comparison Class children show that evaluation of design ideas were not really occurring whilst drawing.

6.6.2f Relating Drawing to Making the Product

These data were obtained by comparison of photographs of finished products with the drawing and also through interview data. My first reaction to this analysis was that it overwhelmingly shows that children at this age can and do make what they draw, despite the folk myth to the contrary.

I was also aware of the possibility of judging more developed drawings more harshly than simple ones. There would be no parity of judgement if an unlabelled line drawing with a product of the same proportions were to be deemed "same" whilst a development of design ideas including materials and construction techniques was judged to be "adapted" if different materials were used. Similar decisions need to be reached about the point at which adaptation became *different*.

The reasons for changes between drawing and product were queried and noted as I took photographs of finished products. If the reasons for change were justified within the remit of the child's own planning (e.g. materials were no longer available) then this was taken as adapted. If reasons were external to their remit (*"I just did this one"*, *"We worked together"*; *"Natasha had that idea"*) then such changes counted as *different*.



The most common adaptations to drawings were proportions, size or colour. Considerations of size, especially where this was important to solving the task, were more frequently dealt with at the making stage, especially of the Suitcase. Children who produced a generic picture of "a suitcase" rather than a detailed drawing of "this suitcase" were more likely to produce an object

of different proportions since they were not, at drawing stage, imaging a specific solution to the problem.

The role of the drawing for the Maze task was demonstrated in its relationship to the object made. The ongoing flow of ideas from planning into making within the Focus Class shows in the number of adaptations of recorded ideas. They have apparently continued to design a solution but fetched the materials at the point at which they were ready to model their ideas in card. The Comparison Class abandoned their drawn ideas because they were only clarifications of the problem and not solutions it.

5.8.3 Emergent Themes from the Techniques Ticksheet

The Ticksheet has provided supporting evidence for many of the themes identified from other layers of the analysis. These will not be repeated here, although reference will be made to Ticksheet evidence in the discussion of these themes within Section 6.

Drawing and Writing for Designing

Many of the techniques used by the children for recording design ideas involved writing and so "Annotation of Drawings" was chosen as the first sub-heading of Section 5.8.2 as my list of features for Communicating Ideas specifically examined the relative use of text and graphics. Ideas about colour, materials, size and construction could all be in writing. These were frequently sensible choices, showing that children could use both drawing and writing interactively as tools to express their ideas. Drawing seemed to be used for developing whole images or ideas. Annotation was added afterwards. Peter was the only child who began by writing. Everyone else drew first and then wrote. The purpose of the writing was initially an aide memoire (labelling the parts, perhaps, where the drawing was unclear) and then further thoughts about their idea (for example, sentences below the drawing) were added upon reflection.

Reality / fantasy

Examining the level of detail on the different drawings led me to conclude that choices about use of drawing were being made partly on the basis of the perception of the reality of task. This would accord with my observation of the Focus Class making very clear drawings throughout the kit-making task where their client was a younger child and then appearing to perform much more poorly on the Maze. Perhaps knowing that this was a school exercise and that we were all

role-playing the status of the model as "helping Theseus to escape" meant that they invested the whole activity with less importance.

Generation, Exploration and Evaluation

Devising the Ticksheet raised issues of the inter-relationship of these three key skills in designing, which impacted on the development of other parts of the analysis instrument.

The reflective journey into the relationship between generation, exploration and evaluation began when my blind-markers reported that they were recording the same thing in several places. A child who produced a row of four different Suitcases could not get a tick in three different Ticksheet categories and then appear to score well on three different Dimensions Continua. They would score well if, and only if, at least one of the ideas were developed towards making, the ideas were basically different or had some novelty value that was being explored, and that a clear evaluative path could be seen across the drawings. Despite the holistic view of designing that the analysis model expressed, the Dimensions could not be allowed to blur into one another.

Relationship to Making

Children who produced well-developed drawings and who subsequently changed their ideas when they made the product frequently did so because they felt their ideas "didn't work." On further questioning it emerged that this often meant that once they had the materials in their hands they would re-image their ideas. However, children who did not develop their recorded ideas and designed something new once they had the materials in their hands would frequently tell me that they were making just what they had drawn. There was a kind of tokenism about their use of drawing, which was quite different from those who had attempted to record an idea which they intended to make.

The sharing and swapping of ideas, adapting someone else's better idea to one's own purpose is such a feature of normal life (and that of design professionals) that it goes unremarked. Lave & Rogoff (1984) contrast this with school life where such helpfulness, especially in test conditions, is forbidden. In Design and Technology, where team-work is part of the overall skill-base to be encouraged, it is anomalous that once an activity is deemed a "test", ideas gleaned from others become "cribbing".

Diversity and richness

Section 5.8.1 set out my hopes for the Ticksheet that it might reveal the diversity and richness of techniques within children's responses to the tasks and, hopefully, that this was more so for my Focus Class. To have explicitly indicated this at every turn along the way would have been

wearisome but a glance back across the techniques produces the summary shown in Table 19 overleaf.

	Prevalence	Range
Annotation	greater	same
Level of detail	greater	greater
Look of the product	greater	greater
Recording materials	same	same
Construction details	greater	same

Table 20 : Diversity & Richness?

Put succinctly, more Focus Class children were using a wider range of techniques to generate and communicate their design intentions.

5.9 Gender Differences

Recent concerns about boys' poorer performance than girls' in school assessment creates the expectation that gender differences in capability be explored in any research into children's learning. Whether there were gender differences in the Exploratory Phase, I do not know, as I was not looking for them. The ethos of the school is of respecting all children equally and as a teacher, I have aimed to promote this by using non-gendered vocabulary. Thus when asked if there was a gendered difference in response to the activities, my immediate answer was "I don't know, they are all just children." So, although perhaps masked by my professional gender-blindness, some differences between boys and girls designing skills were likely to exist.

In both class samples the number of girls to boys was almost a 2:1 ratio (15:9 in the Focus Class and 14:7 in the Comparison Class) as shown by Table 21:

	Focus			Comparison			
	Girls	Boys	All	Girls	Boys	All	
Pizza	12	7	19	13	7	20	
Frosty	14	9	23	11	7	18	
Easter	13	9	22	14	7	21	
Card	14	9	23	12	7	19	
Suitcase	14	9	23	14	7	21	
Maze	11	8	19	14	6	20	

Table 21 : Numbers of Children by Gender

To generalise about boys' and girls' design styles on the basis of so few individuals (especially the seven boys in the Comparison Class) would not be appropriate nor would separating out the results for each layer of the quantified analysis into gender groups. Such low numbers are too small to support the validity of such an exercise (Tversky & Kahneman, 1982). Comparing results on the Purpose Continuum, however, might reveal differences that were less methodologically problematic. As discrete categories, the Drawing Types could also have potential, not being subject to the effects of a few low achievers pulling down an average. My Log Book notes included reflections on gender differences and I had clearly identifiable main boys' and girls' friendship and working groups within the Focus Class.

As I set about the gender analysis, I was curious to know whether, perhaps, some of the anomalies in the data (or even the different routes through the Drawing Types) were gender related. Perhaps Multi-design was more common among boys, or Progressive among girls? It was a possibility that I had not considered.

5.9.1 Quantifiable Differences

Checking the SATs results confirmed that the boys and girls in each class had very similar academic attainments (Table 22). By substituting a 0-5 scale for the SATs scale of "Working Towards" through Level 3 (so that 2C = 2, 2B = 3, 2A = 4 & Level 3 = 5) class averages could be calculated:

		Reading	Writing	Maths
Focus	Boys	2.56	2.00	3.00
	Girls	3.57	2.93	3.57
Comparison	Boys	2.57	1.57	3.29
	Girls	3.43	2.86	3.79

Table	22	:	SATS	Scores	by	Gender
-------	----	---	------	--------	----	--------

The Comparison Class boys' average is low for writing because the child with muscular dystrophy did not achieve Level 1. He was, however, a good designer.



The following observations of Chart 38 can be made:

 There is very little difference between the Focus Class boys and girls, whereas there is a marked difference between the Comparison Class boys and girls;

 There is a greater difference between the Focus Class boys and the Comparison Class boys than between the Focus Class girls and the Comparison Class girls;

• The Focus Class boys out-perform the Comparison Class girls.

Thus, the Focus Class, both boys and girls, developed a better understanding of design drawing as a result of receiving the Programme.

SECTION 5 - Analysis of the Assessment Tasks



The difference in numbers between each gender sample in Chart 38 has been accommodated by making the display the same size but indicating the difference in numbers in the y-axes scales.

Observations:

• There is a high number of Multi-design drawings produced by Focus Class boys. The number of girls' Multi-designs is roughly equal across both classes.

 The boys in both classes produced roughly the same number of Progressive drawings as the Comparison Class girls. The number of Progressive drawings by Focus Class girls is much higher.

 Roughly the same number of boys and Comparison Class girls use Multi-Draw. The Focus Class girls have used more Progressive drawings and less Multi-Draw.

• The number of Single-draws is roughly the same for both genders in both classes, although lowest amongst Focus Class boys.

The line between Multi-Draw and Multi-Design has been marked to indicate the division between design drawings that are static and those in which ideas are travelling. This shows that similar numbers of boys and girls in the Focus Class have ideas on a journey across their paper, whilst this is true only of Comparison Class girls.

5.9.2 Accounting for Differences

My search for reasons for these results led to conclusions drawn from observations of children working and reflection on reading of the literature of gender issues which meshed with understandings established through the literature in the Exploratory Phase.

5.9.2a Discussion whilst Drawing

My de-briefing conversation with Randal after the Flat Stan input (Section 4.4.2c) illustrates the extent to which the Focus Class boys discussed their ideas as they drew. Such design conversations led me to understand that this was not "copying" but co-operation and that the apparent randomness of some of these boys drawings was due to swapping ideas. Craig was always full of helpful suggestions to others. He was observed several times discussing what others could do to develop their ideas and so his input appears in several other children's drawings.

I think this particularly helped Noel who was a slower learner across the rest of the curriculum but included in this friendship group because of his football prowess. His friends' support scaffolded his learning and enabled him to develop into a successful designer.

The main girls' group in the Focus Class tended to chat less whilst they were drawing, but their occasional comments to each other denoted awareness of each others' good ideas. For example, this brief exchange seems almost meaningless:

Natasha (to Ellie) : l'm doing a rabbit. Ellie: That's a good idea.

However, Natasha had developed a detailed plan of how she will have pop-up rabbit's ears coming out of her Easter Egg Holder as it opens. Once this girls' group started making, several (including Ellie) included Natasha's rabbits' ears in their own product. The idea also appeared in some of the boys' work via the itinerant Craig.

Although my observation time in the Comparison Class was limited to the Assessment Tasks, I recorded few instances of co-operative working whilst planning, reaching its most extreme in their almost silent working whilst drawing their Mazes. The effects of this lack of discussion on their ability to Address the Constraints and to Explore the Possibilities of the Task was discussed as an emergent theme from the Dimensions analysis in Section 5.7.4. They did not act as checks on each other when mistakes were made. They did not share good ideas or

suggest things to each other to extend each others design thinking whilst drawing. In the Focus Class I frequently heard children, especially boys, saying "What you could do is.."

This discussion of design ideas whilst drawing was a consistent feature across Teaching Input activities and Assessment Tasks could, therefore, be seen as a factor which enabled the Focus Class boys to outperform the Comparison Class boys and girls.

5.9.2b Designerly Play

Browne & Ross' (1995) typifications of young children in free play activities resonate with my own observations in my Year 1 classroom. Girls tend to gravitate towards drawing and making small items at tables; boys sprawl across the floor playing with construction kits. Missing from the account, however, is the talk that accompanies play.

Browne & Ross typified boys as building models with construction kits, brmm-brmming them around briefly and then taking them apart to make something else. What Browne & Ross missed was that boys play and talk through the construction of complex structures, creating the fantasy and its vehicle together, integral with the social action. Browne & Ross typify girls as tending to make simple structures to support social interaction. My observations of girls at play is that they will create just sufficient play-props to maintain the story line. Whilst sitting making things as a group, they keep a weather eye on what each other are doing, monitoring each others output and appropriating good ideas without comment, often whilst talking about something else.

Both play styles were observed within the design styles of the Focus Class. The outgoing sociable main boys' group talked their way through every activity. They freely shared and commented on each other's ideas in the same way as they would play with a pile of Lego. Such design by discussion facilitated co-operation and sharing of ideas and through peer support enabled boys such as Noel (who had limited language and academic skills) to succeed as a designer. The main girls' group talked less, usually on the practical (sharing pencils or sellotape) or apparently superficial (colour, decorative features) yet shared good ideas appeared within each others' work, whether rabbits ears or mosaic floors.

I had made connections between children's play activities and designing in the Exploratory Phase (Section 3.2.6), having made many observations of Baynes' (1989) "designerly play" in my Year 1 classroom. The truth of his insights were confirmed at a In-service Training Course on using role-play for literacy development which included "designerly" skills such as

SECTION 5 - Analysis of the Assessment Tasks

negotiation, imagination, communication and so on. My observations of children at play had fed into my developing understanding of drawing as metaphor which had in turn led to the recognition of the potential of Lakoff & Johnson's (1980) theories and the adoption of the Container / Journey metaphor to teach design to young children. Thus the holistic at the heart of the Programme delivery was able to tap into children's personal play styles and enabled them to experiment and be creative with ideas on paper.

The Comparison Class were also playing with ideas in the Assessment Tasks but my comments on their playing have not been complementary, since they frequently ignored the rules of the game: the Constraints of the Task. Both gender play styles were there, typified perhaps by the boys snake pits and the girls' cards *"for Mummy because she doesn't like her new job"* (Kathy). But they did not harness these play styles whilst drawing. Experimenting and playing with ideas began in contact with the construction materials, leaving on the paper a static, first reaction to the task. The girls were happy to draw and write about something they might make (Browne and Ross (1995) record drawing as a favoured girls' play activity). The boys, quite simply, failed to see the point of the drawing.

5.9.2c Accessibility of the Container / Journey Metaphor

The results of the quantified analysis suggest that the Container / Journey metaphor was equally well understood by both boys and girls in the Focus Class and that the freedom given to explore ways of using drawing to record that journey was not prejudicial to either group.

The Programme was geared towards teaching for understanding of design in terms of this metaphor, not towards teaching of techniques which might have been more accessible to one gender group than the other. A range of drawing styles were encouraged and celebrated, since I wanted children to use drawing flexibly to support thinking and to develop their ideas. When children's work was shared with each other, I praised and publicly valued designs that showed elements of journeying. I did not attempt to change the way children recorded their ideas in terms of techniques used, nor did I endorse specific recording techniques.

There was no pressure to annotate drawings or add measurements, although many children did so. The teaching was about using drawing to support thinking. This allowed children's understanding to develop, whilst giving them freedom to express that understanding as they felt fitted the task in hand. Secure in my understanding of the role of drawing in designing and in aiming to impart that holistic understanding, I was able to accommodate a wide range of learning styles, which, as a by-product, supported boys and girls equally.

5.9.3 Reflections on Gender Issues

Kimbell et al. (1991) reported that gender differences in attainment were lower within classes of children who had opted to study Craft, Design and Technology than amongst those who had not, thus suggesting that direct teaching of design skills can enable all children to achieve equally well. Recent GCSE examination results, however, appear to suggest that perhaps girls are ahead in the subject. My observations of the social way in which boys design, compared to the way in which girls record their own ideas in greater detail, might suggest that it is the structure and organisation of the 16+ examination itself which is causing the effect, rather than differences in boys' and girls' capabilities.

As stated at the beginning of Section 5.9, gender issues did not feature greatly in my original thinking. However, reflection on Browne & Ross (1995), in pursuit of an answer to the results of my analysis, led to conclusions about the way that designing keys into play styles which connected back to observations and reading during the Exploratory Phase.

That the aim of my Programme was to impart understanding rather than to endorse any particular choice of recording technique, together with public praise of any means of moving design ideas forward, may have contributed to the equal success of boys and girls within the Focus Class.

5.10 Reflections on Analysis of the Assessment Tasks

At four points in Section 5, there has been a summarising section in which emergent themes were identified and discussed (Sections 5.3.2; 5.6.4; 5.7.4 and 5.8.3). This will not be repeated here since these are brought together in Section 6.2. The purpose of this section is to reflect on the aspects of the research that developing the analysis methodology for the Structured Phase brought to light, especially in relation to validation of results.

Assessment Tasks

The Assessment Tasks enabled children's understanding and use of design drawing to be examined in a range of Design and Technology contexts. The tasks represented a range of opportunities for observing and assessing performance.

I discovered that children can be task-sensitive in the way that they use drawing to a greater extent than I realised ahead of analysis of the Assessment Tasks. Product design and problem scenarios elicited a different use of drawing and so it was decided to analyse these separately.

Since I was not aiming for a positivist, pre-input/post-input testing regime, the differences between the Assessment Tasks were able to add richness to the data, rather than detract from it. My sample of children was too small to aim for a true quantitative methodology and, by the very nature of designing as a "wicked" problem (Rittel & Webber,1976), as well as the unpredictable complexities of working with small children, this would not have been a suitable paradigm in which to work.

My teaching experience enabled me to match the tasks appropriately to the maturation of the children, to design activities that would be suitable for the stages along the way through the Programme. I discovered more about children's use of drawing for designing from the diversity within the tasks than if they had all been alike.

Balance between qualitative evaluation and quantified analysis

A major step forward in my conceptual understanding came in realising that the child's understanding of the purpose of drawing for designing was the determining factor in how they chose to use it to support their thinking. Discovering Pascal and Bertram's (1989) evaluation model at about the same time, enabled me to combine the insight and the model to create a representation of my holistic view of the child's experience, which could also function as an analysis tool.

SECTION 5 - Analysis of the Assessment Tasks

Placing the child and their learning at the centre of the process gave me a way of integrating the data that I collected across the course of the Programme. It enabled me to re-establish my belief in the holism of experience and learning as a means of exploiting both qualitative insights and quantitative methods as mutually enhancing views on the whole. The video and audio recordings, observations and reflections in my Log Book, sketches and notes about children's work in progress all gave reality and context to the percentages and radial plots produced through the quantified analysis. At the same time, trends and issues emerged as important that would have passed unnoticed had it not been for the quantification of judgements made about the drawings.

The Multi-layered Analysis Instrument

At the start of the Structured Phase the pragmatics seemed straightforward: the Programme was written ahead of time; and I would deliver the Teaching Input; the drawings would be collected from each Assessment Task and I would assess the results. However, the refining and re-defining of the instrument in the light of developing theoretical understandings was the greater part of my research journey during the Structured Phase.

The re-definition of the quantified analysis instrument in terms of a holistic view of learning and understanding meant that genuine roles for both qualitative evaluation and quantified analysis could be found, with no sense of artificial bonding. Both contributed insights and ways of seeing the children's achievements and their processes.

The range of criteria on which the drawings were evaluated, both between and within the layers of the analysis instrument, enabled a multi-faceted picture of the children's capability to emerge. The inter-relationship of the layers of the instrument, that had evolved as it had been refined and developed, ensured that these layers were mutually supporting and the between-layer relationships within the results explored (for example, relating the understanding of the Purpose of the Drawing to the Drawing Type used).

Developing Theoretical Understandings through Analysis

In the course of devising, refining and re-defining the analysis instrument, I was also reflecting on the way that aspects of designing fit together. This was especially true of the insights that emerged during the development and analysis of the Dimensions of Design Drawing. Across the various iterations and refinements of the instrument, my reflections on observations, reading and analysis enabled me to take apart and piece together a personal belief-system about the nature of design and about design education.

SECTION 6 : Conclusions and Reflections

6.1	Overview Summary of my Research Journey	306
6.2	Key Themes that Emerged from the Research	308
6.3	Recommendations based on the Research	323
6.4	Drawing as a Tool for Thought	334

6.1 Overview Summary of my Research Journey

I began my research with questions about young children's designing at a time when the linear view of the design process, implied by the layout and wording of the National Curriculum documentation, was all that many teachers (including myself) knew. As Kimbell et al.'s (1991) conclusions indicate, the way that the Attainment Targets are viewed (either as sequential steps or domains of capability) makes a significant difference to the way they are interpreted and hence how teachers teach Design and Technology. The application of industrial project management systems to Secondary School practice, which were then extrapolated downwards to Primary Schools, did not fit with the theories of child development of Piaget-steeped Primary practitioners that were coming up to meet it. There was minimal research on what Key Stage 1 could do.

My questions centred on what could (should?) children aged 5-9 be doing in terms of using drawing to support designing.

However, as the research evolved, it was not enough to observe what they could already do. As a teacher, I was keen to discover what more they could be taught to do, and how. Hence the change of emphasis partway through the research from the Exploratory (descriptive, mapping out the field) to the Structured Phase (with hypothesis in hand, attempting to make improvements in performance, "raise standards").

During the Exploratory Phase (described in Section 3), I changed from thinking that Key Stage 1 children would find it very difficult, if not impossible, to use drawing to support their designing to a position of believing that Year 2 children could be taught to use drawing to develop design ideas and that I knew how I could do so. This change came from related literature on cognitive and language development, where I could see parallels with the development of graphic skills, as much as from the literature on children's drawing or designing (Section 3.4)

Early in my research, I became interested in the role of symbolic manipulation in design capability. This threaded through the Taxonomies and the related Concept Webs (Section 3.4.1) and on into the perception of drawing as metaphor, with the belief that when children understand that drawing is symbolic and can manipulate it as symbols, then they can use it powerfully as a design medium.

I increasingly focused on extrapolation skills and the understanding of analogy and metaphor and realised how much teachers rely on children possessing these skills in order to make personal sense of teaching input. I discovered a metaphor for designing which I believed I could make work in a Year 2 classroom: the Container / Journey metaphor, as a way of seeing the drawing as product or process, taking ideas on a journey. (Section 3.4.2).

The transition to the Structured Phase (Sections 4 & 5) was the point at which I felt most daunted at what I had set out to do: how to prove the "big idea" about drawing being metaphorical and that this could be used to teach children to access the genre. I was happy trying things out and seeing what happened, almost on an ad hoc basis. To put together a long-term Programme in the belief that it would enhance children's learning, based only on what I believed I knew, was an act of faith in which I did not always feel very confident.

Perceiving the potential application of Pascal & Bertram's (1989) holistic model for transforming the quantified analysis instrument into a viable tool to assess children's drawings put the purpose of the drawing centre stage (Section 5.4.3). It transformed the elements of the research into an integrated whole. The Container / Journey metaphor taught the children what I believed was the purpose of design drawing (to take ideas across a page and off into the making) and the assessment model placed centrally the children's perception of that purpose. Relating this to the Dimensions of Design Drawing to create discrete Continua (and discovering that I could display these as radial plots from my spreadsheets) enabled a second layer of the integrated analysis system.

The results of the Structured Phase analysis did not just confirm what I had already learnt at the Exploratory Phase. New information came out of the data. For example, the degree of task dependency of drawing use, through comparison of product design tasks to problem scenarios (Sections 5.5.2 & 5.7.2b). New insights emerged, especially the importance of discussion whilst designing and planning to make, both for creativity and addressing the task criteria (Section 5.3.2 & 5.7.4).

The results of both the qualitative and quantified analyses demonstrated that the Focus Class' design capability considerably exceeded that of the Comparison Class, suggesting that specific teaching of the role of drawing for designing via the Container / Journey metaphor was contributory to this success. This suggests that children's ability to learn through metaphorical inference to be quite finely tuned at an early age. Children appear to move from the symbolic towards the realistic, rather than the other way around. In the Exploratory Phase, I observed that the younger children were more likely to pretend that the things they made could work or answer a need. In the Structured Phase, the Comparison Class maintained this pretend way of working to the end of the Programme. The Focus Class did not. They were able to reason within a fantasy scenario yet produce solutions which satisfied the reality of the task.

6.2 Key Themes that Emerged from the Research

At several points along the way, Emergent Themes have been pulled together, from the qualitative evaluation of the Teaching Input and Assessment Tasks and from the results of the quantitative analyses of the Purpose of Drawing, Drawing Types, Dimensions Continua and Techniques Ticksheet (Sections 4.5.2, 5.3.2, 5.5.3, 5.6.4, 5.7.4 and 5.8.3). The aim of this final "themes" section is to pull together the reflections that have been woven throughout these reflective summaries from the major sections of the Structured Phase, whilst also paying more than lip-service to the work in the Exploratory Phase that provided the foundation on which the Structured Phase could be built.

6.2.1 Children can use drawing to plan what they want to make

The Exploratory Phase of the research revealed that, although many design drawings by Key Stage 1 children were rudimentary, there was sufficient potential in them for me to believe that Year 2 children could be taught to use drawing as a design tool (Section 3.3). This view was supported by my reading of parallel literature on language and cognitive development.

The results from both classes in the Structured Phase showed that it is possible for Key Stage 1 children to record their design ideas and to begin to develop them towards a plan for making a product. The far greater capability of the Focus Class demonstrated that with appropriate teaching, such young children can access the genre with considerably greater success.

I believe that being able to use drawing for designing is linked to the development of children's facility with symbolic manipulation and that this was the skill that Piaget identified as emerging around age 7 years. Piaget's critics (e.g. Donaldson, 1978) have linked this skill to the effects of formal schooling rather than innate development and my research would appear to support this position. The Focus Class, who received the teaching input, were able to manipulate a symbol system (drawing) in a way that the Comparison Class could not. This was evident across the ability range. There were more highly achieving Focus Class children and the least able demonstrated greater capability than the lowest achievers within the Comparison Class. Only Emma in the Comparison Class could hold her own with the high achieving group in the Focus Class but she was almost the oldest child in the sample and one of the only two children in the year group (of 95 children) to attain Level 3 on all SATs tests.

6.2.2 The Programme was Successful

As a result of the Programme, the Focus Class learnt to use drawing to support the development of their design solutions and to value drawing as a means of scaffolding ideas and communicating with peers and with me as teacher. They learnt that designing is an iterative process in which ideas change and develop and that the process itself is important and that first ideas are not "wrong" but a staging-post towards a better idea. They learnt the vocabulary of design (sketch, diagram, labelling, develop ideas, etc.), which enabled them to access, use and discuss design techniques.

The observations of children working recorded on video and audio-tape together with Log Book notes (Section 5.3) suggested a greater level of understanding amongst Focus Class than amongst Comparison Class children. These observations were supported by the quantified analysis. The Purpose of Drawing analysis (Section 5.5) showed that the Focus Class children had a clearer *understanding of the role of drawing* to support design thinking as a result of the Programme when compared to the Comparison Class children who had no such input. The Drawing Types analysis (Section 5.6) suggested that the Focus Class used *more sophisticated* Drawing Types earlier and more frequently throughout the Programme than the Comparison Class. The Dimensions Plots (Section 5.7) suggested that the Focus Class had a more overall understanding of *how to use* drawing for designing. The Ticksheet analysis (Section 5.8) showed a *greater range* of techniques in use in the Focus Class.

The differences between the two classes would suggest, therefore, that the Programme had successfully enabled the Focus Class to understand the *usefulness* of drawing to support designing and that the Container / Journey metaphor had been a successful medium by which to teach them to do so. The Comparison Class lacked a central understanding of the purpose of drawing for designing and so could not use it to support their planning in any personally meaningful way, which was evidenced by the lack of connection between the drawing and the making. Where Focus Class children departed from their own plans, it was frequently to incorporate features of someone else's.

6.2.3 Why Children Choose to Draw As They Do

The Exploratory Phase of the research led to the identification of the Drawing Types as a classification system of the drawings produced by the children across the 5-9 age range (Section 3.3.4). There was clearly a progression of some sort within Drawing Types: the youngest produced Pictures and Single-draws, the older children Multi-design and

Progressives, with the occasional Interactive features. However, it was also clear that this was not a simple linear progression.

Through my reflection and observation, I came to the conclusion that there was a competence level inherent in both Multi-design and Progressive drawing ability, in terms of symbolic manipulation skills, which once reached could be exploited at will. Both Multi-design and Progressive Drawings demonstrated an understanding that drawing could be used to represent ideas that could be changed and developed, "seen as" the real object as imagined in the mind's eye, a place-marking from which a design journey could be continued. This related to the Purpose Continuum categories *Using drawing to record design possibilities, Developing design possibilities* and *Working out what will be made and how.* Prior to this realisation, the recording of design ideas is static. A possibility is drawn but does not represent or support the flow of ideas about design possibilities or solutions (Section 5.6.2).

Such understanding enables children to choose to record multiple possibilities if they are swamped with ideas (Multi-design) or to develop their instant "I know what to do" reaction towards a design resolution to see if it would work (Progressive). Understanding of the purpose of drawing for designing as being the recording and development of design ideas, sets children free from having to produce any particular sort of drawing, even to the extent of knowing when not to draw (for example, writing a list of materials). The possibilities recorded as quick sketches (typical of Multi-design) are multiple possible directions the design could go (like roundabout exits). The Progressive drawings, the developments of a single idea towards making is more like the unfolding of a route with few side-turnings.

Multi-draw forms the bridge between Single-draw and these design journeys. However, once the journeying stage was reached, children would also use Multi-draw as a short-hand for situations where they did not feel the need for the support of the drawing. They did not, however, revert to Single-draw.

The round-cornered rectangle in Fig.59 (overleaf) represents this plateau of realisation. The arrows are double-ended to indicate the by-directionality of choice of Drawing Type that the freedom of this cognitive plateau allows. Children who produce Single-drawings have not begun to understand how drawing can be used to develop design ideas. They have learnt the genre (no background, clouds in sky, etc.) but have no grasp that drawing can be used as a tool for the development of design ideas, cut free from its ground anchor and used as the supporting structure for free-flowing ideas.

SECTION 6 : Conclusions and Reflections



Fig. 59 : Developing Understanding of Design Drawing (a)

Relating this to the children's understanding of the purpose of the drawing to support designing, Fig.60 demonstrates the way this plateau of realisation allows children to become far more fluid in their use of drawing to support designing.



Fig. 60 : Developing Understanding of Design Drawing (b)

Crossing the bridge between clarifying the task and designing solutions means that children can then choose the recording technique that they feel most appropriately fits their level of clarity about the task in hand. If they have a rush of ideas but are not sure which to choose, then they use Multi-design. If they settle quickly on one idea but need to develop ideas about its viability for making, they use Progressive drawing.

Arriving on this plateau indicates arriving at an understanding of the Journeying aspect of the genre of design drawing. I think that the Container / Journey metaphor cannot be accessed until some understanding of the purpose of drawing for designing has developed. This process is frequently evidenced by the use of Multi-draw. Chart 12 in Section 5.2.2 shows two different scenarios for Multi-draw. In the Focus Class, its occurrence is divided between the static and journeying categories of purpose perception. In the Comparison Class, its occurrence is almost all in the static category. I think that this difference of function of the same Drawing Type is

evidence of the difference between the two classes in their understanding of the design drawing as a vehicle for developing design solutions.

During the course of conducting the quantified analysis of the Structured Phase, it became clear that children were using drawing differently in problem scenarios to product design tasks. For product design tasks, the Focus Class, particularly, were using the drawing to think through materials, decorative features and construction. For the problem-scenario tasks, the children did not record materials or construction techniques. The priority was to develop a solution. It would seem that they had an awareness of the different status of the 3-D resolution (as product or model) and were able to adjust the level of detail in the drawing accordingly.

6.2.4 Understanding the Purpose of Drawing for Designing

I came to realise that *understanding the purpose of drawing for designing* is the determining factor that allows children to cross the "bridge" into fluency with the genre of design drawing. Once children understand that drawing can be used to develop initial ideas and work out how to make a product that satisfies the design brief, then they are able to choose the kind of drawing that suits their response to the task. That this would not become clear to me until part-way through the process of devising an analysis instrument for the Structured Phase drawings (and after the completion of the Programme) is one of the ironies of my research journey.

By being constantly reflective and maintaining dialogue with those whose research was parallel to my own, I was able to identify and make explicit this underlying, implicit, construct. The appeal of the Container / Journey metaphor was that it answered for me the question *"What is the function of drawing for designing?"* as *"To contain ideas to take on a journey"*.

That this reasoning was commutative (that teaching children that drawing is a container for ideas that can then be taken on a journey would lead them into understanding the purpose of using drawing for designing) was to emerge in the analysis process, rather than preceding the design of the Programme. This was probably no bad thing in terms of the teaching: the children were able to construct their own understanding of the role of drawing in designing. If I had been too explicit it could have hindered rather than aided holistic conceptual development and understanding. My Container / Journey metaphor gave the children an explanation for what they were being asked to do that they could understand and work with.

It was this understanding that distinguished the work of the Focus Class from that of the Comparison Class. Even on the final Maze task, many Comparison Class children were still using drawing to clarify meaning, rather than to develop a design solution, which led to a much lower level of success in terms of design resolution (Section 5.3.1f & 5.7.1). They were less likely to have satisfied the task criteria or to have produced a creative response (Section 5.7.2). This trend was evident also in earlier tasks, surfacing at Easter, immediately after the sharing of the Container / Journey metaphor with the Focus Class (Section 5.7.1).

I think that the Comparison Class saw the drawing as a prelude to the real business of task resolution, whereas the Focus Class saw it as integral. The Comparison Class spent only a short time on their drawings and quickly abandoned drawing once others started making. In the Focus Class, children saw no need to stop discussing and drawing if others had moved on to making. Some children even sat and completed their drawings after all other children around them had come back from collecting their materials and begun making. This would indicate that the Focus Class children saw the drawing as an important part of the whole process.

6.2.5 The Container / Journey metaphor

In reaching the conclusion that this would be suitable to explain design drawing to young children, several metaphorical leaps of my own had been made. It assumed, based on Lakoff and Johnson's (1980) understanding of the role of metaphor in concept construction, that all new knowledge is built by metaphorical connection with previous knowledge. It assumed that all people, including young children, can access metaphor as a concept-building devise and that teachers frequently uses this assumption in practice. It was metaphorical leap (Fig. 61) and not a logical connection between "drawing is a metaphor" and "a suitable metaphor can be used to explain design drawing to young children."



Fig. 61 : Metaphorical Reasoning about Design Drawing

SECTION 6 : Conclusions and Reflections

The Container / Journey metaphor was couched in story form, complete with mime, to demonstrate putting ideas into an imaginary carrier bag which was then carried along on the journey to the next stopping place, unpacked and each imaginary item examined and put down and then the bag repacked and other useful objects found and the process repeated. The way this related to design drawing became part of the seamless narrative explanation, complete with appropriate hand actions between brain and imaginary piece of paper. I think the children appreciated the idea at a sub-linguistic level.

The Focus Class were able, immediately upon being taught this view of designing, to produce drawings that expressed their design intentions more fully than any I had seen before amongst Year 2 children (they outperformed most on my previous Year 3 examples) and since the Focus Class' work continued to exhibit a greater degree of design success than that of the Comparison Class, it must be argued that the metaphor was an appropriate vehicle for conveying understanding of design drawing. The effect on their use of drawing was immediate. It did not grow gradually, indeed some regression occurred amongst those who had not completely understood. It was not a skill that they developed; the Comparison Class did not come to this conclusion about the use of drawing or use it effectively to develop design ideas throughout the course of the Programme.

An additional factor in the success of the teaching of the metaphor was that it was visual as well as aural. Viewing the Snook's Animals PowerPoint presentation set the scene for understanding the features of a design drawing. Reviewing this by starting the Stan cycle by showing the children examples of other children's planning drawings and pictures gave them the information needed to make sense of the explanation and the diagram.

6.2.6 Dimensions of Design Drawing on a Journey

If the endpoint of designing as a teleological act is its realisation as a finished product, then the relationship of the other Dimensions of Design Drawing to "Basis for Making" should reveal something about their relative importance in design journeying and impacts on questions about the Dimensions which had been below the surface throughout the analysis:

- Is there a hierarchy or are all equally important?
- Are some more closely related to each other than others?
- Are some dependent on others / are some independent of others?
- Are any more / less important?
- Which are the ones to stress to get good results?

To address these questions in relationship to taking ideas on a journey, I created a concept map (Fig. 62) based on the emergent themes of Section 5.7 :



The key Dimensions, which have the strongest impact on each other and appeared to be most important for the successful use of drawing for designing are : Generating Developing Design Ideas, Exploring Possibilities, Addressing the Task Constraints and, central to these three, Evaluating whilst Drawing. This has close similarities with Rogers and Clare's (1994) design model (Fig.22 in Section 2.4.2). Planning the Look of the Product is placed at the side. It is either an essential in product design (indicated by arrow) or irrelevant to problem scenarios. Communicating Design Ideas with each other was important for Exploring the Possibilities of the Task and for Addressing the Constraints of the Task (Section 5.7.4). Conversely, a drawing that was easy to understand (perhaps because well-labelled) was not necessarily an indicator that high quality design thinking had taken place.

The Dimensions that have the least impact of the use of drawing for designing are those which can be achieved without the child using the drawing to move the design forward. Many Comparison Class children produced clear construction plans which were subsequently ignored. It was this kind of work that McCormick et al (1994) accused of being ritualistic. For example: Holly produced a single drawing at Easter of a clearly-labelled, lidded cup with no tube in drawing or product; Nicola drew two sets of well-labelled Suitcase instructions neither of which were followed when making. Both of these Comparison Class girls clearly communicated their ideas for construction but did not moved their ideas towards satisfying task criteria (Holly) or connecting the designing with the making (Nicola).

The grappling with design ideas happens within the generating and evaluating of developing ideas within the potentials of the task as set. For example: Hayley's Easter chick, with her

attempt to indicating yellow card wings that would flap or Damian's combination of ideas for his Card which culminated in a pop-up bomb when the card was opened. These Focus Class children had understood design drawing as useful for developing creative ideas about how to solve a specific design problem. Rearrangement of the concept map links the Dimensions more explicitly to taking design ideas on a journey across a sheet of paper and off into making a product:



As the analysis of the Comparison Class' drawings revealed, Planning Construction may not necessarily indicate that ideas are moving towards making the final product (Sections 5.7.1-2). Likewise, Planning the Look of the Product can be a major driver in moving ideas forward but it can equally be a token recording of identical products with different patterns. It is less relevant, of course, to solving problem scenarios.

It does not necessarily follow, however, that a wonderful design journey across a sheet of paper will result in the production of a product to match or that the plans are not abandoned once the materials are to hand and a friend is seen to have had a more appealing idea. However, plans that are well-developed and that represent ideas shared and discussed with peers are more likely to be followed through in the making.

If Addressing Task Constraints and Exploring Possibilities are pulled together as "Awareness of *Task Potential*" and if Planning the Look of the Product is considered as part of Generating and Developing Ideas about a product, then the Dimensions essential for design journeys through drawing appear to be as shown in Fig.64, overleaf:

SECTION 6 : Conclusions and Reflections



Fig. 64 : Contributors to Ideas on a Journey

6.2.7 Generating and Evaluating Creative Solutions

The results from the Structured Phase indicated a primary role for using drawing for Generating and Developing Design Ideas (Section 5.7.1-2). If the drawing is not well used here, further development of ideas through drawing did not take place. Whether or not the children recorded materials or construction techniques, the drawing supported and aided their thinking at the earliest stage of beginning to form images of what they would like to make.

By starting with using drawing to generate ideas, and understanding the process as journey, the Focus Class children took the drawing as far along the track as they felt they needed it to support their thinking. For Comparison Class children, recording how to make a product for which they had a much less clear mental image meant that there was frequently little relationship between the drawing and the final product. Although the Comparison Class could produce detailed working drawings, these did not represent part of the flow of their real design thoughts. This was in contrast to the Focus Class who either recorded several different possibilities, made improvements or indicated how they thought they might make the product. This seems to indicate a different view of the purpose of the drawing. The Focus Class were using drawing to develop a design solution, whereas the Comparison Class were defining the design problem, which made it more likely that the drawing would be subsequently ignored.

Boden (2001), in discussing "exploratory creativity", makes a brief comment in passing that mental spaces are easier to change or adapt than physical ones. It would appear that their greater facility with design drawing had provided the Focus Class with a means of scaffolding their thinking, so that their own thoughts had become visible to themselves and others and thus

available for review and discussion. This enabled a greater creativity of response through playing with ideas which, whilst still in the mind's eye, could be adapted and changed through reference to the drawing. The Focus Class were beginning to use the drawing as a modelling tool, whereas the Comparison Class tended to wait until they were handling the materials before making the real decisions about the product they were going to make.

The Focus Class were much more able to accept task constraints and search for a solution of the task as set. The Comparison Class seemed almost cavalier about the task, abandoning the cardboard tube that was central to the Easter Egg Holder problem, for example. The "tightness" of the early activities of the Programme, strongly teacher-led (for example, the analogy game-making activities) forced the Focus Class to accept my definition of success. The Comparison Class did not experience parallel activities and were, perhaps, given opportunities for freer exploration of their own ideas, rather than being asked to solve specified problems.

The analysis of the Dimensions of Design Drawing led me to realise that creativity in Design and Technology does not come about just by free-play experimentation with the materials. Realisation of the task potential is vital to formulate a creative solution to the problem. This is a balancing act between Addressing the Constraints and Exploring the Possibilities of the task as set. Understanding the purpose of design drawing as supporting the resolution of the tensions between these two apparently opposite demands enables the drawing to be used to move ideas from clarification of the task towards a design solution.

The Focus Class' greater ability to reason within the parameters of the task, was, I believe to do with their greater ability to reason within the rules of the design game, through having a clearer idea of what those rules were and, hence, which ones to bend and which ones must be adhered to more strictly. The design game says that the definitions of reality and fantasy are inherent within the task as set and that the successful conclusion of the task will be a solution which satisfies these parameters and, preferably, be novel or aesthetically pleasing.

In his rejection of the notion of logic as a universal pure, context free principle, Wittgenstein (1969) referred to the construction of meanings within separate domains of human endeavour as "language games," each with their own rules and internal logic from which phenomena are "seen as" and whose interpretation depends on the rules of the language game in question. Polanyi (1958) viewed each subject discipline as having its own tacit knowledge in which the rules make sense and the internal logic cohered but did not transfer to other disciplines or cultures. He saw this as being the reason for science and religion being unable to converse, since neither accepted the rules of engagement of the other's logic system.

SECTION 6 : Conclusions and Reflections

Induction into language games can be seen in most didactic situations. The course writer has a clear view of the aims of the course and the teacher creates a social situation in which the learning of such outcomes can best be achieved and engineers the pupils into learning, not just the facts, but the underlying "seeing as". Teaching is not just of facts but also of how to answer questions posed in certain word frames.

For example:

Key Stage 1 Science SATs question: "How could we help the seeds in the dark cupboard grow better?"

The teacher was irritated with Maria's answer: "Draw a picture of the sun and rain and pin it up inside the cupboard for the seeds to look at."

But I find I want to create a story in which the bad teacher had put some poor seeds in a dark cupboard and then a sweet little girl came along and drew a picture of the sun and rain for them to look at and overnight the seeds grew and grew.... which is the stuff of children's literature, of metaphor and poetry and jumping from reality to fantasy and creating a new world order in which such things are possible. But this is not the right language game for science.

I believe that Wittgenstein's "seeing as" is central to using drawing as a design tool (see Section 3.4.3a) and that this is related to metaphorical perception. Solving a problem appropriately implies understanding and applying the "seeing as" limits of the question. Maria's answer does not necessarily imply that she understands little about science, only that she played the wrong language game with the question. Liddament (1991) applied Wittgenstein's concept of language games to designing and I believe this to be a powerful concept in defining what the skill of designing is. It involves the ability to reason within a set of perceived constraints and yet find a creative solution, in a way that has a family resemblance to a game of football in which players with flair can exploit the game potential whilst remaining within the constraints of the game's rules.

In design terms, this is the ability to juggle conflicts inherent in the problem-space and find a creative solution that satisfies both user and situational constraints. It is the ability to image fantasy onto reality, what might be onto what is, and to accept and reason within the fantasy / reality interface inherent in the design task. Pandy's Suitcase illustrates this dilemma: the children were asked to design a "real" suitcase for a toy panda "going on holiday" into which his real plastic mac would fit and which was the right size for Pandy to "carry".

I think this "seeing as" skill is also parallel to the ability to create paracosms, a complete fantasy world that has its own cultural and social history. Successful problem-solving depends on this paracosm ability, the ability to set up, reason and imagine within a clearly defined mind-space. Its beginnings are in the role-playing of young children.

The Programme enabled the Focus Class children to play with ideas within the design paracosm as defined to them through the Container / Journey metaphor. By defining a design play-space with the children that was clearly linked to the purpose of the design drawing, this became the basis of discussion of design reasoning. The shared vocabulary, the discussion of process, the externalisation of the conceptual foundations, all re-inforced the development of meta-cognition and the ability to think about design thinking.

One of my colleague Sue Hammond's powerful ideas about developing children's writing skills was that children should be encouraged to see themselves as writers and that as they role-played *being writers*, they would *become writers*. Children will role-play *being designers* if they know what designers do. They need to be told the rules of the game.

6.2.8 Multiple literacies: Discussing, Drawing and Writing

A contributing factor to the difference between the two classes' performance was the point at which the children discussed their ideas with each other. The Focus Class talked as they drew. The Comparison Class did not (they drew in silence for the Maze). I believe that this had an impact on the quality of ideas recorded in the drawings. It would indicate the importance of language for idea development and clarification. Ideas flowed in talk and the drawing interacted with the both the telling and the thinking. When children said *"What I'm going to do is"* to their neighbour as they drew, then the idea became more clear to themselves. I felt that this had an improving effect on the Focus Class' capability in using drawing to model design ideas and also to encourage creativity whilst keeping each other on-task (Section 5.3.2). It also demonstrated their understanding of the role of the drawing.

These observations are in accord with current research by Mantell (2000) into the role of language in designing and her observations of the way language frees us from the here and now and makes possible the imaging of the future. Such observations of the roles of drawing and language would lend support to Schon's assertion that:

"Drawing and talking are parallel ways of designing and together make up ... the language of designing." (Schon; 1987: 45) Koutsides (2001), quoting the work of Hennesey & Murphy (1995) suggests positive effects on problem solving skills in an environment in which co-operation and discussion is encouraged, where "concrete models and graphical representations play and important mediating role" (Koutsides, p. 57).

The ability to Address the Constraints of the Task was also aided by this talk. In discussion, ideas not only flowed but were challenged. The individualised view of recording in the Comparison Class meant little cross-pollination of ideas, discussion of the possible and adaptations in light of peer suggestions.

The conversation between some of the Comparison Class children when they were in the Reception class (Section 1.2.2) illustrated the way they co-operated, negotiated and built on each others ideas as they conversed. I hoped that I might see similar conversations across design drawings. In the Focus Class this was so but in the Comparison Class discussion was whilst making, not drawing. They were then less able to be reflective of their ideas since they had committed themselves to cutting into the materials. If this talk had happened at planning stage, then better ideas (more creative and task appropriate) may well have ensued.

A clear link emerged between *Generation of Design Ideas* and *Evaluation whilst Drawing*, linked in turn, I believe, to the level of peer discussion. The Comparison Class frequently drew a single undeveloped idea, with no changes or improvements recorded. I felt that their inability to address the task as set was exacerbated by their lack of discussion whilst drawing and planning. It was clear that they had no expectations of discussing each others' ideas at this stage, of presenting their ideas to each other for critical review or of offering evaluative feedback to each other. It was difficult to know which was cause or effect: whether they did not evaluate because they did not see answering the task criteria as vital to success, or whether they lost sight of the task because of their low level of evaluation.

Both classes used annotation of drawings to clarify and develop their design ideas. The choice to do so was frequently based on common sense (words such as "green" or "shiny" or "lots of *little stars cut out*") which was used as a short-hand for recording ideas that would be inappropriate or take too long to draw. Only Peter (Comparison Class) did not draw at all and his choice of writing hampered the development of design ideas. He was not imaging a solution, only writing about the problem. Most children had no difficulty combining text and graphics and I concluded that their facility with the genre was probably enhanced by the study of non-fiction texts in Literacy Hour.

SECTION 6 : Conclusions and Reflections

The relative roles assigned to writing and drawing seemed to be that the drawing was used to capture inner images and writing for descriptives. In the Sandals project, the Focus Class chose to write what they liked about the sandals they were shown and drew their ideas for sandals they would like to make. When they first put pencil to paper to begin designing, almost all children begin by drawing, very few even put their name on the paper before beginning to draw. Writing is then added for clarification, annotation and comment. Occasionally thoughts extending their ideas will be recorded in writing but it is more likely that the child will add another drawing, even if they are only changing the colour or pattern.

Whereas in the past, reading and writing were the essentials of education, in our fast-changing world, graphics have become equally powerful means of communication. Children live in a world in which they are surrounded by multiple literacies through television, advertising, computer screens as well as in books, posters and road signs. Increased foreign travel requires a greater facility with non-language-based communication. The rapid growth in computing power has enabled animation and special effects to make communication of sophisticated concepts less dependent on traditional text-based systems. The language of our world is no longer just speech and text. In such a cultural and social context, therefore, it becomes vital for children to be able to manipulate graphic symbols for developing and explaining their thoughts and ideas, not just for Design and Technology lessons, but as part of their repertoire for recording their own thoughts and in communicating with others.

The following, attributed to Wittgenstein, challenges all educators to provide children with tools for thought:

The limits of my language mean the limits of my world.

I would argue that the ability to use drawing to take ideas on a journey, both within and beyond a design context, extends thought beyond the barriers and limits imposed by the language of words and into the realm of visual imagery that can be produced and manipulated, turned through three (or even four) dimensions, simulated and digitised. The children of our world know that this is possible; they see it daily on their televisions and consoles. It is our duty as educators to provide the tools by which they can enter into this world as creators and not just consumers. Teaching young children to use drawing as a tool for thought seems, to me, to be an early step along that path.

6.3 Recommendations based on the Research

The following recommendations come from a position of caring deeply about both the education of young children and about the professional development of their teachers. Although my recommendations fall within the framework of the research into children's design drawing skills that I have conducted, I believe there are applications beyond its remit that affect the daily lives of our most important stake-holders in education, the children. Their needs are implicit in each of my recommendations and, although I address recommendations to teachers, managers, policy makers and curriculum writers, teacher educators and professional supporters, and parents, I believe that the needs of the children are central to education.

However, from my experience as a teacher conducting research, there are issues I would like to raise first concerning teachers as researchers.

6.3.1 Teachers as Researchers

When I delivered the paper "Participant Research from the Perspective of a Participant Researcher" at the IDATER 2001 conference, I told of the time that I walked back to class after lunch one day just after the introduction of the National Curriculum in 1989 and thought to myself "Research is dead, then." It seemed to me that once national directives about what and how we were to teach came into force, there was no point in anyone conducting research into children's learning. That the documents were called "Orders" seemed ominously significant. Since then, the orders have been framed ever more tightly, culminating in the Literacy Hour clock, and even the current apparent freedom in foundation subjects is countered by the QCA schemes of work that were sent to all schools. In a climate of such prescription, with target setting dominating the landscape and defining achievement, the teacher who wants to try something new and different in their classroom is doomed to a long and lonely walk.

As a teacher, I should feel most comfortable addressing other teachers but when I have conducted Inservice Training (Inset) sessions, I have found that the questions they bring with them relate to how to deliver a very narrowly conceived curriculum or published scheme of work. Fear of inspection failure appears to drive teachers into wanting to be provided with the formula for inspection success, in terms of which lessons to deliver and which practical skills to teach children in order to ensure achieving prescribed objectives. A culture change is needed within Primary education from "delivery" to "reflective practice." This change must come from policy-makers and implementers. It cannot be initiated by teachers, since they do not have the political power to do so.

SECTION 6 : Conclusions and Reflections

I found it daunting to be the only Primary teacher at research conferences and yet I worked within a school community in which informal reflective discussions on children's learning and cognitive theory took place. I found myself becoming increasingly suspicious of the deficit model of action research that was emerging under Performance Management . It seemed that teachers could conduct research on how to improve our delivery of the National Curriculum and this could feed into our Performance Management. We were not good enough yet, but we could get grants to conduct research in our classrooms to find out how to become better at delivering the National Curriculum. Action research should be about competent teachers researching how to extend the boundaries of knowledge about how to improve and enhance children's learning in all directions, not just towards the unquestioned "targets" within an unassailable national framework.

There needs to be, too, a shift in perception of the relative roles of researcher and researched. Welch & Mueller (2002) speak of the need for teachers to become the creators of knowledge, not just the consumers. I believe that unless teachers are creators of knowledge, then educational standards can only decline as teachers seek the "one right answer" to passing Ofsted inspections and surviving the pressures of achieving externally set targets.

Teachers are increasingly being encouraged to undertake study at Master's Level, as a means to promotion in management roles, and yet largely remain outside the community of contributors to policy. If research into children's learning was recognised as part of the teacher's role and the findings of such research were respected by policy makers, then a transformation of teachers' perceptions and self-esteem would result. However, time and resources would need to be allocated to research activities and there would need to be some guarantee that outcomes would lead to implementation.

As summary to each sub-section of Section 2 Research Methodology, I applied the issues to myself: *Where do I fit?; Myself as both Teacher and Researcher; Applicability to my Research Methodology.* So, in my final comments here, how did the research process affect my perception of myself as a teacher conducting research?

My pessimism in 1989, that research was dead, was coupled with a realisation that, as a Primary teacher, not only that I did not have a voice, but I could not even ask the question: on what research evidence was the Design and Technology National Curriculum for Key Stage 1 based? Nearly ten years later, with a Master's degree and registered with Goldsmiths College to begin my M.Phil./Ph.D, I was able to ask Richard Kimbell that question.
Since its introduction, that curriculum document has changed in response from pressure from industry and from teachers' professional organisations. However, the relationship between policy and research into children's learning remains weak. The tinkerings are at the pragmatic rather than the pedagogical level, teachers are encouraged to think about "what works in classrooms", rather than develop their own understanding about how children learn.

After two years away from the classroom, I could not go back to having to accept government directives about what and how to teach or to working within a straight-jacket interpretation of how children are to be taught. I could not go back to non-research based teaching. I have been saddened to see one of my moderation panel leave full-time teaching and the other two accept unreflectively the adoption of all the QCA schemes of work for all foundation subjects in both Key Stage 1 & 2. Our home-grown creative ideas are now the programme for D&T Club for Key Stage 2 children after school.

In my final term in school, we had an Ofsted inspection. I received a good report of the Design and Technology lesson that was observed. The inspector was impressed that I was teaching Design and Technology to the whole of my Year 1 class without a teaching assistant : "This is very well organised, I usually see chaos." He was less interested in my credentials as Co-ordinator, dismissing my interim research findings, conference papers, photographs of exhibitions taken to international conferences: "We get shown photographs of children's work at every school. What I am interested in is your planning and assessment policy."

Such devaluing of teachers' contributions to knowledge speaks to me of an education system in crisis. Richard Kimbell's Keynote address at the DATA International Millennium Conference, entitled "Creativity in Crisis" gave strong warnings of the culture that Ofsted inspection teams were encouraging through their emphasis on management and their ignoring of creativity and innovation. Without the freedom to experiment, the recognition of innovation or the respect for professional creative curiosity, the most reflective practitioners will seek something more interesting to do.

My first recommendation, therefore, is that *research-based practice* should become the dominant model on which curriculum development should be based and areas of the curriculum that are under-researched should be funded by government. This should replace the politically motivated, policy-driven model that has dominated the educational landscape in the last two decades and is threatening to stifle the creation of knowledge about pedagogy and children's learning. My research in the development of children's design drawing skills has inevitably led to reading and reflection on other areas of children's learning across the

curriculum, but the structure of the imposed curriculum, as well as its tightly defined content, was prohibitive.

Thus to Headteachers and School Governors, I would say that teachers conducting research need support, patronage and their results incorporated into school policy. Ofsted Inspectors visiting schools need to respect these initiatives and see these as signs of vibrant life within the school community, not something to be measured against a one-size-fits-all yardstick of what teaching and learning looks like. The Ofsted team that visited my school in October 2001 were not informed that doctoral research was being conducted into an area of the curriculum. The management team wanted to show how we were conforming, not how some of us might be delving into theories of learning.

Such a radical change in outlook would require national government to put aside the belief that education can (or even should) be centrally controlled and monitored. It requires a re-appraisal of the nature of teachers as professionals, whose own motivation and curiosity can be relied upon to move children's education forward.

6.3.2 Developing Children's Design Drawing Skills

The recommendations that I would make regarding the development of children's design drawing skills are addressed to the many contributors to design education:

• View design as a holistic, creative process involving generating and evaluating, communicating and developing, exploring ideas which creatively answer a design problem, into which drawing can be fitted wherever it is helpful to move children's design ideas forward. This needs to be imparted in documentation issued by policy-makers and curriculum writers. The way in which information is presented is important. Pascal and Bertram's (1991) model is child-centred, holistic and adaptable.

The desire of managerial policy-makers to make everything easily auditable, does not allow for the creative (at times anarchic) nature of innovation and design development. There needs to be a change from a bureaucratic model of curriculum development and assessment to one that acknowledges that a creative process such as designing cannot be cut into manageable lesson bites with predicted outcomes for prescribed grades of pupils. Headteachers and subject co-ordinators need to examine carefully the view of designing embedded within curricular guidance and published schemes of work, to ensure that it adequately represents the spirit of the current National Curriculum documentation and also reflects the creative heart of Design and Technology activity.

Those who are involved in teacher education and professional development must seek to counter the linear view of the design process, with drawing "my design ideas" at the beginning and writing "my evaluation" at the end, which still lingers from the days of numbered Attainment Targets. It is perpetuated by publications that set out schemes of work organised as design-make-evaluate series. This can only be addressed by a rejection of such schemes by schools, which, hopefully, will lead to a more creative approach to scheme writing, that does not provide an off-the-shelf lesson for every week of the year, but a range of options that affirm that designing cannot progress without evaluating, that both designing and evaluating are continuing during making, and that evaluating after the making is completed could result in re-designing.

A change of language in curricular documentation, from "Targets" or "Levels" to "Dimensions", would also help to affirm that each aspect of design capability is interrelated and is used in varying degrees in different situations. "Targets" imply that these must all be met and are equally important. "Levels" implies linear progression and that using a strategy on a "lower" Level implies regression, rather than choice of appropriate strategy for the circumstances, prior knowledge and experience with materials or techniques. In the Structured Phase of my research, the children's different responses to problem scenarios and product designs showed that Year 2 children could choose different and appropriate strategies to achieve a design solution.

• View drawing as one medium through which design ideas can be recorded and developed, but not the only medium. Drawing is, I believe, a very powerful medium for design development, but children should not be expected to draw as a matter of course in every design project. My research has focused on drawing but I was at pains to ensure that the Focus Class did not receive a drawing course. In the Summer term, the children did very little drawing and yet this did not prove detrimental, rather it allowed them the freedom to apply their understanding of ideas development into other media and then re-apply insights gained there back again to drawing.

The marriage of assessment to accountability, viewed as requiring paper evidence ("something in their books") militates against using drawing creatively and also against using other media for ideas development. Research into children's use at different ages of constructions kits or plastic material (such as clay) to model ideas, for example, might find parallels to my Drawing Types, or cast more light on children's view of the interface between model and product.

Management and monitoring of children's progress needs to be subject-appropriate. Although my research has focused on analysis of children's design drawings, the design drawing is not *by itself* an appropriate assessment tool for the broader educational concept of design capability. Children's design intentions within the Structured Phase were also judged by their articulation about their drawing and making, the video recordings and also by examination of the final products. Assessing design capability needs to be multi-stranded, with recognition of the centrality of modelling as the core skill. Research into the way in which children use a range of media (including drawing, writing and discussion) to scaffold and develop design ideas needs to be conducted. Policies on curricular review should include a research agenda that seeks to discover both what children at different ages can currently do and then attempts to find ways to enable improvements.

• Explain the purpose of drawing for designing to the children. Children do not use drawing to support designing because they have not been shown how, not because they are incapable of understanding. And they cannot second-guess that it might be useful. The Comparison Class did not do so, despite taking part in a similar range of practical activities. The design capability of the Focus Class was higher than any children I had previously taught, including the Year 3 class that I taught weekly in 1998-9. The difference was the use of the Container / Journey metaphor to explain the nature of design and the role of drawing. The Focus Class became better users of drawing to support their design thinking because they understood its purpose, not because they were taught particular techniques.

My Programme worked because I had a personal construct of what design drawing is for, and which I could convey to the children through the use of the Container / Journey metaphor. Those who provide professional support and guidance for teachers must aim to convey understanding of what designing is about and the role of drawing within it, not just imparting information about how to deliver a new method of working. This is not a quick-fix, bolt-on, tips-for-teachers method that will instantly enable children to produce design drawings. It has to come from the teachers' own personal understanding of the nature of design and how drawing can be used to support design development. This needs to become integrated into teacher education and professional development for Design and Technology.

• Allow children to use drawing to support their thinking to the extent to which they find it *helpful*. It is counter-productive for children to be engaged in an activity to which they see no point and that might inhibit the flow of genuinely creative ideas. Drawing for designing should support designing, not be a substitute for it so that the teacher can see "evidence" in books.

SECTION 6 : Conclusions and Reflections

There should be no requirements on children to draw in a particular way. The Programme contained no lessons on Multi-designing or how to do Progressive drawings. I suspect that I would have had superficially more impressive results if it had. However, I am a teacher, not an instructor, and I wanted the result to be real learning and for drawing to be useful to support the children's designing, and not the product of having learnt a range of techniques. I would certainly have learnt less myself about children's ability to adapt their use of drawing to their perception of task needs. The Comparison Class had learnt to draw labelled diagrams prior to the Easter Egg Holder task but it inhibited, rather than aided, their use of drawing to develop design ideas. Teachers need to teach for understanding, not specific performance.

Those who write books or create web-sites on how to enhance children's design skills need to resist the temptation to provide ready-made frameworks and "design sheets" which tell the children to use drawing to record particular stages of the design development in particular ways. I only used design sheets in the early stages of the Programme to scaffold the children's design skills and to enable them to have focused discussions about their developing ideas.

The Card Assessment Task also had a "design sheet" that appeared to aid the Focus Class more than the Comparison Class. I believe this was probably because my underlying assumptions about the nature of design, and the way drawing can be used to support it, was similar to that of the designers of the sheet. The Comparison Class could not second-guess these underlying assumptions and so were little aided by the sheet. Thus, the provision of photocopiable design sheets will not by themselves enhance children's design drawing capabilities. However, such sheets can be useful for assessment, as the cues on the sheets trigger responses based on understanding of underlying processes.

• Encourage children to discuss their work as they draw. This aids clarification of their ideas, peer-checks that ideas answer the task and sparks off new good ideas. Communication clarifies thought and drawing enables communication. It is not just to place-hold personal good ideas. From the start of the Programme, the Focus Class were encouraged to discuss the ideas they had drawn with one another and the group-work in the final term explicitly required the children to produce drawn plans that had been produced in discussion. The contrast between the two classes' approach to the Maze task centred on the difference in the role and time of talk.

Part of teachers' problems with group-work and discussion is related to assessment: how will we know what a child can do if half of their ideas come from their friends? I would counter this with another question: Is education about teaching and learning or testing and tracking? The emphasis on accountability in the current political climate has become detrimental to collaborative learning. All learning takes place in a social context and is improved by working

with, rather than against, the social needs of children. Knowledge and understanding is deepened by observation and questioning, bouncing ideas back and forth, parading possibilities before each other's eyes and having castles in the air built up or knocked down.

Constructing a group solution to a design problem requires greater facility with communication skills than constructing one's own solution. In the marble run activity (Teaching Input, Summer term) the least successful products came from pairs who divided the board "your half / my half" and the most socially adept pair (Randal and Carl) continued building, adapting and re-inventing their construction long after the photo-shoot that was intended to end the activity. The skills of joint decision-making, negotiating, and being able to allow others to critique one's ideas are essential high-level work-force skills. Writers of future Design and Technology curricular guidance should take steps to ensure that such essential skills are fostered (but not as another list of targets to achieve).

Model the use of drawing for a range of purposes, including designing. Children often only see teachers writing, which, despite being surrounded by multi-media texts on-screen as well as on paper, teaches them that this is the most highly-valued form of communication for educational success. Teachers should use charts and diagrams, concept webs, cartoons, story-boards, colour and pattern, to make the interaction of text and graphics appear natural and useful. Teachers often feel inhibited drawing in front of children as they feel they are "no good at drawing" but however poor an adult's drawing might be, it is better than a child's, and everyone improves with practice.

Children need to be shown examples of design drawings in order to know what the teacher is talking about. The introduction of the Container / Journey metaphor to the Focus Class was successful because they were shown and asked to discuss features of drawings that demonstrated the development of design ideas to a greater and lesser extent. Suitable examples can be collected from a range of sources, including books intended for the home market, television programmes and videos. All these can be a rich source of encouragement to children's imaginations and enthusiasm for making things and can be used to show children that ideas can be developed through drawing. The role of parents in encouraging their children's design skills needs to be affirmed.

Children need to see adults using drawing to support thinking in a range of circumstances, both within school and at home. Parents frequently believe that linear text, "writing", is the form of communication that they should encourage children to practice at home, since this was the preferred medium in their own school experience. They also have limited knowledge of drawing genres and, although surrounded by graphic messages, do not see the need to help children

read them and create their own. Parents' Information Evenings to encourage parents to help children use a range of media for developing thinking will probably need to be given titles such as "Extending the Boundaries of Literacy" in order to attract a large audience.

• View the whole of knowledge, understanding and learning as holistic. Divisions into subject areas are convenient organisational devices but Design and Technology does not neatly sit within the classic division of arts vs. sciences but inconveniently and creatively straddles both. In teaching children to use drawing to support design thinking, a tool is being provided that can be used across the curriculum. Teaching young children to plan ahead of engagement with materials in a design context can enhance their thinking in all forms of writing, encourage consideration of cause and effect when planning science investigations, as well as providing understanding of the rationale behind making preliminary sketches prior to painting, collage or 3-dimensional art-work. Since I taught my Focus Class for only one afternoon a week, I was unable to make these cross-curricula connections. However, Mrs. R., their class teacher, maintained that they were better able to plan their stories than other Year 2 children she had taught, and attributed this to the design drawing input.

I believe strongly in a joined-up, rather than dis-jointed curriculum for young children, in using a wide range of teaching strategies to engage the interest and incorporate the learning styles of all pupils. An adjunct to this is that a range of recording methods and styles should also be encouraged and that children should be taught to choose appropriately from as wide a range of techniques and media as possible, of which drawing is one of such multiple possibilities.

This cannot be a matter for individual teachers to address since it requires the active support of school management to do so. A radical restructuring of any aspect of the Primary curriculum is difficult for individual schools to undertake alone, firstly, because children need to transfer to Secondary Schools with similar levels of understanding and knowledge, and, secondly, innovation has to be successful in order to maintain the school's profile. It is too big a risk to introduce radical innovative pedagogy. Local clusters of schools, therefore, need to work together, with the support of Local Education Authority advisory staff, to identify the holistic design skills that all children should acquire by transfer age. These can then be promoted across the curriculum.

For policy writers of the future (and those who write curricula based on those policies) the time has come to stop further fragmenting of the Primary curriculum in terms of subject areas. A better way forward would be to build on the identification of Key Skills, of which design capability would be one, and allow development of curricula at local level.

6.3.3 Developing Children's Design Capability

Design and Technology should not be something children do on Friday afternoons once the real work of the week is over. Designing and creating technology (as human activity rather than school subject) is, I believe, part of that which distinguishes humans from all other species. Human civilisation is a designed, technological system, in which its members depend on the technological solutions of other members, both past and present, in order to function as part of that system. Creative designers, clever technologists and system analysts are essential to the maintenance of human society and the more complex and more technology-dependent that society becomes, then the more of these people are needed.

To push Design and Technology education into the Friday afternoon slot, alternating half-termly with Art, and to put a rigidly defined curriculum for literacy and numeracy skills in the front line as the measure of school success, is creating a generation ill-equipped to solve the complex technological problems of the designed world of the future. The principle of every child's entitlement to Design and Technology education from first entry to school was a bold move in the design of the National Curriculum, despite the lack of consultation with Key Stage 1 teachers. The current statement of Design and Technology curriculum content is an equally bold and brave statement that needs translating into creative and exciting practice. Not slippers in Year 6 for ever.

"Design and Technology prepares pupils to participate in tomorrow's rapidly changing technologies. They learn to think and intervene creatively to improve quality of life. The subject calls for pupils to become creative and autonomous problem-solvers, as individuals and members of a team. They must look for needs, wants and opportunities and respond to them by developing a range of ideas and making products and systems. They combine practical skills with an understanding of aesthetics, social and environmental issues, function and industrial practices. As they do so, they reflect on and evaluate present and past design and technology, its uses and effects. Through design and technology, all pupils can become discriminating and informed users of products, and become innovators."

(DfEE/QCA 1999: 90)

The research I have conducted addresses the ways to "develop.. a range of ideas" that can enable children to become "creative and autonomous problem-solvers." The use of drawing as a tool for thought development, as a powerful means of modelling ideas, a discussion document, communication system and presentation device, needs to be taught as part of design education.

6.4 Agenda for Future Research

I do not see this as the end of my research. I will continue to research children's design drawings. I have both a theoretical framework and assessment instrument for doing so and I believe that these would stand me in good stead for future work.

I would like to :

• collect and examine older children's drawings, to see whether the Drawing Types persist into Upper Key Stage 2 and how direct teaching of drawing techniques in Secondary Schools affects children's more intuitive use of drawing to support design thinking;

• try a different, shorter Programme encapsulating the Container / Journey metaphor with different aged children in different kinds of schools serving different demographic areas to test the transferability of this way of teaching the purpose of design drawing;

• share my insights with teachers to see if they have similar success to mine through embedding the Container / Journey metaphor in a Programme of their own devising;

• look at other means of modelling design ideas (through construction kits, perhaps) to find out whether my observations about the way in which children use drawing has parallels to other modelling media and whether the Container / Journey metaphor would transfer across media;

 extend the area of discussion and exploration to include using drawing as a tool for supporting and developing thinking in other curriculum areas, including literacy, science and geography, viewing drawing as just one aspect of a multi-media approach to idea development and communication;

 work collaboratively with others, especially those whose research interests are close to my own and with whom I have had long and fruitful discussions during the past five years;

• and finally, to add an international dimension and work collaboratively with those colleagues in other countries with whom I have made acquaintance during the course of this research.

6.5 Drawing as a Tool for Thought

Beginning from this phrase, which seemed to say what I meant by design drawing but barely knowing what I meant by it, through a journey of exploration into the achievements of young children who attended one primary school in rural Kent and on into an attempt to improve the capabilities of just one class, via a search of literature relating to cognition and language development as well as design theory and children's drawing skills, I have learnt a great deal.

I have learnt about researching, much of which seemed parallel to the children's designing: trying things out and observing the results, forming hypotheses about a possible solution and designing a product to satisfy the need.

I have learnt even more about children. Firstly, that children's capabilities should never be underestimated. I have gained a deep respect for the adaptability of young children, their eagerness to participate with earnest enthusiasm, and their desire to learn and do well at any activity that meshes with their cognitive and manual competencies and that captures their imagination. Secondly, I have learnt that capability is not fixed. One cannot say "seven year olds can but six year olds cannot" learn a particular skill or see the world in a particular way. My Focus Class were seeing the design tasks in a very different way to the view of the Comparison Class. The input changed their perception; they learnt a new game called "Taking Ideas on a Journey."

My Focus Class also learnt to enjoy their own success. They learnt that I valued their learning and that I thought well enough of their work to take examples to conferences and to include photographs of it in journal articles. They learnt that I wanted to know more about how they learnt and that they were able to tell me about it. They learnt that their growing reflectiveness on their design processes was important and they wanted to tell me about it and for me to write down what they had to say about what they were doing. It became a two-way process, a joint enterprise in which I could genuinely talk to them of what *we* were doing. They knew we were breaking new ground and were delighted to be part of it.

I recently went to their Leavers' Assembly. It was their last morning at First School and they sang songs and recounted their memories of their first days at school. As each child walked across the stage to endure the Headteacher's humour and collect their Record of Achievement, I pulled up mental images of their design drawings and the resultant products.

As they go forward into their Middle Schools and beyond, I wish them every success in life and will remember them always with gratitude and affection.

REFERENCES

Ackoff, R. (1979) The Art of Problem-solving; New York; Wiley

Adams, J. L. (1974) Conceptual Blockbusting - a Guide to Better Ideas; New York; W.W.Norton & Co.

Anning, A. (1993) Learning Design and Technology in Primary Schools; in McCormick, R., Murphy, P.& Harrison, M. Teaching and Learning Technology; Wokingham; Addison-Wesley Publishing

Anning, A. (1997) The First Years at School; Buckingham; Open University Press

Apple, M. (1979) Ideology and Curriculum; LondOn; Routledge & Kegan Paul

Archer, B. (1992) A Definition of Cognitive Modelling in Relation to Design Activity in Roberts, P., Archer, B. & Baynes, K. *Modelling: the Language of Design; Design Occasional Papers No.1*; Loughborough University of Technology, Department of Design & Technology.

Archer, B. & Roberts, P. (1992) *Design & Technological Awareness in Education* in Roberts, P., Archer, B. & Baynes, K. *Modelling: the Language of Design; Design Occasional Papers No.1*; Loughborough University of Technology, Department of Design & Technology.

Arnheim, R. (1969) Visual Thinking; Berkeley; University of California

Arnheim, R. (1995) Sketching and the Psychology of Design; in Margolin, V & Buchanan, R, (eds) The Idea of Design; Cambridge, MA.; MIT Press

Bailey, E.P.M. (1971) *The Springs of Creativity in the Young Child*; in Slater, F. *Education and Creative Work*; Hull; University of Hull

Ball, S.J. (1993) Self-doubt and Soft Data: Social and Technical Trajectories in Ethnographic Fieldwork in Hammersley, M. (ed.) Educational Research: Current Issues; London; Paul Chapman Publishing

Baynes, K. (1989) The Basis of Designerly Thinking in Young Children; in Dyson, A. Looking, Making and Learning; London; Kogan Page

Baynes, K. (1992) *The Ethics of Representation* in Roberts, P., Archer, B. & Baynes, K. *Modelling: the Language of Design; Design Occasional Papers No.1*; Loughborough University of Technology, Department of Design & Technology.

Baynes, K. (1996) *How Children Choose* in Roberts, P., Archer, B. & Baynes, K. *Learning Design: Occasional Papers No.2*; Loughborough University of Technology, Department of Design & Technology.

Baynes, K. (1998) *Draw-um paper*, in Loughborough College of Art & Design *Drawing Papers*; Loughborough University of Technology, Department of Design & Technology.

Blanshard, B. (1964) The Nature of Thought, Vol.2; Atlantic Highlands, N.J.; Humanities Press;

Boden (1994) What is Creativity? in Boden.M.A. (ed.) Dimensions in Creativity; Cambridge, MA.; MIT Press

Boden, M. (2001) *Creativity and Knowledge* in Craft, A., Jeffrey, B. & Leibling, M. (eds.) *Creativity in Education*; London; Continuum

Booth, W.C. (1978) *Metaphor as Rhetoric*; in Sacks, S. (ed.) *On Metaphor*; Chicago; University of Chicago Press

Bowen, R (1998) Graphic Approaches to Describing Action Research Methodology in Educational Action Research Vol.6 No.3 1998

Brewer, J. & Hunter, A. (1989) *Multimethod Research : a synthesis of styles;* London; Sage Publications

Browne, N. & Ross, C. (1995) Girls as Constructors in the Early Years: promoting equal opportunties in maths, science and technology; Stoke on Trent; Trentham.

Bruner, J. (1962) On Knowing; Cambridge, MA; Harvard University Press

Bruner, J. (1966) Towards a theory of instruction; Cambridge, Mass; Harvard University Press

Bruner, J (1985) Narrative & Paradigmatic Modes of Thought in Eisner, E. (ed.) Learning & Teaching the Ways of Knowing; NSSE

Bruner, J (1986) Actual Minds, Possible Worlds; Cambridge, MA; Harvard University Press

Buchanan, R. (1995) *Wicked Problems in Design Thinking* in Margolin,V & Buchanan, R. (eds) *The Idea of Design*; Cambridge, MA.; MIT Press

Campaign for Drawing (2002) Start Drawing; London; NSEAD

Campbell, D. & Stanley, J. (1963) *Experimental and Quasi-experimental Designs for Research on Teaching* in Gage, N. (ed.) *Handbook of Research on Teaching;* Chicago; Rand McNally

Cannon, W.B. (1976) *The Role of Hunches in Scientific Thought* in Rothenberg, A. & Hausman, C.R. (eds.) *The Creativity Question*; Durham, N.C.; Duke Unversity Press

Chadwick, E. (1993) *Linking Science and Technology in the Primary School* Conference Proceedings IDATER 93; Loughborough University, Department of Design and Technology

Chalkley, C. & Sheild, G. (1996) Supermodelling! Developing Designing Skills at Key Stage 2 in The Journal of Design and Technology Education Vol.1. No.1; Wellesbourne, Warwickshire; DATA

Chidgey, J. (1994) A Critique of the Design Process in Banks, F. (ed.) Teaching Technology; Milton Keynes, Buckinghamshire; Open University Press

Ching, J & Hulsbosch, M. (2001) *Developing Design;* in Conference Proceeding, Third International Primary Design & Technology Conference (CRIPT); Birmingham

Clement, R (1992) The Classroom Reality of Drawing in Thistlewood, D. (ed.) Drawing; Research and Development; Harlow; Longmans

Coghill, V. (1989) Making and Playing, the Other Basic Skills; in Dyson, A. Looking, Making and Learning; London; Kogan Page

Constable, H. (1994) A Study of Design and Technology Capability at Key Stages 1 & 2; in *Conference Proceedings IDATER94*; Loughborough University of Technology, Department of Design and Technology

Craft, A. et al (1997) Can you teach Creativity? Nottingham; Education Now

Craft, A. (2001) "Little c Creativity" in Craft, A., Jeffrey, B. & Leibling, M. (eds.) Creativity in Education; London; Continuum

Creswell, J. (1994) Research Design: Qualitative and Qualitative Approaches; Sage; Thousand Oaks

Cross, Naughton and Walker (1986) *Design Method & Scientific Method* in Cross,A. & McCormick, R. (eds.) *Technology in Schools*; Milton Keynes, Buckinghamshire; Open University Press

Cupchik, G. (2001, February). Constructivist Realism: An Ontology That Encompasses Positivist and Constructivist Approaches to the Social Sciences in Forum Qualitative Sozialforschung / Forum: Qualitative Social Research 2(1) (On-line Journal); http://www.qualitative-research.net/fqs-texte/1-01/1-01cupchik-e.htm

Dadds, M. (2002) Research Based Professionalism in Educational Action Research

Denton, H.G. (1993) The Design and Make Task (DMT): some reflections on designing in schools; in Conference Proceedings IDATER 93; Loughborough University, Department of Design and Technology

Denzin, N. (1970). *The Research Act: A Theoretical Introduction to Sociological Methods* (2nd edition); New York; McGraw Hill.

Denzin, N.K. & Lincoln, S.Y. (2002) The Qualitative Enquiry Reader; CA, Thousand Oaks; Sage Publications.

DfEE/QCA (1999) The National Curriculum: Handbook for Primary Teachers in England, Key STages 1 & 2; London; Department for Education and Employment / Qualifications & Curriculum Authority

Donaldson, M. (1978) Children's Minds; Glasgow; Fontana

Donaldson, M. (1992) Human Minds; Harmondsworth, Middlesex; Penguin Books

Duckworth, E. (1987) "The Having of Wonderful Ideas" and Other Essays on Teaching and Learning; New York, Teachers College Press

Dunn, S. & Larson, R. (1990) Design Technology: Children's Engineering; London, The Falmer Press

Edwards, C., Gandini, L. & Forman, G. (eds.) (1993) *The Hundred Languages of Children: The Reggio Emilia Approach to Early Childhood Education*; Norwood, NJ; Ablex Publishing Co.

Egan, B.A. (1995) How do Children Perceive the Activity of Drawing? Some Initial Observations of Children in an Infant School; Conference Proceedings IDATER 95; Loughborough University, Department of Design and Technology

Egan, B.A. (1996) *Purposes in drawing: The significance of children's personal styles for design & technology*; in *Conference Proceedings IDATER 96*; Loughborough University, Department of Design and Technology

Egan, B.A. (1999) Children Talking about Designing: a Comparison of Year1 & Year6 Children's Perceptions of the Purposes/uses of Drawing as Part of the Design Process; in Conference Proceedings IDATER 99; Loughborough University of Technology, Department of Design & Technology.

Egan, B.A. (2001) Constructing Ideas through Drawing; Conference Proceedings CRIPT 2001; Birmingham; UCE

Eisner, E. (1972) Educating Artistic Vision; London; MacMillan

Eisner, E. (1993) *Objectivity in Educational Research* in Hammersley, M. (ed.) *Educational Research: Current Issues*; London; Paul Chapman Publishing

Elliott, J. (1991) Action Research for Educational Change; Milton Keynes; Open University Press

Ely, M., Vinz, R., Downing, M. & Anzul, M. (1997) On Writing Qualitative Research; Living by Words; London; Falmer

Eng, H. (1931) The Psychology of Children's Drawings; London; RKP

Fielding, N. & Fielding, J. L. (1986). *Linking data* in Qualitative Research Methods, Vol.4; London; Sage.

Fielding, N. & Schreier, M. (2001). *Introduction: On the Compatibility between Qualitative and Quantitative Research Methods* in *Forum Qualitative Sozialforschung / Forum: Qualitative Sozial Research*; http://www.qualitative-research.net/fqs-texte/1-01/1-01hrsg-e.htm (accessed February 2003)

Fisher, F. (1990) Teaching Children to Think

Flick, U. (1998) An Introduction to Qualitative Research; London; Sage.

Foray, D. & Hargreaves, D. (2003) *The Production of Knowledge in Different sectors: a Model and some Hypotheses*; London Review of Education, Vol.1, No.1; London; Institute of Education & Carfax Publishing

Freeman, N. (1980) Strategies of Representation in Young Children; London; Academic Press

Fulton, J. (1990) *Design in Primary Education: the Name of the Game* in Thistlewood, D. (ed.) *Drawing; Research and Development;* Harlow; Longmans

Gardner, H. (1984) Frames of Mind ; London; Heinemann Educational Press

Gardner, H. (1993) The Unschooled Mind; Glasgow; Fontana

Gardner, H. & Wolf, D. (1987) *Symbolic Products of Early Childhood* in Gorlitz, D & Wohlwill, J.F. (eds.) *Curiosity, Imagination and Play*; Hove, East Sussex; Laurence Erlbaum Associates

Garner, S. (1992) The Undervalued Role of Drawing in Design in Thistlewood, D. (ed.) Drawing: Research and Development; Harlow; Longmans

Garner, S. (1993) A Comparison of Methods for Researching into Drawing within the Field of Design; in Conference Proceedings IDATER 93; Loughborough University, Department of Design and Technology

Garner, S. (1993) *The Importance of Graphic Modelling in Design Activity in Schools* in McCormick, R., Murphy, P. & Harrison, M. (eds.) *Teaching and Learning Technology*; Wokingham; Addison-Wesley Publishing

Garvey, J & Quinlan, A.(1997) *Why Don't I Just Throw it in the Bin? - Evaluation and Self-esteem*; in The Journal of Design & Technology Education Vol.2.No.1; Wellesbrough, Warwickshire; The Design and Technology Association

Gentner, D. (1982) Are Scientific Analogies Metaphors?; in Miall, D. Metaphor: Problems and Perspectives; Hassocks; Harvester

Gick, M.L. & Holyoak, K. (1985) *Analogical Problem Solving*; in Aikenhead, A.M. & Slack, J.M. (eds.) *Issues in Cognitive Modelling*; Milton Keynes, Buckinghamshire; Open University Press

Gitlin, A., Seigel, M.& Boru, K. (1989) *The Politics of Method: from Leftist Ethnography to EducativeResearch;* in Hammersley, M. (ed.) *Educational Research: Current Issues;* London; Paul Chapman Publishing Ltd

Goldschmidt, G. (1994) On Visual Design Thinking: the Vis Kids of Architecture; in Design Studies Vol.15.No.2; London; Butterworth

Goodenough, F.L. (1926) Measurement of Intelligence by Drawings; New York; World Book Co.

Goswami, U. (1992) Analogical Reasoning in Children; Hove, East Sussex; Laurence Erlbaum Associates

Gordon, W.J.J. (1976) *Metaphor and Invention* in Rothenberg, A. & Hausman, C.R. (eds.) *The Creativity Question*; Durham, N.C.; Duke Unversity Press

Goswami, U. (1992) Analogical Reasoning in Children; Hove, East Sussex; Laurence Erlbaum Associates

Green, S. (2000) Analogy and human sense: an equiry into the nature and development of analogical thinking with a consideration of its educational potential; (unpublished M.A.) University of London; Goldsmiths College, Department of Education

Guilford, J.P. (1970) *Traits of Creativity* in Vernon, P.E. (ed.) *Creativity*; Harmondsworth, Middlesex; Penguin Books

Halford, G. (1992) Analogical reasoning & Conceptual Complexity in Cognitive Development; in Human Development 1992; 35; p.193-217

Hammersley, M. (1991) On the Teacher as Researcher in Hammersley, M. (ed.) Educational Research: Current Issues; London; Paul Chapman Publishing Ltd

Hammersley, M. & Atkinson, P. (1983). Ethnography: Principles in Practice; London; Tavistock

Hammond, S. (1997) Acknowledging the Signposts in Young Children's Writing Development unpublished Postgraduate Diploma paper; Canterbury; Canterbury Christ Church University College

Hammond, S. (1999) *Investigating Young Children's Interactions During Independent Group Work* (unpublished M.A. dissertation); Canterbury; Canterbury Christ Church University College

Harrison, A. (1978) Making & Thinking; Hassocks; Harvester

Harste, J.C., Woodward, V.A., & Burke, C.L. (1984) *Language Stories & Literacy Lessons*; London; Heinemann Educational Press

Heisenberg, W. (1958) Physics and Philosophy; New York; Harper Bros.

Hennessy, S., McCormick, R. & Murphy, P. (1993) *The Myth of the General Problem-solving Capability; Design and Technology as an Example*; in The Curriculum Journal Vol.4 No.1 Spring 1993

Hodgkin, R.A. (1985) Playing and Exploring; London; Methuen

Hope, G. (2000a) *Beyond "Draw one & Make it"* Conference Proceedings, Design and Technology International Millennium Conference (DATA); London

Hope, G. (2000b) Beyond their Capability: Drawing, Designing and the Young Child; The Journal of Design and Technology Education, Vol.5, No.2; Wellesbourne; DATA

Hope, G. (2000c) *Why Draw Anyway? The Role of Drawing in the Child's Design Tool Box*; Conference Proceedings International Design and Technology Educational Research Conference (IDATER 2000); Loughborough

Hope, G. (2001a) *Taking Ideas on a Journey called Designing*; The Journal of Design and Technology Education, Vol.6. No.3; Wellesbourne; DATA

Hope, G. (2001b) *Participant Research from the Point of View of a Participant Researcher;* Conference Proceedings, International Design and Technology Education Research Conference (IDATER 2001); August 2001, Loughborough.

Hope, G. (2003) The Process of Solving Problems: young children exploring the rules of the game in science, mathematics and technology in The Curriculum Journal

Irons, M.L. (2000) http://www.rdrop.com/~half/Creation/Puzzles/Analogies/index/html

Johnsey, R. (1995) *The Place of Process Skill Making in Design and Technology, Lessons from Research into the way Primary Children Design and Make*; in *Conference Proceedings IDATER 95*; Loughborough University, Department of Design and Technology

Judd, C.M., Smith, E.R. & Kidder, L.H. (1991) Research Methods in Social Relations, 6th *Edition*; Florida; Hracourt Brace Jovanovich, Inc.

Jung, C.G. (1966) The Collected Works of J.C.Jung; London; Routledge & Kegan Paul

Kahneman, D., Slovic, P. & Tversky, A. (eds.) (1982) *Judgement under Uncertainty: Heuristics and Biases*; Cambridge; Cambridge University Press

Kahney, H. (1993) *Problem Solving - Current Issues*; Milton Keynes, Buckinghamshire; Open University Press

Kellogg, R. (1959) What Children Scribble and Why; San Francisco; NP Publications

Kemmis, S. (1988) Action Research; in Keeves, J.P. (ed) Educational Research Methodology and Measurement: an International Handbook; Oxford; The Pergamon Press

Kemmis, S. (1993) Action Research; in Hammersley, M. Educational Research: Current Issues; London; Paul Chapman Publishing Ltd

Kimbell, R. (1986) Craft, Design and Technology; Milton Keynes; Open University Press

Kimbell, R. (1998) *Design Thinking for Learning*; Paper given at the World Design Summit, North Carolina State University, 1-3 May 1998

Kimbell, R. (2000) *Creativity in Crisis*; in The Journal of Design & Technology Education, Vol.5, No. 3; Wellesbourne; DATA

Kimbell, R. & Perry, D. (2001) *Design and Technology in a Knowledge Economy*; London; Engineering Council.

Kimbell, R., Stables, K. Wheeler, T. Wosniak, A & Kelly, V.(1991) *The Assessment of Performance in Design and Technology*; London; Schools Examinations and Assessment Council

Koutsides, G. (2001) Using Co-operative Learning in Design & Technology in The Journal of Design and Technology Education, Vol.6, No.1; Wellesbourne; DATA

Koestler, A. (1974) *The Act of Creation*; London; Huchinson Kress, G (1997) *Before Writing: Rethinking the Paths to Literacy*; London; Routledge

Kress, G (1994) Learning to Write; London; Routledge

Kuhn, T.S. (1962) The Structure of Scientific Revolutions; Chicago; Chicago University Press

Lakoff, G. & Johnson, M. (1980) Metaphors We Live By; Chicago; University of Chicago Press

Lave, J. & Rogoff, B. (1984) *Everyday cognition: it dev in social context;* Chicago; University of Chicago Press

Lave, J. (1988) Cognition in Practice; Cambridge, Cambridge University Press

LeCompte, M.D., & Preissle, J. (1993) *Ethnography and Qualitative Design in Educational Research*; San Diego; Academic Press.

Liddament, T. (1991) *Design Talk;* in Design & Technology Teaching Vol.23. no.2; Trentham Books

Liddament, T. (1994) Technological Literacy: from Function to Meaning; in Conference Proceedings IDATER 94; Loughborough University, Department of Design and Technology

Liddament, T. (1996) *The Metamorphosis of the Design Vocabulary*; in Design Studies; Hove, Sussex; Elsevier Science Ltd.

Lieberman, J. N. (1977) Playfulness; New York; Academic Press

Light, P. & Perry-Clermont, A-N. (1990) Social Context Effects in Learning and Testing; in Light, P., Sheldon,S. & Woodhead,M. (eds.) Learning to Think: Child Development in a Social Context Vol.2; London; Routledge/The Open University

Lowenfeld, V. (1947) Creative and Mental Growth; New York; Macmillan

Lowenfeld, V (1952) The Nature of Creative Activity; London, RKP

Mantell, J. (1999) Teaching Designing Skills at Key Stage 2: Is there a Role for Techniques? in Conference Proceedings CRIPT 1999; Birmingham

Mantell, J. (2000) Inverstigating how Children Use Language as a Tool of Thinking in Design and Technology at Key Stage 2; in Conference Proceedings Design and Technology International Millenium Conference; London

Margolis, V & Buchanan, R (eds) (1996) The Idea of Design; Cambridge, MA.; MIT Press

Margolis, V. (1989) Design Discourse; Berkeley; University of California

Matthews, J. (1999) The Art of Childhood and Adolescence: the Construction of Meaning; London; Falmer

McCormick, R., Murphy, P. & Davidson, M. (1994) *Design and Technology as Revelation and Ritual* in *Conference Proceedings IDATER94*; Loughborough University, Department of Design and Technology

McCracken, R.A. & McCracken, M.J. (1985) *Reading is only the Tiger'sTail*; Winnipeg, Canada; Pegius Publishers Ltd.

McGraw, G. & Hofstadter, D. (2000) http://www.cogsci.indiana.edu/farg/mcgrawg/spirit.html (accessed May 2002)

McNiff, J. (1988) Action Research: Principles and Practice; London; MacMillan Education Ltd

Middleton, H. (2000) *Design and Technology: What is the Problem?;* in Conference Proceedings, Design & Technology International Millennium Conference; April 2000; London

Morgan. G. (1983) Research Strategies in Morgan. G. (ed.) Beyond Method; Sage; Beverley Hills

Moyles, J. (ed) The Excellence of Play; Milton Keynes, Buckinghamshire; Open University Press

Newell, A. & Simon, H.A. (1972) Human Problem Solving; New Jersey; Englewood Cliffs.

Newell, R.W. (1986) Objectivity, Empiricism and Truth; London; Routledge & Kegan Paul

Nietzche, F. (1968) The Will to Power; New York; Vintage

Outterside, Y.R. (1993) The Emergence of Design Ability: The Early Years; in Conference Proceedings IDATER 93; Loughborough University, Department of Design and Technology

Outterside, Y.R. (1994) The Emergence of Scientific and Technological Awareness in the Early Years; in Conference Proceedings IDATER 94; Loughborough University, Department of Design and Technology

Outterside, Y.R. (1996) Can Young Children Mentally Rotate an Image on a 3D Block and Consequently make a Prediction Based on this Mental Rotation? in Conference Proceedings IDATER 96; Loughborough University, Department of Design and Technology

Oxman, R. (1997) *Design by Re-representation: a Model of Visual Reasoning in Design*; in Design Studies No.18; Hove, Sussex; Elsevier Science Ltd

Oxman, R. (2001) Design Emergence in Design Studies; Hove, Sussex; Elsevier Science Ltd.

Pascal, C. & Bertram, C. (1989) Evaluating and Improving the Quality of Play; in Moyles, J. Just Playing?; Milton Keynes, Buckinghamshire; Open University Press

Piaget, J. & Inhelder, B. M (1971) *Mental Imagery and the Child* (trans. Chilton, P.A.) London; RKP

Piaget, J. & Inhelder, B. M (1969) *Play, Dreams and Imitation* (trans. Gattegno, C. & Hodgeson, F.M.) London; RKP

Polanyi, M. (1958) Personal Knowledge; London; RKP

Proctor, N. (1986) *Educational Psychology and Curricular Design: a Child-centred Approach* in Cross, A. & McCormick, R. (ed.) *Technology in Schools*; Milton Keynes, Buckinghamshire; Open University Press

Pye, D. (1964) Nature & Aesthetics of Design; London; Barrie & Jenkins

Qualifications and Curriculum Authority (QCA) / Department for Education and Employment (DfEE) (1999) Design and Technology - A Scheme of Work for Key Stages 1 and 2; London; QCA

Ritchie, R. (1995) Primary Design and Technology; London; David Fulton Publishers

Rittel, H. & Webber, M.M. (1974) Wicked Problems; London; Hutchinson & Co.

Roberts, P. (1992) Of Models, Modelling and Design: an Applied Philosphical Enquiry; in Roberts, P., Archer, B. & Baynes, K. Modelling: the Language of Design; Design Occasional Papers No.1; Loughborough University of Technology, Department of Design & Technology.

Roberts, P. (2001) Opening Address to IDATER2000 in Norman, E. & Roberts, P. (eds.) Design and Technology Educational Research and Curriculum Development: The Emerging International Research Agenda; Loughborough; Loughborough University

Robinson, G. (1995) Sketch Books - Explore and Store; London, Hodder and Stoughton

Roden, C. (1995) Young Children's Learning Strategies in Design Technology; in Conference *Proceedings IDATER* 95; Loughborough University, Department of Design and Technology

Roden, C. (1999) *How Children's Problem Solving Strrategies Develop at Key Stage 1* in The Journal of Design & Technology Education Vol.2 No.1; Wellesbourne, Warwickshire; Design and Technology Association

Roden, C. (1997) Young children's Problem-solving in Design & Technology: towards a *Taxonomy of Strategies* in The Journal of Design & Technology Education Vol.4 No.1; Wellesbourne, Warwickshire; Design and Technology Association

Rogers, M. & Clare, D. (1994) *The Process Diary: Developing Capability within National Curriculum Design and Technology - some initial findings;* in *Conference Proceedings IDATER94*; Loughborough University of Technology, Department of Design & Technology

Rogers, M & Stables, K. (2001) Providing Evidence of Capability in Literacy and Design and Technology in both Year 2 and Year 6 children; in Conference Proceedings CRIPT 2001; Birmingham

Rogoff, B. (1996) Developmental Transitions in Children's Participation in Socio-cultural Activities; in Sameroff, A and Haith, M The 5-7 Year Shift; Berkeley; University of California

Ryle, G. (1949) The Concept of Mind; London; Huchinson.

Sameroff, A. & Haith, M. (eds) (1996) The Five to Seven Year Shift; Berkeley; University of California

Samuel, G.C. (1991) They Can Never Make What They Draw; in Conference Proceedings DATER91; Loughborough University of Technology

Schofield, J.W. (1993) Increasing the Generalizability of Qualitative Research; in Hammersley, M. Educational Research: Current Issues; London; Paul Chapman Publishing Ltd

Schomburg, R. (2000) Using Symbolic Play Abilities to Assess Academic Readiness

Schon, D.A. (1987) Educating The Reflective Practitioner; San Francisco, CA; Jossey-Bass

Scrivener, S. (1998) *untitled* in Loughborough College of Art & Design *Drawing papers;* Loughborough University of Technology, Department of Design and Technology

Selfe, L. (1983) Normal & Anomalous Representational Drawing Ability in Children; London; Academic Press

Sieber, S. (1979). The Integration of Fieldwork and Survey Methods in American Journal of Sociology, 78(6), 1335-1359.

Smith, F. (1992) To Think; London; Routledge

Smith, J.S. (2001) The DATA Lecture: The Current and Future Role of Modelling in Design and Technology; The Journal of Design and Technology Education, Vol.6, No.1; Wellesbourne; DATA

Snook, B. (1974) Making Birds, Beasts and Insects; London; BT Batsford Ltd.

Stables, K. (1992a) Issues Surrounding the Development of Technology Capability in Children in their First Years of Schooling (ages 5-7); in Conference Proceedings INCOTE92; Weimar, Germany

Stables, K. (1992b) *The Role of Fantasy in Contextualising and Resourcing Design and Technology Activity*; in *Conference Proceedings IDATER92*; Loughborough University, Department of Design and Technology

Stables, K. (1993) Who are the Clients in School-based Design & Technology Projects?; in Conference Proceedings IDATER 93; Loughborough University, Department of Design and Technology

Stables, K. (1997) Out of the Mouth of Babes: Creative Thinking and Children Designing; in Dingli.S. Creative Thinking: Towards Broarder Horizons; Malta; University of Malta

Stables, K., Rogers, M., Kelly, C. & Fokias, F (2001) *Enriching Literacy through Design and Technology Evaluation Project;* London; Goldsmith Colege; University of London

Stenhouse, L. (1975) An Introduction to Curriculum Research & Development; London; Heinemann

Taylor, S. (1982) The Availability Bias in Social Perception and Interaction in Kahneman, D., Slovic, P. & Tversky, A. (eds.) Judgement under Uncertainty: Heuristics and Biases; Cambridge; Cambridge University Press

Thistlewood, D. (1992) *Observational Drawing and the National Curriculum* in Thistlewood, D. (ed.) *Drawing; Research and Development*; Harlow; Longmans

Tipping, C. (1983) Acquiring Design Skills for Teaching - A self-help Suggestion. Studies in Design Education; Craft and Technology, 16, 1, 12-14

Torrengeau, R. (1982) Metaphor and Cognitive Structure in Miall, D. (ed.) Metaphor: Problems and Perspectives; Brighton; Harvester

Trochim, W. (2000) The Research Methods Knowledge Base; http://trochim.cornell.edu/kb; (accessed June 2003)

Tversky, A. & Kahneman, D. (1982) Belief in the Law of Small Numbers in Kahneman, D., Slovic, P. & Tversky, A. (eds.) Judgement under Uncertainty: Heuristics and Biases; Cambridge; Cambridge University Press

Van Sommers, P. (1984) Drawing and Cognition; Cambridge, MA: Harvard University Press

Veale, T. (1999) *Conceptual Blending*; http://www.compappdcu.ie/~tonyv/papers/CogSci.ps,gz (accessed Jan.2002)

Vernon, M.D. (1962) The Psychology of Perception; Harmondsworth, Middlesex; Penguin Books

Vygotsky,L.S. (1971) The Psychology of Art; Cambridge, Mass.; The MIT Press

Welch, M. & Lim H.S. (1999) *From Stick Figure to Design Proposal: Teaching Novice Deisners to "think on paper";* Conference Proceedings, Second International Primary Design & Technology Conference (CRIPT); Birmingham.

Welch, M. & Mueller, A. (2002) *Professional Development asd an Ongoing Conversation*; in Conference Proceedings, DATA International Research Conference, July 2002; Wellesbourne; DATA

Wells, G. (1986) *The Meaning Makers: children learning language and using language to learn;* London; Hodder & Stoughton

Williams, P.H.M. (1985) Teaching Craft, Design and Technology 5-13; London; Routledge

Wilson, B. (1992) *Primitivism, the Avant-garde and the Art of Little Children*; in Thistlewood, D. (ed.) *Drawing; Research and Development*; Harlow; Longmans

Winner, E. (1982) Invented Worlds; Cambridge, MA: Harvard University Press

Winnicott, D.W. (1971) Playing and Reality; London; The Tavistock Press

Wittgenstein, L. (1969) Philosphical Investigations; the Blue Book; The Brown Book; Oxford, Basil Blackford

Wray, D., Bloom, W. & Hall, N. (1994) Literacy in Action; London, The Falmer Press

Wray, D. & Lewis, M. (1997) *Extending Literacy: children reading and writing non-fiction;* London; Routledge

APPENDICES

Appendix A: Taxonomy of Design Drawing	348
Appendix B: Programme Lesson Plans	350
Appendix C: Target Groups : Pizza	366
Appendix D: Powerpoint Presentation of Snook's Animals	.368
Appendix E: Target Groups : Frosty	369
Appendix F: Focus Class Target Group: Stan series	371
Appendix G: Target Groups : Easter Egg Holder	.372
Appendix H: Easter Egg Holder : all Products	.374
Appendix I: Target Groups : Surprise Card	378
Appendix J: Focus Class : Sandals Project	382
Appendix K:Target Groups : Pandy's Suitcase	.384
Appendix L: Target Groups : Maze	.386
Appendix M: Log Book	.388
Appendix N: Development of Quantitative Assessment Instrument	398
Appendix O: Quantitative Data	.412
Appendix P: Published works arising from Thesis	.425

Gill Hope (2004) Drawing ŝ Û Tool for Thought

YR.1 YR.2 YR.3 YR.4 PURPOSE drawing as "finished product" stan Duncun93 Lowenfeld8Cox92 Fulton90 Paines92 stan96/panda98 stan stan comm'n to others/aide mem for ch Wolverson@Denton91 INVESTIGATING NEED perspective taking panda panda panda;insects Stables93 simple prediction orange juice in bottles conation task Silver78 forward planning stan;panda98 stan;panda;box stan;panda;box;fans stan;panda;box;art cl Scrivener98 CONCEPT DESIGN insect kits brainstorming conv/div;food; boats stan; panda;4Sclay generation of ideas (multi draw) stan:panda stan;panda Ritchie95 stan deciding what to make boats art club 2B buggies; rnd stans recording of the design stage insects;4H rndstan **DETAIL DESIGN** insect kits panda Gardner, I. Garvey&Quilan investigating production method panda panda panda;fans Dunn90 Dunn90 deciding on production method panda - Sam Girt fans recording production method box:insects box box sequencing production food prod seq fans dr w respect to materials stan; buggies;box stan; fans;box stan; panda;box stan; wiggly worm produce product specification fans insect kits TYPES OF DRAWING analogue/iconic/symbolic Alex at age 4 Outterside@Costall85 Baynes92 Eisner symbolic (stereotypical) Stan: worm:house Stan: house Stan:house Stan:house OuttersideSWilson92 Willats85 manip of inner image Fulton90 Outterside96 box box box drawing in the style of ... Glen Parody's Klee Winner82 Sam's Van Gogh; Rhys' Titanic. labelled diag buggies; box; panda panda: box: fan stan:panda:box:art cl Constable94 stans;old sch; house;Miccstan;house;panda98/9 Karmiloff-Smith90 canonical drawing viewpoints; old sch. house; stan; panda Aitchenson(Crook85 Ifa lego;panda; art cl:4S clay:panda Constable Singram85 2D & 3D 3D shapes: old school stan buggies plan drawing/viewpoints panda;box panda;box Constable9Willats92 Dunn90 panda:box appropriate level of detail Van Sommers84 stan stan stan stan **DRAWING & MAKING** make what they draw stan; panda stan; panda stan; panda stan; panda Anning93(xMcCormick Johnsey95 Prentice89 Samuel insects Dunn90p133 pattern development recording of finished product panda; Aaron's drawing of panda panda panda Samuel

LITERATURE

Apriles (File

TAXONOMY OF APPLICATION OF DRAWING IN THE DESIGN PROCESS TAXO-1 What they could/could not produce

EVIDENCE FROM TASKS

USES OF DRAWING

TAXONOMY OF CONTRIBUTARY FACTORS IN THE USE OF DRAWING FOR DESIGN			TAXO-2 - Why they could/could not produce it			
USES OF DRAWING	EVIDENCE FROM TAS YR.1	KS YR.2	YR.3	YR.4	LITERATURE	
COGNITIVE DEVELOPMENT Age/maturity Other cognitive factors	stan;panda;box extreme examples	stan;panda;box extreme examples	stan;panda;box extreme examples	stan;panda;box extreme examples	RitchieV23 Rogoff96 Piaget etc Dyslexia & autism studies	
ADULT INPUT Nature of tasks Explanation given Amount of help given	stan;panda98;worm round stan;old sch insects97;stan96/8	stan;panda;box round stan 2Bflat stan	stan;panda;box;fan old sch 3Wflat stan;3Whouses	stan;panda;box;art cl round stan	Buchanan9Bun&Kiss9hennessey et al 93 Bruner etc	X0
KNOWLEDGE BASE Previous experience w materials Previous exp of designing Knowledge of techniques Visual literacy	Yr1&insects	2P stan scores? 2B labelling - buggy	3W measuring - panda	spirals 4E - panda & box res	Black93 N.Smith 79 Constable94 Green71 Kress&VanLeuwen96	onomy
GRAPHICACITY Ability to draw Understand drawing as proc tool Use dr to dev thoughts & ideas Partial occlusion	all tasks stan;panda98;worm stan viewpoints;old sch	all tasks stan;panda;box stan;panda	all tasks stan;panda;box;fan stan;panda Id sch;Saxon houses-Mid	ali tasks stan;panda;box;art cl stan;panda cca	Arnheim69 Freemna etWillats79 Egan95&6 Garvey&Quinlan Morra	of Desi
MODELLING Understand rel betw dr & make Fluency in symbol manip Use dr to dev sol'n in other med Use metaphor & analogy Reality/fantasy Interact with the drawing	stan;panda98;worm panda	stan;panda box;house rnd stan panda	stan;panda;box;fan box;house rnd stan; zebra crossing panda fans	stan;panda;box;art cł box;house rnd stan;4Sclay;art cł panda art club (discussion)	Archer/Roberts92 Ritchie95 Baynes "Drawum" pape Anning Williams85 Gardner/Wolf87 Clement95 Gardner84 Ritchie95 E.Winner Halford92 Gick&HolycGomrich60 Rumelhart8 Ritchie95 Stables92 & 97 Rogers & CScrivener92 Dunn90p34Oxman97 Liddament9	gn Drawir
IMAGING Make changes from drawing Imagine/record future state Imagine/record future intentions Imagine & record the possible	stan;panda98 stan;panda98;worm stan;panda98;worm stan;panda98;worm	stan;panda box(+stan;panda) box(+stan;panda) box(+stan;panda)	stan;panda box(+stan;panda) box(+stan;panda) box(+stan;panda)	stan;panda box(+stan;panda) box(+stan;panda) box(+stan;panda)	Simonoo Ritchie95 Outterside93 Kimbell98 Stables"Out of the Mou Constable94 Craft77 Weininger88	õ

REL TO OTHER WAYS OF THINKING Linguistic/narrative Creativity/imagination

Page 349 Play

Conation

APPENDIX A Taxonomy of Design J 5a 1931

Bruner62/8!Griffiths35 SueH97 Egan96 Segall91 Koestler74 Bailey71 Hudson66 Boden SuttonSmitlLiebermannPrentice74 Singer&SinBaynes96 Atman

Autumn Term 2000

Assessment Task: Design a pizza

Rationale - A design-and-make activity suitable for children at the beginning of Year2.

Objectives - to use pencil drawing to generate design ideas for a collage of a pizza made from a range of sheet materials, fabrics and small items.

Instructions -

a) discussion of favourite toppings on pizzas and how the toppings are arranged to look appetizing.

b) explanation of using white paper to try out some ideas - up to 3 - and then choose one to make. Progression of ideas across drawings was explained and demonstrated.

c) explanation of making a collage (picture) of the chosen design

Success criteria -

Whether progression of ideas across the planning sheet can be seen (or if each idea stands alone unconnected to what went before) and to what extent the final collage is a reflection of that planning process. It is not expected that the collage will be a carbon copy of one of the plans - some progression of ideas into the making stage is acceptable and desirable - but some children see no connection at all between planning and making.

Assessment procedure - use of drawing to develop design recorded on Holistic Grid and Techniques Ticksheet for each child.

First Teaching Block with the Focus Class; Second half of Autumn Term:

Wagwums & Foozles

Session 1: Figures

Rationale - To encourage the children to make what they have drawn in order to perceive the relationship between drawing and making in the process of designing.

Objectives - To make models of their design drawings using plasticine and matchsticks. It would be teacher-led to begin with until they grasped the idea and were able to produce their own drawings and make models of them.

Activity -

a) The children to be shown the Wagwum picture used for the analogy tests at the end of the first half term, invited to make it from the plastice and matchsticks. The discussion when they had finished would focus on the accuracy of their representation. and the ambiguities in the drawing.

b) Repeat task with Foozle.

c) Design sheet -



Success Criteria - Ability to closely link the drawing to the making

Comments from Log -Showed standard Wagwum figure - asked children to make it. Only about half the children made the figure as shown. Most added bits - saying out loud "I'm doing one with 2 antennae" etc.

I did a "Put your hand up if yours has got ... " - going through the features.

Showed desire to be creative and not accept limitations of problem as set - Donaldson - own agenda more important than mine cf Sue's M.A.

Photographs were taken of the 4 Foozles and Alien(s) for nearly all the children.

Evaluation of Session - Not quite enough time for all children to make all 4 of their own Aliens. I told them to just make their best one for the photo. They were very creative and had all sorts of body shapes.

Session 2: Clothes design

Rationale - To introduce the need to plan ahead and make a pattern of the designed object before making in the real materials.

Objectives - To make a set of clothes for their Alien; involving planning the clothes, choosing the material and making a paper pattern rather than just cutting up the cloth and wrapping it round.

Activity -

a) Previous week's worksheets given out and favourite Alien chosen and made.

b) Design sheet:



Explanation of whole process, with demonstration; each stage written on blackboard:

- Draw Alien dressed in clothes not too complicated.
- Choose cloth for each item of clothing, stick small portion on squares at left, indicate what it is to be used for with lines.
- Make pattern from newsaper and check that it fits.
- Cut out clothing from cloth.

Success Criteria - The instructions are followed & clothing planned and made in paper before cutting into the cloth.

Comments from Log - I had set them to draw the clothes & choose the cloth whilst I gave out newspaper - didn't want to clutter tables too soon. But many children making straight away in cloth -not just choosing their fabrics & then making pattern, so stopped whole class and explained again about why we need to plan - waste of cloth, disappointment when it doesn't fit...

Evaluation of Session - Would possibly have been better to structure the session more tightly all draw, now all choose cloth, now all make pattern etc but would have run the risk of faster workers getting bored and losing their momentum and the slower thinkers being under pressure to complete. Stopping them and explaining again worked. Have observed previously taht repeating instructions once everyone has started is a good move - it makes more sense and they can take it in better once they are engaed on the activity.

Session 3: Transporter

Rationale -To encourage evaluation at planning stage through discussion with partner.

Objectives - Development of single design idea to make a vehicle for plasticine Alien, to be made from recycled materials (plastic cartoons, etc) and range of small items such as matchsticks, cardboard wheels, etc.

Activity -

a) Planning sheet: a single fan-folded strip.

- Section 1 record one good idea for a vehicle.
- Section 2 how this might be made discuss it with neighbour
- Section 3 what a real one would be like discuss it with neighbour
- Section 4 final version to be made, taking discussions with neighbour into account and also the materials each child to be given a carton or box with which to make vehicle. Option given to change the design or swop carton / box.

b) Making the vehicle - range of supporting materials available.

Success Criteria -

Good discussion with partner to explain and justify design choices.

Use of the issues raised in the discussion to inform designing.

Clear progresssion from designing into making.

Comments from Log -

Told to do quick sketch of first idea they had, in first section of paper. Quite a few did 3 on 3 sections of the paper.

Discuss with partner - OK but they tended just to say "I'm going to make a.." without describing it and the partners did not ask questions - eg How, what with?

Then told to imagine it was real, not to be made with boxes etc and to draw it & add details eg doors - then discuss with partner. - didn't really discuss. many just sat there

Finally get box and draw again in light of shape of box - many went straight into making at this stage.

Evaluation of Session -

Will produce a planning sheet for next week with designated sections. Amazing that their reaction was to draw lots when they were told to draw one. At least it shows that the multiple ideas for the pizzas wasn't just a fixation on me saying "Have about 3 attempts". Mrs.M commented afterwards - they don't listen - told to do one but did 3. They don't seem to appreciate they are being given instructions to follow.

Session 4: Wagwum's house

Rationale -To brainstorm ideas and then choose one to develop into a final product, using discussion with partner to aid evaluation.

Objectives - To make a house for the Alien model from recycled materials (boxes) *Activity* - A3 paper - told to fold it in 4 & open out and label the 4 sections as per example on blackboard:

My first ideas	The one I like best
How I will make it	What it looked like

Demonstration - several different houses on 1st section - stereotypical square house with triangle roof as first and said "not this" - then demonstrated different ideas.

Sections 1-3 to be completed. Discussion with partner to decide which idea to be developed from the range resulting from initial brainstorming and also how model is to be made.

b) Making of model from cardboard boxes

c) Completion of section 4 of worksheet.

Success Criteria - Fluency of initial ideas.Quality of discussion between partners. Development of the chosen idea into final product.

Comments from Log - Filling in the "How I will make it" box proved a problem - the children didn't know what to draw - Natasha asked Mrs.M if it could be written - yes - I should have told them all to do this originally.

Evaluation of Session - Need to be less fixated about recording in drawing if writing in sentences is more natural for the children and appropriate for the activity. They had told each other how they would make it - why not write that down?

Spring Term 2001

Assessment Task : Frosty the Snowman

Rationale - Assessment of designing skills for comparison between the two classes.

Objectives - Produce drawn design and make a model to demonstrate a means of getting Frosty the Snowman's shopping to him on his hilltop if the shop is on the opposite hilltop and there is a lake in between.

Activity - Showed Frosty (model based on card tube made by daughter), explained he was stuck on his hill in the snow & shop on other hill. How will he get his shopping? Showed Lighthouse keeper's lunch picture as "clue", plus reminder & demo how to roll newspaper to make structure.

Success Criteria - Use of drawing to develop ideas and then use of ideas thus developed to inform making model of problem solution.

Assessment Procedure -

Individual profile spreadsheet; Video of Target Group from Focus Class, photographs from Comparison Class.

Evaluation of Sessions -

Focus Class - amazingly good since I was alone for this session and operating video. Got on sensibly and completed activity without fussing. From personal awareness of what was being videoed - very little interaction about their ideas once making. Discussed ideas at drawing stage, but once they had decided what to make, they just worked without consulting each other. *Reflection in Log book* -

Looking at drawings prior to analysis - Many children appear to have reverted to single-draw. Does this show that I was too specific in my explanation, that showing them the Lighthouse Keepers Lunch stopped them from having their own ideas, or is the use of drawing activity-dependent? Perhaps they had their ideas in their heads and discussed them but didn't draw them. Perhaps they simply opted for their first solution and didn't attempt to think of others. Viewing of tape indicates that Noel & Craig discussed ideas as they drew - and did not put pencil to paper to record the ideas they were discussing.

Focus Class Activities in First Half of Spring Term:

Session 1: Pictures from abstract shapes

Rationale - Development of using what they see to stimulate ideas. Objectives - Drawing activity - circles - ideation fluency Picture activity - develop shape into whole picture.

Activity -

a) the square - whole class. Square drawn on blackboard, child invited to make it into a picture of something. 2nd, 3rd, 4th square ditto.

b) the circles - individual work. The children are provided with a worksheet with rows of circles. They make as many as possible into pictures in a time limit of 5 minutes. The range of ideas will be discussed. There is frequently a run of ideas, i.e. once a child thinks "apple" a whole range of fruit often follows. These features of association of thoughts will be discussed.

c) the horseshoe shape. This idea comes from the RIOTT programme by McCracken (1985).

The children were each be supplied with a card horseshoe shape to use as the basis of a picture.

Success Criteria - Circles - fluency of ideation; Horse shoe - development of single good idea into a picture.

Comments from Log -

Circles - Told them to look around the room & see what they could see that was round. Gave a couple of hints to children who were stuck. Some children did not transfer the principle to the new activity & drew a square inside the circle and put a figure inside the circle. Once Mrs. M. and I had explained again to individuals and they had the idea, they were away. Overall very impressive.

Horse shoe - Complete list of what they each did is in handwritten log book.

Evaluation of Session - We were impressed with their creativity with the horseshoe shape.

Session 2: Visual analogies

Rationale - To develop their understanding of and facility with visual analogies, specifically of the type a>b:c>?

Objectives -To play a card game based on Iron's visual analogy tasks and then to develop their own.

Activity - the children play a card game in order to understand how the analogies work. Three different sets were required so that the children could work in pairs on tables of 6.

Required - Boards based on Iron's example for the Puzzle Series 1, one per pair of children, three different puzzles per table:



Source cards



Target cards for each board (example for left hand board, above)



Rules of game:

The source cards are placed face down in a pile at the top of the card and the target cards showing part filled shapes are laid out face up on the table.

The children take turns to turn over the source cards and place them on the board. They then try to match one of the target cards to it using analogical reasoning. Although taking turns, it is non-competitive.

They then shuffle the source cards and replace them at the top of the board, lay out the target cards and then move round the table to another card set.

Booklet - an A4 sheet folded to make an A5 booklet in which they designed their own analogy puzzle card game. Page 1 to be completed with the puzzle they had just solved. Inside the booklet were two similar boards with examples they had not seen before: The first had the puzzle example at the top and the source shapes. The next had the puzzle example but no shapes - their own choice. The back cover had a blank board - not even the puzzle example. Spare card was available for them to make a game using their back page game and play with their partner.

Success Criteria -

Understanding the analogy rule for each game and then being able to use the understanding of the rules of the game to create their own puzzle.

Comments from Log - Only got as far as completing booklet - no making of own game.

Therefore - Session 3: 2nd attempt at Iron's analogies

Rationale - as before

Objectives - To succeed in the activity designed for last week in the light of observations re children's misunderstanding of the rules of the game.

Activity - Almost as last week. Changes - each pair was given a wrong set as well as a right set of target cards.

Success Criteria - as before - will analyse and compare development booklets from each session and check that final game is same as development booklet p.4.

Comments from Log -

Most children instantly laid out and matched the correct set, ignoring the other one - only Holly & Alisa used the wrong set. To make sure, I told them to hide the cards they had put on the board among all the others laid out on the table and the all swopped seats to find a puzzle they had not done. Mostly successful in solving second puzzle. I repeated the change rule with blackboard demo.

Development booklet - fine. Making own game - mostly successful.

Session 4: Analogy development using letter shapes.

This is an application of the Letter Spirit Project (*internet site*). Originally planned as a 2 week activity, it was reduced to 1 because the Iron's Anologies were repeated.

Rationale - Application of analogical perception of letter shapes.

Objectives - To complete alphabet in demonstrated font and then create name-plate for bedroom door.

To transfer a pattern by scribbling across the back of the design drawing.

Activity - The children to be provided with sheets with letters a - f in 3 different fonts, based on a square, triangle & hexagon. They practice copying these onto squared paper and then choose their preferred font to write the rest of the alphabet. They write their name in their chosen style, scribble across the back and transfer the name to stout card. This can then be drawn over in coloured pens and decoration added.

Success Criteria - Maintaining font throughout alphabet. Using the font to write name. Transferring the design successfully onto card.

Comments from Log -The children were allowed to take these home to complete.

Evaluation of Session - No problem with fonts. They had never used scribbling across the back as a technique before and were surprised and delighted when it worked.

Second half of term:

Session 1: Snooks Animals

Rationale - Understanding of what a clear design drawing looks like.

Objectives - Viewing of examples, leading to discussion and then trying to make clear draiwngs.

Activity - PowerPoint presentation of Snook's Animals, as basis for discussion of what is needed for clear deisgn draiwngs - instructions, material lists, expansion to show small details, construction details etc. Children then given piece of found materials (e.g. driftwood) which they were to imagine how to make into an animal & then draw it "so clearly that someone else could make it".

Success Criteria - Clear, well-labelled drawings, lists, drawn as if from materials of construction.

Comments from Log -The children wanted to make them! Much clearer drawings. Good annotation.

Evaluation of Session - Message seems to have got across - fingers crossed for Stan next week - can we do travelling ideas too?

Session 2: Flat Stan -

Rationale - To introduce the children to the Container / Journey metaphor of design drawing. Objectives -

a) For the children to understand and apply the metaphor to their design drawings to generate and develop ideas about Stan's clothing.

b) to produce a puppet of Flat Stan to fit in an A5 envelope

Activity - The children shown 2 examples of drawings for design and 2 in narrative genre and asked to identify the differences. The way the ideas have travelled across the page in the design drawings indicated. The Container / Journey metaphor drawn on the flip chart and explained.

The children to be reminded of the "Flat Stanley" story and told that they will be making a model of Flat Stanley to go into an A5 envelope. A3 paper to be distributed and children sent to tables to begin drawing. Pink card and materials to come after children begin drawing (deterrant to instant making). Envelope to be available for storage of drawing and model when finished.

Success Criteria -

Puppet is flat and fits in envelope. Children have modelled their ideas on paper and that a clear progression of ideas can be traced through the drawngs and on to the making.

Assessment Procedure -

Although originally intended as a non-assessed activity, I could not resist building a set of spreadsheets identical to the Purpose of Drawing grids to be used for assessment activity results. Flat Stan has too long a history not to be assessed! I also built a collation spreadsheet to compare the Stans, both between the three types to see if there was progress and also to look at the 1998 Stans on the new assessment instrument.

Comments from Log - Got Stacey & Lauren to hold up two examples each of narrative & design drawing. The children Had difficulty spotting the significant difference between the two types. Needed to spell it out & explain. Seemed to be OK. All drew intently for about 15 mins - same as previous presentations.

I re-explained what to do with the card (final idea to be made with this) but I usually have to do this. This time I explained this in terms of the Container / Journey Metaphor. No one tried to throw away their drawing. I had not realised I had automatically trained them into giving me the drawing. At least some Flat Stan drawings usually need rescuing from the bin! *Evaluation of Session -*

Overall, pleased. Results look good. Will need to compare with a previous Yr2 presentation - perhaps 2D (done before Christmas) as well as a Yr2 from 1998

Video of debriefing is definitely more worthwhile than trying to video them working. Get much more data.

Will not video them again until assessment activity or they will get good at being debriefed.

Session 2: Round Stan -

Rationale - To develop understanding of the Container / Journey metaphor. Objectives -

a) To begin to use drawings to develop ideas about construction - materials, different viewpoints, expanded diagrams of small details.

b) To design and make a puppet of Round Stan from toilet roll middle.

Activity - Reminder of Container / Journey metaphor diagram.

Show good examples from last week which illustrate ways of moving ideas further through drawing into making decisions about materials, construction etc. The aim here is to begin to move children into using progressive drawings - sketches to develop ideas.

The Round Stan story to be told - At Christmas, Stanley tried to blow up a balloon and all the air rushed back into him and he became round. They are to use a toilet roll middle to be his body. They can use any of the materials in the trays to make the rest of him.

Success Criteria - Drawings showing clear progression of ideas across the paper. Comments from Log -

Reminded children of Container / Journey diagram and how we can take our ideas on journeys across the page. Re-iterated the process in terms of this metaphor as I showed examples by Jade, Michael, Richard, Connor Best (2D), as examples of different ways to take your ideas on a journey.

Then I told children Ray's story of Round Stan and drew picture of this as I told it. This was a mistake - quite a few thought they had to do a Christmas picture (including Emma W, Megan, Natasha). Carl: "What are we making?" I should have had a toilet roll middle in my hand as I explained to them what they were to do. Shows how much they rely on seeing, as opposed to just listening (Donaldson). I repeated the explanation of the activity.. Once sorted, fine, Good work.

Not all finished and I had to leave them for Chris to finish off sometime - since I was expecting to see Gerry at 2.30 re fixed term exclusion for J in my class. Over weekend, I decided I would go & get these unfinished ones on Monday - they will have forgotten their ideas & I have got the bit boxes back in my room now.

Session 3: Customised Stan -

Rationale - To reinforce understanding of the Container / Journey metaphor and to extend this understanding to thinking about materials of construction.

Objectives -

a) To create own adventure figure based on Stan saga

b) to think about matreials and try drawing figures as if made from the materials available *Activity* - There were no corks in school and none in Boots - so went to Pound Shop and got plastic cups and pegs - each child had cup and 2 pegs instead of corks & nails as per previous presentations of DIY Stan.

Recap of Stan; explanation of activity - they would decide Stan's next adventure.

I explained the metapor diagram on the blackboard. Then told them to have 2 or 3 ideas and try to draw them as if made from the materials - showed them plastic cup & 2 pegs (other materials available off-side). I did a demo of what I meant on the board.

Success Criteria - Could they think about the materials as they designed and draw the figure as if made of these? Could they produce more than one idea done like this? Is their designing improving and developing or do they now think they know what Stan looks like & so are not developing their ideas?

Comments from Log - Re-iterated Cont /Journey diag on blackboard with them in their places rather than gather on carpet.Showed them to cups & pegs. Said that they wer going to take thier ideas on journey into what it would look like made of these things. Showed them on blackboard, having ideas (2) and then trying out what they looked like made out of the cup & pegs. I didn't mind whether they had the several ideas first or had one & then tried it in the stuff & then had another new idea. But wanted them to take their ideas on the journey into thinking about how it would look made form these things & draw it like it to try it out. They found this hard. Lots of good character ideas eg. Chloe - wizard. They found drawing as if made form materials hard. And to think of several, very hard.

Assessment Task : Easter Egg Holder

Rationale - To assess the children's use of drawing to develop design ideas and to ascertain possible emerging differences between the two classes.

Objectives - To design and make a holder for an Easter Egg from a cardboard tube and other recycled materials.

Activity -

Explanation that they would be making a holder for an Easter Egg - wording purposely left vague since I did not want to guide their thoughts into "box" or "carton" etc

The children were shown a large cardboard tube (length approx.120 cm, width approx.100 cm) and box of hard-boiled eggs (explaining that that they would be using these for sizing, not chocolate eggs - too messy) and I demonstrated that the tube was far too big for the egg. They had to design something attractive which would hold the egg. Explanation also of using paper to work out ideas, talking about developing ideas across the page, avoiding use of words "design journey" since I did not want to specifically cue 2R into using a technique - it being their understanding of designing and not the use of a particular technique which I wanted to compare.

Success Criteria - Use of drawing to support and develop design ideas. Comments from Log -

Focus Class: Managed to have my log book under my arm & write down their comments. Also sketched some of their work as they did it.

Comparison Class: The children immediately began to say "I know what I'm doing" and seemed to want to make an instant choice rather than go away and come up with a range of ideas. I spent quite a lot of time helping children to cut holes in things. They seemed to want a lot of practical help - couldn't cut through card tube etc.

Summer Term 2001

Session 1: Designing a marble run.

Rationale - To evaluate each other's ideas and create a joint product. Objectives -

a) To create a marble run from plasticine, lolly sticks, strips of card and plastic bottle tops.b) To work collaboratively and share ideas and plans.

Activity- Different to original planning - after seeing mess made by Yr4 with paper mache and there was no modroc in school, decided that 2R would design playgrounds in discussion with partner and choose best of each to make on shared board with plasticine, lolly sticks, plant sticks & bottle tops.

Success Criteria - Joint project which reflects a contribution from each partner, adapted through discussion and collaboration.

Comments from Log -Took photos.

Went well, except that some children needed help with co-operating with their partner - quite a few helped themselves to their own board - needed correcting. Carl dominated Jack who did not contribute much. Other pairs in trouble sharing were mostly the ex-2S. Michael found difficulty sharing with Zac - each had their won half of the board. Generally they stayed close to their drawn ideas but developed them as they combined ideas - fair enough.

Session 2: Necklaces

Rationale - To do no drawing at all!

Objectives - To make bead necklaces from paper. *Activity* -

Again, change of plan because of lack of resources - no clay in school. I took home some paper to try this out and found that long triangles worked best, wound around a pencil, with just

a blob of glue on the end. Played with long-ways beads as well. had difficulty finding pictures of necklaces.

Used old travel brochures - left over from 1H role play.

Showed the children how to make the beads & they got on with it. They spent a happy afternoon doing this. I circulated and had a go as well, giving my beads to the slower workers. At the end they all had a string of beads which they wore round their necks to story!

They had fun and had a string of beads at the end. Totally low-key. They all wanted to wear their necklaces to story, so they definitely had fun.

Sessions 3& 4: Beach Buggy (2 weeks plus finishing in class-time)

Rationale -

a)To plan and make a longer project, since most activities, although related, have been single session.

b) To use drawing part-way through the design and make process rather than at the beginning. Objectives -

a) Plan and make a beach buggy based on folding a sheet of card into a box shape

b) To viualise and draw the sides, front, back & top view of the buggy once the basic box is made. In this way it is akin to the Box Car assessment activity conducted on previous occasions in that they need to visualise the different viewpoints and they need to think of a logo or graphic for the sides etc.

Week 1

Activity- Began with flat card and showed the children how to make a box, which they did step-by-step along with me and then put axles through it and fitted on the wheels - Someone (unknown) said "Oh its a car" at this point - despite my previous demo of how the wheels fitted nicely on the axles!

Drawing done at this point, to design sidea, front, back & top view. they were unsure what I meant by top view & I demonstrated on the board.

Success Criteria - Hoping for - completed box car plus seats and drawing of sides etc ready for next week to do outsides.

Evaluation of Session -

Explanation of what we were going to make - did this verbally - Therese reported that Lauren did not understand what she was making at all - words "beach buggy" meant nothing. Therese talked to her about Barbie vehicles. Others also lost sight of what they were making I think I should have had a picture or two.

I was hoping for good modelling of ideas here, since they had been involved in making the basic box cart immediately before, but it did not seem to be all that ckear to everyone. Making seats from egg-boxes was OK.

End of lesson - resolved to start next lesson with new explanation.

Week 2

Objectives -

a) Completing inside of buggy.

b) Planning graphics on sides etc of buggy and tranferring these to buggy using range of materials and colouring techniques.

Activity- Had all children on carpet and reminded them about containers & journeys - everyone seemed satisfyingly au fait with the terminology and seemed to understand what I was talking about as I did the diagram.

Did demo - I had done a side, front & back design for my beach buggy and took this along. "Since it was too small for everyoone to see" I redrew it on the flip chart and explained to them how my thoughts had gone on a journey..

New paper given out as well as their design sheet from last week. Some of them still had seats etc to do.

Success Criteria - Planning the graphics & getting buggy finished (no one finished). Comments from Log - Time wizzed by - at 2.15 we packed up and I promised to ask Chris if they could finish them with her tomorrow and take them home. I did not take photos this time.

Evaluation of Session - They worked purposefully and well - where did the time go? Many of them made really good seats etc inside, but some had them in funny places - lower than the axles. Interesting that they don't really think about where parts of a car are in relation to each other. Steering wheels were far away from seats. No thought of an engine - the steering wheel went into the front of the car.

Second half of term:

Sandals

Session 1: Deciding what to make.

Rationale -

a)To begin from product analysis

b) To engage the children in the reality of sandal production rather than the fantasy aspects of design creation.

c) To work on an extended project, which would put together the aspects of designing so far encountered in single sessions.

d) To use drawing part-way through the design and make process rather than at the beginning.

Objectives - To look closely at a range of sanadals and to identify through discussion the features of the sandals, record responses and generate ideas for sandals to be made in future sessions.

Activity- The children were asked to bring in sandals and beach shoes for this session so that they could look at different styles. The vocabulary of shoe parts were provided so that they could discuss in groups the features of the shoes, how they were made and how feasible it would be to try to make each type. They cannot pretend that paper is waterproof - they needed to engage their knowledge of real materials here and also to consider the real demands made on sandals in use. It was anticipated that the children would need some guidance and some demonstrations of the way real sandals accommodate foot movements when walking. They were encouraged to produce and record different ideas for each of the soles and uppers. It was anticipated that they would probably have some ideas about what they would like the finished sandal to look like and so they could record these ideas as well.

Success Criteria - Good, purposeful discussions; identification of material requirements for shoe parts; recording of ideas which can inform planning to make next week.

Comments from Log -

The children chose to record more than I had anticipated. They wrote about the sandals that they observed and they drew their ideas for sandals they would like to make. Interesting - words used for observations; drawings for ideas generation, futuring.

Session 2: Making templates and choosing materials

Rationale -

a) To understand how to make a sandal that will fit their own foot

b) To engage in the reality of template making

c) To choose suitable materials for soles and uppers

Objectives -

a) To make templates of their feet from card, both soles and uppers

b) To make choices about materials and record these ready for next week. Activity -

Initial discussion reminded them of the need of shoes in use. Discussion of how the upper is fixed to the sole: The simplest and most secure fixing solution is to make an upper which is a strip of cloth sandwiched between two layers of the sole and which passes across the top of the foot; need to allow for fabric to go under foot when making template. Introduction of template-making: Thin card provided to make a template of one foot only, both sole and upper, to be cut out and shown to me to check that it fits. Then sole template to be drawn round onto thick card for stiffener for sole of sandal, canvas for base and fabric (upholstery samples) for
Appendix B Programme Lesson Plans

inner lining of sole; template for upper to be drawn round onto chosen fabric (either to match or complement fabric chosen for inner sole). Diagram on blackboard to aid understanding (I told them it was like a sandwich!):



Success Criteria - Completion of template for sole and upper. Materials chosen and some parts of sandals cut out ready to assemble next time.

Comments from Log - Difficulty getting them to understand they only needed one template not two. Some made the templates into sandals & wanted to decorate them by colouring etc. Not understanding we were making real sandals? Not understanding role of template? Most children a long way from completion so arranged with Chris that they could continue to work on the sandals during the week.

Session 3: Making sandals Conducted by Mrs.R.

Session 4: Completing construction of sandals

Rationale and Objectives - To complete sandal-making activity

Activity - Some children had almost finished making their sandals apart from decoration of uppers but some still had a long way to go. Therefore those who finished quickly were encouraged to help those who were still working. I took photos as and when they finished. *Success Criteria* - Completion of pair of sandals which fitted their feet.

Comments from Log - Rather a lot of feathers chosen for decoration - not brought by me - hope Mrs.R. wasn't going to use these for something tomorrow - I was pre-occupied with my photo taking & didn't notice.

Assessment task - Pandy's Suitcase

Rationale - To compare the use of drawing to support design thinking by the Focus and Comparison Classes

Objectives -

a) To use drawing to generate ideas, model solutions

b) To relate this to the real problem of making a suitcase for the Panda

c) To make a full size mock-up of the suitcase which fits over the Panda's paw, is appropriate size for Panda to carry and which can contain the plastic mac.

Activity - Design a travel bag for the toy Panda to take on holiday, using identical activity and script from Pandy99.

Success Criteria -

a) Suitcase fulfils design criteria: big enough for Pandy's plastic mac and hat to fit in, but small enough for him to be able to carry, and the handle must go over his paw so he could carry it.
b) Drawing used to model design ideas

c) Full-size mock-up made from card to demonstrate chosen idea.

I hope that the Focus Class will be comparable to the Year 3 children in the 1999 study.

Assessment Procedure - Data to be collected: Log book notes made during session, drawings and mock-ups.

Log book notes will be collated via Log Book database.

Drawings will be assessed using Assessment Grid.

Mock-ups will be used to support analysis of relationship between drawing and making.

Appendix B Programme Lesson Plans

Comments from Log -

Details in database. Clear differences in each class' way of working. Significant number of Comparison class did not use the design sheet as such - some even drew and cut out this sheet & made their mock up from it. The younger children in Comparison class (not part of study) really struggled, whereas the Focus class ones didn't - perhaps I should have left them in after all. But as individuals they have greater learning difficulties anyway, so it could just be individuals involved - 2 of them usually have LSA.

Evaluation of Session - Both sessions went smoothly. Wait & see what the analysis shows up!

Autumn Term 2002

Sessions 1-3: Food.

Week 1: Menu Planning

Rationale -

a) To extend design thinking into menu planning and presenting a food-related diarama.

b) To encourage team-work and use drawing to communicate ideas to each other and to record group decisions.

Objectives -

a) To look at recipe books to design a healthy meal.

b) To produce a menu, an advertising poster and a list of materials needed to make a model of the meal, using waste materials, as a 3-dimensional advertising feature.

Activity -

Discussion about foodstuffs appropriate to a well-balanced healthy meal. Working in groups of 4, the children looked at recipe books, listed ideas and devised a first draft menu. These ideas would be recorded on one A2 sheet of paper per group. By the end of the session they needed to have decided what they would use to make their models and provide me with a list of materials.

Success Criteria - Production of clear idea of what they were going to make plus materials list.

Comments from Log - ange of group working methods. Some argued their way through the whole menu. Other groups divided the workload and decided who would choose and design each course. One group did this but each memebr produced a list of ideas in drawn form from which the rest of the group made the final menu choice.

Evaluation of Session - Good discussion throughout, clearly focussed on task. Main fault - little recording of materials, although I stressed this loudly several times.

Week 2: Diaramas

Rationale -

a) To transfer design ideas recorded in previous lesson to construction of model.

b) To work as a team to realise ideas recorded in drawing and writing during previous lesson.

c) To develop meta-cognitive awareness of design processes

Óbjectives -

a) To create an advertising diarama of a healthy meal.

b) To make oral presentation to peers to give account of working method of the group Activity -

The children were reminded of the activity as set last week and shown how to make a card tray to arrange their work on. Paper plates, bowls and cups plus straws were provided; Demonstration of fixing these to tray with split pin prior to filling with "food". They were then to choose the materials as listed by themselves the previous week and make the model. Embarrassed giggles ensued when they realised they had forgotten what they were going to use and had not recorded it on the paper. Relief was expressed when I produced the materials I had brought "in case they might need them". They were encouraged to discuss and record materials before beginnning to make the models. Models based on the plans from previous

Appendix B

Programme Lesson Plans

lesson were made. At end of session, each group presented their planning sheet and finished work to the rest of the class, described their work and their working method.

Success Criteria -

a) A completed "meal" on a tray plus a menu and advertising poster was originally intended as outcome, however, due to only spending two sessions on this topic, the menu and poster were not completed.

b) Ability to coherently explain their healthy meal and their working method to the rest of the class.

Comments from Log -

No time to make menus. Only one group finished in time to complete poster started previous lesson. Others had made attempt at drawing pictures of their meal on to paper provided and writing a title but generally poorly done (in contrast to diaramas which showed imagination and good team work)

Evaluation of Session - Presentation by groups good - able to give account of working methods and how they had worked as a group. I felt this was more valuable than the details of their models. Ran out of time - needed to allow 15 mins for this, not all groups got enough feedback through having to rush through last three groups

Week 3 - Sandwich-making

Rationale - Not sure - see QCA

Objectives - To give Focus Class same fun activity as rest of Year Group (they had spent some weeks working up to the grand tasting; we did it all in one!)

Activity - Organised with military precision to avoid mess on floor!

Variety of ready sliced breads and rolls (excluding Tescos economy sliced), range of spreads (butter, Olivio etc. peanut butter clearly labelled) and fillings (Bovril, jam, chocolate spread, fish paste, etc), all brought by children and myself. "Tasters" to be prepared by children from ingredients supplied to their group on trays previously laid up by me. All children in group to taste these and to decide on personal favourite. Each child then provided with two slices of chosen bread to make round of sandwiches using chosen spread and filling. These were to be arranged on a large serving plate per group but the children simply arranged them on the plate on which they had made them (they wanted to eat their own, not someone else's!). Whilst they were doing this, I covered spare tables with a paper and poured orange squash. The children brought their sandwiches to me and were not impressed when I removed these from their individual plates and arranged them on the large ones and sent them back to their seat. We then enjoyed our buffet.

Success Criteria - Minimum mess, maximum fun!

Evaluation of Session - Very messy but clearly enjoyed by children. I was too neurotic about margarine on the carpet in someone else's classroom to enter into the spirit of it all. On balance, however, glad I did it.

Second half of term:

Session 1-4: Kit-making.

Week 1: Introduction of activity, brainstorming in small groups and initial sketches of ideas.

Rationale -

a) To generate ideas; b) To consider what would be suitable for younger children to make as a present for Mum

c) To work in teams; d) To critically review each other's ideas Objectives -

a) Each group to produce a clear plan of what they would make and present idea to peers

b) To record materials (didn't for food)

Appendix B

Programme Lesson Plans

Activity - Initial introduction and discussion of task: what might be a suitable gift, what Year 1 children can make, what materials are available, what techniques would be suitable. Split into groups and discuss ideas, coming to agreement about gift to be made, use paper to model and communicate ideas. Each group to produce a plan which could then be shared with rest of class in poster form. Time allocation for plenary needs to be 15 mins (food plenary ran out of time)

Success Criteria - Clear and realistic plan of item to be made; Materials considered this time (food plans didn't)

Comments from Log - They got the point re the materials - as soon as I mentioned this in the intro they all groaned & nodded etc - so had learnt this from the food sessions - good. Although they discussed together, most groups made individual "poster" which were really neat versions of their working sheets. They tended to discuss in groups, some children recording ideas or drawing them to show others what they meant and then each member produced their own neat version. I guess they all wanted a "product" from the session.

Weeks 2 & 3: Making the prototype.

Rationale - a) Prototyping their product; b) Evaluation of own work; c) Team work

Objectives - To make their own version of the gift they planned last week

Activity - Explanation and reminder of the task and then children working at own speed (stressed helping each other). The components of the finished task (own gift, poster, instructions and kit of parts for Year 1) written on white board. Their poster drawing had to look like their gift (not what they would like it to be) or the Year 1 customer would be confused. The poster needed to include simple drawn instructions, which could be labelled but they needed to remember Year 1s can not read very well. If time, they could make a box or tray as packaging in which to send it to Year 1.

Success Criteria - Completion of own gift and poster. The poster drawing must look like their product; it must be achievable by a Year 1 child

Comments from Log - Easier to provide materials - they had all specified them. Some who finished early did not understand they were making a poster to be used as instructions, some re-drew their plans neatly on a clean sheet of paper. Interesting gap in understanding between the class and the two boys who had joined in September. They did not understand the purpose of the drawings at all. Although the Summer-born children are mostly still trailing the others, the gap is greater between these two new boys and all of the others.

Evaluation of Session -

Not everyone finished their own gift. They wanted a perfect gift for their own Mum (understandable) rather than prototyping something for someone else to make. Time needs to be given next time to finish.

Week 4: Assembly of the kit.

Rationale - To address needs of a client Objectives -

a) To produce kit of parts for Year 1 child to make their product

b) To produce poster, instructions for final version and kit of parts in envelope

Activity - Despite some not having finished own gift last week, I decided that I would insist that they put together the kit for Year 1 or the younger children would not have time to make theirs. Year 3 could finish their own gift with own teacher at another time (I wanted to observe Year 1 following the instructions next week)

Success Criteria -

Poster & instructions which are suitable for Year 1 to make and follow instructions It could be made by year 1 - Evaluation by Year 1 to follow!

Comments from Log -

Panic ensued as they realised that their work really was going to Year 1 and that the younger children would have a view on their success!

Appendix B Programme Lesson Plans

Spring term 2002

Final Assessment Task : Theseus' Maze

Rationale -To compare the use of drawing to support design thinking by Focus and Comparison Classes

Objectives - To design and make a model of King Minos' maze to help Theseus find the Mniotaur and escape safely.

Activity - Discussion of the myth, which the children had heard in Literacy Hour. The children were then shown a pop-up book with a 3-dimensional maze. I stressed that theirs did not need to be pop-up but that it did need to be 3-dimensional. They should plan the positions of the walls on paper before making the model in card and I demonstrated this on the whiteboard, showing them a ground-plan view and saying this is how someone would plan out where the walls would go before they were built. They might change their minds (demonstrating rubbing bits out) or want to show that arches and doorways would be here (indicated by arrows) and, of course, they might change their mind completely and start again (demonstrating doing another drawing next to the first). Once the ground plan decided on, then thick yellow card is for the base and thinner red and blue card is for the walls. String is available for showing Thesus the route in and out.

Success Criteria - Ideas worked out on paper; Drawing informs making.

Comments from Log - Comparison Class (done first) - not a clue half of them. They were not making real mazes! A lot of them drew a maze on the yellow card and made walls round the drawing. The few who really engaged with the task frequently did not use drawing to plan it - only a very few made what they drew into something 3-dimensional (inc. Myscular Dystrophy child who was without his helper today - has constant adult support scaffolded meaning for him in a whole range of ways?). On to Focus Class feeling the activity had failed dismally - but much better - they were using the drawing to support real thinking about a solution (with only 3 exceptions) and they made 3-D mazes, many of which bore some resemblance to what they had drawn.

APPENDIX C Target Groups : Pizza

Focus Class:



APPENDIX C Target Groups : Pizza





APPENDIX E Target Groups : Frosty



Kara































Focus Class:







Beth

Martin







Carl

Natasha









Hayley

Damian





Craig









Ellie

Comparison Class:



Emma









Kathy

Garth

Sophie





Wendy





Peter

Matthew

Carys



Robert





Zara

Alice

Chloe



APPENDIX I Surprise Card

Page 378

APPENDIX I Target Groups : Card



APPENDIX I Target Groups : Card

Focus Class (cont.)



Comparison Class :





How I will make my card

1000

Randal

genta real run

🕅 The pr

in my ci Hitita

Laco tools will you

APPENDIX I Target Groups : Card

Comparison Class (cont.) :









APPENDIX J Focus Class : Sandals Project











APPENDIX J Focus Class : Sandals Project







APPENDIX K Target Groups : Suitcase

Focus Class: Adara wanter and and Craig X A 1346 Ċ Birth Martin Hayley Randal Kara Maria



APPENDIX L Target Groups : Maze















APPENDIX L Target Groups : Maze



APPENDIX M

Appendix M contains the Log Book Index by which Log Book entries were catalogued and the transcript of the video of the Focus Class Target Group during the Frosty Assessment Task.

Log Book Index

<u>Name</u>	<u>Class</u>	<u>Task</u>	Observation
Ellie	F	Pizza	Cut out chosen design from design sheet
Louise	F	Pizza	Pipe cleaner bent round = pepperoni
Chloe	F	Pizza	Decorated design sheet
Noel	F	Pizza	Has stuck black tissue underneath pizza and nothing on top.
Maria	F	Wag figure	did an open frame structure
Noel	F	Wag figure	looks like a Mr. Man - no distiction between head & body.
Martin	F	Wag clothes	wanted to go straight to the cloth and cut it up immediately after making the model
Emily	F	Wag clothes	wanted to go straight to the cloth and cut it up immediately after making the model
Carl	F	Wag clothes	designed football coat, bots & hat - has no.20 on coat, but can't make it.
Kara	۴	Wag clothes	cut out a triangle to be a skirt - I got Natalie to come & show how a skirt is made - by turning round etc - skits go all round, not just stuck on front.
Stacev	F	Wag clothes	made paper template and drew round it onto the cloth
Connor	F	Wag clothes	figure is Superman
Martin	F	Wag buggy	still drawing long after everyone else is making
Randal	F	Wag buggy	drew one rocket. Insisted that the single drawing contained everything he needed to know, but when it came to the making, said he didn't know what to mal
Craig	٣	Wag buggy	model - technically good - slots for rocket foils to fit into.
Michael	F	Wag house	"Bars so they can't break in" (his motorbike was stolen last week)
Randal	F	Wag house	liked the shape of the house he'd drawn
Lewis	F	Wag house	liked the shape of the house he'd drawn
Martin	F	Wag house	like Millenium dome
Natasha	F	Wag house	Asked if she could write how to make the house
Craig	F	Wag house	identified a problem - he's drawn a round house - cereal packet won't do - I tell him to go & find something round in the bag. Comes back with 2 round pots
Maria	F	Wag house	is making a 2D collage on the inside of a piece of the box.
Lewis	F	Wag house	has drawn on inside of the box "So he can cut"
Martin	F	Wag house	using paper to make a curved roof over the top of the cereal packet - original - and no one copied this idea
Randal	F	Wag house	cut up the box to make shapes - might as well have had falt card.
Carl	F	Wag house	really good - just like the picture.
Craig	F	Wag house	really good - just like the picture.
Kate	F	Wag house	knows how she will make "scrunchy bits"
Maria	F	Wag house	realised Wagwum can't go in house - not 3D. Goes looking for a box - I let her use the lid of the paper box I'd brougt some of the materials in. This is then
Lewis	F	Wag house	still flat. & looking like a donkey
Randal	F	Wag house	still flat. Richard gets in a paddy with it. He has now cut out sides for it, but no time to make it - we're clearing up.
Martin	F	Wag house	is playing with the whole set - Wagwum, house & buggy
Damian	F	Wag house	excellent - has quietly worked away and hardly asked for any help.
Natasha	F	Wag house	inside the box is better than the outsdie - fireplace, chair, picture, mat.
Michael	F	Circles	drew a square inside the circle and put a figure inside the circle.
Noel	F	Circles	drew a square inside the circle and put a figure inside the circle.
Emily	F	Circles	using circle as framework only.
Maria	F	Irons 1	very naughty throughout, even when sent to book corner.
Stacey	F	Irons 1	With Katie, having trouble finding right shapes in game.
Kara	F	Irons 1	With Katle, having trouble finding right shapes in game.
Louise	F	Irons 1	explaining to Rosle and Katle.
Hayley	F	Irons2	With Alisa, used the wrong set of cards to match
Lisa	F	Irons2	With Alisa, used the wrong set of cards to match

Page 389

LouiseFFlat Stanonly child to Single-drawJoleneFFlat Stan he had drawn and wanted to make a puppet - popping out of the envelope - hence cutting hole in envelopeGraigFRound StanWhat are we making?KateFDIY StanIm going to do a wizard.StaceyFDIY Stanhas worked out what heres will look like - drawn if as pot with cotton wool feet.MichaelDIY Standrawn and picture frame round his designLouiseFDIY Standrawn and picture frame round his designLouiseFDIY Standrawn and picture frame round his designLouiseFEastertof the about eas after she't done other drawings. Added them to No.2 when I said I'd forget.RandalFEasterCan I do that one? (No.1 - he dd)KaraeFEasterThat one's wong (1's at atempt)CraigFEasterthas cut the tube and bent it round to make it smaller. His drawing shows this.MariaFEasterthas cut the tube and bent it round to make it smaller. His drawing good ideas going on.CraigFEasterThere is a great deal of negotation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotation & sharing of ideas & copying good ideas going on.KaraeFEasterThere is a great deal of negotation & sharing of ideas & copying good ideas going on.Ka
JolenFF lat Staneven brainstormed in wordsMartinFF lat Stan- he hod drawn and wanked to make a puppet - popping out of the envelope - hence cutting hole in envelopeCraigFRound StanWhat are we making?'KateFD IY StanIm going to do a wizard.StaceyFDIY Stanheres will look like - drawn it as pot with cotton wool fest.MichaelFDIY Standrawn a picture frame round his designLouiseFDIY Stanwanke the eno on the left. Drow this first and then the one on the lrght. Liked the first idea best.NatashaFEasterbold the about ears after she'd done other drawings. Added them to No. 2 when I said I'd forget.RandalFEasterCan do that one? (No 1 - he dit)GraigFEasterthat one's wrong (1st attemptCraigFEasterthat ben it round to make it smaller. His drawing shows this.MariaFEasterthat ben a ben it round to make it smaller. Use a sport diese going on.CraigFEasterthat is a great deal of negotiation & sharing of ideas & copying good ideas going on.EtiteFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sha
Mark Graig GraigF FFlat Stan F- he had drawn and wanted to make a puppet - popping out of the envelope - hence cutting hole in envelopeGraig GraigFDIY StanIm going to do a wizard.StaceyFDIY Stanhase worked out what heres will look like - drawn it as pot with cotton wool feet.MichaelFDIY Standrawn and picture frame round his designLouiseFDIY Stanwante the one on the left. Drew this first and them the one on the right. Liked the first idea best.NatashaFEasterCan I do that one? (No.1 - he did)KaraFEasterCan I do that one? (No.1 - he did)KaraFEasterThat one's wrong (1st attempt)CraigFEasterthat one's wrong (1st attempt)CraigFEasterthat one's wrong (1st attempt)CraigFEasterthat one's wrong (1st attempt)CraigFEasterthere is a great deal of negotiation & sharing of ideas & copying good ideas going on.CraigFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EliteFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of i
CraigFRound StanWhat are we making?KaleDIY StanImoging to a wirard.StaceyFDIY Stanhas worked out what heres will look like - drawn it as pot with cotton wool feet.MichaelFDIY Standrawn a picture frame round his designLouiseDIY Stanwarse the one on the left. Drew this first and then the one on the inght. Liked the first idea best.NatashaFEastertol that tone? (No.1 - he did)KaraFEasterThat one's worng (1st attemptCraigFEastertol that one? (No.1 - he did)KaraFEasterthat one's worng (1st attemptCraigFEasterthat one's worng (1st attemptCraigFEasterthat one's worng (1st attemptCraigFEasterthat one's worng (1st attemptCraigFEasterthat e is a great deal of negolation & sharing of leas & copying good ideas going on.EliteFEasterThere is a great deal of negolation & sharing of ideas & copying good ideas going on.EliteFEasterThere is a great deal of negolation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negolation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negolation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negolation & sharing of ideas & copying good ideas going on.NatashaFEaster
KateFDIY StanI'm going to da wizard.StaceyFDY Stanhas worked out what heres will look like - drawn it as pot with colton wool feet.MichaelFDIY Standrawn a picture frame round his designLouiseFDIY Stanwants the one on the left. Drew this first and them the one on the light. Liked the first idea best.NatashaFEasterCan I do that one? (No.1 - he did)KaraFEasterThat one? wrong (I st at temptCraigFEasterthat one? (No.1 - he did)KaraFEasterthat one? wrong (I st at temptCraigFEasterthas on design journey, but she says be is doing No2 In fact she is doing Stripes - No.1.CraigFEasterthe advort cars at design journey, but she says be is doing No2 In fact she is doing ot ideas at conving oppet so is not a design journey, but she says be is doing No2 In fact she la doing stripes - No.1.CraigLEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.HayleyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.HayleyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of
StaceyFDIY Stanhas worked out what heres will look like - drawn it as pot with cotton wool feet.MichaelFDIY Stanwans bit one on the left. Drew this first and then the one on the irght. Liked the first idea best.NatashaFEastercon it do that one? (No.1 - he did)RandalFEasterThat one's wrong (1st attemptCraigFEasterThat one's wrong (1st attemptCraigFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EllieFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.BlackFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.LikeFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEaste
MichaelFDIY Standrawn a picture frame round his designLouiseFDIY Stanwants the one on the left. Drew this first and then the one on the lright. Liked the first idea best.NatashaFEasterCan I do that one? (No.1 - he did)RandalFEasterCan I do that one? (No.1 - he did)CraigFEasterThat one's wrong (1st attemptCraigFEasterthat one's wrong (1st attemptCraigFEasterthat one's wrong (1st attemptCraig & LFEasterthas cut the tube and bent it round to make it smaller. His drawing shows this.MariaFEasterthas cut the tube and bent it round to make it smaller. His drawing shows this.Craig & LFEasterthas cut the tube and bent it round to make it smaller. His drawing shows this.Craig & LFEasterthare is a great deal of negotiation & sharing of ideas & copying good ideas going on.EllieFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.LouiseFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.LouiseFEasterThere i
LouiseFDIY Stanwants the one on the left. Drew this first and then the one on the left. Liked the first idea best.NatashaFEastertold me about ears after she'd done other drawings. Added them to No.2 when I said I'd forget.RandalFEasterCan I do that one? (No.1 - he did)KaraFEasterThat one's wrong (1st attemptCraigFEasterhas cut the tube and bent It round to make it smaller. His drawing shows this.MariaFEasterthas one and the round to make it smaller. His drawing shows this.Cari (FEasterthas drawn chicken laying eggs but has simply covered the tube. I query where the good idea has gone. He makes a pink (sic) chicken to go inside the tubeCraig (F)EasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EllieFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of i
NatashaFEasterIold me about ears after she'd done other drawings. Added them to No.2 when I said I'd forget.RandalFEasterCan I do that one? (No.1 - he did)KaraFEasterThat one's wrong (1 st attemptCraigFEasterthas cut the tube and bent it round to make it smaller. His drawing shows this.MariaFEasterthas cut the tube and bent it round to make it smaller. His drawing shows this.CarlFEasterthas drawn chicken laying eggs but has simply covered the tube. I query where the good idea has gone. He makes a pink (sic) chicken to go inside the tubeCarlFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.HayleyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.LouiseFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.LouiseFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.LouiseFEasterThere is a great deal of
RandalFEasterCan I do that one? (No.1 - he did)KaraFEasterThat one's wrong (1st attemptCraigFEasterhas cut the tube and bent it round to make it smaller. His drawing shows this.MariaFEasterthas cut the tube and bent it round to make it smaller. His drawing shows this.CariaFEasterthas drawin chicken laying eggs but has simply covered the tube. I query where the good idea has gone. He makes a pink (sic) chicken to go inside the tubeCraig & LFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EllieFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.JoleneFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.JoleneFEasterDid not finish product.RandalFMarble runwas good - shame they changed it after I said I liked it.CraigF
KaraFEasterThat one's wrong (1st attempt)CraigFEasterhas cut the tube and bent it round to make it smaller. His drawing shows this.MariaFEasterdrawing appears to show a design journey, but she says she is doing No2. In fact she is doing stripes - No.1.CarlFEasterhas drawn chicken laying eggs but has simply covered the tube. I query where the good idea has gone. He makes a pink (sic) chicken to go inside the tubeCraig & LFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EllieFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.LouiseFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.JoleneFEasterhas egg holder inside at bottom - not visible in photo.JoleneFEasterDid of thish product.CraigFMarbie rundominated Jack who did not contribute muchMartinFMarbie rundominated Jack who did not contribute muchMartinFBeac
CraigFEasterhas cut the tube and bent it round to make it smaller. His drawing shows this.MariaFEasterhas drawing appears to show a design journey, but she says she is doing No2. In fact she is doing stripes - No.1.CarlFEasterhas drawn chicken laying eggs but has simply covered the tube.Craig & LFEasterThere is a great deal of negotilation & sharing of ideas & copying good ideas going on.EllieFEasterThere is a great deal of negotilation & sharing of ideas & copying good ideas going on.HayleyFEasterThere is a great deal of negotilation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotilation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotilation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotilation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotilation & sharing of ideas & copying good ideas going on.LouiseFEasterThere is a great deal of negotilation & sharing of ideas & copying good ideas going on.JoleneFEasterThere is a great deal of negotilation & sharing of ideas & copying good ideas going on.JoleneFEasterDid not finish product.RandalFMarble runwas good - shame they changed it after I said I liked it.DarnianFMarble runwas good - shame they changed it after I said I like
MariaFEasterdrawing appears to show a design journey, but she says she is doing No2. In fact she is doing stripes - No.1.CarilFEasterhas drawn chicken laying eggs but has simply covered the tube. I query where the good idea spoing on.Craig & LFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EllieFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.Image: StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.Image: StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.Image: StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.Image: StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.Image: StaceyFEasterThere is a great deal of negotiation & shari
CarlFEasterhas drawn chicken laying eggs but has simply covered the tube. I query where the good idea has gone. He makes a pink (sic) chicken to go inside the tubeCraig & LFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.HayleyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.LouiseFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.LouiseFEasterred on drawing means red baseLouiseFEasterDid not finish product.RandalFMarble runwas good - shame they changed it after 1 said 1 liked it.DamianFMarble runwas good - shame they changed it after 1 said 1 liked it.MartinFBeach buggyOh its a carChloeFBeach buggyWarled to put guns on - no!ChloeFBeach buggywas call found.MarinFCardWarled to put guns on - no!ConnorFBeach buggywarled to put guns on - no! <tr< td=""></tr<>
Craig & LFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EllieFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.HayleyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EmilyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.LouiseFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.JohneFEasterned or awing means red baseLouiseFEasterDid not finish product.RandalFMarble runwas good - shame they changed it after I said I liked it.DamianFMarble runwas good - shame they changed it after I said I liked it.CraigFMarble runMichael found difficulty sharing with Zac - each had their won half of the board.UnknownFBeach buggywanted to put guns on - no!did bench seats all round.wanted to put guns on - no!CraigFBeach buggywanted to put guns on - no!MariaFcardno looking at big book - is she list
EllieFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.HayleyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EmilyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.LouiseFEasterred on drawing means red baseLouiseFEasterhas egg holder inside at bottom - not visible in photo.JoleneFEasterDid not finish product.RandalFMarble runwas good - shame they changed it after I said I liked it.DamianFMarble runwas good - shame they changed it after I said I liked it.CraigFBeach buggyOh its a carChiceFBeach buggyOh its a carChiceFBeach buggyvery good seats.CraigFBeach buggywanted to put guns on - no!did bench seats all round.did bench seats all round.MariaFcardnot looking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelistEmilyFcardto looking at big book - is she list
HayleyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EmilyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EmilyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.JoleneFEasterhas egg holder inside at bottom - not visible in photo.JoleneFEasterDid not finish product.RandalFMarble runwas good - shame they changed it after I said I liked it.DamianFMarble rundominated Jack who did not contribute muchMartinFBeach buggyOh its a carChiceFBeach buggyvery good seats.CraigFBeach buggywanted to put guns on - no!ConnorFBeach buggydid bench seats all round.MariaFcardnot loking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelistEmilyFcardnot loking at big book - is she listening? -chris says her dad left at weekend. Me to table to support & ensure she did thelist
StaceyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EmilyFEasterred on drawing means red baseLouiseFEasterhas egg holder inside at bottom - not visible in photo.JoleneFEasterDid not finish product.RandalFMarble runwas good - shame they changed it after I said I liked it.DamianFMarble runwas good - shame they changed it after I said I liked it.CraigFBeach buggyOh its a carChoeFBeach buggyOh its a carChoeFBeach buggywanted to put guns on - no!CraigFBeach buggywanted to put guns on - no!ConnorFBeach buggywanted to put guns on - no!MariaFcardin otoking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelistEmilyFcardI tol draw rather than write "love hearts & flowers" - got rather into it "Miss. I've got a pattern"
KateFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EmilyFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EmilyFEasterred on drawing means red baseLouiseFEasterhas egg holder inside at bottom - not visible in photo.JoleneFEasterDid not finish product.RandalFMarble runwas good - shame they changed it after 1 said 1 liked it.DamianFMarble runwas good - shame they changed it after 1 said 1 liked it.CraigFMarble runMoinated Jack who did not contribute muchMartinFMarble runMichael found difficulty sharing with Zac - each had their won half of the board.UnknownFBeach buggyVery good seats.CraigFBeach buggywanted to put guns on - no!ConnorFBeach buggydid bench seats all round.MariaFcardnot looking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelistEmilyFcardI told her to draw rather than write "love hearts & flowers" - got rather into it "Miss. I've got a pattern"
NatashaFEasterThere is a great deal of negotiation & sharing of ideas & copying good ideas going on.EmilyFEasterred on drawing means red baseLouiseFEasterhas egg holder inside at bottom - not visible in photo.JoleneFEasterDid not finish product.RandalFMarble runwas good - shame they changed it after I said I liked it.DamianFMarble runwas good - shame they changed it after I said I liked it.CraigFMarble rundominated Jack who did not contribute muchMartinFMarble runMichael found difficulty sharing with Zac - each had their won half of the board.unknownFBeach buggyOh its a carChloeFBeach buggywasted to put guns on - no!ConnorFBeach buggydid bench seats all round.MariaFcardnot looking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelistEmilyFcardI told her to draw rather than write "love hearts & flowers" - got rather into it "Miss. I've got a pattern"
EmilyFEasterred on drawing means red baseLouiseFEasterhas egg holder inside at bottom - not visible in photo.JoleneFEasterDid not finish product.RandalFMarble runwas good - shame they changed it after I said I liked it.DamianFMarble runwas good - shame they changed it after I said I liked it.CraigFMarble rundominated Jack who did not contribute muchMartinFMarble runMichael found difficulty sharing with Zac - each had their won half of the board.unknownFBeach buggyOh its a carChloeFBeach buggyvery good seats.CraigFBeach buggywanted to put guns on - no!ConnorFBeach buggydid bench seats all round.MariaFcardnot looking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelistEmilyFcardi told her to draw rather than write "love hearts & flowers" - got rather into it "Miss. I've got a pattern"
LouiseFEasterhas egg holder inside at bottom - not visible in photo.JoleneFEasterDid not finish product.RandalFMarble runwas good - shame they changed it after I said I liked it.DamianFMarble runwas good - shame they changed it after I said I liked it.CraigFMarble rundominated Jack who did not contribute muchMartinFMarble runMichael found difficulty sharing with Zac - each had their won half of the board.UnknownFBeach buggyOh its a carChloeFBeach buggywanted to put guns on - no!ConnorFBeach buggydid bench seats all round.MariaFcardnot looking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelistEmilyFcardI told her to draw rather than write "love hearts & flowers" - got rather into it "Miss. I've got a pattern"
JoleneFEasterDid not finish product.RandalFMarble runwas good - shame they changed it after I said I liked it.DamianFMarble runwas good - shame they changed it after I said I liked it.CraigFMarble rundominated Jack who did not contribute muchMartinFMarble runMichael found difficulty sharing with Zac - each had their won half of the board.unknownFBeach buggyOh its a carChloeFBeach buggyvery good seats.CraigFBeach buggywanted to put guns on - no!ConnorFBeach buggydid bench seats all round.MariaFcardnot looking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelistEmilyFcardI told her to draw rather than write "love hearts & flowers" - got rather into it "Miss. I've got a pattern"
RandalFMarble runwas good - shame they changed it after I said I liked it.DamianFMarble runwas good - shame they changed it after I said I liked it.CraigFMarble rundominated Jack who did not contribute muchMartinFMarble runMichael found difficulty sharing with Zac - each had their won half of the board.unknownFBeach buggyOh its a carChloeFBeach buggyvery good seats.CraigFBeach buggywanted to put guns on - no!ConnorFBeach buggydid bench seats all round.MariaFcardnot looking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelistEmilyFcardI told her to draw rather than write "love hearts & flowers" - got rather into it "Miss. I've got a pattern"
DamianFMarble runWas good - shame they changed it after I said I liked it.CraigFMarble rundominated Jack who did not contribute muchMartinFMarble runMichael found difficulty sharing with Zac - each had their won half of the board.unknownFBeach buggyOh its a carCholeFBeach buggyvery good seats.CraigFBeach buggywanted to put guns on - no!ConnorFBeach buggydid bench seats all round.MariaFcardnot looking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelistEmilyFcardI told her to draw rather than write "love hearts & flowers" - got rather into it "Miss. I've got a battern"
Craig F Marble run dominated Jack who did not contribute much Martin F Marble run Michael found difficulty sharing with Zac - each had their won half of the board. unknown F Beach buggy Oh its a car Chole F Beach buggy very good seats. Craig F Beach buggy wanted to put guns on - no! Connor F Beach buggy did bench seats all round. Maria F card not looking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelist Emily F card I told her to draw rather than write "love hearts & flowers" - got rather into it "Miss. I've got a battern"
Marbie run Michael found difficulty sharing with Zac - each had their won half of the board. unknown F Beach buggy Oh its a car Chloe F Beach buggy very good seats. Craig F Beach buggy wanted to put guns on - no! Connor F Beach buggy did bench seats all round. Maria F card not looking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelist Emily F card I told her to draw rather than write "love hearts & flowers" - got rather into it "Miss. I've got a battern"
unknown F Beach buggy On its a car Chloe F Beach buggy very good seats. Craig F Beach buggy wanted to put guns on - no! Connor F Beach buggy did bench seats all round. Maria F card not looking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelist Emily F card I told her to draw rather than write "love hearts & flowers" - got rather into it "Miss. I've got a pattern"
Chie F Beach buggy very good sears. Craig F Beach buggy wanted to put guns on - no! Connor F Beach buggy did bench seats all round. Maria F card not looking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelist Emily F card I told her to draw rather than write "love hearts & flowers" - got rather into it "Miss. I've got a battern"
Craig F Beach buggy wanted to put guns on - nor Connor F Beach buggy did bench seats all round. Maria F card not looking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelist Emily F card I told her to draw rather than write "love hearts & flowers" - got rather into it "Miss. I've got a battern"
Maria F card not looking at big book - is she listening? -Chris says her dad left at weekend. Me to table to support & ensure she did thelist Emily F card I told her to draw rather than write "love hearts & flowers" - got rather into it "Miss. I've got a pattern"
Emily F card I told her to draw rather than write "love hearts & flowers" - dot rather into it "Miss. I've dot a battern"
Copport E part Are you allowed to do your Mum & your Dad? Me, you is avoid to do your Cad?
Tarmin E card drew Maran's rotande, told to ut and the yes, texpected.
Tashini T card User megana tectanging - tota una du do nei own.
Limity i card frie drawing is the blue spiral denois pop-up at top, we, prompt to minimated intern,
general c i card
Carle F card two bins beer negative model. Reader in the other & folded "And I'm going to keep on going down in size"
Michael E card drawing the demo cards & mechanisms, bash't really designed something of bis own
Damian F card drawing a low to make - most writing - dito CRiag
Natasha F card hers opens like the booklet - 2 flaps revealing a central page
Fmily F card has carried on designing into box6
Mia F card the drawing is of card & pencils
Michael F card Me: what is your message? Max: A frog & a mouse

APPENDIX M Log Book Index

<u>Name</u>	<u>Class</u>	<u>Task</u>	Observation
Tasmin	F	card	put "best" in "make better" space
Martin	F	suitcase	Immediately meauring mac with a ruler
Ellie	F	suitcase	also get ruler to measure mac
Louise	F	suitcase	also get ruler to measure mac
Natasha	F	suitcase	also get ruler to measure mac
Lisa	F	suitcase	also get ruler to measure mac
Tasmin	F	suitcase	measuring Panda's arm
Maria	F	suitcase	done 3 small drawings & shown me -Natalie: You haven't done much have you?
Natasha	F	suitcase	measuring again
Martin	F	suitcase	measuring again
Ellie	F	suitcase	looking on
Natasha	F	suitcase	says "7"
Emily	F	suitcase	drawn Panda
Maria	F	suitcase	Now has mac at table & is measuring it
Ellie	F	suitcase	I am going to make a box because Mrs Reed taught us how to make a box.
Jolene	F	suitcase	l know what we could do, we could make him a little model car.
Connor	F	suitcase	I'm going to pretend he's just bought it from Tescos.
Havley	F	suitcase	Could make some sweets to go in his bag.
Randal	F	suitcase	wheels - you can get these bars & the bar goes up into the suitcase
Randal	F	suitcase	Skateboard with bag attached. Or thing that goes around the skateboard waving arms about to show me & Amy
Jolene	F	suitcase	Has folded over the edges of the piece of card she has cut out - wil the mac still fit?
Natasha	F	suitcase	drawing round mac onto card
Stacev	F	suitcase	shown me rectangle & mac - it's too small. What to do? Me - cut out another.
Maria	F	suitcase	Miss, it works, Me; good, You just need to cut out the other side now, don't you. Megan grins - seems to understand.
Stacev	F	suitcase	2 pieces of card, right size - mac sandwiched - how to join together? thinks she will use sellotape.
Jolene	F	suitcase	still too small - does she realise?
Jolene	F	suitcase	now playing with hers - making into letterbox, trolley etc
Maria	F	suitcase	I don't know how to make the handle. Me: draw round the inside of the sellotape & cut in half.
Noel	F	suitcase	Made one card one & realised it was too small - tried the mac in it
Kate	F	suitcase	made box by glueing strips to rectangle base
Jolene	F	suitcase	Me; does the mac fit? Have you tried it? - no to both
Kara	F	suitcase	has 2 rectangles -what next? me:stick them together. Katie not sure. Told her to look at Jade's
Carl	F	suitcase	has stuck 2 halves together - will mac fit? yes
Craig	۴	suitcase	has made really good bag - but will the mac fit? bit small - came apart - making another
Jolene	F	suitcase	now finished larger bag - it works
Maria	F	suitcase	about to abandon hers - disappointed
Lisa	F	suitcase	mine fits! mac slides right through tube-like structure
Stacev	F	suitcase	put bits of string at sides so it opens further to get mac in.
Ellie	F	suitcase	really large box
Natasha	F	suitcase	same as Beth only smaller
Emily	F	suitcase	cut out 2 very large squarish shapes - now making handle out of string
Chloe	F	suitcase	long cuboid object - flap at end to put things in & out Handle on top.
Lewis	F	suitcase	single piece of card folded up rather than 2 separate sides.
Michael	F	maze	did outside view - told to do top/down
Maria	F	maze	did outside view - told to do top/down
	•		

Name	<u>Class</u>	<u>Task</u>	Observation
Connor	F	maze	You can go out of the first maze and into the second maze
Craig	۴	maze	I need two parts joined by a bridge
Hayley	F	maze	doing the one on top left of paper.
Lisa	F	maze	Can I make slots - I can make slots
Randal	F	maze	Can we change it cos it's quite hard to make that.
Craig	F	maze	has two pieces joined by bridge
Connor	F	maze	making castellations on outside walls
Noel	F	maze	measuring his wall against the plan. Craig: You don't stick it on there, mate. {was he?] Nathan now cutting out more green to make it same length as yellow
Stacey	F	maze	has lost her way. I say "Look back at drawings - remind yourself of what you were ding" Katie - sorted her out - folded edge of card.
Ellie	F	maze	large box structures
Natasha	F	maze	large box structures
Carl	F	maze	large box structures
Emily	F	maze	checkerboard floor pattern - copied from Lauren?
Tasmin	F	maze	checkerboard floor pattern - copied from Lauren?
Lisa	F	maze	holes in card are for minotaur to got hrough - bits left are the bridges
Michael	F	maze	drawing on yellow.
Chloe	С	Pizza	Copied design onto pizza by drawing it on pizza. But did so independently - not sitting with others.
Alistair	C	Pizza	wants a deep crust pizza
Wendy	C	Pizza	Copied design onto pizza by drawing it on pizza.Sitting together.
Ellís	C	Pizza	Copied design onto pizza by drawing it on pizza. Sitting together.
Alice	C	Pizza	Copied design onto pizza by drawing it on pizza.Sitting together.
Robert	C	Pizza	Stuck things directly onto design sheet
Zara	C	Pizza	Stuck things directly onto design sheet. Wendy gave orange paper. Sh:"Is this really what I have to do it on?"
Zara	C	Pizza	Arranged matchsticks on pizza, marked their position "So if anyone knocks it fill know where they're meant to be"
Kirsty	C	Pizza	Me "Which one are you going to make?" S "The cheese one" Me "Which one is that?" S pointed to first. Me "What are the others?" S "Forgotten now"
Emma	C	Pizza	To Wendy "Is this pepperoni colour?" Wanted me to cut circles - now many? Counted the circles on her design drawing.
Gann	C	Pizza	Did not draw pizza on orange sheet, just cut out a freehand cicle. "It's not a circle." Me "Make it into one then." He cut a bit off so it was.
Gann	C	Pizza	i can't make it that size pointing to drawing.
Zara	C o	Pizza	I ned to put red tissue between match sticks (already stuck on) to be the sauce. To vvendy: "I want to do more detail."
Allce	Č	Pizza	Looking for something to be nam.
vvenay	C C	Easter	
matthew	Č	Easter	going to label materials for me
Dobort	ĉ	Easter	violated early office and teams from early to devide this. I don't think about that
Aliotair	ĉ	Easter	walked some only stores - got some from claim's store.
Rhiannor	ĉ	Easter	while is a hit like Sonhie's You can put the end inthere or there
7ara	č	card	has drawn rund the blue mechanism provided
∆lietair	č	card	has drawn noting nichture
Garth	č	card	Nichelle is talling him exectly what to do & showing him on a scran of paper
Cassy	c C	card	her flower vase is to be like the demo and with the wohly had
Sophie	č	card	her first drawing is of the green mechanism example
Peter	č	card	Can I write it? - he has done all writing
Kathy	č	card	Me - what's inside? K - I forgotto do the back of it
Chloe	č	card	A little arl on the front - words on inside left, picture on right
Alistair	č	card	For Gran & Grandad - they like birds - when you open it there will be a bird which springs out really quickly and come alive & fly away. It's going to come ou

Name	<u>Class</u>	<u>Task</u>	Observation
Jordan	С	card	front - Who's a monkey; inside - you!
Nicola	С	card	top bit is a flip-up football
Carys	С	card	has changed her design after working out how to make it
Robert	C	card	Me - what do you like about your card? R: the birthday cake; so he has done a range of ideas but not developed a final solution
Zara	С	suitcase	wanted to go & get her teddy & make a suitcase for her teddy
Lee	С	suitcase	sent outside for rudeness
Alistair	С	suitcase	Are we allowed to cut out the paper? - he has drawn a single decorated item. Got card - It can't be this big can it? waving card about
Robert	С	suitcase	is cutting up the paper
Alistair	С	suitcase	is using his drawng as a transfer
Lee	С	suitcase	shown me white paper - Shall I cut it out?
Peter	С	suitcase	doing all writing - no drawng
Wendy	С	suitcase	drawing & making a carry stick - you put things in & hang it over your shoulder
Holly	С	suitcase	has made a box shape
Lee	С	suitcase	came to me with cut out - it doesn't fit over Pandy's arm: "I've done a little hole" Me - Does the mac fit in it? - Don't know haven't tried it.
Lee	С	suitcase	brought the mac & white single sided cut out: "It fits" Me: What are you going to tdo now? - Make it properly
Lee	С	suitcase	Then went on to make a 2 sided bag - so was the cut out a prototype? But final product is much larger.
Chloe	С	suitcase	Pink paper stuck to cut out. Me: Did it fit? - I don't know I didn't try it.
Sophie	С	suitcase	single sided - Me: Does it fit? - yes - Me: What are you goingto do now? : Put little footballs, like on the drawing.
Chloe	С	suitcase	middle bit fits side bits don't. Me: Do you want to try again & make it bigger? - yes
Wendy	С	suitcase	single sided - fits over paw, went over shoulder - not happy - trying again.
Rhiannor	С	suitcase	super little bag but too small. I suggest making another one bigger
Peter	С	suitcase	folding the mac to try to fit onto white paper - it just about fits onto there.
Alice	С	suitcase	The mac fits"
Peter	С	maze	has drawn rope as well - really good drawing
Alistair	С	maze	wanted to do trap doors etc at the drawng stage - I nearly stopped him
Robert	С	maze	has drawn a maze on the card and a single door
Garth	С	maze	has shown me on the drawing which bit the card he has cut out is
Ellis	С	maze	I've just worked out how mine's going to work
Wendy	С	maze	Has cut out one piece of red - put glue on - doesn't like it - now drawn the plan on red card
Alistair	С	maze	brilliant "This is really, really hard. Look I've got all that to do. I'll have to do all that."
Kathy	С	maze	Will you draw the walls or make it? me: whatever you like K: I'm drawing it and then put walls
Zara	С	maze	I'm going to bend the walls so they stick to the walls so I don't have to stick them to the base
Lee	С	maze	Doing the walls, following his plan.
Zara	С	maze	This is for the crocodile to come in (blue) - Liam"It's a minotaur' Sh: No a crocodile in the garden; I'm doing a boat, walls & a door.
Kathy	С	maze	I'm going to make figures and maybe a little boat so he can get back [she has no internal walls] so I can put my bits inside.
Chloe	С	maze	Also only has walls - I'm going to make people
Jordan	С	maze	waterfall (blue) for minotaur to have a drink
Robert	С	maze	blue fan-folds - bones on top - where the minotaur has left bones
Ellis	С	maze	blue = gate
Chloe	С	maze	finished - drawn maze not made walls
Holly	С	maze	is 3D but doesn't look much like her drawing
Peter	С	maze	blue triangle in flat base "skulls"
Zara	С	maze	playing with it - walking the minotaur about

video Jan2001 - video

Time	<u>Child</u>	<u>falking</u> (<u>Saying</u>	Action1		
1	Noel	Craig	Discussing his drawing; explaining the features to Crai	(Pointing to parts of the drawing as he talks; runs pencil around valley		
1	Craig	Noel	you could do	Pointing with pencil on Noel's drawing		
1	Martin			not engaged with the other 2 boys.		
2	Noel	Craig	cont dicsission re drawing			
2	Craig	Noel	about his drawing			
2	Martin			still doing own drawing; no communication with others		
3	Randal			reaching for sellotape to wrap around rolled up newspaper		
3	Damian	Noel	discussing Damian's newspaper roll			
3	Noel	Damian		cutting length of wool to same length as drawing laying it on the drawing and carefully cutting it to same length		
3	Jolene	don't kno	prabout drawing	pointing at parts of drawing with pencil as she talks		
3	Natasha			got up to fetch other materials		
4	Randal	Damian	negotiating use of sellotape	using sellotape to fix newspaper rolls		
4	Natasha	Randal &	Rnegotiating use of sellotape	getting the sellotape from Randal & Damian		
4	Damian	Randal	negotiating use of sellotape	using sellotape to fix roll of newspaper		
4	Jolene			rolling up newspaper & fixing with sellotape	11 11	
5	Noel			looking at wool stuck to paper and considering what other children are doing - not sure he has done the rgiht thin	Ó	
5	Damian			fxing sellotape around rolls	Ũ	
5	Jolene	within ar	cnedotiations over use of sellotape	making newspaper rolls & fixing with sellotape	Ct .	
5	Natasha	within gr	cnegotiations over use of sellotape	making newspaper rolls & fixing with sellotape	~ 2	2
6	Randal			leaves table to fetch materials - returns with straws.	$< \tau$)
6	Damian	within ar	cnegotiations over use of sellotape	making newspaper rolls & fixing with sellotape)
6	Noel		· ·	looking at others working & at his own paper, realising others have not stuck things to design sheet		1
6	Natasha	within gr	cnegotiations over use of sellotape	making newspaper rolls & fixing with sellotape	XZ	i
6	Jolene	within gr	cnegotiations over use of sellotape	making newspaper rolls & fixing with sellotape		i
7	Craid	Jolene 8	concerning sellotage to fix newspaper rolls	wants sellotpe round model - Jolene does this for him whilst he holds it		, s
7	Jolene	Craig &	concerning sharing of sellotape	puts sellotape around Craig's model for him	$\mathbf{z} \times$	J A
7	Noel	me to No	You're not just making a picture of it, you're making a m			
7	Natasha			rolling and sellotaping second roll	$R \leq$	i
7	Martin	me to M	awhy dont you just use the whole cork?	off-side, cutting cork in half with scissors (so is Chris)	v)	
8	Damian			making newspaper roll		
8	Noel			has fetched sheet of newspaper to begin making model	9 19 19 19 19 19 19 19 19 19 19 19 19 19	
8	Jolene			now has 2 rolls completed	Q	
8	Martin	Chris	about cutting cork	both Chris & Martin are trying to cut cork in half with a pair of scissors - I have already suggested to Martin that	6 -9 ·	
9	Randal	Noel	Asks for the sellotape	From this point Randal and Damian develop thir ideas together		
9	Damian	Randal	use of sellotape	from this point Randal & Damian develop their ideas together		
10	Martin	me	explaining what the cork is for and what else is going to	cutting up the cork		
10	Natasha			cutting up straws, measuring them against each other & cutting them to the same length		
10	Megan			laid out a sheet of newspaper on the table, has some straws in her hand and is laving the straws out on the new		
11	Randal	Damian	about what Randal is going to use the straws for			
11	Damian	Randal	about what Randal is oing to use straws for and whether			
11	Jolene			tving string around newspaper roll		
11	Natasha	me	about the length of the straws	comparing the proportions of her straws to the drawing, considering them and going off to find some more		
11	Martin			still trying to cut up the cork (so is Chris)		

I.

Time	Child	[alking t	<u>saying</u>	Action1		
12	Randal			playing with the flattened and bent roll and rushing off to get more materials		
12	Noel			flattening newspaper roll		
12	Damian			watching and considering what Noel is doing with his newspaper roll		
13	Martin			still cutting the cork with Chris		
13	Megan	group	Whose is this?" as she picks up a newspaper roll - is	picks up Noel's roll.		
13	Randal	Damian	discussing straws and the cotton reel which they are fil	tlooking through bit box with Noel		
13	Noel			looking through bit box with Randal		
13	Damian	Randal	discussing straws and the cotton reel which they are fil	waiting for Randal & Noel to return from fetching materials		
13	Jolene					
14	Megan			picks up Noel's roll but it is taken from her		
14	Randal	Damian	about the cotton reel going down the roll	experimenting with fitting a cotton reel down the centre of the roll		
14	Damian	Randai	about the cotton reel going down the roll	experimenting with fitting a cotton reel down the foll		
14	Niegan		Lester at you little bealsof!	self-absorbed, include end of selfotape to wind found newspaper foil		
14	Jolene	me	Look at my little basket			
14	Natacha	mo	about measuring backet to the drawing			
15	Martin	IIIC	about measuring pasket to the thaming	and still cutting the cork. Lask if he is going to make anything, another child line in & comments. Martin still obliv	11	
15	Randal	Martin	about his cork cutting	looking at his newspaper roll and considering the straws and cofton reels inside	2	
15	Crain	TALCH CIT	about the corr cuting		ŭ	
15	Noel	Damian	about Neel's construction	looking at what Randal and Damian are doing	, Č	
15	Damian			fitting two straws end to end	\leq	D
15	Jolene			making pizza to go in the basket	<	σ
16	Randal	Damian a	Discussion re what to do with straws	Putting straw down centre of Noel's roll		σ
16	Noel	Randal 8	Discussion re what he is doing with the newspaper roll			m
16	Damian	Randal 8	Discussion re what to do with straws		ň	
16	Martin			putting a collection of materials on the table where he sat to do the drawing, despite the fact that Randal, Damiar	<u> </u>	1
16	Jolene			putting string on the basket	and the second	RECURA
17	Jolene			tying string - basket?	ລ	\times
17	Natasha			looking at the drawing & working out what to do	5	
17	Damian			waiting for Randal to return	ហ	Income in
18	Noel			waiting for Randal to return	Ô	
10	Joiene			lixing basket to ropeway	, 1929 1	
10	Nool			sent by Damian to fetch materials	\mathbf{O}^{*}	
18	Damian	Pandal 8	There's some over there?" directing them to find materia		entre de la constante de la co	
18	Martin	Nanuaro		aving out the newspaper on the table where Noel was working & stariting to out it up		
19	Damian			Waiting for Randal & Noel to return & then runs off to join them on other side of the room		
19	Martin			cutting newspaper		
19	Noel			returning to table with Damian		
19	Craid	me	I ask where he's been; indicates other table;has corks e			
20	Craig		· · · · · · · · · · · · · · · · · · ·	gone to fetch something		
21	Martin			beginning to roll up newspaper		
21	Craig			trying out the position of two lolly sticks on a toilet roll middle; has abandoned the cork structure		
21	Jolene			attaching basket to ropeway		
21	Martin			measuring his two newspaper rolls against each other		

Time	<u>Child</u>	falking t	<u>saying</u>	Action1		
21 22 22 22 23 23 23 23	Megan Randal Craig Damian Martin Randal Damian Noel	Damian Randal	about Damian's model "Yeah, Do that" asking Randal's opinion	walking across room with sellotape holding the flattened roll which he had at the start and manipulating it whilst looking at Jolene's finished product cutting through sellotape holding lollysticks & waiting for turn of sellotape to stick them on again measures his roll against Randal's cutting more newspaper to size (ready for rolling) putting two straws end to end & beginning to fix them togeter with sellotape		
23	Martin			laying three rolls on the table in bridge shape & starting to cut more newspaper		
24	Randal	Damian	discussing how they will use the lollysticks	Randal & Damian now have lolly sticks		
24	Damian	Randal	discussing how they will use the follysticks	Randal & Damian now have folly sticks		
25	Negan			wrapping sellotape around end of roll takes sellotape from Megan to use to join follusticks		
25	Damian			holding straws end on & looking at them		
25	Martin			cutting newspaper		
26	Martin			has cut hill-shaped piece of newspaper	andend	
26	Jolene			making a pizza to go in the basket		
26	Noel			sellotaping two straws together end to end	Ő	
26	Randal	Megan 8	about use of sellotape	negotiating & sharing sellotape	Ŋ	
26	Damian			brushing debris off his drawing so he can still see it.	K	Þ
21	Noel			ining more newspaper	Real Property lies	Ð
27	Megan			tryba to get the sellotabe started		ð
27	Martin			still cutting up lots of newspaper	Q.	m
27	Randal	Damian	about the rolls & straws etc		^w	
27	Damian			looking at Noel fitting straws up the rolled newspaper	<u> </u>	n
27	Noel			has gone around the table & has joined Randal & Damian; investigating fitting his two end-joined straws up the c		
27	Craig			tying string on to model	ഖ	$\boldsymbol{\times}$
21	Martin			aligned already cut shape of newspaper to another sneet, picked up pencil, checked & adjusted alignment and the		
28	Crain			holds out rope bridge to check be is happy with it	S	P. C. S.
28	Randal			ioining lollysticks end to end	Ω	
28	Damian			picks up roll and inspects it; he has bundle of lollystcks in one hand; observes what Randal is doing with his	19 13 155555 14 155555	
28	Noel			removes two rolls out of his way; assuming them to be Martin's he moves them in Martin's direction	Q A	
28	Jolene			close-up of pizza in basket	•••	
29	Martin			aligning two halves of a "hill" cut from two flat sheets of newspaper		
29	Damian			plying straws in his hands whilst watching; waiting for sellotape?		
29	Craig			has returned with two wheels, his cork structure and string; he is tying string onto th cork structure		
30	Craid	mo	claims be's finished. Lauen, where's the backet -how w	wants the selicitable to secure top piece to roll be cark construction prefere this to the two rolls be had earlier.		
30	Medan	me	ounto he o miloned, r query where o the paoret -100 W	cutting ap repeatings, notion much happen with the conclusion of preferences the to the two rolls field earlier		
30	Randal			attempting to stick sellotabe to end of straw		
30	Damian			aligning straw with roll; then waiting for sellotape		
31	Martin			tying string round end of one roll		
31	Megan			twists roll; wanders around; returns with piece of string		
Time	e Child	falking to	Saying	Action1		
----------------	----------------------------	------------	--	---		
31 31 31	Damian Randal Noel			unpeeling sellotape; has straws in other hand sticking straws to come out of end of roll now has three straws all sellotaped end to end		
32 32	Noel Damian			trying straw up inside of roll (same as Noel, but more tentative)		
32 32	Randal Craig	Noel	"It can go zoom" complete with action - this confirms th	fitting straws in the end of the roll so that they are poking out. Was this idea his & the others copied & improved		
33 33 33	Martin Damian Randal			trying to stolk the two halves of his hill together unsuccessfully - sticks to drawing instead fitting straws into roll; considers them & fiddles with pulling the straws in & out bending the straws down & prodding it on the table to see if this will work: the straws are the "legs" and the roll is		

Development of Quantitative Assessment Instrument

CONTENTS :

AN1 Ex	cploratory Phase Databases	399
	Introductory Explanation	
	AN1.1 Form view of Stan database	
	AN1.2 Spreadsheet of Stan data	
	AN1.3 Form view of Panda & Box database	
	AN1.4 Spreadsheet of Panda data	
AN2 St	ructured Phase Spreadsheets & Databases	405
AN2 St	tructured Phase Spreadsheets & Databases	405
AN2 St	ructured Phase Spreadsheets & Databases Introductory Explanation AN2.1 Individual Marking Grid	405
AN2 St	ructured Phase Spreadsheets & Databases Introductory Explanation AN2.1 Individual Marking Grid AN2.2 Technique Ticksheet 1	405
AN2 St	tructured Phase Spreadsheets & Databases Introductory Explanation AN2.1 Individual Marking Grid AN2.2 Technique Ticksheet 1 AN2.3 Purpose Grid	405
AN2 St	Tructured Phase Spreadsheets & Databases Introductory Explanation AN2.1 Individual Marking Grid AN2.2 Technique Ticksheet 1 AN2.3 Purpose Grid AN2.4 Technique Ticksheet 2	405
AN2 St	Introductory Explanation AN2.1 Individual Marking Grid AN2.2 Technique Ticksheet 1 AN2.3 Purpose Grid AN2.4 Technique Ticksheet 2 AN2.5 Purpose & Dimensions Continua	405

Development of Quantitative Assessment Instrument

AN1 Exploratory Phase Databases

Introductory Explanation

AN1 contains the databases created to quantify the observations and evaluations made of the Exploratory Phase whole school surveys of children's design drawing skills, as detailed in Section 3.3.2.

AN1.1 Form view of Stan database:

Table 2 in Section 3.3.2a shows the form view of the Flat Stan part of this database, together with an explanation of how it was used to attempt to analyse the child's capability with design drawing. AN1.1 shows both pages of the form view of one child's record for both Flat and Round Stan tasks. There were 371 records on the database.

AN1.2 Spreadsheet of Stan data:

The numerical data were collated from the Stan database into this spreadsheet, from which charts such as Charts 1-3 in Section 3.3.2a were created.

AN1.3 Form view of Panda & Box database:

This shows the form view of the same child's record from the Panda and Box database as that shown in AN1.1. The Panda database shows the addition of fields that were to aid my understanding of children's ability to engage in the juxtaposition of fantasy and reality, which became an emergent theme from the Structured Phase. The Box task is referred to briefly in Section 3.3.2 and Ex.6 in Section 3.2.4b is taken from this activity. Its function was as an introductory activity prior to Pandy's Suitcase. No product was made and results of analsys are not reported in the main text.

AN1.4 Spreadsheet of Panda data:

The numerical data from the Panda database were collated onto this spreadsheet, from which the prevalence of the Drawing Types identified through Stan98 could be seen. It was also realised through this analysis, that categories beyond Multi-design were needed.

Development of Quantitative Assessment Instrument AN 1.1 Form view of Stan database 1998

FIRST NAME: Ellis M/F: M COHORT CODE: 96X SECOND NAME: Munday CLASS 97/8: 2B

FLAT STAN: 6 F.EXP.; with supply teacher

F.CRITERIA: 2 F FITS: 1 FLAT: 1

F.DRAW SCORE: 2

F DR DET: 1 F PRE-DR: 1 F MULT DR: 0 F MULT DES: 0 F VIEWPOINTS: F EXPANDED: F LABEL:

F.RECORD MATS: ⁰ F DRAW MATS: ⁰ F WRITE MATS: ⁰ F COLOURS: ⁰ F INSTRUCTS: F EQUIP:

F.MAKE SCORE: 2 F MATS: 1 F RESEMB: 0 F EXACTLY: 0 F MK DETAIL: 1

ROUND STAN: 9 R.EXP: shown examples of design drawing

R.CRITERIA: 2 R FITS: 1 3DMODEL: 1

R.DRAW SCORE: 2 R PRE-DR: 1 R MULT DR: 0 R MULT DES: 0 R DR DET: 1 R VIEWPOINTS: R EXPANDED: R LABEL:

R.RECORD MATS: ² R DRAW MATS: ¹ R WRITE MATS: ¹ R COLOURS: ⁰ R INSTRUCT: R EQUIP:

R.MAKE SCORE: ³ R MATS: ¹ R RESEMB: ¹ R EXACTLY: ¹

if 2D - RMK.DETAIL:

FLAT STANS	1	4L		4M		ЗR		38		ЗW		2D		2P		2B		¥1		T		YR4		YR3		yr2		all Fla	ats
	scores		0%		0%		0%		0%		0%		0%		0%		0%		0%	s	scores		0%		0%		0%		
	0		0		0		0		0		0	0	0	0	0	0	0	0	0		0	0	0	0	0	0	0	0	0
	1		0		0		0		0		0	0	0	0	0	0	0	4	10		1	0	0	0	0	0	0	4	2
	2		0	1	3		0		0		0	0	0	5	18	0	0	3	8		2	1	2	0	0	5	6	9	3
	3		0	0	0		0	2	6		0	1	3	2	7	0	0	4	10		3	0	0	2	2	3	4	9	3
	4	2	14	1	3	1	3	4	13	4	13	4	14	4	14	0	0	4	10		4	3	7	9	10	8	10	24	9
	5	1	7	2	7	5	16	З	9	3	10	11	38	4	14	4	14	2	5		5	3	7	11	12	19	23	35	14
	6	4	29	8	28	8	25	8	25	7	23	6	21	З	11	8	28	6	15		6	12	28	23	24	17	20	58	22
	7	2	14	6	21	7	22	10	31	3	10	5	17	3	11	7	24	9	23		7	8	19	20	21	15	18	52	20
	8	1	7	2	7	7	22	3	9	9	30	3	10	З	11	4	14	6	15		8	3	7	19	20	10	12	38	15
	9	1	7	4	14	4	13	1	3	0	0	0	0	3	11	0	0	0	0		9	5	12	5	5	3	4	13	5
	10	2	14	2	7		0	1	3	2	7	0	0	0	0	2	7	0	0		10	4	9	3	3	2	2	9	3
	11	0	0	3	10		0		0	1	3	0	0	0	0	0	0	0	0		11	3	7	1	1	0	0	4	2
	12	1	7		0		0		0	1	_3	0	0	1	4	0	0	0	0		12	1	2	1	1	1	1	3	1
NO OF CH.		14		29		32		32		30		29		28		29		39				43		94		83		258	

AN1.2 Spreadsheet of Stan data

ROUND STANS	4H		4M		ЗR		3RC/	D	ЗW		38		2B		YR 1		all R	NDs	all S	tans
scores		0%		0%		0%		0%		0%		0%		0%		0%				
1		0		0		0		0		0		0	0	0	0	0	C	0	0	0
0		0		0		0		0	1	4		0	0	0	0	0	1	1	5	1
1	1	3	2	7		0		0	2	8	2	7	1	4	3	11	11	6	20	4
2	2	7	4	14	1	5	1	9	1	4	3	11	2	7	7	25	21	11	30	7
3	2	7	4	14	4	20	0	0	2	8	7	26	3	11	9	32	31	16	55	12
4	1	3	2	7	3	15	3	27	5	21	9	33	4	14	3	11	30	15	65	14
5	7	23	3	11	5	25	2	18	2	8	4	15	6	21	2	7	31	16	89	19
6	2	7	2	7	4	20	7	64	2	8	1	4	4	14	3	11	25	13	77	17
7	6	20	2	7	3	15	2	18	7	29	1	4	4	14	1	4	26	13	64	14
8	2	7	4	14		0		0	1	4		0	5	18	0	0	12	6	25	5
9	5	17	2	7		0		0	1	4		0	0	0	0	0	8	4	17	4
10	2	7	З	11		0		0		0		0	0	0	0	0	5	3	9	2
11		0		0		0		0		0		0	0	0	0	0	0	0	3	1
12																				
NO OF CH.	30		28		20		11		24		27		28		28		196		459	

Development of Quantitative Assessment Instrument APPENDIX N

Page 401

FLAT STANS				114		20		20		310/		20		20	·/	28		V1		Vr	1	Vr 3		Vr 2		<u>əll</u>	
FLAT STANS		"?L.	0%		0%	01	0%	00	0%	000	0%	413	0%	<u>~1</u>	0%	20	0%	11	0%	11	0%	6	0%	11.2	0%	an	0%
CRITERIA	0	2	14	4	14	0	0	2	6	1	3	1	3	3	11	0	0	7	18		5 14	1 3	3	4	5	20	8
OT CITES OF	1	4	29	12	41	2	6	19	59	10	33	24	83	18	64	õ	ō	14	36	16	37	7 31	33	42	51	103	40
	2	8	57	13	45	30	94	11	34	19	63	4	14	7	25	29	##	17	44	21	49	60	64	40	48	138	54
DRAW SCOR	EO	0	0	0	0	0	0	0	0	0	0	0	0	8	29	0	0	8	21	() 0	0 0	0	8	10	16	6
	1	1	7	4	14	З	9	2	6	1	3	1	3	2	7	3	10	4	10	ŧ	5 12	? 6	6	6	7	21	8
	2	5	36	6	21	11	34	9	28	8	27	5	17	6	21	18	62	12	31	11	26	28	30	29	35	80	31
	3	4	29	12	41	12	38	12	38	15	50	15	52	6	21	7	24	11	28	16	5 37	7 39	41	28	34	94	37
	4	4	29	7	24	6	19	9	28	6	20	8	28	6	21	1	3	3	8	11	26	21	22	15	18	50	20
REC MATS	0	7	50	14	48	28	88	30	94	25	83	22	76	14	50	27	93	37	95	21	49	83	88	63	76	204	80
	1	4	29	6	21	4	13	2	6	2	7	6	21	8	29	2	7	2	5	10) 23	8 8	9	16	19	36	14
	2	1	7	3	10	0	0	0	0	3	10	1	3	5	18	0	0	0	0	2	9	3	3	6	7	13	5
	З	2	14	6	21	0	0	0	0	0	0	0	0	1	4	0	0	0	0	8	19	0	0	1	1	9	4
MAKE SCORE	Ξ0	2	14	4	14	0	0	4	13	2	7	4	14	3	11	0	0	6	15	(5 14	6	6	7	8	25	10
	1	3	21	3	10	5	16	5	16	2	7	15	52	9	32	1	3	12	31	e	5 14	12	13	25	30	55	21
	2	3	21	9	31	18	56	11	34	17	57	8	28	13	46	14	48	4	10	12	28	46	49	35	42	97	38
	З	4	29	10	34	2	6	12	38	5	17	З	10	2	7	8	28	12	31	14	- 33	19	20	13	16	58	23
	4	1	7	3	10	5	16	0	0	4	13	0	0	1	4	6	21	4	10		9	9	10	7	8	24	9
ROUND STAN	VS	4H		4M		3R		3RC/	ND	ЗW		38		2B		Y1 T	·P	1H		Yr.4		Yr.3	·,	Yr.1		all	
			0%		0%		0%		0%		0%		0%		0%	patra.	0%		0%		0%	6	0%		0%		0%
CRITERIA	0	0	0	7	25	3	15	0	0	3	13	18	67	0	0	1	4	2	7	7	25	5 6	28	3	11	16	12
	1	13	43	12	43	17	85	4	36	20	83	8	30	7	25	12	43	4	14	25	86	5 41	205	16	57	82	59
	2	17	57	9	32	1	5	5	45	1	4	2	7	21	75	2	7	8	29	26	89	ə 7	55	10	36	43	31
DRAW SCOR	EO	4	13	1	4	0	0	0	0	2	8	0	0	1	4	З	11	2	7	5	17	7 2	8	5	18	12	9
	1	1	3	1	4	5	25	1	9	0	0	4	15	З	11	6	21	6	21	2	7	76	34	12	43	20	14
	2	16	53	18	64	2	10	3	27	9	38	11	41	21	75	5	18	2	7	34	118	3 14	75	7	25	55	40
	3	4	13	4	14	12	60	4	36	8	33	10	37	2	7	0	0	З	11	8	28	3 24	130	З	11	35	25
	4	5	17	4	14	1	5	3	27	5	21	2	7	1	4	0	0	1	4	ç	31	9	53	1	4	19	14
REC MATS	0	4	13	3	11	11	55	0	0	12	50	25	93	10	36	8	29	14	50	7	24	1 23	105	22	79	52	37
	1	8	27	11	39	5	25	10	91	7	29	2	7	7	25	4	14	0	0	19	66	5 22	145	4	14	45	32
	2	17	57	4	14	4	20	1	9	5	21	0	0	11	39	2	7	0	0	21	71	10	50	2	7	33	24
	З	1	3	3	11	0	0	0	0	0	0	0	0	0	0	0	0	0	0	4	14	1 0	0	0	0	4	3
MAKE SCORE	ΕO	0	0	10	36	0	0	2	18	3	13	2	7	1	4	0	0	2	7	10	36	5 5	31	2	7	17	12
	1	11	37	4	14	8	40	4	36	7	29	7	26	16	57	8	29	9	32	15	51	19	106	17	61	51	37
	2	10	33	4	14	8	40	3	27	8	33	14	52	4	14	З	11	З	11	14	48	19	101	6	21	39	28
	3	9	30	10	36	4	20	0	0	6	25	4	15	7	25	З	11	0	0	19	66	6 10	45	3	11	32	23
	4	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0) 0	0	0	0	0	0

Development of Quantitative Assessment Instrument

APPENDIX N

AN1.2 Spreadsheet of Stan data (cont.)

Page 402

APPENDIX N Development of Quantitative Assessment Instrument Form view of Panda and Box database 1999

CLASS 98/9: 3W

PANDA: 9 P_PRE_DR: 1

P_DRAW_SCORE: 3 P_MULT_DRAW: 1 P_MULT_DES: P_VIEWPOINTS:

SINGLE FACE: 1 3d DRAWING: P MAKE DIAG:

P_MEASURE: 1 P_LABEL: P_EXPAND:

P_MAKE_SCORE: 3 P_RESEMBLE: 1 P_EXACTLY: 1 P_MULTIMAKE: P_MADE_MEAS:

SINGLE_SIDED: FLAT_BAG: 1 BOX_BAG: BOX_SHAPE: SUITCASE:

P_INSTRUCTS: 3 P_STATE: 1 P_HOW_TO: 1 DESC_PRODUCT: 1 P WORDSEQ: 1 P PICSEQ: P DIAG: 1 P INTSR MEAS:

BOX: 5 BX PRE DR: 1

BX VIEW: ³ BX CHOICE VIEW: side

BX_TOP: 1 BX_FRONT: 1 BX_REAR: 1

BX_DRAW_MATS: 1 BX_1ST VIEW: 1 BX_TOP_MATS: BX_FRONT_MATS: BX_REAR_MATS:

BX_REC_MATS: 1 (extra to box) BX_DR MATS: BX_WRITE MATS: 1 BX_EXPAND: BX_INSTRU: BX_LABEL: BX_INSTR_WDS:

AN 1.4 Pan	aa s	Suit	case	1333	1										
	4E		4S		3P		3R		ЗW		2B		2D		
		0%		0%		0%		0%		0%		0%		0%	
P-MULTI-DR	14	50	18	53	13	48	31	100	20	67	19	61	19	66	
P-MULT-DES	4	14	2	6	7	26	10	32	3	10	2	6	3	10	
P-EXPAND	3	11	0	0	0	0	0	0	0	0	0	0	0	0	
P-LABEL	19	68	2	6	2	7	1	3	4	13	1	3	1	3	
P-MEASURE	6	21	17	50	5	19	0	0	17	57	0	0	2	7	
SINGLE-FACE	15	54	30	88	26	96	23	74	28	93	26	84	25	86	
3D-DRAWING	6	21	1	3	0	0	5	16	1	3	2	6	0	0	
MAKE-DIAG	9	32	3	9	0	0	4	13	1	3	1	3	1	3	
MULTI-VIEWS	8	29	4	12	1	4	3	10	0	0	0	0	1	3	
P-RESEMBLE	19	68	29	85	24	89	26	84	27	90	24	77	22	76	
P-EXACTLY	10	36	12	35	12	44	19	61	15	50	14	45	14	48	
P-MULT-MK	3	11	5	15	2	7	2	6	0	0	4	13	1	3	
P-MD-MEAS	2	7	11	32	1	4	0	0	5	17	0	0	0	0	
SING-SIDED	0	0	1	3	5	19	4	13	1	3	7	23	13	45	
	15	54	16	47	14	52	18	58	18	60	19	61	13	45	
BOX-BAG	q	32	15	44	6	22	1	3	7	23	1	3	2	7	
TRUE BOX	2	7	10	3	ŏ	0	2	6	5	17	1	3	1	3	
SUITCASE	2	7	0 0	õ	ō	Õ	2	6	ō	0	1	3	Ó	Ō	
00110/102	Garg	•	•	~	, in the second s	_	x	X	_			_	_		
P-STATE	15	54	33	97	21	78	16	52	28	93	27	87	17	59	Instructions
P-HOW-TO	11	39	23	68	12	44	5	16	13	43	15	48	11	38	How-to
DESC-PROD	10	36	17	50	9	33	3	10	10	33	8	26	8	28	Accurate
WORD-SEQ	6	21	19	56	11	41	10	32	20	67	8	26	7	24	Words
PIC-SEQ	1	4	2	6	4	15	1	3	6	20	4	13	1	3	Picture
P-DIAG	6	21	3	9	1	4	0	0	8	27	5	16	4	14	Diagram
P-INST-MEAS	3	11	6	18	0	0	0	0	3	10	1	3	0	0	Measurements
TOTAL CH	28		34		27		31		30		31		29		

Development of Quantitative Assessment Instrument

APPENDIX N

AN 1.4 Panda's Suitcase 1999

Page 404

Development of Quantitative Assessment Instrument

AN2 Structured Phase Spreadsheets & Databases

Introductory Explanation

AN2 illustrates the development of the quantitative analysis instrument used in the Structured Phase of the research. Throughout the evolution of the analysis instrument, the individual child's record comprised a three paged spreadsheet: page 1 was the Grid (AN2.1, 2.3 or 2.5), a Collation sheet onto which the data on the grid was collated, and the Ticksheet on page 3.

The records shown are of one child, Craig, from the Focus Class Target Group, whose work can be found in Appendices C - L. All recording was done through entry of 1 = yes.

AN2.1 Individual Marking Grid:

This is the first page of the first version of the analysis instrument, as shown in Table 9, Section 5.4.1b, but with the recording cells that were used to collate the numerical data. The columns on the Grid represent the features of the Drawing Types and the rows represent the Aspects of design drawing, to create a grid of descriptor cells. Within each descriptor cell, six *entry cells* were used to record the occurence of the descriptor in relation to the drawing produced in response to each Assessment Task. As in all versions of this Grid, the letters P,F,E,C,S & M refer to entries for each of the Assessment Tasks (Pizza, Frosty, Easter, Card, Suitcase and Maze). Placing all tasks on one sheet in this way reduced the spreadsheet to three pages, whilst still being able to see the criteria against which each drawing was to be evaluated. Since the Grid was revised in Spring 2001, only the Pizza and Frosty tasks are recorded on this example of it.

AN2.2 Technique Ticksheet 1:

This first version of the Ticksheet is as shown as Table 10 in Section 5.4.1b.

AN2.3 Purpose Grid:

This second version of the Grid is as shown in Table 12, Section 5.4.2a, with the addition of the sub-cells by which the child's achievements were recorded for each task. Since the analysis instrument underwent major rethinking, again, not all tasks are recorded on this example.

AN2.4 Technique Ticksheet 2:

This second version of the Ticksheet is shown as Table 15 in Section 5.4.3b(iii). The analysis in Section 5.8 was based on the collation of these Ticksheet records. To 1=yes and $0 = n_0$, were added x= not possible based on previous 0 and N= not applicable to task.

AN2.5 Purpose & Dimensions Continua:

This shows the final version of the analysis instrument, as shown in Tables 13a & 14, Section 5.4.3b(i), with the addition of the recording cells.

AN2.6 Collation Grid:

This was page 2 of the individual child record and onto which all data for each child were collated. Each *entry cell* of the Marking Grid or Continua related to a *collate cell* on this page of the spreadsheet. Thus an entry under "drawing a picture of an object, not designing" for the Pizza Assessment Task, for example, would result in the appearance of a 1 in the first row of the first column on the Pizza Collate Grid. This created a Grid for each task, although it lost the advantage of being able to see the criteria. The Total column on each Collation Grid weighted the collation cell entries according to the numerical scale show in Tables 13a & 14. This created a profile for each child for each task that was then collated into class records on which the spreadsheets shown in Appendix O were based.

AN 2.1 Individual Marking Grid : Craig

	PICTURE			1	SING	F-DR	NAN		T	MU	LTI-	DRA	W		T	MU	JI TI-	DES	IGN		T	PR	ŌĠ	RES	SIVE	02002.0000000000	T	IN'	TFR	ACTI	VF	
GENERATING &	drawing a pictur	e of an		single	sim	ole			sev	eral	atten	nots I	o im	prove	sev	eral	unrel	ated	draiv	/nas	bro	ares	sion	of id	eas		US	es di	rawin	as re	flecti	velv
DEVELOPING IDEAS	object not desig	nina		drawi	nd				dra	wina	of si	nale	idea		sho	wind	i rano	le of	ideas	3	acr	055.0	draw	inas			to	aene	irate i	aew ir	deas	
	a product	, ang									01 01						, ,	ye o,										30110	i alto i	1911 1	1040	
	PEEC	S	M	p i	F F	C	5	M	P	F	F	С	S	М	p	F	F	С	S	М	p	F	F	C.	S	N/I	p	F	F	C	S	м
		- <u>1</u>	T	† j	`	T	Ť	- <u>1</u>	<u> </u>	т <u>́</u>	_ <u></u>	Ť	Ť		1 1	T 1		Т ~ -	Ť	T	f	† –	<u> </u>	1	Ť	191	-ť	Ť		Ţ	Ť	
	not evoluting		.l	minin	al us	a of di	awin	<u>а</u>	aim	ing t) imr	TOVA	the	drawin	dhrai	insto	rmin		o nar)er	1100	s dr	awin	ator	leve!	lon a		mhin	as nr		s ide	as with
	decian problems			toev	Jore i	doae	Contra 1	9	not	avnli	vae	alutic	ne	Ci ci i i i i	harith	nut	davab	nnin	u a ci	nution	das	ian e	solul	ion	10101	Jop u	noi	1000	os to	nrod	unat	ha haei
FROBLEM	design probleme	3		IU CA	1010 1	1600				enpi	JIG 0	oradic	113			lout	4640	opin	g a si	Julion	luca	gna	20101	1011			sol	ution	3010	prou	100 (ne pesi
	PFEC	S	М	PI	E E	С	s	М	P	F	Е	С	s	М	Р	F	Е	С	S	М	P	F	E	С	S	М	Р	F	E	С	S	М
		1	T	† †			T		† · · ·	Ť	1	T	1	T	1	1	1	T	T	T	Ť.	Ť	1	1	Ť	T	1-	Ť	T	T	T	T
ADDRESSING	minimal underst	anding	-	draw	ng co	nveys	partia	al	dra	wing	show	ws ur	nders	standin	gsev	eral	ideas	for	satisf	ying	the	clier	nt's r	needs	and	I wants	з the	clier	nt's n	eeds	and	wants
CLIENT NEEDS	of client needs a	Ind		unde	stand	ing of	add	ressin	dof tl	he ne	eds	of th	e clie	ent	clie	nt ne	eds	are r	ecord	ed	are	con	side	ed as	3 the	desig	nare	trea	ited a	s par	t of	
	wants			client	need				1						but	not o	devel	opec	1		pro	ceed	s				the	itera	ative	proce	SS	
	PFEC	S	Μ	PI	E	С	S	Μ	Ρ	F	Е	С	S	M	Ρ	F	E	С	S	М	P	F	E	С	S	M	P	F	E	С	S	M
					1										1	<u> </u>								1					T			
APPEARANCE OF	views the drawir	ig as		minin	nal co	nsider	ation	of	only	/ one	ove	rall fir	nishi	ng	exp	erier	nents	s with	1		idea	as at	oout	finist	ies a	ire	ide	as al	bout (finish	es d	evelop
PRODUCT	the product			final a	ppea	rance	of pro	oduct	sch	eme	cons	sidere	ed		sevi	eral i	finish	ling s	schen	ies	add	ed to	o de	sign d	Jurin	g	inte	eracti	ively '	withir	i ove	rall
				being	desig	ned															the	drav	ving	stage	;		des	sign (devel	opme	ent	
	PFEC	S	Μ	P I	E E	С	S	М	Р	F	E	С	S	M	P	F	E	С	S	М	Р	F	Е	С	S	M	Р	F	E	С	S	Μ
						1									1									1								
COMMUNICATING	use of narrative	or		minin	nal rec	ording	g of d	esign	sev	eral o	irawi	ings	ហ		quic	ck sk	etch	es of	a		con	veys	ser	se of	obje	ect	cle	arly c	conv∉	eys id	eas (about
IDEAS	other drawing ge	enre		ideas					san	ne ide	∋a				rang	ge of	idea	s			to b	e ma	ade,	e.g. I	by lai	belling	, obj	ect tr	o be i	made	e.g.	
																					inst	ructi	ons	etc			mu	ltiple	view	pints		
	PFEC	S	M	PI	<u> </u>	Ç	S	M	P	F	E	С	S	<u>M</u>	P	F	E	С	S	M	P	F	E	C	S	M	P	F	E	<u>C</u>	S	M
			1						1		L					1				<u> </u>				1								
PLANNING	not planning to r	nake		no ev	idence	e of m	ateria	ils or	min	imal	cons	sidera	tion	of	indi	cates	s whi	ch ic	lea wi	11	mat	erial	s or				cor	istru	ction	al iss	ues	
CONSTRUCTION	the object drawr	1		const	ructio	n issu	es		con	struc	tion				be r	nade	e, but	not	how		con	istru	ctior	n feat	ures	i	cor	iside	red e	n rou	te to	final
																					sho	wn c	on dr	awing	J		des	sign				
	PFEC	S	M	P I	<u> </u>	C	S	М	P	F	E	<u> </u>	S	M	P	F	E	C	S	M	P	F	E	C	S	M	P	F	E	<u>C</u>	S	<u>M</u>
				1	1			<u> </u>			<u> </u>	1						1		L	 			1								1
EVALUATING	no evaluation			minin	nal eva	aluatio	n		reje	cted	early	ater	npt(s	s)	con	sidei	red a	nd			dec	ision	is m	ade a	bout	the	cha	anges	s mar	de as	a re	sult of
WHILST DRAWING									at d	rawir	۱g sii	ngle i	dea		reje	cted	rang	e of	ideas		obje	ect w	hilst	draw	/ing		refl	ectin	ig upr	on pre	eviou	S
									LINE COLUMN]												des	sign d	drawi	ngs		
	PFEC	S	M	PI	<u> </u>	<u> </u>	S	M	P	F	E	C	S	M	P	F	E	C	S	M	P_	F	E	<u> </u>	S	M	Р	F	_ <u>E</u>	<u>_C</u>	S	M
			1						Ļ		<u> </u>				1	1		<u> </u>			ļ			1	\bot			L				<u> </u>
RELATIONSHIP	making an object	t is		minin	nal rela	ations	hip be	etweer	lopie	ect m	ade i	is the	san	ne	obje	ect m	ade i	is on	е		prog	res	sion				use	s dr	awing	js as	reso	urce
TO MAKING	viewed as a sep	arate,		drawi	ng an	d mak	ing		as t	he ol	oject	draw	/n		of th	ne id	eas d	Irawi	n		fron	n dra	wing	g into	mak	ling	dur	ing r	nakin	g		-
	new activity]																	
	PFEC	S	Μ	PI	<u> </u>	<u>C</u>	S	M	Р	F	E	<u>_C</u>	S	M	P	F	E	C	S	M	Р	F	E	C	S	M	P	F	E	C	S	M
nimitat			1							1													1 .						1		1	

Development of Quantitative Assessment Instrument APPENDIX N

Page 406

Development of Quantitative Assessment Instrument

AN 2.2 Technique Ticksheet 1: Craig

	PIZZA	FROSTY	EASTER	CARD	SUITCASE	MAZE
SATISFIES CRITERIA OF DESIGN BRIEF				·		
as set by teacher	·	ann - 77 77 a 1800 ann an 1810 ann 1710 ann	8-70-14-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0-0			
begins from task requirements	1	1	1			
maintains task requirements into msking	1	1	1		1	
as required by client		·!	<u> </u>	L		
begins from client needs		1	1	1	1	
maintaine aliant peads into making		4	1		++	
	TUDEO	<u> </u>	<u> </u>	L.,		
RELATIONSHIP BETWEEN WORDS & PIC	IURES	4	1		1	
bictures only	}.		4			
minking recorded mostly in pictures		<u> </u>	l			
Ininking recorded mostly in words		+				
single words or phrases relating to picture			1			
list (e.g. of materials)						
full sentences to describe planned product		<u> </u>				
labelled diagram (with) arrows or lines			1			
words/pictures interact to record process						
engthy verbal explanation						
instructions - words only						
instructions - pictures with few/no words						
instructions - labelled diagrams		1		<u> </u>		
FEATURES OF THE DRAWING						
appropriate level of detail	1	1	1			
shows dynamics/movement			1			
various viewpoints						
expansions to show small details						
cut away diagram to show inside of product		1	·····			
idicates how parts will fit together		<u>}</u>	1			
APPEARANCE OF PRODUCT (FINISH)						
indicates pattern or motif	1	X	1	Ţ	1	
indicates colour		X				
considers more than one finish	1	T X		<u> </u>		
justifies choice of finish	1	Y	1			
	L	^	1	1		
Indicatos materials by list		NAMES OF A DESCRIPTION OF	}	T		······
indicates materials by list						
molicates materials by labeling drawing						
materials indicated are available			1			
indicates cuts, folds &/or fixings			ļ	<u> </u>		
indicates measurements						
indicates equipment needed			1	1		
indicates parts to be assembled	<u> </u>		1	<u> </u>		
can be made in time available (roughly)	<u> </u>		1	ļ		
technically realistic (child can make it)	1	<u> </u>	11	<u> </u>	<u> </u>	
EVALUATING WHILST DESIGNING						
choices related to appearance of product	1	<u></u>				
choices related to design specification	1	1	1			
choices related to client needs	1	1	1	1		
choices related to material constraints			1			
choices related to construction			1			
justifies choices made	[
RELATIONSHIP TO MAKING						
same/adapted object	1	1	1			
same/adapted shape	1		1	1		
same/adapted size/proportions	1	David and a static	1	1		[
same/adapted colour				1		
same/adapted pattern decoration nicture etc.	1		1			
same/adapted pattern, decordition, poldre etc		-	1			
came/adapted figing technique	<u> </u>	+	4			
pamerauapteu nxing teorinique				+		<u> </u>
able to justify changes	1	1	1 1	1		1

All a.o I dipor	20 OI DIAWING OIL	v. viug				
	PICTURE	SINGLE-DRAW	MULTI-DRAW/		PROGRESSIVE	INTERACTIVE
CHILD'S VIEW OF	Draw the object	Record an idea of what mic	Make the drawing as good a	Brainstorming	Develop chosen idea and	To work out what will be
PURPOSE OF THE	and the object.	be made, to show	as nossible before		idicate how it might	made and how to make it.
DRAWING		the teacher.	showing the teacher.		be made.	
GENERATING &	drawing a picture of an	single, non-stereotypical.	several attempts to improve	several unrelated draiwngs	progression of ideas	uses drawings
DEVELOPING IDEAS	object not designing	drawing	drawing of single idea	of a range of ideas	across drawings	reflectively
hale there are stated to make a	a product	and thing	and the good and a same	di a miligio di mono	adiovo alatituga	to generate new ideas
			1	1	1 1	
EXPLORING THE	design problems are not	minimal use of drawing	aiming to improve the drawir	brainstorming onto paper	uses drawing to develop a	combines previous
PROBLEM	adressed in the drawing	to explore ideas	rather than explore solutions	without developing a solution	design solution	ideas withnew ones
	())au (y)					to produce best solution
			1	1 1	1	
ADDRESSING	minimal understanding	drawing conveys partial	drawing shows understandir	several ideas for satisfying	the client's needs and wants	the client's needs and wants
CLIENT NEEDS	of client needs and	understanding of addressir	of the needs of the client	client needs are recorded	are considered as the desig	are treated as part of
	wants	client need		but not developed	proceeds	the iterative process
			1	1 1	1	
APPEARANCE OF	views the drawing as	minimal consideration of	only one overall finishing	experiements with	ideas about finishes are	ideas about finishes develop
PRODUCT	the product	final appearance of product	scheme considered	several finishing schemes	added to design during	interactively within overall
		being designed		-	the drawing stage	design development
				1	1 1	
COMMUNICATING	use of narrative or	minimal recording	several drawings to express	demonstrates range of ideas	conveys sense of object	clearly conveys ideas about
IDEAS	other drawing genre	of design ideas	same undeveloped idea	often through series of quick	to be made, e.g. by labelling	object to be made e.g.
		-		sketches	instructions etc	multiple viewpints
			1		1 1	
PLANNING	not planning to make	no evidence of materials or	minimal consideration of	indicates which idea will	drawing indicates	constructional issues
CONSTRUCTION	the object drawn	construction issues conside	construction	be made, but not how	consideration of materials or	considered en route to final
					construction features	design
			1 1			
EVALUATING	no evaluation relating to	minimal evaluation	rejected early atempt(s)	considered and	decisions made about the	changes made as a result of
WHILST DRAWING	designing the product		at drawing single idea	rejected range of ideas	object whilst drawing	considering and discussing
						design drawings
			1		1 1	
RELATIONSHIP	making an object is	minimal relationship betwee	object made is the same	object made is one	clear progression	uses drawings as resource
TO MAKING	viewed as a separate	drawing and making	as the object drawn	of the ideas drawn	from drawing into making	during making
	new activity		· · ·			
					1	
1						

Development of Quantitative Assessment Instrument APPENDIX N

AN 2.3 Purpose of Drawing Grid: Craig

Development of Quantitative Assessment Instrument

AN 2.4TECHNIQUE TICKSHEET 2	Crai	g				
	P	F	Ē	С	S	M
SATISFIES CRITERIA OF DESIGN BRIEF						
as set by teacher						
begins from task requirements	1	1	1	1		
maintains task requirements into msking	1	1	1	x		
as required by client				******		
begins from client needs	1	1	1	1		
maintains client needs into making	1	1	1	x	<u> </u>	1
RELATIONSHIP BETWEEN WORDS & PICTUR	ES	مىيىتى <u>م</u>	L	J		
pictures only	1	1	1	1		
thinking recorded mostly in pictures						1
thinking recorded mostly in words			1			
single words or phrases relating to picture			1		1	
list (e.g. of materials)				<u> </u>	1	
full contences to describe planned product					1	<u> </u>
abelled diagram (with) arrows or lines		<u> </u>	†		<u>†</u>	<u> </u>
worde/pictures interact to record process			+	<u> </u>	+	
words/pictures interact to record process					<u> </u>	
			+		+	
instructions - words only			4			
nsulucions - pictures with rew/no words			1			
		L	11	I	l	l
FEATURED OF THE DRAWING	4	1			<u> </u>	1
	1	<u> </u>	11	11		
snows dynamics/movement				1		
various viewpoints		ļ		1	<u> </u>	ļ
expansions to show small details		ļ		1		<u> </u>
cut away diagram to show inside of product						ļ
idicates how parts will fit together		<u> </u>	1	11	1	<u> </u>
APPEARANCE OF PRODUCT (FINISH)				T		γ
indicates pattern or motif	1	X	1	1		
indicates colour		X		ļ	ļ	ļ
considers more than one finish	1	X	<u> </u>	<u> </u>	ļ	ļ
ustifies choice of finish		X				<u> </u>
PLANNING CONSTRUCTION		·				. .
indicates materials by list						
indicates materials by labelling drawing			1			<u> </u>
materials indicated are suitable			1	1		
materials indicated are available			1	1		
indicates cuts, folds &/or fixings			1	1		
indicates measurements						
indicates equipment needed					1	
indicates parts to be assembled		1	1	1		
can be made in time available (roughly)		1	1	1	1	
technically realistic (child can make it)		1	1	1	1	1
EVALUATING WHILST DESIGNING						
choices related to appearance of product	1	Τ	1	11	1	T
choices related to design specification	1	1	1	1	1	-
choices related to client needs		1	1	1	1	
choices related to material constraints		<u> </u>	1	1	1	\uparrow
choices related to construction		+	1	1	+	+
ustifies choices made			+	+ '	+	+
				1	<u>.</u>	1
came/adapted object		1	1		1	Т
pamerauapted object		+	1		+	+
pame/adapted site/seconding		+		X	+	+
		+	+	X		+
same/adapted colour			+	X	+	+
same/adapted pattern, decoration, picture etc			$\frac{1}{1}$	X	+	
same/adapted materials			1	X	+	
same/adapted fixing technique	<u> </u>		1	X	+	
able to justify changes	1	1	1	X	1	

Development of Quantitative Assessment Instrument

AN2.5 Purpose & Dimens	ions Continua	NAME: Carl	CLASS: F

CF	HLD	SV	IEW	I OF	тн	EP	'UR	POS	E I	OF 7	THE	DR	AW	ING	3													
Vi	ews	the				Re	cor	d an	ide	a of	;	Us	ing -	dra	wing	to		De	vel	op ci	hose	ən id	lea	То	woi	rk ol	it wha	at
dr	awin	ig a:	s the	,		wh	at n	nigh	t be	e ma	de	rec	cord	de	sign			and	d in	dica	te h	ow it		wi	ll be	ma	de an	d
pr	odu	- ct						-				po.	ssib	ilitie	es.			mic	aht	be n	nade	9		ho	w to	ma	ke it	
ŕ	F	E	С	s	M	P	F	Ε	<u>c</u>	s	M	Ŕ.	F	E	С	S	M	P	F	E	C	S	М	P	F	E	<u>c</u> s	; M
							Ĺ				<u> </u>	1	1			1	1			1	1	<u> </u>						
GI	ENE	RAT	FING	i & I	DEV	EL	OPI	NG	IDE	AS								· · · · · ·										
dr	awin	g a	pictu	ıre,		sim	nple	ske	tch,	,		des	sign	ide	as			рго	gre	ssio	n of	idea	s	use	s di	rawii	ngs	
no pri	t des oduc	signi :t	ing a	3		shc be	owin mac	ıg ob de	vjec	t to		ger de\	herai /elor	ted ped	but i l	not		acri dra	oss win	s or v Igs	vithi	n		refi ger	ectiv vera	vely te no	to ew ide	as
Р	۴	E	С	s	M	P	F	Е	<u>c</u>	s	M	Р	F	E	С	s	M	Р	F	E	С	s	М	Р	F	Е	c s	; M
L											1	1	1			1				1	1							
E)		DRII	NG 1	THE	PO	SSI	BIL	ITIE	s c	DF T	HE	TAS	SK															
de	sign	pos	sibil	ities		ste	reot	ypic	al r	espo	onse	rec	ordi	ng	poss	sible		usir	ng	draw	ing	to		cor	nbir	ies r	novel	
ar	e not	t adı	ess	ed		sho	owin	ıg liti	tle c	creat	ive	cre	ativo	e so	olutio	m(s)	ł	dev	/elo	p no	vel			sol	utio	ns to) prod	uce
'n	the o	draw	/ing			tho	ugh	rt				to t	he t	ask	<			des	sign	ı solı	ition	ı(s)	1	inn	ovat	ive c	tesigr	ו
P	F	E	<u>c</u>	s	M	P	F	E	<u>c</u>	s	M	P	F	E	<u>_</u>	S	M	Ρ	F	E	c	S	M	P	F	E	<u>c</u> s	<u>\$ M</u>
	1		1		ļ	1	L	1		1		<u> </u>	1		1	<u> </u>	1				<u> </u>				1			
A	DR	ESS	SING	; TH	EC		STE	RAIN	VTS	<u>5 OF</u>	TH	E TA	ASK	<u>(</u>										L				
m	nima	al				dra	IWIN	g sh	10W:	S SOI	me	rec	ords	3 W	ay to	•		tasi	K CI	onsti	aint	S		tas	k co	instr	aints	are
μn	ders	stand	ling	of		unc	ders	tanc	ling	j of ti	he	add	ires	s ta	ask 8	/or		are	co	nsid	ered	as		trea	ated	as j	oart	
tas	sk / i	lser	nee	ds		tas	k co	onstr	ain	ts		clie	ent n	ieed	ds/wa	ants		the	de	sign	pro	ceed	IS	of i	tera	tive	proce	ss
P	F	<u> </u>	<u> </u>	<u>s</u>	<u>M</u>	P	F	E		<u> </u>	<u>M</u>	P	F	E	<u></u>	s	<u>M</u>	P	F	E	<u></u>	<u>s</u>	<u>M</u>	P	F	E		<u>; M</u>
	0.01	LIAL				VC			00		ICT	1	1	<u> </u>		<u> </u>	1	l	L	1	1				<u> </u>	<u> </u>		
	POOL	AILAA			.00	n C	<u>/Г 1</u>				1 2/2										4 6 m							
ah	pear	di lu	e or	d		DILLIN OF 1	e co Final	Insid	ield		_	OVE	ः व॥		h			lae	dS i	abou	L 1111	isne	s	iae	as a	uod.	t ninis dutu	nes
pr	oauc	x is	not				nnai	app	lear	ance	3	aea	cora	tive	: scn	eme	:	are	e ac	iaea	το c	iesig	ļn	lae(/eloj		.กเถ 	
CO	nsia	erec	2	~	3.4		proc	JUCI	~	~		COL	nsia:	ere	a	-		wn	nist 		wing			ove	srall -	desi	igning	
F		<u>–</u>	<u> </u>	$\frac{S}{1}$		۳_	۲	T T	$\frac{c}{1}$	5		P 1	+	Ě		S		۳_	F		T	5	<u>M</u>	٣	F	E T		5 M
C	OMN	AUN		TIN	G D	ESI	IGN	IDE	AS	<u>L</u>	-l	J		1			J	1	L	_1	J	-L	1	L	L	.i	<u> </u>	
us	e of	nari	rativo	2		sim	nple	unia	abe	lled		cor	nvev	is s	ome	sen	se	cor	ive	VS SE	ense	of.		cie	are	กอมเ	ah for	
or	othe	r dr	awin	a		ske	etch	(es)	: re	lvina	on	of e	obie	ct t	o be	mac	je:	ob	iec	t to b	e m	ade		sor	neo	ne e	lse to	
ae	nre			3		sh:	arec	1 me	anir	nas	••••	e a	i inc	tica	ating	mat	erial	er	7 14	nrkii	na d	iann	am	m	ake	the r	orodu	ct
P	F	Е	С	s	M	P	F	E	C.	S	M	P	F	E	<u> </u>	S	M	P	F	E	Č	S	M	Ρ	F	E	C §	3 M
						1	1			1	1									1	1							
P	AN	NIN	GC	ONS	TR	UCT		N																				
nc	it pla	nnir	ng to	,		mir	nima	al co	nsi	dera	tion	dra	awin	g ir	ndica	tes		dra	wir	ng de	emoi	nstra	ates	cor	nstra	uctio	nal is	sues
m: as	ake i dra	the o wn	objec			of o wh	con: nilst	struc drav	ction vinc	n 1		sor	me o nstri	con ucti	isider ion	ratio	n of	cor cor	nsic nstr	lerat uctio	ion d on	of		cor to	nside fina	ered	l en ro sian	oute
ρ	F	Е	С	s	М	Р	F	Е	С	s	М	P	F	Е	С	s	М	P	F	E	с	s	М	Р	F	Ε	C S	s M
F	İ	1	T	T	T	1	1	T	Ť	1	T	1	†–	T	Ť	T	1	1	Ť.	1	1	T	T	i	Ť.	T	Ē	
E		TAU	ING	WH	HLS	TC		WIN	١G	<u>-</u>	_ _	. .	J	<u> </u>			<u> </u>		.k		i		L.,	J	J		.ll	
Le	t to i	defir	ne th	e		mi	nim:	alev	alu	ation		COL	nsid	pre		d		der	risi	onsi	mad	e ah	nut	ch	ange	es m	ade a	is.
L.	sinr	ı tas	k	0		at	drav	vina	nh	350		rei	ecte	ol c	ande	of		nro	vdu	ct we	nilst	drav	vinn	rac	uit i	of oc	mside	rina
	oig.						a.a.	mg	P . 14			lide			ungo	•.								de	sinn	dra	wings	
p	F	F	C	s	М	D	F	F	С	2	M	p	ц Е	F	C	s	М	p	F	F	C	s	ßЛ		ھر ا	F	C S	-
ŀ	Ť	T	Ť	Ť	1	†_	†	Ť	Ť	Ť	11	1	$\frac{1}{1}$	Ť	$\overline{1}$	$\overline{1}$	T	t	Ċ	1	Ť	Ť	141	<u> </u>	Ļ.	T	ŤŤ	2 191
A	BAS	SIS F	OR	MA	KIN	G						7																
m	akin	g an Is a	obje	ect is	3	pro)duc	st rel	iate rde	sto din		ob	ject the i	ma ide	ide is ac dr	i one	2	cle	ar ('h #	jevel hrou	nqoi nh d	nent Irawi	na	use	as d	rawi	ngs a	is
ne	w a	.s a ctivit	зері У			dra	awir	ig		ч ш			ane i		10 UI	44411		int	:0 m	nakir	ig Ig		y	ma	ikina	Je u J	unny	
P	F	E	<u></u>	S	M	P	F	Ē	<u>c</u>	S	M	P	F	E	C	s	M	P	F	E	č	s	M	P	F	E	<u>C</u> §	<u>3 M</u>
H 4	1 4	1	1	1	1	1	1	1	1	1	1	1	1	1		1 1	1 1	1	1	1 1	1	1	1	1	1	1	1	3

				1	1	, , , , , , , , , , , , , , , , , , ,	d .
ADDRTASK CC			1			3	
LOOK OF PRO			1			3	
COMM IDEAS		1			L	2	
PLAN CONSTR		1				2	
EVAL W DRAW			1			3	
REL TO MAKIN	1					1	
TASK	Easter Egg	Holder			ى ئەلەر يەرىپىيە يەرىپىيە يەرىپە بەلەت يە بەرە ئەلەر يەرىپەر يەرە يەرە يەرە يەرە يەرە يەرە يەرە ي	TOTAL	
PURPOSE				1		4	
GEN& DEV IDE				1		4	
EXPL THE PRC		1				2	
ADDRTASK CC				1		4	
LOOK OF PRO				1		4	
COMM IDEAS				1		4	
PLAN CONSTR			[1		4	
EVAL W DRAW				1		4	
REL TO MAKIN				11		4	
							1 1
TASK	Pandy's Su	itcase		-	The second s	TOTAL	
PURPOSE			1			3	
GEN& DEV IDE			1			3	
EXPL THE PRO		1		<u> </u>	<u> </u>	2	
ADDRTASK CO		1		L		2	
LOOK OF PRO	1					1	

1

1

TOTAL

3

З

2

2

2 З

3

TASK	Frosty			*****		TOTAL
PURPOSE			1			3
GEN& DEV IDE			1			3
EXPL THE PRO			1			3
ADDRTASK CC			1			З
LOOK OF PRO	Х	Х	X	Х	Х	Х
COMM IDEAS		1				2
PLAN CONSTR		1				2
EVAL W DRAW			1			3
REL TO MAKIN	1					1

		the second s		the second se	the second s	and the second se
TASK	Card					TOTAL
PURPOSE				1		4
GEN& DEV IDE				1		4
EXPL THE PRO			1			3
ADDRTASK CC				1		4
OOK OF PRO					1	5
COMM IDEAS				1		4
PLAN CONSTR				1		4
EVAL W DRAW			1			3
REL TO MAKIN	Х	X	Х	Х	Х	Х

TASK	Maze					TOTAL
PURPOSE			1			3
GEN& DEV IDE		1				2
EXPL THE PRC			1			3
ADDRTASK CC			1		:	3
LOOK OF PRO	X	x	X	X	Х	Х
COMM IDEAS		1				2
PLAN CONSTR			1			3
EVAL W DRAW		1				2
REL TO MAKIN			1			3

EXPL THE PRO

Gill Hope (2004) Drawing as a Tool for Thought

AN 2.6 Collation Grid : Craig

1

1

1

1

1

COMM IDEAS

PLAN CONSTR EVAL W DRAW

REL TO MAKIN

CONTENTS

Introductory	[•] Explanation	
--------------	--------------------------	--

AO1 Collation of Numerical Data from Individual Child Records ... 414

AO1.1 Purpose of Design Drawing AO1.2 Purpose Continuum Scores - all children AO1.3 Drawing Types - all children AO1.4 Dimensions of Design Drawing AO1.5 Ticksheet Data

AO2	Children's	Records	419
-----	------------	---------	-----

AO2.1 Collation Sheets - all children

Introductory Explanation

Appendix O contains the spreadsheets from which the charts in Sections 5.5 - 5.-9 were produced.

AO1 is organised according to the 4 layers of the analysis instrument and contains the sheets onto which the data from each child's records were collated.

AO2 contains the Collation Sheets for all of the children, from which the sheets in AO1 were collated.

Quantitative Data Collation of Numerical Data from Individual Child Records: AO1.1: Purpose of Design Drawing

Sheet 1: Mean scores on Purpose Continuum

	Focus	Comparison
Pizza	2.35	2.20
Frosty	2.30	2.28
Easter	3.45	2.67
Card	3.39	2.47
Suitcase	3.30	2.71
Maze	3.00	2.45

Sheet 2: Purpose of Drawing Profiles Focus

	Fo	cus	Co	mparison	
Pizza	Scores	Nos.	0%	Nos.	0%
n=20	0	0	0	O	0
	1	0	0	0	0
	2	13	65	16	80
İ .	3	7	35	4	20
	4	0	0	σ	0
	5	0	0	0	0
Frosty	0	0	0	0	0
F n=23	1	1	4	3	17
C n=18	2	14	61	8	44
	3	8	35	6	33
{	4	0	0	1	6
	5	0	0	0	0
Easter	0	0	0	1	5
F n=22	1	0	0	0	0
C n=21	2	3	13	10	48
	3	8	35	4	19
	Ą	9	39	6	29
L	5	2	9	0	0
Card	0	0	0	1	5
F n=23	1	0	0	0	0
C n=20	2	4	17	11	55
	3	9	39	4	20
	4	7	30	2	10
[5	3	13	11	5
Suitcase	e 0	0	0	0	0
F n=23	1	0	0	1	5
C n=21	2	2	9	9	43
	3	15	65	6	29
	4	3	13	4	19
<u> </u>	5	3	13	0	0
Maze	0	0	0	0	0
F n=19	1	D	0	1	5
C n=20	2	6	32	9	45
1	3	9	47	10	50
	4	2	11	0	0
1	5	2	11	0	0

Sheet 3 : Growth in Understanding of Purpose across time

As differneces in mean scores									
1	Focus		Comparison	Difference					
Problem		0.70	0.17	0.52					
Product		0.95	0.51	0.44					
As %									
	Focus		Comparison	Difference					
Problem		13.91	3.44	10.47					
Product		19.09	10.29	8.80					

APPENDIX O Quantitative Data AO1.2 Purpose Continuum Scores

Name	Class	<u>Pizza</u>	Frosty	Easter	Card	Suitcase	<u>Maze</u>
Lisa	F	2	3	5	5	5	3
Mia	F	3	2	2	2	3	А
Louise	F	А	2	4	2	3	А
Ellie	F	2	2	4	4	4	4
Craig	F	3	3	4	4	3	3
Kate	F	2	2	3	3	2	Ā
Carl	F	2	1	3	3	3	2
Connor	F	2	2	3	3	3	3
Damian	F	2	3	3	5	3	3
Tasmin	F	2	2	2	3	3	3
Havleigh	F	3	2	4	4	4	4
Jolene	F	2	3	5	4	5	4
Kara	F	3	2	3	3	3	2
Michael	F	A	2	2	3	2	2
Maria	F	2	3	4	3	2 A	2
Martin	F	3	3	4	5	5	2 4
Natasha	F	3	3	4	4	ר ג	3
Noel	F	2	2	3	2	3	Δ
Lewis	F	3	2	<u>з</u> Д	3	2	3
Randal	F	2	3	3	3	3	2
Stacev	F	2	2	4	4	3	2
Emily	F	Ā	2	Å	?	3	3
Chloe	F	2	$\overline{2}$	3	4	ĩ	3
Alistair	C	2	1	3	2	2	2
Rhiannon	Ċ	2	3	2	3	4	3
Sophie	С	A	3	2	2	2	3
Cassy	С	3	3	3	2	2	3
Chloe	С	2	А	2	2	2	2
Emma	С	3	4	3	5	2	3
Wendy	С	2	2	2	А	3	3
Carys	С	2	2	2	4	2	2
Ellis	С	2	3	4	А	2	2
Garth	С	2	2	4	2	3	2
Holly	С	3	2	2	2	4	2
Jordan	C	2	1	3	2	3	2
Alice	С	3	А	4	3	3	3
Kathy	C	2	A	2	2	2	3
Lee	С	2	2	2	2	3	3
Matthew	Ċ	2	2	4	2	4	A
Nicola	С	2	3	4	4	4	3
Peter	Ċ	2	2	0	0	2	2
Robert	Ċ	2	1	2	3	1	1
Zara	Ċ	2	2	4	3	4	3
Kirsty	С	2	3	2	2	3	2

APPENDIX O Quantitative Data AO1.3 Drawing Types

Name	<u>Class</u>	<u>Pizza</u>	Frosty	<u>Easter</u>	<u>Card</u>	<u>Suitcase</u>	Maze
Lisa	F	3	4	5	5	5	4
Mia	F	4	2	3	2	4	А
Louise	F	А	2	5	3	5	А
Ellie	F	3	2	5	5	5	5
Craig	F	4	4	5	3	4	2
Kate	F	3	2	3	3	2	А
Carl	F	1	1	4	3	4	2
Connor	F	2	2	4	3	4	4
Damian	F	2	4	4	6	5	4
Tasmin	F	2	2	3	4	4	2
Havleigh	F	4	2	5	5	6	5
Jolene	F	2	4	5	5	5	5
Kara	F	4	2	4	3	4	2
Michael	F	A	2	3	4	2	3
Maria	F	3	4	5	4	5	3
Martin	F	4	4	5	6	5	5
Natasha	F	4	4	5	5	2	2
Noel	F	2	2	ĩ	2	3	Ã
Lewis	F	4	2	5	4	4	2
Randal	F	2	5	4	4	4	2
Stacev	F	2	2	5	5	4	4
Emily	F	A	2	Ā	4	4	5
Chloe	F	2	2	5	, 5	4	4
Alistair	Ĉ	2	2	5	5	Д	3
Rhiannon	C	r r	2 4	4	4	5	4
Sonhie	Č	A	4	3	3	2	3
Cassy	Č	лх Д	4	2	2	2	Д
Chloe	C	2	Δ	2	3	3	4
Emma	c	4	5		6	<u>з</u>	т Д
Wendy	Č	2	2	3	Δ	3	Δ
Carve	č	2	2- A	2	3	2	2
Filie	C C	2	-r /t	5	Δ	2	2
Garth	c	2	3	5	2	3	2
Ually	C	Л	נ ה	2	2	- 1 5	2
Tordan	C	4	2	4	2	<u>л</u>	2
Joruan	C	1	1	4	5	4	2
Ance	C	4	A	2	2	4	4
Kathy	C	2	A.	2	3	2	4
Lee	C	2	4	5 E	3	4. E	4 A
Matthew	C	2	<u>_</u>) 5	4) _	A
INICOIA	C C	2	4	5	2	2	4
reter	U C	2	2	0	0	2	2
Kobert	C	5	l	2	4	3	l
Lara	C	1	2	5	4	5	4
Kirsty	С	2	4	2	2	4	2

Collation of Numerical Data from Individual Child Records:

AO1.4 Dimensions of Design Drawing

Sheet 3: Percentages of children achieving each band on the Dimensions Continua

Pizza														
Focus	G Ex A	0000000	1 5 0 10	2 81 67 57	3 14 33 33 33	400000	5 0 0 0	Comparison G Ex A	0000	1 10 0 5	2 70 80 80	3 20 20 15 45	400000	5 0 0 0
	C P Ev M	00000	0 10 5 29	100 90 67 43	0 0 29 29		0000	C P Ev M	00000	10 10 10 10 35	40 90 90 85 45	0 0 5 20	00000	00000
Frosty	,													
Focus	G EX A C P EV M	000000000000000000000000000000000000000	1 4 4 4 4 4 4 4 17	2 61 61 57 96 87 61 30	3 35 30 35 0 9 30 52	4 0 4 4 0 4 0 4 0	5 0 0 0 0 0 0 0 0 0 0 0 0	Comparison G Ex A C P Ev M	• • • • • • • • • •	1 17 17 17 11 17 17 39	2 44 50 33 78 67 44 17	3 33 22 44 6 11 33 39	4 6 11 6 6 6 6 6	5 0 0 0 0 0 0 0 0 0 0 0 0
Easter	•													
Focus	G Ex A L C P EV M		1 0 0 18 0 0 14	2 14 41 45 9 41 45 27 23	3 27 32 27 14 18 14 23 27	4 55 27 27 45 27 27 27 45 36	5 500 14 14 14 50	Comparison G Ex A L C P Ev M	0 500000000000000000000000000000000000	1 0 5 33 62 10 5 14 25	2 48 81 43 10 38 24 57 40	3 14 0 10 19 29 33 10 30	4 33 10 14 10 24 38 19 5	5 0 0 0 0 0 0 0 0 0
Card														
Focus	G Ex A L C P Ev	0 00000000000000000000000000000000000	1 22 0 4 0 0 0 0	2 30 35 26 4 35 35 39	39 39 30 61 35 30 26	4 9 22 39 26 26 26 26 26	5 0 4 0 9 4 9 9	Comparison G Ex A L C P Ev	0 500000000000000000000000000000000000	10555555	2 58 58 47 0 53 63 63	3 21 21 37 58 26 21 16	4 11 11 5 32 16 11 11	5 555005
Suitca	ise													
Focus	G Ex L C P Ev M	• 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1 13 0 9 4 0 0 0 13	2 57 52 9 70 70 17 26	3 22 48 13 52 13 9 57 39	4 9 0 22 30 13 13 22 22	5 0 4 4 4 9 4 0	Comparison G Ex A L C P Ev M		1 0 5 29 48 10 10 5 24	2 38 71 52 14 62 57 48 38	3 38 14 5 24 10 10 29 24	4 24 10 14 14 10 19 19 14	50000000000000000000000000000000000000
Maze														
Focus	G Ex C P Ev M		1 0 5 21 0 0 0 37	2 37 60 37 63 53 53 21	3 32 30 21 21 42 32 26	4 26 5 16 16 5 11	5 5 0 5 0 5 0 5 0 5 0 5 0	Comparison G Ex A C P Ev M		1 5 0 3 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	2 45 47 40 80 45 45 20	3 50 47 30 15 50 50 35	4 0 5 0 0 0 0	50000000000000000000000000000000000000

AO 1.5 Ticksheet Data

Focus Class

Comparison Class

Sheet 1 : Generation & Development of Ideas

Pizza	Frosty	Easter	Card	Suitcas	Maze	Pizza	Frosty	Easter	Card	Suitcas	Maze
45.00	65.22	0.00	13.04	8.70	36.84	60.00	33,33	33.33	42.11	23.81	35.00
0.00	0.00	4.55	17.39	4.35	0.00	0.00	5,56	19.05	5.26	0.00	5.00
20.00	0.00	22.73	26.09	8.70	10.53	20.00	11.11	28.57	5.26	19.05	5.00
e 0.00	0.00	40.91	26.09	21.74	10.53	0.00	0.00	28.57	26.32	23.81	0.00
35.00	30.43	31.82	39.13	60.87	47.37	20.00	50.00	9.52	26.32	33.33	60.00
5 0.00	0.00	4.55	8.70	17.39	0.00	0.00	5.56	0.00	10.53	4.76	0.00
0.00	0.00	4.55	0.00	0.00	0.00	0.00	5.56	4.76	5.26	4.76	0.00
0.00	4.35	50.00	39.13	30.43	10.53	0.00	5.56	9.52	21.05	19.05	0.00
i 0.00	4.35	54.55	47.83	34.78	31.58	0.00	11.11	33.33	42.11	33.33	5.00
: 0.00	0.00	4.55	8.70	4.35	0.00	0.00	0.00	0.00	5.26	0.00	0.00
i 0.00	0.00	4.55	0.00	4.35	0.00	0.00	0.00	0.00	5.26	0.00	0.00
	Pizza 45.00 0.00 20.00 0.00 35.00 0.00 0.00 0.00 0.00 i 0.00 i 0.00 i 0.00 	Pizza Frosty 45.00 65.22 0.00 0.00 20.00 0.00 35.00 30.43 0.00 0.00 0.00 0.00 0.00 4.35 0.00 4.35 0.00 4.35 0.00 0.00 0.00 0.00 0.00 0.00	Pizza Frosty Easter 45.00 65.22 0.00 0.00 0.00 4.55 20.00 0.00 40.91 35.00 30.43 31.82 0.00 0.00 4.55 0.00 0.00 4.55 0.00 4.35 50.00 0.00 4.35 54.55 0.00 0.00 4.55 0.00 0.00 4.55 0.00 0.00 4.55 0.00 0.00 4.55	Pizza Frosty Easter Card 45.00 65.22 0.00 13.04 0.00 0.00 4.55 17.39 20.00 0.00 22.73 26.09 0.00 0.00 40.91 26.09 35.00 30.43 31.82 39.13 0.00 0.00 4.55 8.70 0.00 0.00 4.55 0.00 0.00 0.00 4.55 0.00 0.00 4.35 50.00 39.13 1 0.00 4.35 54.55 47.83 0.00 0.00 4.55 8.70 0.00 0.00 4.55 0.00	Pizza Frosty Easter Card Suitcas 45.00 65.22 0.00 13.04 8.70 0.00 0.00 4.55 17.39 4.35 20.00 0.00 22.73 26.09 8.70 20.00 0.00 40.91 26.09 21.74 35.00 30.43 31.82 39.13 60.87 0.00 0.00 4.55 8.70 17.39 0.00 0.00 4.55 8.70 17.39 0.00 0.00 4.55 8.70 17.39 0.00 0.00 4.55 0.00 0.00 0.00 0.00 4.55 0.00 0.00 0.00 4.35 50.00 39.13 30.43 1 0.00 4.35 54.55 47.83 34.78 0.00 0.00 4.55 0.00 4.35 0.00 4.35	Pizza Frosty Easter Card Suitca:Maze 45.00 65.22 0.00 13.04 8.70 36.84 0.00 0.00 4.55 17.39 4.35 0.00 20.00 0.00 4.55 17.39 4.35 0.00 20.00 0.00 42.73 26.09 8.70 10.53 0.00 0.00 40.91 26.09 21.74 10.53 35.00 30.43 31.82 39.13 60.87 47.37 0.00 0.00 4.55 0.00 0.00 0.00 0.00 0.00 4.55 0.00 0.00 0.00 0.00 0.00 4.55 0.00 0.00 0.00 0.00 4.35 50.00 39.13 30.43 10.53 1 0.00 4.35 54.55 47.83 34.78 31.58 0.00 0.00 4.55 0.00 4.35 0.00 0.00 0.00 4.55	Pizza Frosty Easter Card SuitcatMaze Pizza 45.00 65.22 0.00 13.04 8.70 36.84 60.00 0.00 0.00 4.55 17.39 4.35 0.00 0.00 20.00 0.00 4.55 17.39 4.35 0.00 0.00 20.00 0.00 22.73 26.09 8.70 10.53 20.00 20.00 0.00 40.91 26.09 21.74 10.53 0.00 35.00 30.43 31.82 39.13 60.87 47.37 20.00 5 0.00 0.00 4.55 8.70 17.39 0.00 0.00 0.00 0.00 4.55 0.00 0.00 0.00 0.00 0.00 0.00 4.35 50.00 39.13 30.43 10.53 0.00 0.00 4.35 54.55 47.83 34.78 31.58 0.00 0.00 0.00 4.55 0.00	Pizza Frosty Easter Card Suitca:Maze Pizza Frosty 45.00 65.22 0.00 13.04 8.70 36.84 60.00 33.33 0.00 0.00 4.55 17.39 4.35 0.00 0.00 5.56 20.00 0.00 22.73 26.09 8.70 10.53 20.00 11.11 0.00 0.00 40.91 26.09 21.74 10.53 0.00 0.00 35.00 30.43 31.82 39.13 60.87 47.37 20.00 50.00 0.00 0.00 4.55 8.70 17.39 0.00 0.00 5.56 0.00 0.00 4.55 0.00 0.00 0.00 5.56 0.00 0.00 4.55 0.00 0.00 0.00 5.56 0.00 4.35 50.00 39.13 30.43 10.53 0.00 5.56 0.00 4.35 54.55 47.83 34.78 31.58	Pizza Frosty Easter Card Suitcat/Maze Pizza Frosty Easter 60.00 33.33 33.00 33.33 33.33 30.00 5.56 19.05 4.35.00 3.0.43 31.82 39.13 60.87 47.37 20.00 5.56 <	Pizza Frosty Easter Card SuitcatMaze Pizza Frosty Easter Card 45.00 65.22 0.00 13.04 8.70 36.84 60.00 33.33 33.33 42.11 0.00 0.00 4.55 17.39 4.35 0.00 0.00 5.56 19.05 5.26 20.00 0.00 22.73 26.09 8.70 10.53 20.00 11.11 28.57 5.26 0.00 0.00 40.91 26.09 21.74 10.53 0.00 0.00 28.57 26.32 35.00 30.43 31.82 39.13 60.87 47.37 20.00 50.00 9.52 26.32 5 0.00 0.00 4.55 8.70 17.39 0.00 5.56 0.00 10.53 6 0.00 4.55 0.00 0.00 0.00 5.56 9.52 21.05 6 0.00 4.35 50.00 39.13 30.43 10.53 0.00	Pizza Frosty Easter Card SuitcatMaze Pizza Frosty Easter Card SuitcatMaze 45.00 65.22 0.00 13.04 8.70 36.84 60.00 33.33 33.33 42.11 23.81 0.00 0.00 4.55 17.39 4.35 0.00 0.00 5.56 19.05 5.26 0.00 20.00 0.00 4.55 17.39 4.35 0.00 0.00 5.56 19.05 5.26 19.05 20.00 0.00 40.91 26.09 21.74 10.53 0.00 0.00 28.57 26.32 23.81 35.00 30.43 31.82 39.13 60.87 47.37 20.00 50.00 9.52 26.32 33.33 5 0.00 0.00 4.55 8.70 17.39 0.00 0.00 5.56 0.00 10.53 4.76 0.00 0.00 4.55 0.00 0.00 0.00 5.56 9.52 21.05 19.05

Sheet 2: Annotation of drawings

oneer 2. Miniotation of arawingo	m -		-	~ ·	• • • • •		D ¹	F . 1		^	~	
pictures only	F1228 #####	91.30	50.00	13.04	60.87	36.84	100.00	88.89	28.57	10.53	61.90	60.00
single words or phrases relating to pic	0.00	8.70	0.00	52.17	21.74	52.63	0.00	5.56	19.05	21.05	19.05	30.00
labelled diagram (with) arrows or lines	0.00	0.00	22.73	13.04	17.39	21.05	0.00	5.56	33.33	10.53	4.76	0.00
list (e.g. of materials)	0.00	0.00	0.00	17.39	0.00	10.53	0.00	0.00	0.00	15.79	0.00	0.00
full sentences to describe planned pro	0.00	0.00	30.36	47.83	8.70	5.26	0.00	0.00	14.29	57.89	14.29	15.00
words & pictures used interactively	0.00	0.00	9.09	4.30	8.70	10.55	0.00	0.00	4.10	0.00	9.52	0.00
Sheet 3: Level of detail	Easter	Card	Suitca	se			Easter	Card	Suitcas	se		
various viewpoints	18.18	78.26	17.39				23.81	47.37	14.29			
expansions to show small details	22.73	8.70	8.70				0.00	5.26	0.00			
cut away diagram to show inside of pro	36.36	73.91	4.35				0.00	0.00	0.00			
indicates how parts will fit together	36.36	17.39	17.39				28.57	15.79	23.81			
Sheet 4: Decorative features												
	Easter	Card	Suitca	se			Easter	Card	Suitcas	se		
indicates picture, pattern or motif	40.91	69.57	60.87				33.33	94.74	42.86			
indicates colour	9.09	60.87	47.83				14.29	89.47	28.57			
considers more than one finish	13.64	21.74	30.43				4.76	15.79	14.29			
Sheet 5: Recording Materials & Co	Pizza	Frosty	Easter	Card	Suitca	Maze	Pizza	Frosty	Easter	Card	Suitcas	Maze
materials	0.00	0.00	31.82	60.87	17.39	26.32	0.00	5.56	61.90	42.11	19.05	10.00
construction	0.00	0.00	52.17	52.17	30.43		0.00	5.56	38.10	52.63	33.33	
materials:												
list	0.00	0.00	0.00	30.43	0.00	10.53	0.00	0.00	0.00	15.79	0.00	0.00
sentences	0.00	0.00	18.18	26.09	8.70	0.00	0.00	0.00	19.05	10.53	9.52	5.00
drawing	0.00	0.00	9.09	0.00	0.00	5.26	0.00	0.00	14.29	10.53	0.00	0.00
labelling	0.00	0.00	13.64	4.30	8.70	15.79	0.00	5.56	38.10	5.26	9.52	5.00
construction.	0.00	0.00	77 77	56 52	8 70	0.00	0.00	0.00	10.05	26.32	10.05	0.00
cuis, iolus, ixings	0.00	0.00	27.27	0.52	26.70	0.00	0.00	0.00	19.00	20.32	19.00	0.00
equipment	0.00	0.00	9.00	26.00	20.09	0.00	0.00	5.56	4.76	31.58	14.29	0.00
narts	0.00	0.00	45 45	30.43	26.09	0.00	0.00	0.00	28.57	26.32	33.33	0.00
	0,00	0.00	10.10	00.10	20.00	0.00	0.00	0,00	20.01	20.02	00.00	0.00
Sheet 6: Evaluation whilst Drawing	Pizza	Frosty	Easter	Card	Suitca	Maze	Pizza	Frosty	Easter	Card	Suitcas	Maze
attempts to improve drawing of single	20.00	0.00	18.18	17.39	4.35	10.53	20.00	0.00	28.57	42.11	14.29	5.00
considered a range of ideas	32.50	30.43	18.18	26.09	56.52	47.37	22.50	50.00	14.29	26.32	33.33	60.00
development of single initial idea	0.00	4.35	50.00	39.13	21.74	10.53	0.00	0.00	33.33	10.53	23.81	0.00
developing one of two/several ideas	0.00	0.00	13.64	0.00	8.70	5.26	0.00	0.00	0.00	0.00	4.76	0.00
complication of ideas	0.00	0.00	0.00	0.70	4.30	0.00	0.00	0.00	0.00	J.∠U	0.00	0.00
changes related to appearance of pro-	: 32.50	0.00	0.00	17.39	17.39	5.26	22.50	0.00	0.00	0.00	5.26	14.29
changes related to task specification	0.00	4.35	31.82	34.78	34.78	10.53	0.00	0.00	0.00	19.05	10.53	19.05
changes related to client/user needs	0.00	4.35	31.82	39.13	39.13	15.79	0.00	0.00	0.00	14.29	10.53	0.00
changes related to material constraint	e 0.00	0.00	9.09	4.35	4.35	5.26	0.00	0.00	0.00	0.00	10.53	14.29
changes related to construction issue	: 0.00	0.00	22.73	17.39	17.39	5.26	0.00	U.00	0.00	9.52	0.00	0.00
Sheet & : Relationship to Product	Pizza	Frostv	Easter	Suitca	:Maze		Pizza	Frosty	Easter	Suitca	Maze	
Different	35.00	30.43	13.64	26.09	42.11		30.00	33.33	38.10	9.52	50.00	
Same	45.00	39.13	50.00	43.48	10.53		40.00	22.22	33.33	61.90	30.00	
Adapted	15.00	34.78	45.45	34.78	47.37		20.00	44.44	33.33	33.33	25.00	

AO2.1 Collation Sheets

Focus Class

Girls							Boys						
Lisa	PIZZA FF	ROS	EASTI CA	RD (SUITC MAZI	Ξ	Craig	PIZZA FI	ROS E	EASTICA	RD	SUITC MA	λZE
PURPOSE	2	3	5	5	4 3	;	PURPOSE	3	з	4	4	3	З
GEN& DEV IDEA	2	З	4	4	4 4	ŀ	GEN& DEV IDEA	2	3	4	4	3	2
EXPL THE PROB	3	3	4	3	2 2	2	EXPL THE PROB	2	3	2	3	2	З
ADDR TASK REC	3	3	3	4	4 4	ŀ	ADDR TASK REC	2	3	4	4	2	3
PLANNING LOOP	2 X		4	3	зх		PLANNING LOOP	2 X	-	4	5	- 1 X	-
COMM IDEAS	2	2	5	4	3 4	I.	COMM IDEAS	2	2	4	4	2	2
PLAN CONSTR	2	2	5	5	4 3	3	PLAN CONSTR	2	2	4	4	2	3
EVAL W DRAW	3	3	4	4	4 3	{	EVAL W DRAW	2	3	4	3	3	2
BASIS FOR MAKI	1	3	4 X	·	4 1		BASIS FOR MAK	2	1	4 X	Ŭ	3	3
	,	-						-	•			-	•
Mia	PIZZARC	STY	ASTEICAI	ת בר	ITCASMAZE		Carl	PIZZARO	DSTA	STEFCA	RD I	TCASMA	ZE
PURPOSE	3	2	2	2	3 0)	PURPOSE	2	2	3	3	3	3
GEN& DEV IDEA	1	2	2	2	3 0)	GEN& DEV IDEA	2	2	3	4	3	3
EXPL THE PROB	2	2	2	2	з С)	EXPL THE PROB	2	2	З	2	3	3
ADDR TASK REC	2	2	2	2	2 0)	ADDR TASK REC	2	2	2	З	3	2
PLANNING LOOF	1 X		1	3	4 X		PLANNING LOOP	2 X		4	З	ЗХ	
COMM IDEAS	2	2	2	2	2 0)	COMM IDEAS	2	2	2	2	2	3
PLAN CONSTR	1	2	2	2	з ()	PLAN CONSTR	2	2	2	2	2	2
EVAL W DRAW	1	2	2	2	з ()	EVAL W DRAW	2	2	3	2	З	З
BASIS FOR MAK	1	2	2 X		2 0)	BASIS FOR MAK	3	2	1 X		1	3
			ACTE CON	~~ ′		-	0			OTTO A			
DUDDOSE	PIZZARC	, i c (ASTERUAL	<i>י ב</i> א	n Casiviaze	2	DUDDOSE	PIZZARU	JSTA	SIEFUR	2	n casivia o	2E
	0	2	4	4) \	PURPUSE	2	1	3	3	3	2
GEN& DEVIDEA	0	2	4	2	3 (, ,	GEN& DEVIDEA	2	1	3	3	3	2
EXPL THE PROB	0	2	2	2	2 (ر د	EXPLINE PROB	3	1	4	3	3	4
ADDR TASK REC	0	3	4	2	3 (נ	ADDR TASK REC	2	1	3	3	2	1
PLANNING LOOP	ΟX	_	3	3	3 X		PLANNING LOOP	2 X		3	3	ЗХ	
COMM IDEAS	0	2	3	3	3 ()	COMM IDEAS	2	1	3	3	2	2
PLAN CONSTR	0	3	2	3	3 ()	PLAN CONSTR	2	1	2	2	2	2
EVAL W DRAW	0	2	4	2	2 ()	EVAL W DRAW	3	1	2	2	3	2
BASIS FOR MAK	0	3	ЗХ		з ()	BASIS FOR MAK	3	1	2 X		2	1
Ellie	PIZZARO	STV	ASTEFCA	RD /	ITCASMAZE	Ξ	Damian	PIZZAR	OSTY	STEFCA	RD	ITCASMA	ZE
PURPOSE	2	2	4	4	4 4	1	PURPOSE	2	3	3	5	3	3
GEN& DEV IDEA	2	2	4	4	4 4	4	GEN& DEV IDEA	3	3	3	5	4	3
EXPL THE PROB	2	2	2	4	2 2	2	EXPL THE PROB	3	4	3	5	3	3
ADDR TASK REC	2	2	4	4	5 4	1	ADDR TASK REC	2	3	3	4	З	3
PLANNING LOOP	2 X		4	5	2 X		PLANNING LOOF	2 X		4	4	4 X	
COMM IDEAS	2	2	5	4	4 3	3	COMM IDEAS	2	2	2	5	2	2
PLAN CONSTR	2	2	5	4	4	3	PLAN CONSTR	2	2	2	5	2	2
EVAL W DRAW	2	2	4	4	4	3	EVAL W DRAW	3	3	3	5	3	3
BASIS FOR MAK	1	3	4 X	•	4 2	2	BASIS FOR MAK	3	2	зх		3	2
Kate	PIZZARO	DST'.	ASTEFCA.	RDI	ITCASMAZE	Ξ	Michael	PIZZAR	OSTA	STEFCA	RD /	ITCASMA	ZE
PURPOSE	2	~	3	3	2 (י ר	PURPUSE	0	~	2	ວ າ	2	2
GEN& DEVIDEA	. 2	2	4	2	2 (ע -	GEN& DEVIDEA		2	~	3	2	2
EXPL THE PROB	2	2	2	2	2 (ر -	EXPL THE PROB		2	2	3	2	2
ADDR TASK REC	2	2	4	3	2 (ر	ADDR TASK REC	. 0	2	2	2	1	1
PLANNING LOOP	2 X	_	4	3	ЗХ	_	PLANNING LOOP	· ox	-	2	3	3 X	-
COMM IDEAS	2	2	3	3	2 (C	COMM IDEAS	0	2	2	3	2	2
PLAN CONSTR	2	2	3	3	2 (5	PLAN CONSTR	0	2	2	3	2	2
EVAL W DRAW	2	2	2	2	2 (3	EVAL W DRAW	0	2	2	3	2	2
BASIS FOR MAK	2	3	2 X		3 ()	BASIS FOR MAK	0	3	1 X		1	1

Focus Class (Cont.)

Girls	-	-					Boys						
Tasmin	PIZZARC	STV	STEFCAL	RD II	ITCASMAZE	Ē	Martin	PIZZARC	STV	STEFCA	RD II	TCASMA	ZE
PURPOSE	2	2	2	3	з з	3	PURPOSE		З	4	5	5	4
GEN& DEV IDEA	2	2	2	3	3 3	3	GEN& DEV IDEA	3	з	4	5	4	4
EXPL THE PROB	2	2	2	3	2 2	2	EXPL THE PROB	3	З	3	4	2	З
ADDR TASK REC	2	2	2	2	2 2	2	ADDR TASK REC	З	3	2	4	4	4
PLANNING LOOF	2 X		1	4	4 X		PLANNING LOOF	2 X		4	2	4 X	
COMM IDEAS	2	2	2	2	2 3	3	COMM IDEAS	2	2	4	3	5	4
PLAN CONSTR	2	2	2	2	2 2	2	PLAN CONSTR	2	2	4	4	5	4
EVAL W DRAW	2	2	2	3	3 2	2	EVAL W DRAW	3	3	4	5	4	4
BASIS FOR MAK	1	2	2 X		3 1	l	BASIS FOR MAK	1	3	4 X		4	4
Havleigh	PIZZARO	OSTY	STEFCA	RD II	ITCASMAZE	Ē	Noel	PIZZARO	STY	ASTEFCA	RD (I	TCASMA	ZE
PURPOSE	3	2	4	4	4 5	5	PURPOSE	2	2	3	2	2	0
GEN& DEV IDEA	2	2	4	4	5 4	1	GEN& DEV IDEA	2	2	3	2	2	0
EXPL THE PROB	2	2	4	4	3 3	3	EXPL THE PROB	2	2	2	2	2	0
ADDR TASK REC	2	2	2	4	2 2	2	ADDR TASK REC	2	2	3	2	2	ō
PLANNING LOOP	2 X	-	5	4	5 X	-	PLANNING LOOP	2 X		1	3	- З X	-
COMMIDEAS	2	2	4	4	2 3	2	COMMIDEAS	2	2	3	2	2	Ο
PLAN CONSTR	2	2	4	4	2 3	2		2	2	3	2	2	ñ
	2	2		7	5	л Л		2	2	2	2	2	0
BASIS FOR MAK	2	2	4 X	-1	4 4	4	BASIS FOR MAK	1	1	э́х	-	3	0
1.1						-	5 7					TOACLAR	75
Joiene	PIZZARC	517	ASTERCA	יעא	II CASMAZE	-	Lewis	PIZZARU	517	4SIEFCA	יי שא ה	II CASIVIA	ZE
PURPOSE	2	3	5	4	5 5	2	PURPOSE	3	2	4	3	3	3
GEN& DEVIDEA	2	3	5	4	5 5	2	GEN& DEV IDEA	2	2	4	3	3	3
EXPL THE PROB	3	3	4	3	3 3	3	EXPL THE PROB	2	2	2	2	3	2
ADDR TASK REC	2	3	3	3	4 :	S	ADDR TASK REC	1	2	4	3	2	2
PLANNING LOOP	2 X	_	5	3	4 X		PLANNING LOOP	2 X	_	4	4	ЗХ	
COMM IDEAS	2	2	4	2	4 4	4	COMM IDEAS	2	2	4	3	2	2
PLAN CONSTR	2	2	5	3	4	3	PLAN CONSTR	2	2	4	3	2	3
EVAL W DRAW	3	3	4	2	4 5	5	EVAL W DRAW	2	2	4	З	3	2
BASIS FOR MAK	3	2	4 X		4 4	4	BASIS FOR MAK	2	3	2 X		3	3
Kara	PIZZAR	OSTY	ASTEICA	RD I	ITCASMAZI	E	Randal	PIZZAR	DSTV	ASTEFCA	RD /	ITCASMA	ZE
PURPOSE	3	2	З	3	3 2	2	PURPOSE	2	З	3	3	3	2
GEN& DEV IDEA	2	2	3	3	з :	2	GEN& DEV IDEA	2	З	3	З	3	2
EXPL THE PROB	2	2	З	З	З 2	2	EXPL THE PROB	2	З	З	З	3	2
ADDR TASK REC	2	2	2	3	4	1	ADDR TASK REG	2	3	2	1	1	3
PLANNING LOOP	• 2 X		З	3	4 X		PLANNING LOOP	2 X		4	3	2 X	
COMM IDEAS	2	2	2	2	3 3	2	COMM IDEAS	2	2	2	2	2	2
PLAN CONSTR	2	2	2	3	2 3	2	PLAN CONSTR	2	2	2	2	2	2
EVAL W DRAW	2	2	З	2	3 3	2	EVAL W DRAW	2	4	3	З	З	2
BASIS FOR MAK	2	2	1 X		1 :	2	BASIS FOR MAK	2	1	3 X		3	З
Maria	DI77ADI	0070	ASTEICA	RD	ITCASMA7	F							
	7	20, / 2	401210/1	2	A	2							
	2	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	4	â		2							
	2 2	3	2	2	 	2							
	· ·	2	∠ ?	4 2	<u>د</u> د	2							
	י ∠ ג ייע	J	<u>ح</u> ۱	2		4							
COMMAIDEAS	2 1	n	, ,	2		2							
	4	4	4	с С	5	ム つ							
	2	4 7	4	4 7	. د ۸	∠ ?							
BASIS FOR MAN	1 2	2		0	- - 2	<u>۔</u> ۱							
	ູ່	0	J / J		4								

Focus Class (Cont.)

Girls	•					
Natasha	PIZZARC	STA	STEFCA	RD II	TCASMA.	ZE
PURPOSE	З	З	4	4	3	3
GEN& DEV IDEA	2	З	4	4	3	З
EXPL THE PROB	2	3	4	4	2	2
ADDR TASK REC	2	4	2	4	2	3
PLANNING LOOF	2 X		5	4	ЗХ	
COMM IDEAS	2	2	5	4	2	2
PLAN CONSTR	2	3	4	4	2	3
EVAL W DRAW	2	з	4	4	2	2
BASIS FOR MAK	2	3	4 X		2	3
Stacey	PIZZARC	STA	STEFCA	RD II	TCASMA	ZE
PURPOSE	2	2	4	4	3	2
GEN& DEV IDEA	2	2	4	4	3	2
EXPL THE PROB	2	2	4	3	2	2
ADDR TASK REC	1	2	4	4	2	1
PLANNING LOOF	1 X		4	4	ЗХ	
COMM IDEAS	2	2	4	4	2	2
PLAN CONSTR	1	2	4	4	2	2
EVAL W DRAW	2	2	4	4	3	2
BASIS FOR MAK	2	3	4 X		2	1
Emily	PIZZARC	STA	STEFCA	RD (I	ITCASMA	ZE
PURPOSE	0	2	0	2	3	З
GEN& DEV IDEA	0	2	0	2	3	4
EXPL THE PROB	0	2	0	2	2	2
ADDR TASK REC	0	2	0	2	2	2
PLANNING LOOF	0 X		0	3	3 X	
COMM IDEAS	0	2	0	2	2	2
PLAN CONSTR	0	2	0	2	2	2
EVAL W DRAW	0	2	0	2	3	3
BASIS FOR MAK	0	З	0 X		2	1
Chloe	PIZZARC)ST`A	STEFCA	RD I	ITCASMA	ZE
PURPOSE	2	2	0	2	3	3
GEN& DEV IDEA	3	2	0	2	З	4
EXPL THE PROB	з	2	0	2	2	2
ADDR TASK REC	2	2	0	2	2	2
PLANNING LOOP	2 X		0	З	зх	
COMM IDEAS	2	2	0	2	2	2
PLAN CONSTR	2	2	0	2	2	2
EVAL W DRAW	З	2	0	2	3	3
BASIS FOR MAK	3	з	οx		2	1

Comparison Class

Girls						Boys						
Rhiannon	PIZZA FR	os e	EASTICA	RD	SUITC MAZE	Alistair	PIZZA	FROS E	ASTI CA	RC :	SUITC MA	AZE
PURPOSE	2	З	2	3	4 3	PURPOSE		1	3	2	2	2
GEN& DE∖	2	3	2	3	4 3	GEN& DE∖	2	1	3	2	2	2
EXPL THE	2	2	2	2	2 3	EXPL THE	2	1	2	3	2	2
ADDR TAS	2	3	2	2	4 2		2	1	2	2	2	2
PLANNING	2 Y	Ŭ	2	4	1 X			X,	2	â	2 X	2
COMMINE	2	2	2	2	5 2	COMMINE	2,	ົ່າ	~	2	2	3
	2	2	2	2	5 2			<u>ح</u>	4	2	2	3
FLAN CON	2	4	2	4	3 2	FLAN CON	2	1	4	4	2	3
EVAL W DI	2	3	4	4	4 2	EVAL W D		1	2	2	2	2
BASIS FOF	3	З	1 X		4 2	BASIS FOR	• 1	2	ЗХ		2	3
Sophie	PI77A FR	nos P	EASTI CA	RC	SUITC MATE	Garth	P1774	FROSE	ASTICA	RD	SUITC MA	47F
PURPOSE	0	3	2	2	2 3	PURPOSE	///////////////////////////////////////	2	4	" つ	3	2
GEN8 DEV	õ	2	2	2	2 3	CENS DEV	2	2		ົ	3	2
	0	2	2	2	2 0		2	~	4	2	2	2
	0	3	2	2	2 2	EXPLINE	2	4	2	2	2	2
ADUR TAS	U	3	2	3	1 3	ADDR TAS	2	2	3	3	2	2
PLANNING	0 X	_	1	3	2 X	PLANNING	: 3.	X	1	З	1 X	
COMM IDE	0	2	2	2	2 2	COMM IDE	2	2	3	2	2	2
PLAN CON	0	2	3	2	23	PLAN CON	2	2	4	2	2	З
EVAL W D	0	3	2	2	2 3	EVAL W D	2	2	2	2	3	2
BASIS FOF	0	З	ЗХ		3 1	BASIS FOR	2	1	зх		2	З
Cassy	PIZZA FF	ROSE	EASTICA	IRD	SUITC MAZE	Jordan	PIZZA	FROS E	EASTICA	ARC .	SUITC MA	AZE
PURPOSE	3	3	2	2	2 3	PURPOSE		1	3	2	3	2
GEN& DE∖	3	3	2	2	23	GEN& DE\	, 1	1	3	2	3	2
EXPL THE	3	4	2	2	2 4	EXPL THE	2	1	4	2	З	2
ADDR TAS	3	3	2	2	2 3	ADDR TAS	3 2	1	1	2	2	3
PLANNING	зх		2	3	1 X	PLANNING	1	х	1	3	4 X	
COMM IDE	2	2	4	2	2 2	COMM IDE	: 1	1	2	2	2	З
PLANCON	2	2	4	2	2 3	PLAN CON	. 1	1	2	2	2	ŝ
	2	2	- -	2	~ ~ ~			4	2	2	2	2
EVAL VV DI	3	2	∠ 2 V	2	2 3		- 1	4	3 2 V	Z	3	2
DAGIG FUF	I	4	2 ^		2 J	BASIS FUR	- 1	1	2 8		I.	I
Chloe	PIZZA FF	ROSI	EASTI CA	RD	SUITC MAZE	Lee	PIZZA	FROS E	EASTICA	ARD	SUITC M	AZE
PURPOSE	2	0	2	2	2 2	PURPOSE		2	2	2	3	З
GEN& DEV	2	0	2	2	2 2	GEN& DE	2	2	2	2	3	3
EXPL THE	2	0	2	2	2 2	EXPL THE	2	2	2	2	3	3
	2	ō	2	2	1 1		: 2	2	1	2	2	2
	2 1	0	2	2	1 Y			v	-	2	1 V	0
COMMUNE	2 ^	~	2	5	2 2			^ ^	1	2	1	~
	2	0	2	~	2 2			2	1	4	۱ م	2
PLAN CON	2	0	2	4	2 2	PLAN CON		2	2	3	2	3
EVAL W D	2	0	2	2	2 3	EVAL W D	2	2	2	2	3	3
BASIS FOF	1	0	2 X		3 1	BASIS FOR	F 1	1	1 X		1	3
5					SINTE MATE	S faith and	0774	CROS	ACT C		SULTC M	170
Emma	PIZZAFI	1001	EASTICA 2	1RL 5	SUITEMALE		FIZZA	-RU31	ASTICA	4RL 2	SULL IVI	AZE
FURFUSE	3	4	ر ۸	5	2 3			~ ~	~~;	2		0
GEN& DEV	د ، -	4	4	5	3 3	GEN& DE	v 2	2	4	2	4	0
EXPL THE	3	4	2	5	2 3	EXPL THE	. 2	2	2	2	2	0
ADDR TAS	3	4	2	5	1 1	ADDR TAS	5 2	2	2	2	1	0
PLANNING	3 X		4	5	2 X	PLANNING	2	Х	4	3	1 X	
COMM IDE	2	4	3	4	32	COMM IDE	E 2	2	3	2	З	0
PLAN CON	2	4	3	4	1 3	PLAN CON	v 2	2	4	2	4	0
EVAL W D	3	4	З	5	2 3	EVAL W D	1 2	2	4	2	2	0
BASIS FOR	2	4	2 X		2 3	BASIS FO	F 1	1	2 X		З	0

Comparison Class (Cont.)

Girls							Boys						
Wendy PIZ	ZAFR	DS EA	STI CA	RC SU	ТС МА	ZE	Peter	PIZZA FR	OS EA	STI CA	RD SU	ТС МА	ZE
PURPOSE	2	2	2	0	3	з	PURPOSE		2	0	0	2	2
GEN& DEV	2	2	2	0	3	з	GEN& DE∖	2	2	0	σ	2	2
EXPL THE	2	2	2	0	4	2	EXPL THE	2	2	1	1	2	2
ADDR TAS	2	2	1	0	1	1	ADDR TAS	2	2	1	1	2	2
PLANNING	2 X		1	0	2 X		PLANNING	2 X		1	1	1 X	
COMM IDE	2	2	2	0	2	2	COMM IDE	2	2	1	1	2	2
PLAN CON	2	2	3	0	2	2	PLAN CON	2	2	1	1	3	2
EVAL W D	2	2	1	0	3	3	EVAL W D	2	2	1	1	2	2
BASIS FOF	2	3	1 X		3	1	BASIS FOF	1	2	1 X	•	2	1
Carys PIZ	ZA FR	OS EA	STI CA	RC SU	ІТС МА	ZE	Robert	PIZZA FR	OS EA	STI CA	RC SU	ITC MA	ZE
PURPOSE	2	2	2	4	2	2	PURPOSE		1	2	3	1	1
GEN& DE∖	2	2	2	4	2	2	GEN& DE∖	2	1	3	3	2	1
EXPL THE	2	2	2	3	2	2	EXPL THE	2	1	2	3	1	1
ADDR TAS	2	3	2	2	2	2	ADDR TAS	2	1	1	3	1	1
PLANNING	2 X		3	4	3 X		PLANNING	3 X		1	4	1 X	
COMM IDE	2	2	2	4	2	2	COMM IDE	2	1	3	2	1	1
PLAN CON	2	2	2	3	2	2	PLAN CON	2	1	3	2	1	1
EVAL W D	2	2	2	4	2	2	EVAL W D	2	1	2	з	1	1
BASIS FOF	2	3	2 X		1	1	BASIS FOF	2	1	ЗХ		1	1
Ellis PIZ	ZA FR	OS EA	STICA	RE SU	ITC MA	ZE							
PURPOSE	2	3	4	0	2	2							
GEN& DE∖	2	3	4	0	3	2							
EXPL THE	2	3	2	0	2	2							
ADDR TAS	2	3	4	0	2	2							
PLANNING	2 X		1	0	3 X								
COMM IDE	2	3	4	0	2	3							
PLAN CON	2	3	4	0	3	З							
EVAL W DI	2	3	4	0	2	2							
BASIS FOF	2	3	2 X		2	1							
Holly PIZ	ZA FR	OS EA	STICA	RD SU	ITC MA	ZE							
PURPOSE	3	2	2	2	4	2							
GEN& DEV	3	2	2	2	4	2							
EXPL THE	3	2	2	2	2	2							
ADDR TAS	2	3	1	3	4	3							
PLANNING	3 X	_	1	3	4 X	_							
COMM IDE	2	2	3	3	5	2							
PLAN CON	2	2	3	3	4	3							
EVAL W D	3	2	2	2	4	2							
BASIS FOF	3	1	ЗХ		4	3							
	774 60	00 F 4	0T. 0.4		1170 000	1-7							
Alice Pl2	.ZA FR	OSEA	STICA	RE SU	II C MA	AZE							
PURPOSE	3	0	4	3	3	3							
GEN& DEV	3	0	4	3	3	3							
EXPL THE	3	0	4	4	3	3							
ADDR TAS	2	0	4	3	2	З							
PLANNING	3 X	_	1	4	ЗХ								
COMM IDE	2	0	3	3	2	2							
PLAN CON	2	0	4	3	2	3							
EVAL W D	2	0	4	3	3	З							
BASIS FOF	3	0	ЗΧ		2	2							

Comparison Class (Cont.) Girls

Kathy	PIZZA P	ROS E	ASTI CA	ARD S	UITC M	IAZE
PURPOSE	2	0	2	2	2	З
GEN& DEV	2	0	2	2	2	3
EXPL THE	2	0	2	2	2	2
ADDR TAS	3	0	1	2	2	2
PLANNING	зх		1	З	зх	
COMM IDE	2	0	2	2	2	2
PLAN CON	2	0	з	2	2	2
EVAL W D	2	0	1	2	2	З
BASIS FOF	2	0	1 X		2	1
Nicola	PIZZA P	-ROS E	ASTI C	ARC S	UITC N	IAZE
PURPOSE	2	З	4	4	4	3
GEN& DEV	2	З	4	4	4	3
EXPL THE	2	з	2	4	2	3
ADDR TAS	2	З	2	4	3	2
PLANNING	2 X	2	1	4	1 X	
COMM IDE	2	2	4	4	4	2
PLAN CON	2	З	4	4	4	2
EVAL W D	2	3	2	2	4	3
BASIS FOR	2	3	2 X		3	2
Zara	PIZZA I	FROS E	ASTI C	ARC S	SUITC N	1AZE
Zara PURPOSE	PIZZA I 2	FROS E 2	ASTIC	ARC S 3	SUITC N 4	1AZE 3
Zara PURPOSE GEN& DEV	<i>PIZZA I</i> 2	-ROS E 2 2	ASTIC 4 4	ARC S 3 3	SUITC N 4 4	1AZE 3 3
Zara PURPOSE GEN& DEV EXPL THE	<i>PIZZA I</i> 2 1 2	FROS E 2 2 2	ASTi C. 4 4 4	ARE S 3 3 3 3	SUITC N 4 4 4	1AZE 3 3 3
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS	<i>PIZZA I</i> 2 1 2 5 1	FROS E 2 2 2 2 2	ASTIC. 4 4 4 4	ARC S 3 3 3 3 3	SUITC N 4 4 4 4	1AZE 3 3 3 1
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING	<i>PIZZA I</i> 2 1 2 5 1 2 2 2	FROS E 2 2 2 2 2	ASTi C. 4 4 4 4 1	ARC S 3 3 3 3 4	SUITC N 4 4 4 4 1 X	1AZE 3 3 3 1
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE	<i>PIZZA I</i> 2 1 2 1 2 1 2 2 1	=ROS E 2 2 2 2 2 3	ASTIC. 4 4 4 4 1 4	ARE S 3 3 3 3 4 3	SUITC N 4 4 4 4 1 X 4	1AZE 3 3 3 1
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN CON	<i>PIZZA I</i> 2 1 2 1 2 2 1 2 2 2	FROS E 2 2 2 2 2 2 2 2 2 2 2 2	ASTIC. 4 4 4 4 1 4 4	ARC S 3 3 3 4 3 2	SUITC M 4 4 4 4 1 X 4 4	1AZE 3 3 1 2 2
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN CON EVAL W D	<i>PIZZA I</i> 2 1 2 1 2 1 2 2 1 2 2 1 2 2	FROS E 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	ASTIC. 4 4 4 1 4 4 4 4	ARC S 3 3 3 4 3 2 3	SUITC M 4 4 4 1 X 4 4 4	1AZE 3 3 1 2 2 3
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN CON EVAL W D BASIS FOR	PIZZA I 2 1 2 1 2 2 1 2 2 1 2 2 3	FROS E 2 2 2 2 2 2 2 2 2 1	AST C. 4 4 4 1 4 4 4 4 4 X	ARC S 3 3 3 4 3 2 3	SUITC M 4 4 4 4 1 X 4 4 4 4	1AZE 3 3 1 2 2 3 3
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN CON EVAL W D BASIS FOR	PIZZA I 2 1 2 1 2 1 2 2 1 2 2 3	FROS E 2 2 2 2 2 2 2 2 2 1	AST, C. 4 4 4 4 1 4 4 4 4 4 4 X	ARE S 3 3 3 4 3 2 3	SUITC N 4 4 4 4 1 X 4 4 4 4 4	1AZE 3 3 1 2 2 3 3
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN CON EVAL W D BASIS FOR Kirsty	PIZZA I 2 1 2 1 2 1 2 1 2 1 2 3 7 1 2 3 7 1 2 2 3	FROS E 2 2 2 2 2 2 2 2 1 5 FROS E	ASTIC. 4 4 4 1 4 4 4 4 4 4 X SASTIC	ARC S 3 3 3 4 3 2 3 3 ARC S	SUITC M 4 4 4 1 X 4 4 4 5 UITC M	IAZE 3 3 1 2 2 3 3 3 1 AZE
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN CON EVAL W D BASIS FOR Kirsty PURPOSE	PIZZA I 2 1 2 1 2 1 2 1 2 3 7 1 2 3 7 1 2 2 3	FROS E 2 2 2 2 2 2 2 2 1 5 FROS E 3 2	ASTIC. 4 4 4 4 1 4 4 4 4 4 X ASTIC 2	ARE S 3 3 3 4 3 2 3 3 ARE S 2	SUITC M 4 4 4 1 X 4 4 4 5 UITC M 3	IAZE 3 3 1 2 2 3 3 1 AZE 2 2
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN CON EVAL W D BASIS FOR Kirsty PURPOSE GEN& DEV	PIZZA I 2 1 2 1 2 1 2 1 2 3 7 2 7 2 2 2 2	FROS E 2 2 2 2 2 2 2 2 1 5 7 5 7 7	ASTIC. 4 4 4 1 4 4 4 4 4 X 5 ASTIC 2 2	ARE S 3 3 3 4 3 2 3 3 ARE S 2 2	SUITC M 4 4 4 1 X 4 4 4 4 5 5 0 1 7 0 7 0 7 0 7 0 7 0 7 0 7 0 7 0 7 0	IAZE 3 3 1 2 2 3 3 <i>IAZE</i> 2 2
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN CON EVAL W D BASIS FOF Kirsty PURPOSE GEN& DEV EXPL THE	PIZZA I 2 1 2 1 2 2 1 2 2 3 7 7 7 2 2 2 2 2 2	FROS E 2 2 2 2 2 2 2 2 1 5 7 5 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	ASTIC. 4 4 4 1 4 4 4 4 4 4 5 ASTIC 2 2 2 2	ARE S 3 3 3 4 3 2 3 4 3 2 3 4 2 2 2 2 2	SUITC M 4 4 4 1 X 4 4 4 4 5 0 0 0 7 0 7 0 7 0 7 0 7 0 7 0 7 0 7 0	1AZE 3 3 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN CON EVAL W D BASIS FOF Kirsty PURPOSE GEN& DEV EXPL THE ADDR TAS	PIZZA I 2 1 2 1 2 2 1 2 2 3 PIZZA I 2 2 2 2 2 2 2	FROS E 2 2 2 2 2 2 2 2 1 5 7 5 7 5 7 7	ASTIC. 4 4 4 1 4 4 4 4 4 5 ASTIC 2 2 2 2 2	ARE S 3 3 3 4 3 2 3 4 3 2 2 2 2 2 3 2	SUITC M 4 4 4 4 4 4 4 4 4 4 3 3 2 2 2	1AZE 3 3 1 2 2 3 3 1 1 AZE 2 2 1
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN CON EVAL W D BASIS FOR Kirsty PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING	PIZZA I 2 1 2 1 2 2 1 2 2 3 7 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	FROS E 2 2 2 2 2 2 2 2 1 5 FROS E 3 3 3 3 3 3 3 3	ASTIC. 4 4 4 1 4 4 4 4 4 4 4 4 5 ASTIC 2 2 2 2 2 3 0	ARE S 3 3 4 3 2 3 4 3 2 3 2 2 2 3 3 3	SUITC M 4 4 4 4 4 4 4 4 4 4 3 3 2 2 2 4 X	IAZE 3 3 1 2 2 3 3 IAZE 2 2 1
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN CON EVAL W D BASIS FOR Kirsty PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE	PIZZA I 2 1 2 1 2 1 2 1 2 3 PIZZA I 2 2 2 2 2 2 2 2 2 2 2 2 2	FROS E 2 2 2 2 2 2 2 2 1 5 7 5 7 5 7 7 7 7 7 7 7 7 7 7 7 7 7 7	ASTIC. 4 4 4 4 1 4 4 4 4 4 4 2 2 2 2 2 3 2 2	ARE S 3 3 4 3 2 3 4 3 2 2 2 3 3 3 3 3	SUITC M 4 4 4 4 4 4 4 4 4 4 3 3 2 2 4 X 2	IAZE 3 3 1 2 2 3 3 <i>IAZE</i> 2 2 1 2 1
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN COM EVAL W D BASIS FOR Kirsty PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN COM	PIZZA I 2 1 2 1 2 1 2 1 2 3 PIZZA I 2 2 2 2 2 2 2 2 2 2 2 2 2	FROS E 2 2 2 2 2 2 2 2 1 5 7 3 3 3 3 3 3 2 2 2 2	ASTIC. 4 4 4 4 4 4 4 4 4 4 4 4 4 2 2 2 2 2 3 2 2 2 2	ARE S 3 3 4 3 2 3 4 3 2 2 2 3 3 3 2 2	SUITC M 4 4 4 4 4 4 4 4 4 4 3 3 2 2 4 X 2 2 4 2 2	IAZE 3 3 1 2 2 3 3 <i>IAZE</i> 2 2 1 2 2 1
Zara PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN CON EVAL W D BASIS FOF Kirsty PURPOSE GEN& DEV EXPL THE ADDR TAS PLANNING COMM IDE PLAN COM EVAL W D	PIZZA I 2 1 2 1 2 1 2 1 2 3 PIZZA I 2 2 2 2 2 2 2 2 2 2 2 2 2	FROS E 2 2 2 2 2 2 2 2 1 5 7 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3	ASTIC. 4 4 4 4 1 4 4 4 4 4 4 4 4 4 2 2 2 2 2 2	ARC S 3 3 4 3 2 3 4 3 2 2 2 3 3 3 2 2 2 2 3 3 2 2 2	SUITC M 4 4 4 4 4 4 4 4 4 4 3 3 2 2 4 X 2 3 3	IAZE 3 3 1 2 2 3 3 <i>IAZE</i> 2 2 1 2 2 1 2 2 2

APPENDIX P Publications arising from the Research

Journal Articles:

Summer 2000: Beyond their Capability: Drawing, Designing and the Young Child; The Journal of Design and Technology Education, Vol.5, No.2; Wellesbourne; DATA

Autumn 2001: *Taking Ideas on a Journey called Designing* The Journal of Design and Technology Education; Vol.6. No.3 Wellesbourne; DATA (originally presented as a paper for the Design and Technology International Conference, Coventry, July 2001)

January 2003: The Process of Solving Problems: young children exploring the rules of the game in science, mathematics and technology in The Curriculum Journal

Conferences for which Proceedings were published:

April 2000: Beyond "Draw one & Make it" Design and Technology International Millennium Conference (DATA); London

August 2000: Why Draw Anyway? The Role of Drawing in the Child's Design Tool Box; International Design and Technology Educational Research Conference (IDATER 2001); Loughborogh

June 2001: The Emergence of Understanding of the Relationship between Planning and Designing amongst Young Children; The Third International Primary Design and Technology Conference (CRIPT); Birmingham

July 2001: *Taking Ideas on a Journey called Designing*; Design and Technology International Conference (DATA), Coventry (subsequently published in DATA Journal, see above)

August 2001: Participant Research from the Perspective of a Participant Researcher; International Design and Technology Educational Research Conference (IDATER 2001); Loughborogh

July 2002: Questioning the Design and Technology Paradigm; Design and Technology International Conference (DATA); Coventry (extensive reference was made to this paper in Kimbell, R. (2003) Reflections on the DATA Conference "Paradigm" Debate; in The Journal of Design & Technology Education, Vol. 8 No.1.; Wellesbourne; DATA

June 2003: A Holistic View of Assessing Young Children's Designing; The Third International Primary Design and Technology Conference (CRIPT); Birmingham

Conferences without Published Proceedings:

April 2002: Learning something new: Problem-solving and the development of knowledge; Commonwealth Association for Science, Technology and Mathematics Education; St. Juliens, Malta

September 2002: Ideas on a Journey: Design and Research; British Educational Research Association (BERA); Exeter (published on-line)

September 2003: *Taking Ideas on a Journey : a model to support young children's thinking processes;* British Educational Research Association (BERA); Exeter (published on-line and also this paper is to be included in a published collection by the Centre for Educational Research, Canterbury Christ Church University College)

APPENDIX P Publications arising from the Research

Requests to use Published Work in other contexts:

Re. Beyond "Draw one & Make it"

TORQUE is the quarterly bulletin of the Technology Association of South Africa. We are small fry - we have about 200 members. I would very much like to publish a precis of the paper in our bulletin. I would also like to use it at Stellenbosch University where I lecture to the B Ed students on Technology (our version of your D&T).

Andre Goosen Technology Association 49 Newlands Road Claremont, 7708 South Africa

I have recently received a request from the Open University to publish a copy of your article "Beyond Draw One and Make it - Developing Better Strategies for the Use of Drawing for Design in Key Stages 1 and 2", which was presented at the Millennium DATA Conference. The OU wish to publish this on their Teach and Learn website. The Teach and Learn project is one of the Open University's current major initiatives, involving the on-line delivery of continuing professional development for teachers and other educational staff in primary and secondary schools. The web site will present a customised learning environment which will include a comprehensive range of both generic and subject-specific resources to support staff in their professional development.

Lucy Rose, Editorial Administrative Officer DATA, 16 Wellesbourne House, Walton Road, Wellesbourne, Warwickshire,

Web Publishing:

www.designdrawing.net - personal website, established with intention that it should become forum for discussion of design drawing development

www.kented.org.uk/ngfl/primary.html - The Kent National Grid for Learning Site. I have a Design & Technology site within this much larger web space, for which I was allocated funding and time. The Stan series of lessons are on this site.

Book To be published:

Teaching Design & Technology 3-11; with Continuum Publishers; series editor: Mark O'Hara. This is to form part of the "Reaching the Standards" Series. Manuscript submission date: Dec.2003